Queensland Competition Authority

Final recommendation

Part B: Queensland Rail declaration review

March 2020

Level 27, 145 Ann Street, Brisbane Q 4000 GPO Box 2257, Brisbane Q 4001 Tel (07) 3222 0555 www.qca.org.au

© Queensland Competition Authority 2020

The Queensland Competition Authority supports and encourages the dissemination and exchange of information. However, copyright protects this document.

The Queensland Competition Authority has no objection to this material being reproduced, made available online or electronically but only if it is recognised as the owner of the copyright and this material remains unaltered.

Contents

1	INTRODUCTION	1
1.1	The existing declaration	1
1.2	Queensland Rail's below-rail network	1
1.3	Structure of Part B	3
1.4	Final recommendation	4
1.5	The NCC's recommendation in the Port of Newcastle matter	4
2	THE SERVICE AND FACILITY	6
2.1	Introduction	6
2.2	Identifying the service	7
2.3	Identifying the facility	11
3	CRITERION (A)—PROMOTE A MATERIAL INCREASE IN COMPETITION	13
3.1	Introduction	13
3.2	The QCA's approach to assessing criterion (a)	14
3.3	Dependent markets for the service as a whole	14
3.4	Comparing a future with and without declaration	16
4	QUEENSLAND RAIL'S DEED POLL AND ACCESS FRAMEWORK	17
4.1	Introduction	17
4.2	QCA approach to assessing the deed poll and access framework	17
4.3	The deed poll as an appropriate counterfactual	18
4.4	Effectiveness of the deed poll and access framework as a constraint on conduct	21
4.5	Conclusion	27
5	CRITERION (A)—THE NORTH COAST ROUTE SERVICE	29
5.1	Part of the existing declared service and the dependent markets	29
5.2	Geographical description of the North Coast Route	29
5.3	Dependent markets	29
5.4	Above-rail freight haulage market on the North Coast Route	30
5.5	Queensland Rail's ability and incentive to exercise market power	36
5.6	Competition in the above-rail freight haulage market in a future with and without declaration	55
5.7	Conclusion	69
6	CRITERION (A)—THE MOUNT ISA ROUTE SERVICE	70
6.1	Part of the existing declared service and the dependent markets	70
6.2	Geographical description of the Mount Isa Route	70
6.3	Dependent markets	70
6.4	North West Queensland minerals tenements market	71
6.5	Queensland Rail's ability and incentive to exercise market power	76
6.6	· · · · · · · · · · · · · · · · · · ·	

6.7	Conclusion	91
7	CRITERION (A)—THE WEST MORETON ROUTE SERVICE	92
7.1	Part of the existing declared service and dependent markets	92
7.2	Geographical description of the West Moreton Route	92
7.3	Dependent markets	92
7.4	West Moreton region coal tenements market	93
7.5	Queensland Rail's ability and incentive to exercise market power	97
7.6	Competition in the West Moreton region coal tenements with and without declaration	102
7.7	Conclusion	107
8	CRITERION (A)—THE SOUTH WESTERN ROUTE SERVICE, THE WESTERN ROUTE SERVICE AND THE CENTRAL WESTERN ROUTE SERVICE	108
8.1	Parts of the existing declared service and the dependent markets	108
8.2	Geographical description of the South Western, Western and Central Western Routes	109
8.3	Dependent markets	109
8.4	Above-rail freight haulage markets on the agricultural systems	110
8.5	Queensland Rail's ability and incentive to exercise market power	121
8.6	Competition in the above-rail freight haulage markets in a future with and without declaration	130
8.7	Conclusion	135
_		
9	CRITERION (A)—THE TABLELANDS SYSTEM SERVICE	136
9.1	Part of the existing declared service and dependent markets	136
9.2	Geographical description of the Tablelands system	136
9.3	Dependent markets	136
9.4	Above-rail passenger market on the Tablelands system	137
9.5	Queensland Rail's ability and incentive to exercise market power	139
9.6	Competition in the above-rail passenger market on the Tablelands system	142
10	CONCLUSION FOR CRITERION (A)	143
11	CRITERION (B)—MEET TOTAL FORESEEABLE DEMAND AT LEAST COST	144
11.1	Introduction	144
11.2	The QCA's approach to assessing criterion (b)	145
11.3	The structure of criterion (b) analysis	145
11.4	The market	146
11.5	Period for assessing total foreseeable demand	148
11.6	Total foreseeable demand over the declaration period	151
11.7	At the least cost compared to any two or more facilities	152
11.8	Conclusion	153
12	CRITERION (C)—STATE SIGNIFICANCE	154
12.1	Introduction	154
12.2	The QCA's approach to assessing criterion (c)	154
12.3	The structure of criterion (c) analysis	157
12.4	Stakeholder submissions on criterion (c)	157

12.5	Data on Queensland Rail's rail systems	159	
12.6	Criterion (c)—the facility for the service as a whole	170	
12.7	Criterion (c)—the facilities for each part of the service	172	
13	CRITERION (D)—PROMOTE THE PUBLIC INTEREST	177	
13.1	Introduction	177	
13.2	The QCA's approach to assessing criterion (d)	178	
13.3	The structure of criterion (d) analysis	179	
13.4	Investment in facilities	179	
13.5	Investment in markets that depend on access to the service	185	
13.6	Administrative and compliance costs incurred by the provider of the service	195	
13.7	Other relevant matters	203	
13.8	Conclusions	210	
14	DECLARATION PERIOD	213	
APPEND	IX A: THE HOLD-UP PROBLEM	214	
Overviev	N	214	
Why doe	es hold-up occur?	215	
Case stu	dy: General Motors–Fisher Body	223	
APPEND	APPENDIX B: SERVICE AND FACILITY DEFINITIONS 225		

1 INTRODUCTION

1.1 The existing declaration

The use of Queensland Rail's below-rail network, as specified in s. 250(1)(b) of the QCA Act, is taken to be declared under Part 5, division 2 of the QCA Act (see Box 1).

The regulatory framework for the existing declaration is governed by the QCA Act, as well as Access Undertaking 1¹ (AU1), which was approved by the QCA on 11 October 2016 and expires on 30 June 2020. AU1 sets out the terms and conditions under which Queensland Rail provides access to its service. AU1 also addresses the process required for an access seeker to negotiate access to the service, and the way in which any disputes in relation to access are to be resolved.

Box 1: The declared service

Section 250(1)(b) of the QCA Act provides that the following service is taken to be a service declared under Part 5, division 2 of the QCA Act:

the use of rail transport infrastructure for providing transportation by rail if the infrastructure is used for operating a railway for which Queensland Rail Limited, or a successor, assign or subsidiary of Queensland Rail Limited, is the railway manager

The meaning of 'railway manager' and 'rail transport infrastructure' are defined in the *Transport Infrastructure Act 1994* (Qld), schedule 6.

1.2 Queensland Rail's below-rail network

Queensland Rail manages a rail network that extends more than 6,600 kilometres across Queensland and is used by freight and passenger trains.²

Queensland Rail provides the below-rail service as well as above-rail passenger services on its network. Pacific National, Aurizon Operations, Linfox and Watco (from December 2019³) provide above-rail freight services on various parts of the network. A map of Queensland Rail's network is in Figure 1.⁴

¹ Queensland Rail, Access Undertaking 1, approved by the QCA 11 October 2016, https://www.queenslandrail.com.au/business/access/Access%20Undertaking%20and%20related%20documents/ Queensland%20Rail%20Access%20Undertaking%201%202016.pdf.

² Queensland Rail, sub. 8, p. 2.

³ Watco Companies, Rail Services: Australia, accessed 7 February 2020, https://www.watcocompanies.com/services/rail/australia/.

⁴ This map is a simplified representation of the main railway track, and it does not show all rail transport infrastructure as defined within the meaning of the existing declaration.



Figure 1 Queensland Rail network

Source: QCA website, https://www.qca.org.au/Rail/Queensland-Rail/Qld-Rail-rail-systems.

Queensland Rail describes its network as comprising seven regional 'systems' and the Metropolitan 'system'.⁵ These systems are highlighted in Figure 1, and based on Queensland Rail information, can be summarised as follows:

- Tablelands system (dark blue on the map above)—comprises two corridors from Cairns to Forsayth and Normanton to Croydon. The system is currently used exclusively by tourist services.⁶
- North Coast Line (purple)—extends between Nambour (north of Brisbane) to Cairns.⁷ It services major population centres in Brisbane, Bundaberg, Gladstone, Rockhampton, Mackay, Townsville and Cairns as well as various ports along Queensland's eastern coastline. The line primarily transports intermodal/containerised freight, agricultural products and various regional passenger services.⁸
- Mount Isa Line (dark green)—extends from Stuart (near Townsville) to Mount Isa, and includes the Flynn to Phosphate Hill branch line. The line carries bulk minerals from the North West Minerals Province east to the Port of Townsville and carries mining and

⁵ Queensland Rail, sub. 8, p. 2.

⁶ Queensland Rail, sub. 33, p. 73.

⁷ The Parana to Rocklands and Kaili to Durroburra sections of the North Coast Line (in the Central Queensland Coal Network region) are currently managed by Aurizon Network.

⁸ Queensland Rail, sub. 33, pp. 63–64.

industrial inputs west into the Mount Isa region. It also carries a small amount of livestock as well as four one-way passenger services per week.⁹

- Central Western system (orange)—comprises the main line from Nogoa to Winton and the Clermont branch (from Emerald to Clermont). The system carries agricultural products including grain and livestock, as well as four one-way passenger services per week.¹⁰
- Western system (light blue)—comprises the main line from Miles to Quilpie, and a number of branch lines, including Dalby to Meandarra (Glenmorgan), Miles to Wandoan, and Westgate to Cunnamulla.¹¹ The system carries agricultural freight including grain and livestock, as well as four one-way passenger services per week (on the 'Westlander' from Brisbane to Charleville).¹²
- West Moreton system (pink)—extends from Rosewood to Miles. The system primarily carries coal, as well as agricultural products and regional passenger services.¹³
- Metropolitan system (light green)—radiates from the Brisbane central business district. It is bounded by Rosewood to the west and Nambour to the north, and extends south to Varsity Lakes station in the Gold Coast region, as well as south-west to the Acacia Ridge Terminal (where it connects with the interstate rail system to New South Wales).¹⁴ It is primarily used for commuter passenger services and is also an important rail freight connection, including for coal and agricultural products travelling from the west to the Port of Brisbane; intermodal freight travelling interstate and between Brisbane and north Queensland; and livestock travelling from the central west and north west regions to processing facilities in Brisbane.¹⁵ Currently, all interstate traffic (i.e. to/from New South Wales) must travel on the Metropolitan system and transfer to the interstate rail system at Acacia Ridge.¹⁶
- South Western system (yellow)—extends from Toowoomba south-west to Thallon, including the Wyreema to Millmerran and Warwick to Wallangarra branch lines. This system primarily carries agricultural products including grain.¹⁷

1.3 Structure of Part B

The assessment of the service that Queensland Rail provides is set out in this part (Part B) in the following order:

- Service and facility
- Criterion (a)
- Criterion (b)
- Criterion (c)

⁹ Queensland Rail, sub. 33, pp. 65–66.

¹⁰ Queensland Rail, sub. 33, pp. 72–73.

¹¹ Queensland Rail information shows that currently, the Charleville to Cunnamulla, Miles to Wandoan and Tycanba to Jandowae branch lines on this system are non-operational: Queensland Rail, sub. 33, p. 69, para. 352.

¹² Queensland Rail, sub. 33, pp. 69–70.

¹³ Queensland Rail, sub. 33, pp. 66–67.

¹⁴ ARTC, sub. 22, p. 2. The section of rail track extending from the Acacia Ridge Terminal south to the Queensland-New South Wales border is currently leased by the Australian Rail Track Corporation (ARTC).

¹⁵ Queensland Rail, sub. 33, attachment A, p. 5, table A1.

¹⁶ There is a change of rail gauge at Acacia Ridge, as Queensland rail networks are narrow gauge (1,067 mm), and New South Wales rail networks are standard gauge (1,435 mm).

¹⁷ Queensland Rail, sub. 33, p. 70, para. 380; sub. 33, attachment A, p. 4.

• Criterion (d).

1.4 Final recommendation

The QCA's final recommendation is that parts of the below-rail service provided by Queensland Rail, as described in s. 250(1)(b) of the QCA Act¹⁸, be declared. Specifically, the QCA recommends that the following parts of the service, each of which is itself a service within the meaning of s. 72 of the QCA Act, be declared:

- the North Coast Route service
- the Mount Isa Route service
- the West Moreton Route service
- the Central Western Route service
- the Western Route service
- the South Western Route service.

The QCA recommends that each of these parts of the service be declared for a period of 15 years.

The QCA recommends that the Tablelands system service not be declared.

Each of these parts of the service is defined in Appendix B.

1.5 The NCC's recommendation in the Port of Newcastle matter

On 24 September 2019, the National Competition Council (NCC) released its recommendation to revoke the declaration of the service at the Port of Newcastle.¹⁹

The NCC found that criterion (a)²⁰ was not satisfied, and that the Minister could reasonably form the view that criterion (d)²¹ was not satisfied. In making these recommendations, the NCC considered the declaration criteria in s. 44CA of the *Competition and Consumer Act 2010* (Cth) (CCA). These declaration criteria are substantially similar to the access criteria applied by the QCA under s. 76 of the QCA Act.

The QCA notes that the service provided by Queensland Rail is materially different from the service provided by the Port of Newcastle Operations (PNO). Queensland Rail provides access to its below-rail services, which is used by its customers to haul freight across the state, whereas PNO provides the right to access and use the shipping channels at the port, by which ships may enter, exit, load and unload at the relevant port terminals.²² The users of the Port of Newcastle service are generally homogenous (e.g. coal miners or their shipping agents), in contrast with the users of the Queensland Rail service, who are likely to be varied and operate in a variety of dependent markets, including mining, industry, agriculture, retail and logistics.

¹⁸ That is, 'the use of rail transport infrastructure for providing transportation by rail if the infrastructure is used for operating a railway for which Queensland Rail Limited, or a successor, assign or subsidiary of Queensland Rail Limited, is the railway manager'.

¹⁹ The NCC released its recommendation following the decision of the Minister (the Federal Treasurer) to revoke the declaration of the service at the Port of Newcastle on the same day.

²⁰ Section 44CA(1)(a) of the CCA.

²¹ Section 44CA(1)(d) of the CCA.

²² NCC, *Revocation of the declaration of the shipping channel service at the Port of Newcastle*, final recommendation, July 2019, para. 3.1.

The NCC considered that charges for the Port of Newcastle service were likely to remain a small proportion of the international spot prices of coal with or without declaration of the service. In contrast, the QCA considers that due to the nature of the services provided by Queensland Rail, and given submissions from stakeholders, it is likely that the access charges of Queensland Rail would be a material part of the overall supply chain costs of access seekers/users. Moreover, Queensland Rail would have an ability and incentive to exercise market power at time of contract renewal.

The NCC considered that there were a range of factors that would mitigate the risk of hold-up, including the impact on PNO's reputation and its lease agreement with the State. In the context of the submissions received, the QCA is not persuaded by Queensland Rail's arguments that hold-up would not occur. In particular, in relation to the Queensland Rail service, the QCA considers that long-term contracts and the prospect of reputational damage are not effective solutions to hold-up.

In considering whether the threat of reputational damage is an effective constraint on Queensland Rail's ability and incentive to exercise market power, the QCA had regard to whether Queensland Rail's conduct is transparent to other access seekers. Relevantly, in a future without declaration, there would not be oversight, or transparency, of Queensland Rail's access terms, including pricing terms.²³ It is likely that Queensland Rail would continue to bilaterally negotiate access terms with individual customers.

In contrast, the NCC noted that PNO advertises its fees for accessing its service in a 'Schedule of Service Charges' which is available on its website.²⁴ While PNO appears to charge different amounts for different types of users of the service, its Schedule of Service Charges provides for the same navigation service charge, wharfage charge and port security charge rates to be imposed on all coal vessels, regardless of the operator or whose coal is being carried.²⁵

The QCA considers that the different conclusions reached by the NCC and the QCA are likely due to the differences between the services provided by PNO and that provided by Queensland Rail.

²³ The QCA considers the effect of Queensland Rail's access framework in Part B, Chapter 4.

²⁴ NCC, *Revocation of the declaration of the shipping channel service at the Port of Newcastle*, final recommendation, July 2019, para. 7.145.

²⁵ NCC, *Revocation of the declaration of the shipping channel service at the Port of Newcastle*, final recommendation, July 2019, para. 7.146.

2 THE SERVICE AND FACILITY

2.1 Introduction

The starting point of the review of Queensland Rail's declaration is s. 250(1)(b) of the QCA Act:

the use of rail transport infrastructure for providing transportation by rail if the infrastructure is used for operating a railway for which Queensland Rail Limited, or a successor, assign or subsidiary of Queensland Rail Limited, is the railway manager

'Railway manager' and 'rail transport infrastructure' are defined in the *Transport Infrastructure Act 1994* (Qld) (TI Act).²⁶ Specifically, 'rail transport infrastructure' is defined in schedule 6 of the TI Act²⁷ as:

facilities necessary for operating a railway, including-

- (a) railway track and works built for the railway, including, for example-
- cuttings
- drainage works
- excavations
- land fill
- track support earthworks; and
- (b) any of the following things that are associated with the railway's operation-
- bridges
- communication systems
- machinery and other equipment
- marshalling yards
- notice boards, notice markers and signs
- overhead electrical power supply systems
- over-track structures
- platforms
- power and communication cables
- service roads
- signalling facilities and equipment
- stations
- survey stations, pegs and marks
- train operation control facilities
- tunnels
- under-track structures; and

²⁶ Section 250(5) and schedule 2 of the QCA Act.

²⁷ Reprint current from 11 April 2019.

- (c) vehicle parking and set down facilities for intending passengers for a railway that are controlled or owned by a railway manager or the chief executive; and
- (d) pedestrian facilities, including footpath paving, for the railway that are controlled or owned by a railway manager or the chief executive;

but does not include other rail infrastructure.²⁸

2.2 Identifying the service

2.2.1 Stakeholder submissions

Stakeholders did not dispute that the service is defined in s. 250(1)(b) of the QCA Act. However, stakeholders commented on whether Queensland Rail's service should be analysed as a single service or a series of smaller services.

Queensland Rail said that it provides eight services, being the use of the following eight sections of its network:

- the Mount Isa Line, being that part of the network bounded to the east by (and including) Stuart and to the west by (and including) Mount Isa and including all branch lines comprised in that part of the network;
- the North Coast Line, being those parts of the network bounded to the south by (and including) Nambour station, to the north by (and including) Cairns and to the west by (but excluding) Stuart and including all branch lines, comprised in that part of the network, including those in the Maryborough area and Taragoola to Graham;
- the West Moreton System, means that part of the network comprising the rail corridor from (and including) Rosewood to Miles, excluding all branch lines not directly connecting coal mine loading facilities to that rail corridor;
- the Western System, being those parts of the network bounded to the east by (and including) Miles and to the west by (and including) Quilpie and including all branch lines comprised in that part of the network, but excluding those parts of the network that are part of the West Moreton System;
- the South Western System, being that part of the network bounded to the west by (and including) Thallon, to the north by (and including) Toowoomba and to the south by (and including) Wallangarra and including all branch lines comprised in that part of the network including Wyreema to Millmerran branch line;
- the Central Western System, being that part of the network bounded to the east by (and including) Nogoa, to the north by (and including) Clermont and to the west by (and including) Winton and including all branch lines comprised in that part of the network;
- the Tablelands System, being those parts of the network bounded to the west by (and including) Normanton and to the east by (and including) Cairns and including all branch lines comprised in those parts of the network; and
- the Metropolitan System, being that part of the network bounded to the north by (and including) Nambour station and to the west by (and including) Rosewood and including all branch lines comprised in that part of the network.²⁹

While Queensland Rail acknowledged that each service entails the use of rail transport infrastructure for providing transportation by rail, it said such a definition (as set out by the QCA in the draft recommendation) lacks the specificity required for regulation under the QCA Act.³⁰

²⁸ The definitions of terms within the meaning of 'rail transport infrastructure' (for example, 'other rail infrastructure', 'railway' and 'railway manager') also appear in schedule 6 of the TI Act.

²⁹ Queensland Rail, sub. 33, p. 11, para. 52.

Queensland Rail argued that each service being assessed should be defined by reference to the sections of railway line used to provide this service.³¹ It said this approach is:

- implicit in the specification of the service in s. 250(1)(b) of the QCA Act, where the service is defined by reference to the railway for which Queensland Rail is the railway manager (i.e. the whole of Queensland Rail's narrow gauge network)
- implicit in the QCA's approach in the draft recommendation to identifying parts of the service, and specifically the eight rail systems constituting separate services to be assessed
- required, given the significant variation in the systems' supply chain dynamics, rail corridor characteristics and geographic locations.³²

In addition, Queensland Rail noted that in the draft recommendation, the QCA considered some services 'together' for the purpose of assessing criterion (a), namely the services provided using the West Moreton system and Metropolitan system, and the services using the North Coast Line and Metropolitan system. Queensland Rail submitted that such an approach is unnecessary, as access applications can be made in respect of more than one service. However, it said that if the QCA is minded to consider the Metropolitan system together with the North Coast Line and the West Moreton system, it is appropriate to amend the definition of services offered to capture the relevant infrastructure used to provide the service. That is:

- the North Coast Line service would also include the use of the Metropolitan system from Nambour to Roma Street and to Fisherman Islands, Moolabin or Acacia Ridge
- the West Moreton system service would also include the use of the Metropolitan system from Rosewood to the Fisherman Islands via Corinda and Yeerongpilly.³³

Queensland Rail further submitted that if such an approach was adopted by the QCA, the relevant service should be defined so that use of the Metropolitan system not in conjunction with the North Coast Line or the West Moreton system, is not captured.³⁴

Queensland Rail noted its submissions address the eight services it has identified, however, it also noted that those arguments apply equally to the service as a whole and its analysis is not changed if the alternate definitions for the North Coast Line service and West Moreton system service are adopted.³⁵

The South West Producers said the scope of the declaration means that access to all of Queensland Rail's rail network constitutes part of the declared service. While some of those rail lines are shown as distinct pieces of infrastructure (as in Figure 1), the South West Producers said it is important to note:

freight to and from a number of parts of the network travel across other parts of the network, particularly including the South East Queensland network. That is, the 'services' provided are best understood on an origin-destination basis, which do not necessarily correspond to Queensland Rail's designation of its own lines and systems.³⁶

³⁰ Queensland Rail, sub. 33, p. 10, para. 46.

³¹ Queensland Rail, sub. 33, p. 10, para. 47.

³² Queensland Rail, sub. 33, p. 10, paras 48–50.

³³ Queensland Rail, sub. 33, p. 12, paras 54–57.

³⁴ Queensland Rail, sub. 33, p. 12, para. 57.

³⁵ Queensland Rail, sub. 33, p. 12, para. 58.

³⁶ South West Producers, sub. 4, p. 9.

On this basis, from the perspective of the service provided on the West Moreton and Metropolitan systems, the South West Producers said the relevant service is either:

- the entirety of the Queensland Rail declared service as defined in the QCA Act (i.e. the entirety of the Queensland Rail network); or
- the use of rail transport infrastructure for providing transportation of coal to the Port of Brisbane by rail consisting of the West Moreton network, future extensions or expansions to it, and relevant parts of the South East Queensland network, including the dedicated dual gauge track from Lytton Junction to Fisherman Islands (the West Moreton corridor coal rail access service).³⁷

The South West Producers noted this definition is closer to Queensland Rail's definition (than to the QCA's)—but is more appropriate, as it defines the service by reference to what the customer is acquiring, rather than being unnecessarily prescriptive about particular points in the Metropolitan system (which the South West Producers consider has the potential for making any declaration ineffective if Queensland Rail makes changes to how coal services utilise the Metropolitan system to access the Port of Brisbane during the declaration period).³⁸

In addition, the South West Producers said that the rail access service utilised by coal producers with mines in the West Moreton system is properly considered as a single service involving access to parts of both the West Moreton and Metropolitan systems, given that:

- travelling only along the West Moreton system is of very limited utility on its own
- access is currently applied for and contracted on the basis of a service from a coal producers' rail load out to the QBH coal terminal at the Port of Brisbane (it is not contracted as separate services for each system)
- while it might be theoretically possible to apply for access separately, it would create additional risks (operationally and contractually) if access for each system had to be contracted separately.³⁹

Glencore said the Mount Isa Line is a service in itself and should be assessed separately (as Queensland Rail has submitted). However, it said the service should be properly defined, including the rail links to the Port of Townsville, via the Jetty branch line—Queensland Rail's definition does not include the section of rail between Stuart and Townsville and this should form part of the service.⁴⁰

2.2.2 QCA analysis

The QCA considers that the 'service' about which it must make a recommendation under s. 87A of the QCA Act is the service as a whole (i.e. use of the entire Queensland Rail network) as described in s. 250(1)(b). There is nothing in the text of the QCA Act that indicates the service currently declared consists of one or more separate rail systems.

If the QCA is not satisfied that the service described in s. 250(1)(b) of the QCA Act satisfies each of the access criteria, it may still recommend that a part of this service be declared if that part of the service is itself a service and satisfies each of the access criteria.⁴¹ The QCA interprets

³⁷ South West Producers, sub. 4, p. 4; sub. 31, p. 4; sub. 40, p. 9.

³⁸ South West Producers, sub. 40, p. 9.

³⁹ South West Producers, sub. 40, p. 9.

⁴⁰ Glencore, sub. 41, p. 2.

⁴¹ Sections 87A and 87C of the QCA Act.

s. 87A(1)(b) as contemplating a recommendation that one or more parts of a service may be declared (i.e. the reference to the singular 'part' includes the plural).

The QCA notes the description of services submitted by Queensland Rail, which sought to frame the services by reference to Queensland Rail's definitions of the various rail systems making up its network, being the:

- North Coast Line
- Mount Isa Line
- West Moreton system
- Metropolitan system
- South Western system
- Western system
- Central Western system
- Tablelands system.

The QCA considers that Queensland Rail's description of its rail systems is a convenient way to analyse the various components of the Queensland Rail network, since this is how the rail systems are widely described and understood by Queensland Rail and its users.

However, neither the separation of the network into these rail systems, nor the description given to each system by Queensland Rail, necessarily means that use of these systems constitute separate 'services' for the purpose of analysis under s. 87A of the QCA Act. Furthermore, these descriptions are not necessarily based on the economic activity in dependent markets that are potentially affected by the use of this rail infrastructure, as different parts of the Queensland Rail network are often used in combination with each other to provide rail access to customers in different dependent markets.

The operation of rail access services may necessitate the use of rail infrastructure in one or more rail systems. As a result, unless the provision of the service is confined to an individual system, the QCA has identified and defined services in terms of the use of 'routes', which encompass the relevant rail system(s) and other infrastructure that are necessary to operate that service.

This is central to the QCA's approach to identifying and defining the different 'parts' of the declared service, namely identifying the assets that are used to provide rail access services to customers in different dependent markets. Because of this approach to analysing the declared service, and the different parts of this service, the QCA has started with its assessment of criterion (a) in relation to the Queensland Rail service, rather than criterion (b).

2.2.3 Summary of QCA conclusions

In applying the access criteria, the QCA has first analysed the service as a whole—that is, the service as described in s. 250(1)(b) of the QCA Act. As discussed in section 3.3, the QCA considers that criterion (a) is not satisfied for the service as a whole. This precludes a recommendation that the whole of the service described in s. 250(1)(b) of the QCA Act be declared.

Therefore, the QCA has considered whether one or more parts of the service, which are themselves services, satisfy the access criteria. In summary, the QCA identified the following parts of the service on which it has conducted its assessment:

- the North Coast Route service
- the Mount Isa Route service
- the West Moreton Route service
- the Central Western Route service
- the Western Route service
- the South Western Route service
- the Tablelands system service.

The QCA considers that each of these parts of the declared service is itself a 'service' within the meaning of s. 72 of the QCA Act. Definitions of these terms are contained in Appendix B.

2.3 Identifying the facility

2.3.1 Stakeholder submissions

Stakeholders did not dispute that the facility for the service was defined in the QCA Act, but they also provided views on defining the facility by reference to rail systems.

Given that what is to be declared is the service rather than the facility, Queensland Rail submitted that it provides eight services (by reference to its rail systems) and the QCA is required to:

- identify the 'facility for the service' according to each of the eight services
- consider what the facility is (or facilities are), based on the minimum bundle of assets required to provide the relevant individual services (as in the approach outlined by the Australian Competition Tribunal in the Sydney International Airport case).⁴²

Queensland Rail said this approach is consistent with the policy rationale of ensuring an access regime only facilitates access where required and does not unduly stifle competitive forces, and in the present case is simplified, given the services are defined by reference to the relevant parts of Queensland Rail's network.⁴³ With respect to identifying the facility, Queensland Rail argued that 'the QCA has not engaged with the question required of it' in the draft recommendation. When this inquiry is made, Queensland Rail said, it is clear that there are eight facilities.⁴⁴

Other stakeholders also provided their views on how the facility should be defined, by reference to the infrastructure that is needed to provide the service:

- The South West Producers provided a preferred definition for the West Moreton corridor coal rail access service, with reference to infrastructure that they (or a similar customer) would need to use for the service.⁴⁵
- Glencore considered the Mount Isa Line and said the service should be properly defined by reference to the rail access service for the Mount Isa Line including the rail links to the Port of Townsville, via the Jetty branch line.⁴⁶

⁴² Queensland Rail, sub. 33, pp. 12–14.

⁴³ Queensland Rail, sub. 33, p. 13, para. 62.

⁴⁴ Queensland Rail, sub. 33, p. 13, paras 60–66.

⁴⁵ South West Producers, sub. 40, p. 9.

2.3.2 QCA analysis

The QCA considers that the facility for the existing declared service is the facility described in s. 250(1)(b) of the QCA Act, that is, *rail transport infrastructure* if the infrastructure is used for operating a railway for which Queensland Rail Limited, or a successor, assign or subsidiary of Queensland Rail Limited, is the railway manager.

'Rail transport infrastructure' is defined in schedule 6 of the TI Act to mean 'facilities necessary for operating a railway', and includes not only the railway track, but also 'things that are associated with the railway's operation', such as bridges, marshalling yards, overhead electrical power supply systems and service roads. Thus, the facility for the declared service is all facilities included in the statutory definition of 'rail transport infrastructure'.

The QCA considers that the facilities for any part of the declared service (which is itself a service) will be the assets (that is, the rail transport infrastructure) used to provide rail access services to customers in different dependent markets.

2.3.3 Summary of QCA conclusions

Given the QCA's approach to identifying the relevant service(s) (see section 2.2), the QCA has first analysed the facility as a whole, and secondly identified the following facilities for each part of the service:

- the North Coast Route
- the Mount Isa Route
- the West Moreton Route
- the Central Western Route
- the Western Route
- the South Western Route
- the Tablelands system.

Each of these facilities should be understood to refer to all relevant rail transport infrastructure, and not just the railway track. Definitions of these terms are contained in Appendix B.

⁴⁶ Glencore, sub. 41, p. 2.

3 CRITERION (A)—PROMOTE A MATERIAL INCREASE IN COMPETITION

3.1 Introduction

Section 76(2)(a) of the QCA Act is expressed as follows:

that access (or increased access) to the service, on reasonable terms and conditions, as a result of a declaration of the service would promote a material increase in competition in at least 1 market (whether or not in Australia), other than the market for the service

A summary of key matters raised by stakeholders with respect to criterion (a), as well as the QCA's final recommendations, are set out below in Table 1.

Criterion (a)			
Issue	Queensland Rail	Other stakeholders	QCA final recommendation
Queensland Rail's service as a whole (i.e. as defined in s. 250(1)(b) of the QCA Act)	Criterion (a) is not satisfied in relation to the Queensland Rail service as a whole	Stakeholders discussed a variety of dependent markets across the various Queensland Rail systems	Criterion (a) is not satisfied. Analysis of parts of the service undertaken See section 3.3
Queensland Rail's access framework	In a future without declaration, the access framework ensures access will be provided on reasonable terms and conditions As such, declaration would not promote a material increase in competition in any dependent markets	The access framework will cause uncertainty of pricing and non-pricing terms of access, which will damage competition in a number of dependent markets	See Part B, Chapter 4
North Coast Route service	Criterion (a) is not satisfied in relation to any parts of Queensland Rail's	Pacific National and Linfox considered criterion (a) was satisfied	Criterion (a) is satisfied See Part B, Chapter 5
Mount Isa Route service	service	Glencore considered criterion (a) was satisfied	Criterion (a) is satisfied See Part B, Chapter 6
West Moreton Route service	-	The South West Producers considered criterion (a) was satisfied	Criterion (a) is satisfied See Part B, Chapter 7
Central Western Route service Western Route service	-	Watco and GrainCorp considered criterion (a) was satisfied with respect to the use of the Central Western, Western and South Western systems	Criterion (a) is satisfied See Part B, Chapter 8
South Western Route service		Linfox focused on the Central Western system and considered criterion (a) was	

Table 1 Summary of key positions—s. 76(2)(a) of the QCA Act

Criterion (a)			
	satisfied		
Tablelands system service	No stakeholders made submissions relating specifically to the Tablelands system	Criterion (a) is not satisfied See Part B, Chapter 9	

3.2 The QCA's approach to assessing criterion (a)

The QCA's approach to assessing criterion (a) has broadly involved the following. First, the QCA has identified relevant dependent markets (other than the market for the service).⁴⁷ Subsequently, the QCA has assessed whether access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration, would promote a material increase in competition in at least one dependent market. This involves comparing the likely future with declaration and the likely future without declaration.

In particular, in comparing the future with declaration and the future without declaration, the QCA has assessed whether Queensland Rail has the ability and incentive to exercise market power, and if so, whether an exercise of that market power would have a material adverse effect on competition in a dependent market.

3.3 Dependent markets for the service as a whole

In assessing criterion (a), the QCA has first considered the service as a whole. Therefore, the QCA has sought to identify whether there is a market, other than the market for the service, that is dependent on access to the service as a whole.

Stakeholders identified a range of markets as separate from the market for Queensland Rail's service, but did not make detailed submissions to support the assessment of a single market that is dependent upon access to the whole of the service.

One possible market that is dependent upon access to the service as a whole may be an aboverail freight haulage market. The QCA has considered whether such a market exists, and based on the information before it, the QCA is not satisfied that such a market exists for the whole Queensland Rail service, for the purpose of assessing criterion (a).

A number of third party operators provide freight services on Queensland Rail's network. The QCA is satisfied that the market for these above-rail freight services is separate from the market for the below-rail access service. In considering whether this is a single Queensland-wide above-rail freight haulage market, and whether this market is dependent on access to the entire Queensland Rail network, the QCA notes:

- There is minimal cross-system traffic across the network (while some customers traverse more than one rail line to reach their destination, traversing the network more broadly is not required).
- The predominant types of freight hauled on each particular rail system are different, for instance:

⁴⁷ The market for the service is discussed in section 11.4.

- the North Coast Line carries predominantly intermodal freight
- the Mount Isa Line carries predominantly non-coal minerals and other bulk freight (e.g. acid and fertiliser)
- the West Moreton system carries predominantly coal.
- A range of customers depend on above-rail haulage services, and the market conditions these customers face vary markedly, depending on the type of freight hauled.
- The infrastructure requirements vary between the various rail systems, meaning that an above-rail operator will often need to configure bespoke rollingstock depending on the particular rail system on which they operate. For example, the Metropolitan system is rated A grade track (20–30 tonne axle loads), the North Coast Line and Mount Isa Line are rated B grade track (20–22 tonne axle loads), whereas the West Moreton system, the Central Western system, the South Western system and parts of the Western system are rated D grade track (15.5–18 tonne axle loads).⁴⁸
- Freight is not hauled on the Tablelands system. Queensland Rail submitted that the condition of the Tablelands system is such that there is little scope to operate freight services by rail.⁴⁹

Given this, the QCA has not identified a single above-rail market (or any other market warranting further analysis of criterion (a)) that is dependent on the use of the service as a whole. Therefore, the QCA is unable to conclude that criterion (a) is satisfied for the Queensland Rail service as a whole.

However, it is apparent from the above that there are a range of markets dependent on access to parts of the service. These are discussed in detail with respect to the part of the service on which they depend. Table 2 provides a brief summary of possible dependent markets that the QCA identified, the parts of the service that the QCA has identified and their corresponding facility. The definitions of each service and facility is in Appendix B.

Possible dependent markets	Description of the relevant part of the declared service	Facility
The above-rail freight haulage market	North Coast Route service	North Coast Line Metropolitan system (together, the North Coast Route)
The North West Queensland minerals tenements market	Mount Isa Route service	Mount Isa Line Those parts of the North Coast Line
The above-rail freight haulage market		that interconnect the Mount Isa Line and the Port of Townsville
The market for mining inputs in the North West Queensland minerals region		(together, the Mount Isa Route)

Table 2 Parts of the service, facilities and possible dependent n	narkets
---	---------

⁴⁸ Queensland Rail, Brisbane Metropolitan System Information Pack, October 2016, p. 30, https://www.queenslandrail.com.au/business/acccess/Documents/Brisbane%20Metropolitan%20System%20Infor mation%20Pack%20-%20Issue%203%20-%20October%202016.pdf.

⁴⁹ Queensland Rail, sub. 33, attachment A, p. 5.

Possible dependent markets	Description of the relevant part of the declared service	Facility
The market for coal tenements in the West Moreton region	West Moreton Route service	West Moreton system Metropolitan system
The above-rail freight haulage market		(together, the West Moreton Route)
The Port of Brisbane coal handling services market		
The above-rail freight haulage	Central Western Route service	Central Western system
market		North Coast Line that interconnects the Central Western system and the Metropolitan system and the Port of Mackay
		Metropolitan system
		(together, the Central Western Route)
The above-rail freight haulage	Western Route service	Western system
market		West Moreton system that interconnects the Western system and Metropolitan system
		Metropolitan system
		(together, the Western Route)
The above-rail freight haulage	South Western Route service	South Western system
market		West Moreton system that interconnects the South Western system and Metropolitan system
		Metropolitan system
		(together, the South Western Route)
The above-rail passenger market	Tablelands system service	Tablelands system

3.4 Comparing a future with and without declaration

In assessing whether access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a material increase in competition in at least one dependent market, the QCA has considered the likely future with declaration and the likely future without declaration.

An important consideration in assessing the likely future without declaration is the presence of other access arrangements that may apply, including the deed poll and access framework developed by Queensland Rail.

As Queensland Rail's deed poll and access framework are expressed to apply broadly⁵⁰, the QCA has first considered the operation of Queensland Rail's deed poll and access framework (Part B, Chapter 4), and then analysed criterion (a) in detail for each part of the service identified by the QCA (Part B, Chapters 5 to 10).

⁵⁰ That is, over the 'services provided using the North Coast Line, Mount Isa Line, West Moreton System and Metropolitan System' (Queensland Rail, sub. 33, p. 5, para. 15).

4 QUEENSLAND RAIL'S DEED POLL AND ACCESS FRAMEWORK

4.1 Introduction

Queensland Rail has developed an access framework to apply in a future without declaration, in the form of an annexure to an executed deed poll.⁵¹ The executed deed poll is dated 11 March 2019.

Queensland Rail said that the executed deed poll is irrevocable, which 'means that a new access framework will be legally binding on Queensland Rail and enforceable by a specified class of third parties in the future without declaration for services provided using the North Coast Line, Mount Isa Line, West Moreton System and Metropolitan System'.⁵²

Queensland Rail considered that the access framework will ensure that access will be available on reasonable terms and conditions in the future without declaration, such that declaration will not promote a material increase in competition in any dependent markets.⁵³ Queensland Rail stated:

The Access Framework is based on the QCA approved 2016 Access Undertaking that is currently in force and accordingly provides as much regulatory certainty for access seekers and access holders as currently exists and as much regulatory certainty as would exist in the future with declaration.⁵⁴

Queensland Rail's consultant, HoustonKemp, considered that there will be no difference in market outcomes between a future with and without declaration, given the similarity between the access framework and current regulation.⁵⁵

In contrast, the South West Producers and Glencore said that the deed poll and access framework will not constrain Queensland Rail's behaviour.⁵⁶ Pacific National did not consider the deed poll and access framework provided as much regulatory certainty for access seekers and access holders as currently exists under declaration.⁵⁷

The QCA has considered the degree to which Queensland Rail's deed poll and access framework affect the extent to which access as a result of declaration would promote a material increase in competition in a dependent market compared to a future without declaration. In doing so, the QCA has assessed the extent to which it would constrain Queensland Rail's ability to exercise market power in the absence of declaration.

4.2 QCA approach to assessing the deed poll and access framework

The QCA's approach, as a matter of principle, to how it has considered the deed poll and access framework is explained in Overview—Chapter 2.

In line with that approach, the QCA has considered:

• whether the deed poll is an appropriate counterfactual in the absence of declaration

⁵¹ Queensland Rail, sub. 33, attachment E, parts 1–3.

⁵² Queensland Rail, sub. 33, p. 14, para. 73.

⁵³ Queensland Rail, sub. 33, p. 44, para. 215.

⁵⁴ Queensland Rail, sub. 33, p. 5, para. 16.

⁵⁵ Queensland Rail, sub. 33, attachment B, p. 28.

⁵⁶ South West Producers, sub. 40, p. 5; Glencore, sub. 41, p. 20.

⁵⁷ Pacific National, sub. 37, p. 6.

how effective the deed poll and access framework is as a constraint on Queensland Rail's ability to exercise market power—including Queensland Rail's ability to expropriate rents (or sunk costs)⁵⁸ from access seekers at the time of renegotiating access contracts.⁵⁹

4.3 The deed poll as an appropriate counterfactual

4.3.1 Stakeholder submissions

Queensland Rail submitted that it had executed an 'irrevocable' deed poll.⁶⁰ In Queensland Rail's view, the deed poll 'will give rise to a binding and enforceable Access Framework in the future without declaration of services on the North Coast Line, Mount Isa Line, West Moreton System and Metropolitan system'.⁶¹

In addition to (and independently of) the constraints described above, to the extent Queensland Rail's services are not declared, Queensland Rail will provide open access to the Mount Isa Line, North Coast Line, West Moreton System and Metropolitan System on substantially the same terms as it currently does. Queensland Rail will provide services, for the purposes of both freight and passenger rail services, in accordance with a binding and enforceable access framework in the future without declaration.⁶²

A number of other stakeholders did not consider it appropriate for the QCA to consider the deed poll and access framework in assessing criterion (a).

Glencore, Pacific National and the South West Producers considered that the deed poll and access framework should be rejected as a 'contrived' attempt to circumvent the access criteria.⁶³

Pacific National submitted that acceptance of the access framework as the relevant counterfactual 'could potentially set a precedent which allows access providers to construct a counterfactual to declaration in a way that best suits their case, while leaving flexibility to amend the framework in future'.⁶⁴ Pacific National submitted that this ability to amend the framework without QCA oversight undermines its relevance as a counterfactual for the analysis of criterion (a).⁶⁵

The South West Producers and Glencore submitted that the deed poll was not legally effective, given that there has been no reliance upon and no acceptance of Queensland Rail's deed poll. The South West Producers expressly rejected the deed poll.⁶⁶ Both parties noted the New South

⁵⁸ For certain dependent markets (e.g. the above-rail freight haulage market), given the nature of the market, the expropriation may be of the firms' sunk costs rather than of rents. However, the hold-up problem also arises in these markets. Given the varied nature of the dependent markets analysed in this recommendation, the term 'rents' is used generally in this analysis of the access framework and deed poll.

⁵⁹ If an access seeker undertakes considerable sunk investment to enter (or expand in) a dependent market, the access seeker's willingness to pay will increase—affecting its negotiation power—at the time of contract renewal. This gives Queensland Rail the ability to expropriate the value of their customers' sunk investments at the time of contract renewal.

⁶⁰ Queensland Rail, sub. 33, p. 14, paras 72–73; sub. 33, pp. 39–40, paras 193–198.

⁶¹ Queensland Rail, sub. 33, p. 40, para. 199.

⁶² Queensland Rail, sub. 33, p. 39, para. 193.

⁶³ Glencore, sub. 41, pp. 23–24; Pacific National, sub. 37, p. 2; South West Producers, sub. 40, p. 38.

⁶⁴ Pacific National, sub. 37, p. 2.

⁶⁵ Pacific National, sub. 37, p. 2. Pacific National made similar submissions in relation to the DBCT access framework.

⁶⁶ South West Producers, sub. 40, p. 37.

Wales Court of Appeal decision in *Burns Philp Hardware Ltd v Howard Chia Pty Ltd*, that to be effective, a deed poll must first be accepted or relied upon by the intended beneficiaries.⁶⁷

4.3.2 QCA analysis

The QCA considers that the extent to which, in a future without declaration, the operation of the deed poll and access framework affects competition in dependent markets, compared to a future where there is access or increased access on reasonable terms as a result of declaration, is a relevant consideration.

The QCA has assessed the deed poll and access framework on its terms. In this regard, the relevant issue is whether the deed poll and access framework collectively represent a suite of arrangements that will, in the absence of declaration, effectively constrain Queensland Rail's ability to exercise market power.

Are access arrangements under the deed poll binding and irrevocable?

There are divergent views on whether the deed poll is binding on Queensland Rail and irrevocable. The QCA has carefully considered the submissions received and the issues raised on this question.

The QCA considers that Queensland Rail intends to be legally bound by the deed poll and access framework such that it considers it to be irrevocable, albeit that it will not impose access obligations unless and until the service ceases to be declared and a relevant service is not declared. By its terms, the access framework is effective from 9 September 2020 for five years, unless a relevantly defined service is declared (in which case it will end on that date).⁶⁸ This intention is evident from the terms of Queensland Rail's submissions during the QCA's consultation process, together with the terms of the deed poll and access framework.

The QCA has not assessed the deed poll and access framework on the basis that it is 'contrived'. While the deed poll has been produced in the context of the declaration review, having been executed, it should be assessed on its terms.

A party's intention to be legally bound by a deed can either be absolute or subject to fulfilment of a condition. The intention in question is the intention of the person said to be bound, rather than a mutual intention of the person bound and the putative beneficiaries of the deed. Where the intention is conditional, the deed is immediately irrevocable but becomes binding according to its terms once the condition is satisfied (although nothing further need be done by the party who delivered it).⁶⁹

Ultimately, the proposition raised by the South West Producers and Glencore—that the deed poll is not legally effective without acceptance or reliance—raises a question of legal principle in respect of which it is unnecessary for the QCA to form a concluded view. The deed poll, by its terms, will apply to access seekers only where those parties complete the required forms specified in the access framework. Where this is done, the factual foundation for the proposition that there is no acceptance or delivery will fall away.

The argument advanced by the South West Producers and Glencore appears to contemplate the possibility that, until there is acceptance or reliance, Queensland Rail can, in effect, change its

⁶⁷ Glencore, sub. 41, p. 20; South West Producers, sub. 40, pp. 36–37.

⁶⁸ Queensland Rail, sub. 33, attachment E, part 2 (access framework), cl. 1.1 and the definitions in cl. 7.1.

⁶⁹ Beesly v Hallwood Estates Ltd [1961] 1 Ch 105; Alan Estates Ltd v WG Stores [1982] 1 Ch 511; Monarch Petroleum NL v Citco Australia Petroleum Ltd [1986] WAR 310 at 357.

mind and repudiate the obligations it has taken upon itself through the deed poll. Regardless of whether the law permits this, Queensland Rail has submitted that the deed poll is irrevocable. Were Queensland Rail simply to reverse this position, after the declaration of the Queensland Rail service has expired, it would face the prospect of a fresh application for declaration, which would be founded, in part at least, on the ability of the service provider to repudiate commitments pursuant to a deed to prospective users apparently entered into in good faith. The QCA considers that, in reality, the prospect of Queensland Rail repudiating the deed poll is remote.

The QCA considers the deed poll is a part of the appropriate counterfactual in circumstances where prospective access seekers seek access or increased access in a future where the relevant service is not declared.

Non-compliance and disclaimer

Even though the deed poll is part of the counterfactual, there may be circumstances where, in the absence of declaration, the deed poll and access framework would not determine the basis upon which access or increased access to the service would be provided.

Under the terms of the access framework, if an access seeker does not agree to 'unconditionally and irrevocably' comply with the framework and deed poll, Queensland Rail may refuse to accept an access application (cl. 2.1.1(c)). It appears that in these circumstances, Queensland Rail would have broad discretion to refuse to accept the access application. Should an application be refused, in the sole discretion of Queensland Rail, the access seeker would have no right to request access and Queensland Rail would have no obligation to negotiate.

Alternatively, the QCA notes that the beneficiary of a deed poll may unilaterally disclaim the benefits under the deed poll.⁷⁰ If this occurred, the rights and obligations would cease to have effect between the maker of the deed and the beneficiary.

Even if Queensland Rail was prepared to negotiate with an access seeker that had disclaimed the benefits of the deed poll, or had not agreed to comply with its terms, Queensland Rail would be under no obligation to give access in accordance with the access framework. Access would, in effect, be offered by Queensland Rail on a voluntary basis. The access seeker would have no rights under the deed poll.

A prospective user who had disclaimed the benefits of the deed poll or refused to comply with the access framework would still have the option of seeking declaration of the relevant service in order to seek access on terms it deemed acceptable. The QCA notes that, in the event of a declaration application, the QCA would be required to consider the environment with and without declaration to assess the effect of declaration on competition in dependent markets. In these circumstances, where the application was made during the term of the deed poll and access framework, the deed poll and access framework would be part of the counterfactual. The QCA does not consider that, in any such assessment, it would be appropriate to disregard the available terms (and their effect on competition in dependent markets) on the basis that one or more access seekers had chosen to reject them.

In summary, the QCA considers that in a future without declaration, the deed poll is an appropriate part of the counterfactual. On this basis, the QCA has continued its assessment of

⁷⁰ *FCT v Cornell* (1946) 73 CLR 394; Seddon, N, *Seddon on Deeds*, 1st edn, Federation Press, Alexandria, NSW, para. 7.9.

the effectiveness of the deed poll and access framework as a constraint on Queensland Rail's conduct below.

Application of the access framework

Queensland Rail submitted that the access framework ensures that access to the North Coast Line, Mount Isa Line, West Moreton system and Metropolitan system will be available on reasonable terms.

Relevantly, there are a range of markets, such as rail freight haulage markets, that are dependent on access to parts of the service that are not subject to the same access arrangements outlined in the access framework and deed poll. This is due to rail freight haulage operators transporting goods across routes that span multiple lines, not all of which are covered by the deed poll and access framework.

In any case, the QCA has assessed the effectiveness of the deed poll and access framework as a constraint on Queensland Rail's conduct for those services which are subject to the access framework and deed poll.

4.4 Effectiveness of the deed poll and access framework as a constraint on conduct

Criterion (a) requires the QCA to determine if access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a material increase in competition in a dependent market. This involves comparing a future with declaration to a future without declaration, including a scenario in which the deed poll has been executed and is operative. This does not necessarily entail a clause by clause analysis of the deed poll and access framework, rather it is a question of whether there are any particular features that are relevant in comparing competitive conditions in a dependent market in a future with and without declaration.

In assessing criterion (a), the QCA considers that in a future without declaration, Queensland Rail has the incentive and ability to exercise market power in order to maximise profits in a way that may adversely affect competition in dependent markets.⁷¹ In considering a future with and without declaration, the QCA has assessed the extent to which the deed poll and access framework would constrain Queensland Rail's ability to exercise market power in a way that could potentially cause competitive harm in dependent markets.

The QCA has identified a number of aspects of the deed poll and access framework's operation that are particularly relevant to the assessment of its effectiveness as a constraint on Queensland Rail's ability and incentive to exercise market power. Each of those matters is considered in turn, namely:

- boundaries around price negotiation
- access negotiations
- compliance and enforcement of access arrangements
- the ability to amend the access framework.

⁷¹ The QCA's analysis of criterion (a) is in Part B, Chapters 5 to 10 below.

4.4.1 Boundaries around price negotiation

The access framework establishes access price bounds to guide negotiations—setting a revenue ceiling to limit Queensland Rail from earning revenues that exceed the efficient economic cost it incurs in providing the service. Queensland Rail stated:

Queensland Rail will thus be constrained in the future without declaration from imposing excessive access charges, even if it otherwise had the ability and incentive to do so (which it does not).⁷²

The QCA has considered the extent to which the pricing arrangements in the access framework are an effective constraint on Queensland Rail's ability to exercise market power in setting access charges.

The pricing arrangements that Queensland Rail has implemented in its access framework provide it with very broad discretion to set prices between a floor and a ceiling, where the latter reflects a standalone price of providing the service.⁷³ The QCA considers that the proposed revenue ceiling limit will not sufficiently constrain the extent to which Queensland Rail is able to expropriate rents from access seekers, such that it does not have a material impact on competition in dependent markets.

Where the revenue currently obtained from negotiated access charges is significantly below the ceiling limits⁷⁴, there is a high risk that access charges set at the ceiling would exceed what many participants in the various dependent markets that are serviced by Queensland Rail would be willing and/or able to pay. At the time of renewing an access agreement, that is, after the access seeker has undertaken sunk expenditure, the access seeker's willingness to pay for access can be expected to increase and Queensland Rail may be able to increase access charges as a result.

Moreover, while there are limitations on access charge differentiation, Queensland Rail has not sought to constrain the extent to which it is able to vary access charges for a user relative to the user's existing charges, at the time of contract renewal.⁷⁵ In this regard, the boundaries around price negotiation in the access framework are not an effective constraint on Queensland Rail's ability to expropriate rents from access seekers at the time of renegotiating access contracts.⁷⁶

The QCA's concern is not merely that Queensland Rail may have an ability to expropriate rents from users at the time of contract renewal, but that an access seeker, alerted to this risk, may decide not to contract with Queensland Rail at all. Put another way, the access framework, by creating a risk of hold-up for access seekers at the time of contract renewal, could deter utilisation of Queensland Rail's services and hinder competition in dependent markets.

⁷² Queensland Rail, sub. 33, p. 40, para. 197.

⁷³ Queensland Rail, sub. 33, attachment E, part 2 (access framework), cl. 3.2.

⁷⁴ Queensland Rail, sub. 33, p. 52, para. 255.3.

⁷⁵ Limits around pricing to constrain the extent to which Queensland Rail is able to vary access charges from one contract to a user's next contract at the time of contract renewal are not included in the access framework. The standard access agreement outlines that any rights that the access holder may have in relation to the renewal of this agreement will be as expressly provided in the access framework (see Queensland Rail, sub. 33, attachment E, part 3 (standard access agreement), cl. 1.2).

⁷⁶ In a future with declaration, there is scope to constrain Queensland Rail's ability to expropriate rents from access seekers at the time of renegotiating access contracts. For instance, an approved access undertaking may require renewal rights or set a reference tariff (as provided by s. 101(4) of the QCA Act) to mitigate the risk of hold-up for access seekers.

It follows that the QCA does not consider that the ceiling limit, by itself, will constrain Queensland Rail's ability to exercise market power in future contracting periods, such that this could not have a material impact on competition in dependent markets.

Additionally, the South West Producers submitted that the predictability of the ceiling would be severely impacted, such that the negotiate–arbitrate regime would not serve its purpose of facilitating commercial resolution.⁷⁷

The ceiling is calculated by Queensland Rail using the depreciated optimised replacement cost (DORC) methodology.⁷⁸ This requirement is not prescribed by the QCA Act, nor is it in Queensland Rail's approved 2016 access undertaking. The QCA notes that adopting a DORC valuation approach for setting the ceiling will involve matters of judgment, embodying multiple subjective assumptions.

Determining a regulatory asset base (RAB) using a DORC methodology has proven to be highly contentious in the context of gas pipeline regulation, and has been the subject of lengthy appeals. The QCA notes that the outcome of a RAB revaluation using a DORC methodology could result in widely different estimates depending on the approach and assumptions adopted by Queensland Rail, and could lead to a material increase or decrease in access charges.⁷⁹ Queensland Rail is to calculate the DORC; it is unclear whether access seekers will be provided with sufficient information to understand or determine the likely grounds to dispute the valuation undertaken. This fails to provide access seekers with certainty in relation to the bounds for future price negotiations.

4.4.2 Access negotiations

The access framework relies on parties to negotiate access rights, with recourse to arbitration if parties cannot reach an agreement on terms. Under the access framework, Queensland Rail is required to negotiate in good faith; not to unfairly differentiate between access seekers; and to consistently apply the access framework for all requests and negotiations for access.⁸⁰

Queensland Rail must provide information that is reasonably required by the access seeker for the purpose of negotiating with Queensland Rail, as well as specific categories of information, including, if requested by the access seeker, the price at which Queensland Rail provides access.⁸¹ Queensland Rail must also provide the access charge for the requested access rights, including the basis for the calculations and details of how Part 3 of the access framework has been applied in those calculations.⁸² Disputes under the access framework are to be resolved by an independent arbitrator (unless otherwise agreed, or resolved earlier in the process), with arbitration conducted in accordance with The Resolution Institute Arbitration Rules.⁸³ The access framework provides guidance to the arbitrator on how to determine an access dispute. This guidance is likely to inform negotiations.

⁷⁷ South West Producers, sub. 40, pp. 34–35.

⁷⁸ See Queensland Rail, sub. 33, attachment E, part 2 (access framework), cl. 3.2.3(c).

⁷⁹ For example, the QCA's decision on Queensland Rail's 2015 draft access undertaking valued Queensland Rail's assets on the West Moreton network having regard to the expected useful lives of assets. Queensland Rail's access framework does not permit such an approach.

⁸⁰ Queensland Rail, sub. 33, attachment E, part 2 (access framework), cl. 1.3.

⁸¹ Queensland Rail, sub. 33, attachment E, part 2 (access framework), cl. 2.7.2.

⁸² Queensland Rail, sub. 33, attachment E, part 2 (access framework), cl. 2.7.2(a)(vi).

⁸³ Queensland Rail, sub. 33, attachment E, part 2 (access framework), cl. 6.1.5(b).

While the access framework is largely based on the 2016 access undertaking, there are differences that are likely to have implications for access negotiations in a future without declaration.

In particular, it is unclear whether access seekers would have visibility of information on Queensland Rail's costs and asset values, which are relevant inputs in determining the price for access. As a result, access seekers may be at a disadvantage in negotiating access, as their ability to form a view about expected pricing outcomes under the access framework may be limited.

Additionally, the QCA notes that:

- Queensland Rail services various dependent markets, with the willingness to pay of users for the service varying considerably, reflecting the different characteristics of these various markets
- Queensland Rail bilaterally negotiates with individual customers, with no transparency of access terms between users, including in relation to pricing terms.

Queensland Rail submitted that the access framework addresses potential information asymmetries as it requires extensive information be provided to access seekers, providing an effective basis for negotiation.⁸⁴ While the access framework contains some obligations requiring information be provided to access seekers (e.g. cls. 2.7.2(a)(i) and (vi)), there are no express requirements for the provision of certain key information (such as cost information) to inform access seekers at the time of negotiating access.

In contrast, in a future with declaration, access seekers would benefit from the obligations in the QCA Act with respect to information provision. Such obligations include requirements for the access provider to give an access seeker information about the costs of providing the service, including the capital, operational and maintenance costs.⁸⁵ Moreover, the QCA Act provides for an access provider or access seeker to ask the QCA for advice or directions in relation to these matters.⁸⁶

Where a dispute is referred to independent arbitration, the access framework prescribes certain matters to which the arbitrator must have regard when making a determination (see cl. 6.1.5(f)).⁸⁷ These include the pricing methodology contained in the access framework.

As outlined above, the QCA is concerned that the proposed pricing rules in Queensland Rail's access framework will not sufficiently constrain the extent to which it is able to expropriate rents from access seekers, particularly at the time of contract renewal. As such, prospective access seekers may view the dispute resolution mechanism as deficient.⁸⁸

The QCA does not consider the overarching obligations on Queensland Rail to negotiate in the access framework will be an effective constraint on Queensland Rail exercising market power when negotiating access charges. In this respect, in a future without declaration, access seekers will be in a less favourable negotiating position with Queensland Rail at the time of renegotiating access charges, after having committed to entering the market and incurring

⁸⁴ Queensland Rail, sub. 33, p. 40, para. 197.

⁸⁵ QCA Act, ss. 101(2), (4).

⁸⁶ QCA Act, s. 101(5).

⁸⁷ Queensland Rail, sub. 33, p. 43, para. 205.

⁸⁸ In a future with declaration, it is open for the QCA to approve an access undertaking that contains terms and conditions of access that constrain Queensland Rail's ability to exercise market power at renewal time. Relevantly, any arbitration undertaken by the QCA cannot be inconsistent with an approved access undertaking (QCA Act, s. 119(1)(a)).

considerable sunk costs. The dispute resolution mechanism does not curtail this risk in circumstances where an arbitrator is required to take into account the pricing methodology in the access framework.

4.4.3 Compliance and enforcement of access arrangements

Effective compliance and enforcement measures provide stakeholders with greater confidence in the integrity of an access regime, and therefore greater certainty underpinning their investment decision-making.

In comparing the 2016 access undertaking and the executed access framework, HoustonKemp considered that Queensland Rail has a strong incentive for material compliance under both regimes given they are ultimately enforceable by the Queensland courts with comparable remedies.⁸⁹

The QCA considers that enforcement by a court or an arbitrator provides a mechanism for holding Queensland Rail accountable for compliance under the deed poll and access framework. However, future access seekers and access holders will likely face a greater degree of uncertainty associated with compliance and enforcement than would be the case with access under declaration.

In particular, it would be up to covenantees of the deed poll to bring proceedings before the court, as there is no independent body with equivalent investigative powers of the QCA to monitor and enforce compliance. No person has equivalent powers to conduct investigations, require information about compliance and take action, as occurs where a service is declared.

Additionally, the QCA Act specifies enforcement mechanisms to resolve access disputes and enforce access determinations; enforce compliance with access undertakings; and prohibit the hindering of access and unfair differentiation.⁹⁰ Under declaration, the QCA (or another person) may apply to the court for an order to enforce an access undertaking.⁹¹

As a result, compared to access with declaration, covenantees may face additional costs associated with enforcing the deed poll and access framework.

The South West Producers considered that in many cases the time delay and cost involved in bringing proceedings will mean that users may be better off accepting breaches of the deed poll and access framework.⁹²

Reliance on legal proceedings requires an access seeker or user who sought to enforce the deed poll, to undertake expensive and potentially protracted court proceedings, with an uncertain outcome. Given this, there may be a disincentive for affected parties to pursue enforcement for a breach.

4.4.4 Ability to amend access arrangements

Queensland Rail is able to amend the access framework, but only in accordance with cl. 6 of the deed poll. This provides for Queensland Rail to make amendments to the access framework that are:

⁸⁹ Queensland Rail, sub. 33, attachment B, pp. 32–33.

⁹⁰ QCA Act, ss. 112, 117, 123, 124, 152, 153, 158A.

⁹¹ QCA Act, ss. 10(ha), 158A.

⁹² South West Producers, sub. 40, p. 35.

- not inconsistent with the objective of the access framework—which is defined as the object of Part 5 of the QCA Act
- appropriate having regard to specified mandatory considerations—which are based on the matters set out in ss. 138(2) and 168A of the QCA Act.

The process for making amendments to the access framework involves a consultation process with access holders and access seekers who have signed an access application or renewal access application. Covenantees may challenge the validity of such amendments by commencing legal proceedings, which is the only avenue for recourse in relation to an access framework amendment dispute.⁹³ Queensland Rail said:

The Queensland courts are the appropriate forum for resolving disputes regarding any amendments to the Access Framework. The courts are well versed in overarching objective provisions and having regard to mandatory considerations and are equipped to efficiently determine a dispute arising under the Deed Poll, including in relation to amendments to the Access Framework.⁹⁴

A number of stakeholders observed that there is uncertainty as to whether the terms of the access framework will remain the same.⁹⁵

The QCA considers that the ability of Queensland Rail to amend price and non-price terms of access within the access framework creates a high degree of uncertainty for access seekers, as the basis for negotiating the terms and conditions of access may change over time. As the access framework sets out the negotiation framework and terms and conditions of access, its terms will be important to future access seekers considering entering into the relevant dependent markets.

The ability of Queensland Rail to amend the terms of access within the access framework will be of particular importance to access seekers, given that they will be required to renegotiate the terms of access at the time contractual arrangements expire—and do not have recourse to contractual remedies to constrain Queensland Rail from expropriating rents at the time of contract renewal. For instance, Queensland Rail has the ability to amend key price and non-price terms that may impact a user's ability to derive value from their sunk investments.

The following factors highlight the risk for prospective access seekers:

- Queensland Rail has considerable discretion under the deed poll in determining whether a change to the access framework is appropriate. While Queensland Rail must consult on proposed amendments and 'review and consider' any comments received, it is not bound to implement stakeholder comments.
- Although the criteria (cl. 6 of the deed poll) for amending the access framework adopts similar language to that of the QCA Act, a critical difference is that, with declaration, it is the QCA—as the independent regulator discharging its statutory functions in accordance with the QCA Act—that weighs the various considerations and determines what is appropriate. Under the deed poll it is Queensland Rail that is the decision-maker regarding changes to the access framework. It is also an interested party that stands to benefit from amendments in its favour and therefore has an incentive and ability to make changes, within the scope permitted by the deed poll, that favour its commercial interests.

⁹³ Queensland Rail, sub. 33, p. 40, para. 201; sub. 33, attachment E, part 1 (deed poll), cls. 6.4.3, 6.4.5.

⁹⁴ Queensland Rail, sub. 33, p. 40, para. 202.

⁹⁵ Pacific National, sub. 37, p. 2; Aurizon Coal, sub. 39, p. 2; Glencore, sub. 41, pp. 21–22; South West Producers, sub. 40, pp. 32–34.

While covenantees have the option of referring a matter to court if they consider an amendment has been implemented in breach of the deed poll, there are limitations on the ability of a covenantee to successfully challenge an amendment to the access framework through the courts. The relevant terms of the deed poll—that an amendment is permitted so long as it is not inconsistent with the framework objective and is appropriate having regard to mandatory considerations (cls. 6.2 and 6.3)—give Queensland Rail a broad discretion, making it difficult to establish that a proposed amendment is invalid. Additionally, there may be circumstances where a potential future user is not a beneficiary (that is, not a 'covenantee' within the meaning of the deed poll) and has no ability to challenge proposed changes to the access framework.⁹⁶

Queensland Rail faces little cost and risk (and therefore disincentive) of pursuing
amendments in its favour. In contrast, if a covenantee believed a proposed amendment to
be inappropriate or invalid, it would have to bear the costs and risks of undertaking legal
action to prevent it. Glencore and South West Producers submitted that Queensland Rail
could continually seek to make changes to the access framework—requiring users to go
through expensive and protracted legal processes to prevent the amendments.⁹⁷

Importantly, Queensland Rail has not sought to include terms in the deed poll (which would be subject to a covenant not to revoke or amend) that 'lock in' the basis for negotiating access, including access charges. This is in contrast to the deed poll executed by DBCT Management, which irrevocably constrained its ability to exercise market power on a key aspect of access. Queensland Rail's ability to modify those terms, and potential limitations on the ability to successfully challenge proposed changes to which an access seeker objects, significantly weakens the effect of the access framework on competition in dependent markets. This difficulty is heightened where an access seeker, negotiating an access agreement under the framework, is faced with the possibility that principles for setting access charges may be modified before it comes to negotiate a renewal of its access agreement (i.e. after its investment is sunk). This risk may serve as a material disincentive to seek access under the framework in the first place.

As outlined above, the QCA is concerned that the boundaries around pricing negotiation applied by Queensland Rail in the access framework provide for the risk of hold-up for access seekers, which will adversely affect competition in dependent markets. The risk of hold-up for access seekers is exacerbated by Queensland Rail's ability to modify the boundaries around pricing negotiation (which is relatively unconstrained), as it increases Queensland Rail's ability to hold up a user seeking to negotiate terms for a new access agreement once a user has incurred sunk costs involved with entering and/or expanding in the market.

4.5 Conclusion

The QCA considers the deed poll is a part of the appropriate counterfactual in circumstances where prospective access seekers seek access, or increased access, to the service in a future without declaration.

In a future without declaration, and in circumstances where the deed poll is operative, the QCA considers that the access framework is not an effective constraint on Queensland Rail's ability to exercise market power, chiefly because the deed poll and access framework do not constrain

⁹⁶ Queensland Rail, sub. 33, attachment E, part 1 (deed poll), cls. 2 and 6.

⁹⁷ Glencore, sub. 41, pp. 21–22; South West Producers, sub. 40, pp. 32–34.

Queensland Rail's ability to expropriate rents from access seekers at the time of renegotiating access contracts.

In this regard, the QCA notes:

- The pricing arrangements in the access framework do not constrain Queensland Rail's ability to exercise market power in future contracting periods, such that it would not have a material impact on competition in dependent markets.
- The overarching obligations on Queensland Rail to negotiate access are not effective constraints on Queensland Rail exercising market power when negotiating access charges.
- Future access seekers and access holders will likely face a degree of uncertainty associated with compliance and enforcement, more so than would be the case in a future with declaration.
- Queensland Rail's ability to modify the access framework exacerbates the risk of hold-up for users at the time of contract renewal.

The QCA considers that the access framework is not an effective constraint on Queensland Rail's ability to expropriate rents from access seekers at the time of renegotiating access contracts. The QCA considers that, in a future without declaration, the risk of hold-up will adversely affect the environment for competition in relevant dependent markets, as discussed below (Part B, Chapters 5 to 10).

5 CRITERION (A)—THE NORTH COAST ROUTE SERVICE

5.1 Part of the existing declared service and the dependent markets

The QCA has assessed the following part of the existing declared service and the following dependent markets:

Dependent market	Part of the existing declared service upon which the market is dependent	Facility for the relevant part of the service
The above-rail freight haulage market	North Coast Route service, that is use of the North Coast Route	North Coast Line Metropolitan system (together, the North Coast Route)

Table 3 The North Coast Route service and the dependent market

5.2 Geographical description of the North Coast Route

The North Coast Line extends south from Cairns along Queensland's eastern coastline to Nambour (approximately 100 km north of Brisbane Central station). From Nambour, the North Coast Line joins the Metropolitan system, which consists of a network of various lines and branch lines across the Brisbane metropolitan area.⁹⁸ Users of the North Coast Line traverse the Metropolitan system to access the crucial supply-chain infrastructure based in Brisbane, such as the import/export terminals at the Port of Brisbane, and the intermodal terminals at Acacia Ridge and Tennyson.

The Metropolitan system joins the interstate rail track (managed by the Australian Rail Track Corporation) at the Acacia Ridge terminal. The rail track changes gauge at Acacia Ridge— Queensland Rail's network is narrow gauge (with the exception of a dual gauge route from Acacia Ridge north to the Port of Brisbane), whereas the New South Wales rail network, including the track from Acacia Ridge south to the Queensland–NSW border, is standard gauge.⁹⁹

5.3 Dependent markets

The QCA considers that a relevant dependent market of the North Coast Route service is the above-rail freight haulage market on the North Coast Route—which is the market for the transportation of freight by rail on the North Coast Route. In this market, beneficial freight owners (or freight forwarders) contract with operators of rollingstock to haul freight via rail from an origin to a destination point along the North Coast Route.

⁹⁸ Queensland Rail, North Coast Line System South Information Pack, October 2016, p. 6, https://www.queenslandrail.com.au/business/acccess/Documents/North%20Coast%20Line%20South%20System% 20Information%20Pack%20-%20Issue%203%20-%20October%202016.pdf; Queensland Rail, North Coast Line System North Information Pack, October 2016, p. 6, https://www.queenslandrail.com.au/business/acccess/Documents/North%20Coast%20Line%20North%20System% 20Information%20Pack%20-%20Issue%203%20-%20October%202016.pdf.

⁹⁹ ACCC, Pacific National/Linfox—Proposed acquisitions of intermodal assets from Aurizon, Statement of Issues, March 2018, p. 19, paras 109–13, https://www.accc.gov.au/system/files/publicregisters/documents/MER18%2B2552.pdf.

Pacific National noted in its submission in response to the QCA's draft recommendation that the above-rail haulage market was a relevant dependent market, stating:

The stable market environment created by declaration promotes efficient investment in the above-rail haulage market and promotes investment in below-rail infrastructure.¹⁰⁰

Queensland Rail did not dispute the identification of the above-rail haulage market as a dependent market, and stated:

While there may be scope to debate the way in which the markets dependent on these services [the North Coast Line, Mount Isa Line, West Moreton system and Metropolitan system services] are defined in this review, the precise market definitions do not change the competition analysis of the likely future with or without declaration in this instance as Queensland Rail will continue to provide access on reasonable terms and conditions in the future without declaration.¹⁰¹

Queensland Rail noted the following, with respect to the distinction between above-rail and below-rail services:

The provision of track services and the running of freight trains can be provided separately and occur in functionally distinct markets. Currently, Aurizon and Pacific National provide freight services on Queensland Rail's systems. Queensland Rail provides below rail services on its systems but does not operate freight trains.¹⁰²

The QCA notes that other dependent markets include the downstream end product markets for the goods hauled on the North Coast Route; however, these markets are varied, given the diverse range of freight carried on the North Coast Route. In addition, Pacific National identified other dependent markets, such as the market for wagon and locomotive maintenance services, but it did not discuss these markets in detail.¹⁰³ The QCA has not examined these markets in detail, as stakeholders did not provide information on these markets and only limited information on them is publicly available.

Notwithstanding these other possible dependent markets, the QCA considers that one major relevant dependent market of the North Coast Route service is the above-rail freight haulage market.

5.4 Above-rail freight haulage market on the North Coast Route

5.4.1 The market

The market for above-rail freight haulage in this context is the market in which beneficial freight owners (or freight forwarders) contract with operators of rollingstock (locomotives and wagons) to haul freight via rail from an origin to a destination point along the North Coast Route (illustrated in Figure 2).

¹⁰⁰ Pacific National, sub. 37, p. 4.

¹⁰¹ Queensland Rail, sub. 33, p. 20, para. 107.

¹⁰² Queensland Rail, sub. 33, p. 21, para. 113.

¹⁰³ Pacific National, sub. 37, p. 5.



Figure 2 Above-rail freight haulage market on the North Coast Route

The suppliers in the above-rail freight haulage market are the operators of rollingstock. The QCA understands that Pacific National and Linfox (formerly Aurizon Operations) currently provide above-rail haulage services on the North Coast Route.¹⁰⁴

The customers in the above-rail haulage market include beneficial freight owners and freight forwarders. Freight forwarders offer a 'door-to-door' origin to destination service for freight owners, and typically contract above-rail haulage services on a wholesale level (and then package it with a pick-up and delivery service). Alternatively, beneficial freight owners may contract directly with above-rail haulage operators for rail haulage services.¹⁰⁵

The geographic dimensions of the above-rail market on the North Coast Route are bounded by the geographic dimensions of the rail systems—Brisbane in the south and Cairns in the north, along Queensland's eastern coastline, including the regional cities along the route (e.g. Bundaberg, Gladstone, Rockhampton, Mackay and Townsville).

5.4.2 The relationship between the market for the service and the dependent market

The market for the North Coast Route service is the market for the below-rail service on these rail systems.¹⁰⁶ In this market, Queensland Rail is the supplier of below-rail services and Pacific National and Linfox are the customers.

A dependent market for the North Coast Route service is the above-rail freight haulage market on this route. In the above-rail freight haulage market, Pacific National and Linfox are the suppliers, and beneficial freight owners and freight forwarders are the customers. Figure 3 illustrates the relationship between the market for the below-rail service and the dependent above-rail freight haulage market.

¹⁰⁴ Pacific National and Linfox provided details of their operations in their submissions—Pacific National, sub. 9, pp. 2–3; Linfox, sub. 50, p. 1.

¹⁰⁵ See ACCC, *Pacific National/Linfox–Proposed acquisitions of intermodal assets from Aurizon*, Statement of Issues, March 2018, p. 6, para. 35.

¹⁰⁶ Specifically, this is the market for the use of rail transport infrastructure for providing transportation by rail. A detailed discussion of the market for the service is in Part B, Chapter 11.
Figure 3 Relationship between the market for the service and the dependent above-rail haulage market



5.4.3 The nature of traffic on the North Coast Route

The main types of traffic carried by above-rail operators on the North Coast Route are:

- containerised (intermodal) freight (86% of total gtk carried on the line in 2017–18¹⁰⁷)
- long distance passenger services (7% of total gtk carried)
- agricultural products (bulk sugar, livestock and grain) (6% of total gtk carried)
- other products (1% of total gtk carried).¹⁰⁸

The North Coast Route primarily carries containerised intermodal freight. The term 'intermodal' is used to describe typically non-bulk or general freight in containerised, palletised and/or parcel configurations (e.g. retail products, manufactured goods, and industrial supplies). Intermodal and 'non-bulk' freight is distinguished from 'bulk' freight, which consists of loose homogenous commodities typically transported in large volumes, such as sugar, grain, acid, fertiliser, coal and minerals.

The range of products transported on intermodal above-rail haulage services on the North Coast Route include:

- retail goods: groceries, alcohol, white goods
- fruit and vegetables for wholesale or export markets
- import and export goods: electrical, home goods, processed meat and agricultural products
- building and construction products
- industrial products in packaged and palletised form.¹⁰⁹

The breakdown of the types of freight carried by rail operators on the North Coast Line is shown in Figure 4.

¹⁰⁷ Gross tonne kilometres (gtk) is a measure of the level of operating activity on a particular rail system. It is the product of the total gross weight of the train (i.e. including the locomotives and wagons used, as well as the goods and passengers carried) and the distance (in kilometres) travelled by the train.

 ¹⁰⁸ Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p. 11, https://www.queenslandrail.com.au/business/acccess/Compliance%20and%20reporting/2017-

^{18%20}QCA%20Annual%20Performance%20Report.pdf.

¹⁰⁹ Queensland Rail, sub. 33, attachment C, p. 69.



Figure 4 North Coast Line freight volumes by commodity (and passengers), 2017–18 (million gtk)

Source: Adapted from Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p. 11.

Intermodal general freight trains operate from the Brisbane region, and the major service destinations are Rockhampton, Mackay, Townsville and Cairns. Some of the services stop en route to drop and pick up freight. Intermodal general freight trains carry containerised freight servicing the domestic market. The intermodal freight services operate throughout the year with peak periods occurring before Easter and Christmas.¹¹⁰

Intermodal shipping freight trains operate to and from Fisherman Islands (Port of Brisbane) in Brisbane and have service destinations of Rockhampton and Mackay. Shipping freight trains carry containerised freight servicing the import/export market for the Port of Brisbane. The shipping freight services operate throughout the year.¹¹¹

In addition to intermodal freight, bulk agricultural freight services also operate on the North Coast Line. Bulk sugar freight services operate from Proserpine and Burdekin area mills to either Mackay or Townsville sugar terminals.¹¹² Bulk grain freight services operate from the southwestern and central-western Queensland regions to the Fisherman Islands, Mackay and Gladstone export terminals. Bulk agricultural services are seasonal.¹¹³

Bulk mining freight trains also traverse the North Coast Line at various locations in order to reach export ports. These include minerals services in transit from the Mount Isa Line and coal services in transit from Aurizon Network's Central Queensland Coal Network. Bulk mining freight services operate throughout the year.¹¹⁴

¹¹⁰ Queensland Rail, sub. 33, attachment C, p. 44.

¹¹¹ Queensland Rail, sub. 33, attachment C, p. 44.

¹¹² Queensland Rail, sub. 33, attachment C, p. 44.

¹¹³ GrainCorp, sub. 52, p. 4.

¹¹⁴ Queensland Rail, sub. 33, attachment C, p. 44.

5.4.4 Entry into the market

In the current market for above-rail freight haulage services on the North Coast Route, Pacific National and Linfox are the above-rail operators.

A potential new entrant into the above-rail freight haulage market on the North Coast Route is likely to incur substantial sunk costs as part of entering the market. For example, Pacific National noted in its submission:

PN has invested well over \$1 billion in above rail infrastructure in Queensland. Much of this infrastructure has very long lives and [is] only recently deployed ... Rail investment is largely sunk, most of the infrastructure could not be deployed elsewhere if PN were to cease operation in Queensland. This is primarily because rollingstock is configured for electric infrastructure (which is unique to Queensland) and a narrow gauge track which is not widely deployed in Australia.¹¹⁵

In this context, a potential entrant's decision may depend on several factors, including:

- anticipated revenue—for example, from sufficient contracts (or sufficient volumes) with major customers who are willing to commit to haulage contracts to underwrite and sustain that entry¹¹⁶
- anticipated below-rail costs—the price charged by Queensland Rail for the use of its belowrail network
- anticipated non-price access terms—for example, the scheduling of train paths by Queensland Rail in a way that allows for delivery times and frequency of service to meet customer needs, particularly given the need to coordinate paths with the existing operators' demand; and scheduling and access to intermodal terminal capacity, freight terminals, marshalling yards and other related below-rail infrastructure in Brisbane and along the North Coast Route.¹¹⁷

5.4.5 The features of the existing market

The starting point for an analysis of the future with and without declaration is to note some salient features of the existing market for above-rail freight haulage on the North Coast Route.

Two above-rail operators

Before 2018, the two above-rail freight operators on the North Coast Route were Pacific National and Aurizon Operations. In August 2017, Aurizon Operations announced its intention to sell its Queensland intermodal business (including operations on the North Coast Line) to Pacific National; in July 2018, this proposed sale was opposed by the ACCC.¹¹⁸ In January 2019,

¹¹⁵ Pacific National, sub. 37, p. 9.

¹¹⁶ South West Producers, sub. 16, p. 14; Glencore, sub. 17, p. 15. The South West Producers and Glencore both mentioned the example of Pacific National who was able to enter the Queensland coal haulage industry through a contract underwritten by volumes from Glencore and Rio Tinto. See also ACCC, *Pacific National/Linfox—Proposed acquisitions of intermodal assets from Aurizon*, Statement of Issues, March 2018, p. 15, para. 91.

¹¹⁷ ACCC, *Pacific National/Linfox—Proposed acquisitions of intermodal assets from Aurizon,* Statement of Issues, March 2018, p. 15, para. 91.

¹¹⁸ ACCC, Pacific National Pty Ltd/Linfox—Proposed acquisitions of Intermodal assets from Aurizon, accessed 10 February 2020, https://www.accc.gov.au/public-registers/mergers-registers/public-informal-mergerreviews/pacific-national-pty-ltd-linfox-proposed-acquisitions-of-intermodal-assets-from-aurizon.

Aurizon Operations completed the sale of its Queensland intermodal business to Linfox.¹¹⁹ Therefore, there will continue to be two above-rail freight operators on the North Coast Route—Pacific National and Linfox.

No evergreen renewal rights in access agreements

The QCA understands that access agreements with Queensland Rail are typically for a period of 10 years. Importantly, the existing access agreements for the Queensland Rail network (including for users of the North Coast Route service) do not provide evergreen renewal rights for the terms of access. Terms under these agreements (e.g. in relation to pricing, capacity allocation or usage of facilities) will progressively expire and must be renegotiated for new contracts within the bounds of an approved access undertaking.¹²⁰

In contrast, the QCA understands that existing access agreements with DBCT Management in relation to access to DBCT contain 'evergreen' renewal clauses, which allow incumbent users the option to extend their agreements and continue to access DBCT based on the terms of access set out in the existing agreements, up to the volumes in these agreements (see Part C, section 3.3.2).

Queensland Rail is not vertically integrated with respect to freight services

Queensland Rail's above-rail business comprises solely passenger services; it is currently the only provider of passenger train services across Queensland.¹²¹ Queensland Rail does not offer any above-rail freight haulage services and as such is not vertically integrated into the above-rail freight haulage market.¹²² Queensland Rail has said that its constitution limits its operations to the provision of rail passenger transport and activities relating to the planning, development and ongoing maintenance of its rail transport infrastructure, and that this constitution can only be amended with the consent of Queensland Rail's shareholding Ministers.¹²³ Thus, the QCA considers it unlikely that Queensland Rail would enter the above-rail freight haulage market in the foreseeable future.

Spare capacity exists on the North Coast Line and the Metropolitan system

According to Queensland Rail's consultant, HoustonKemp, the North Coast Line is not currently operating at capacity:

Figure 4.4 [of HoustonKemp's report] illustrates that none of the sections on the North Coast line are more than 50 per cent utilised. The most utilised section is Mackay to Durroburra at 44 per cent of train path capacity and the least used section Erkala to Mackay Harbour at 4 per cent. Thus, there is significant available capacity.¹²⁴

Queensland Rail repeated in its submissions that it has excess capacity on its network:

¹¹⁹ Aurizon Operations, Aurizon completes sale of Queensland intermodal business to Linfox, media release, 1 February 2019, accessed 4 June 2019, https://www.aurizon.com.au/en/news/2019/Aurizon-completes-sale-of-Queensland-Intermodal-business-to-Linfox; Linfox, sub. 50, paras 1.1–1.2.

¹²⁰ South West Producers, sub. 4, p. 28. See also QCA, *Queensland Rail's Draft Access Undertaking*, decision, June 2016, sections 2.5 and 3.6, https://www.qca.org.au/wp-content/uploads/2019/05/30680_Secondary-Undertaking-Notice-attachment-QCA-Decision-1.pdf, for a discussion of access renewal rights in the context of the Queensland Rail 2015 Draft Access Undertaking process (which resulted in the Queensland Rail 2016 Access Undertaking).

¹²¹ With the exception of a small number of small, localised, private tourist train operators.

¹²² Queensland Rail, sub. 8, p. 1, para. 2; Queensland Rail, sub. 33, p. 15, para. 79.

 ¹²³ Queensland Rail, *Draft Queensland Rail Access Undertaking 1*, explanatory submission, March 2012, pp. 15–17, https://www.qca.org.au/wp-content/uploads/2019/06/8354_r-qrail-qrail2012dau-expdocs2012dau-0412.pdf.
¹²⁴ Queensland Rail, sub. 33, attachment B, p. 22.

Queensland Rail has excess capacity on its network. A non-vertically integrated service provider with excess capacity has strong economic incentives to maximise utilisation on its network (so as to recover some proportion of its fixed costs) and thus has an incentive to promote (rather than limit) competition in downstream markets.¹²⁵

Queensland Rail argued that the existence of excess capacity on its network is one of the reasons that demonstrates it does not have the ability or incentive to exercise market power to adversely affect competition in any dependent market.

5.5 Queensland Rail's ability and incentive to exercise market power

Whether access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote competition in the dependent above-rail freight haulage market depends firstly on whether Queensland Rail has market power that it could use to adversely affect competition in the dependent market; and secondly on whether Queensland Rail has an ability and incentive to exercise that market power, in a future without declaration.¹²⁶

As a business, Queensland Rail has an incentive to maximise profits.¹²⁷ For example, under s. 10 of the *Queensland Rail Transit Authority Act 2013* (Qld), Queensland Rail is required to carry out its functions as a commercial enterprise (except its community service obligations). Similarly, under Queensland Rail's constitution, one of the objects of the company is 'the provision of network rail services and access to rail networks owned or operated by the Company or its subsidiaries for reward'.¹²⁸

The QCA considers that Queensland Rail does have market power that could be used in a way that adversely affects competition in dependent markets, including the above-rail freight haulage market. Queensland Rail is the natural monopoly provider of the service that is fundamental to the operation of the above-rail freight haulage market (among other dependent markets). Queensland Rail has argued that in a future without declaration, it would be constrained in its ability and incentive to exercise that market power to adversely affect competition in any dependent market.¹²⁹

In its submission in response to the QCA's draft recommendation, Queensland Rail raised detailed arguments and provided data to support its position that criterion (a) was not satisfied:

Queensland Rail submits that criterion (a) is not satisfied in relation to any of the services it provides because, contrary to the QCA's preliminary findings, Queensland Rail does not have the ability or incentive to exercise market power to adversely affect competition in any dependent market. This is because:

- Queensland Rail is no longer a vertically integrated service provider of rail transport services ...
- Queensland Rail has excess capacity on its network ...

¹²⁵ Queensland Rail, sub. 33, p. 15, para. 79.2; Queensland Rail, sub. 8, p. 4, para. 25.

¹²⁶ For example, NCC, Declaration of Services, A guide to declaration under Part IIIA of the Competition and Consumer Act 2010 (Cth), April 2018 edn, p. 33, para. 3.26; Duke Eastern Gas Pipeline [2001] ACompT 2 at [116]; Queensland Rail, sub. 33, p. 19, para. 100; South West Producers, sub. 40, p. 19; Glencore, sub. 41, p. 14.

¹²⁷ Or minimise losses.

¹²⁸ Queensland Rail, *Constitution of Queensland Rail Limited*, accessed 27 June 2019,

https://www.queenslandrail.com.au/aboutus/governance.

¹²⁹ Queensland Rail, sub. 33, p. 15, para. 79.

Queensland Rail faces intense and increasing competition from road operators.¹³⁰

Queensland Rail's submissions in support of its argument that it has no ability or incentive to exercise market power can be considered under three broad categories:

- (1) As a non-vertically integrated service provider with excess capacity on its network, Queensland Rail has 'strong economic incentives to maximise utilisation on its network (so as to recover some proportion of its fixed costs) and thus has an incentive to promote (rather than limit) competition in downstream markets'.¹³¹
- (2) Queensland Rail is materially constrained in the provision of below rail services by road freight operators. It said that '[r]oad transportation offers an effective substitute service to rail, which has a significant and direct downward impact on the prices that Queensland Rail negotiates with access seekers'.¹³²
- (3) Queensland Rail is constrained by other factors, such as:
 - (a) its statutory obligations and position as a statutory authority¹³³
 - (b) the threat of regulation or declaration¹³⁴
 - (c) dependent markets are already effectively competitive.¹³⁵

Each of these points, as well as the access arrangements applied by Queensland Rail, are discussed below. The operation of the access framework as a possible constraint on Queensland Rail's ability and incentive to exercise market power is considered in Part B, Chapter 4.

5.5.1 Non-vertically integrated service provider with excess capacity

No relevant vertical integration

Queensland Rail noted that it is no longer a vertically integrated service provider of rail transport services:

Queensland Rail is not (and does not intend to become) vertically integrated in a way that would give it an ability and incentive to leverage any market power into a dependent market.

The provision of track services and the running of freight trains can be provided separately and occur in functionally distinct markets. Currently, Aurizon and Pacific National provide freight services on Queensland Rail's systems. Queensland Rail provides below rail services on its systems but does not operate freight trains. As a result, Queensland Rail is not vertically integrated in a relevant way and has no incentive to leverage any market power in the provision of below rail services to advantage a related entity providing above rail freight transport services. This will not change in the future without declaration.¹³⁶

Other stakeholders did not dispute Queensland Rail's submissions on this issue.

The QCA considers that as Queensland Rail is currently not vertically integrated with respect to above-rail freight haulage services, it has no incentive to leverage any market power to favour a related entity in the provision of above-rail freight services. Queensland Rail is also unlikely to

¹³⁰ Queensland Rail, sub. 33, p. 15, para. 79.

¹³¹ Queensland Rail, sub. 33, p. 15, para. 79.1–79.2.

¹³² Queensland Rail, sub. 33, p. 15, para. 79.3.

¹³³ Queensland Rail, sub. 33, p. 32, paras 152–54.

¹³⁴ Queensland Rail, sub. 33, p. 32, para. 155.

¹³⁵ Queensland Rail, sub. 33, pp. 46–47, para. 225.

¹³⁶ Queensland Rail, sub. 33, p. 21, para. 112–13.

become vertically integrated with respect to above-rail freight haulage services over the period for which the service is recommended to be declared in the future.

However, a non-vertically-integrated firm such as Queensland Rail may nevertheless have an ability and incentive to exercise market power to increase its profits, which may adversely affect competition in a dependent market. Some examples of such an exercise of market power are discussed in section 5.6.2.

Excess capacity

Queensland Rail noted that there is excess capacity on its network and claimed that this excess capacity constrained its ability and incentive to exercise market power to adversely affect competition in a dependent market:

A non-vertically integrated service provider with excess capacity has strong economic incentives to maximise utilisation on its network (so as to recover some proportion of its fixed costs) and thus has an incentive to promote (rather than limit) competition in downstream markets.

... a consequence of spare capacity ... is that an access seeker that can be charged any positive margin over the incremental cost of providing the service represents a contribution to Queensland Rail's substantial fixed cost base. As such, Queensland Rail has an incentive to maximise demand for its services (rather than price).¹³⁷

Data from Queensland Rail demonstrates that it has spare capacity across the whole of its network, including significant spare capacity on the North Coast Line and the Metropolitan system.¹³⁸

As Queensland Rail is not vertically integrated into the above-rail freight haulage market, and has capacity on its network, there may be certain circumstances where Queensland Rail could be incentivised to provide access in order to recover a proportion of the fixed costs of the network (examples of such circumstances are discussed below). In such circumstances, Queensland Rail would arguably have nothing to gain from denying an access seeker entry or restricting access; rather, it could obtain additional revenue from selling unused network capacity, assuming it is not constrained to charging a uniform price.¹³⁹

However, as a general proposition, the presence of spare capacity does not imply that Queensland Rail will not behave in a profit-maximising manner. Put another way, a firm with market power has an incentive to maximise profits, not utilisation of capacity, even with spare capacity.

The QCA considers that a firm with market power would only have incentives to maximise volume in a limited set of circumstances. One such circumstance could be an infrastructure provider that faces previously unanticipated competition from another provider that has recently gained entry into the market. Given the presence of competition for demand, the incumbent provider might have an incentive to decrease its price below the profit-maximising

¹³⁷ Queensland Rail, sub. 33, p. 22, paras 118–19.

¹³⁸ Queensland Rail, sub. 33, attachment A, pp. 9–11.

¹³⁹ Under the 2016 access undertaking, Queensland Rail is not required to charge a uniform price on the North Coast Line or any other system except for West Moreton, where there is a reference tariff for coal trains. On the non-West Moreton systems, prices are negotiated between Queensland Rail and the customers seeking below-rail access.

price in order to gain sufficient revenue to cover (at least) its fixed costs. Importantly, this strategy would require some elasticity of demand for the service in order to expand output.¹⁴⁰

However, this does not characterise the general situation of Queensland Rail. It is the dominant service provider in most of its markets and does not face the prospect of competition. For example, in the West Moreton and Mount Isa regions, rail is the most economical option for the haulage of bulk minerals and coal. In those markets, Queensland Rail faces a relatively inelastic demand for its service, as there is no economically viable long-term substitute for rail to transport bulk minerals and coal. Accordingly, the QCA considers that economic circumstances in these regions are more likely to support the standard profit-maximising incentive.

In addition, 'the hold-up problem' can arise in situations where an access seeker does not make an investment, or underinvests, due to its expectation that the future conduct of the monopoly input supplier (i.e. access provider) may result in the (future) expropriation of part, or all, of the value of the access seeker's investment. Importantly, the incentive is present for the access provider to expropriate part of the access seeker's investment after it is made (i.e. ex post), regardless of whether the access provider has spare capacity. While Queensland Rail is likely to have an incentive to solve this problem prior to the investment, its ability to effectively do so is limited (discussed in section 5.6.3).

The potential for hold-up can explain why regulation may still be necessary for firms that do not even earn a normal rate of return (for example due to excess capacity):

According to the sunk investment hypothesis the presence or absence of monopoly rents is not the primary driver of regulation – rather, it is the scope for hold-up. A firm may have significant scope to hold-up its customers even if it is earning below normal returns – and indeed, the customers of a firm may be particularly exposed to hold-up if that firm receives some external source of funding (subsidies) which can be withdrawn at any time.¹⁴¹

This risk has been highlighted by stakeholders. For example, Pacific National stated:

QR repeats its claim that, due to spare capacity on its network, it would have incentives to maximise demand for below-rail services, and that this would provide some constraint on its ability and incentive to exercise market power.

This submission conflates the provision of access with provision of access on *reasonable terms*. While QR may face incentives to seek customers for its spare capacity, it will not be constrained in setting the terms of access to that capacity. On the contrary, QR will face incentives to maximise profits, and therefore may be expected to set unreasonable terms (as seen above, prices). Therefore the risk of hold-up remains, notwithstanding spare capacity in parts of the QR network.¹⁴²

GrainCorp agreed, stating:

As the QCA correctly observes [in the draft recommendation], while QR may have an incentive to provide access to spare capacity (so long as it is profitable to do so), it will not have an incentive to provide access on reasonable terms and conditions. Rather, as a monopoly service provider, QR may be expected to maximise its profit, which is likely to involve increasing access charges to users such as GrainCorp who rely on access to its infrastructure. Therefore, the risk of

¹⁴⁰ For example, this type of increase in volume (through a reduction in price) took place between an incumbent natural gas pipeline and a new entrant natural gas pipeline (see *Duke Eastern Gas Pipeline Pty Ltd* [2001] ACompT 2).

¹⁴¹ D Biggar, 'Is protecting sunk investments by consumers a key rationale for natural monopoly regulation?', *Review* of Network Economics, vol. 8, no. 2, 2009, pp. 23–24.

¹⁴² Pacific National, sub. 37, p. 11.

hold-up remains for potential entrants (or recent entrants wishing to expand their operations), notwithstanding spare capacity in parts of the network.¹⁴³

The prospect of hold-up, even in the presence of excess capacity, is an important issue. The hold-up problem is discussed in detail in a separate section (section 5.6.2).

5.5.2 Competition between road freight and rail freight transport

Queensland Rail argued that it has no ability or incentive to exercise market power to adversely affect competition in any dependent market because it is materially constrained in the provision of below-rail services by road freight operators:

In the future without declaration, Queensland Rail would be materially constrained in the provision of below rail services for the purposes of transporting freight by ... strong competition from road operators, which provide a closely substitutable service in respect of the transportation of freight, other than some bulk commodities being transported over long distances. Parties requiring freight transportation services can readily shift to moving freight by road in the event of an increase in access price and/or decline in quality of service provided.¹⁴⁴

Road transportation offers an effective substitute service to rail, which has a significant and direct downward impact on the prices that Queensland Rail negotiates with access seekers.¹⁴⁵

The QCA notes that Queensland Rail, in its submission in response to the QCA's draft recommendation, has substantially increased both the volume and detail of the arguments made in its earlier submission. The submission included detailed data and consultants' reports, which were not previously made available to the QCA. In considering this substantial volume of new information, the QCA has revised its analysis in relation to the competition between road and rail freight on the North Coast Line, as detailed below.

The nature of the freight task on the North Coast corridor

In assessing Queensland Rail's submissions, the QCA considers it beneficial to provide a discussion of the nature of the freight task on the North Coast corridor, in particular, the use of road and the use of rail for the transportation of freight.

The 'North Coast corridor' describes the route along Queensland's eastern coastline, bounded by Brisbane in the south and Cairns in the north, which passes through the regional cities of Bundaberg, Gladstone, Rockhampton, Mackay and Townsville. This transportation corridor includes both the Metropolitan and North Coast Line rail systems, as well as the Bruce Highway road infrastructure (which for many parts of the route runs adjacent to the North Coast Line).

The presence of competition between road operators and above-rail haulage operators in providing freight transport services on the North Coast corridor has been acknowledged by various stakeholders. For example, Pacific National stated:

The effect of declaration meant rail haulage providers could grow their business outside of specific tenders through investment. Rail haulage providers could create business opportunities and new freight markets in intermodal and bulk to compete against road transport.¹⁴⁶

Regulated access has also enabled PN to compete effectively with road freight operators in Queensland, particularly on the North Coast corridor and Mt Isa corridor.¹⁴⁷

...

¹⁴³ GrainCorp, sub. 52, p. 9.

¹⁴⁴ Queensland Rail, sub. 8, p. 4, para. 25; sub. 33, p. 25, para. 131.1.

¹⁴⁵ Queensland Rail, sub. 8, p. 1, para. 3; sub. 33, p. 15, para. 79.3.

¹⁴⁶ Pacific National, sub. 37, p. 5.

...

Access to the monopoly below rail infrastructure supports a highly competitive environment for haulage along these freight corridors. In the haulage market, PN competes with other above-rail operators such as Aurizon, as well as other modes of freight transport (e.g. road transport).¹⁴⁸

The Ranbury Management Group (Ranbury), stated in its final report on the North Coast Line capacity improvement study:

The freight on the North Coast Line can be categorised as either contestable freight or noncontestable freight. Contestable freight is defined as the current freight task being transported by rail that is more than likely able to be converted to road transport. Non-contestable freight is defined as the current freight task being transported by rail that cannot easily be subject to a mode shift to road ... This [intermodal] freight task is generally able to be easily switched between rail and road transport ... The non-contestable freight includes nickel ore and bulk products from the Mount Isa line.¹⁴⁹

Additionally, Queensland Rail highlighted previous statements made by Pacific National, Aurizon and the Australian Rail Track Corporation to the effect that rail competes with road for the transport task in the intermodal freight market.¹⁵⁰

The QCA considers that the extent to which the transportation of freight by road can compete with the transportation of freight by rail depends on the extent to which road transport is substitutable for rail transport for the particular type, volume and distance of freight to be transported. A range of factors determines the appropriate mode for any transport task. These include:

- the type, size and volume of product to be transported
- the suitability/capability of the mode to transport and handle the product
- total door-to-door cost
- reliability of the transport mode
- the origin/destination pairing of the task
- the required transit time
- service frequencies offered
- flexibility of service offerings
- level of customer service provided.¹⁵¹

Ranbury also stated:

The relative importance of the various factors will differ from customer to customer, heavily influenced by the features of the supply chain (ease of substitution) and most recent experiences. ARTC market research on freight forwarder and end customer users of rail concluded that price was the major driver of mode choice. Even though other factors such as reliability, frequency and consistency are considered, price or cost to the customer remains the critical factor.¹⁵²

¹⁴⁷ Pacific National, sub. 9, p. 6.

¹⁴⁸ Pacific National, sub. 9, p. 8.

¹⁴⁹ Queensland Rail, sub. 33, attachment C, pp. 47–48.

¹⁵⁰ Queensland Rail, sub. 33, pp. 31–32, para. 150.

¹⁵¹ Queensland Rail, sub. 33, attachment C, p. 73.

¹⁵² Queensland Rail, sub. 33, attachment C, p. 73.

The effect of the product type (bulk vs non-bulk products)

The Bureau of Infrastructure, Transport and Regional Economics (BITRE) found that the suitability of the transport mode depended upon the transport task, saying:

Each mode [road or rail] has attributes that render them more suitable, and generally less costly, for particular transport tasks. For example, the flexibility of road transport for urban goods distribution is unassailable; equally, the scale economies of rail over longer distances and for bulk commodities advantage it, over road, for these tasks.¹⁵³

Road transport is generally preferable for non-bulk goods, including perishable, fragile and highvalue commodities (e.g. food, manufactured goods, retail products), which tend to be more time- and reliability-sensitive. As the ACCC stated:

[R]oad freight is generally faster, as there are no strict cut off times, and in some cases more reliable as it is less prone to delay due to weather or maintenance events. For these reasons, road freight is considered better for goods with a short shelf life or otherwise with highly time sensitive delivery. Transport by road is also considered to be gentler on freight than transport by rail, making it more suitable for moving easily damaged goods. The flexibility of transport by a truck can make road freight more appropriate for products which need to be collected from a farm gate or a destination sufficiently far from the nearest rail terminal, such that the pick-up and delivery component would extend the total price and journey time.¹⁵⁴

The advantages of using road transport as compared against rail transport were also highlighted by Queensland Rail in its submission. For example:

- Road transport involves less complexity, as the responsibility for delivery usually rests on one road service operator. In contrast, rail transport involves the services of below-rail, above-rail and pick-up and delivery operators (although Queensland Rail acknowledges that freight forwarders do offer customers an end-to-end solution using rail transport).
- Road transport provides greater reliability than rail. Given the single track nature of the North Coast Line, planned maintenance and unplanned incidents can cause greater disruption to a customer using rail than a customer using road, particularly given the larger volumes that a full train can carry compared to a full truck, if a service is disrupted.¹⁵⁵
- Road transport options provide a shorter transit time—this factor is relevant for goods that are time sensitive.¹⁵⁶

In contrast, rail transport is generally preferable for bulk freight, which involves large volumes of homogenous product, typically liquid or crushed material (e.g. coal, minerals, sugar, grain), transported in mass quantities, without packaging, and which tend to be relatively non-perishable and non-fragile. As the ACCC noted:

Market feedback has indicated that freight owners moving large volumes of non-perishable or long-shelf life perishable goods view rail as the more appropriate mode over longer distances due to its tonnage and volume capacity, lack of time sensitivity and lower price.¹⁵⁷

¹⁵³ BITRE, *Road and rail freight: competitors or complements?,* information sheet 34, Australian Government, July 2009, p. 1.

¹⁵⁴ ACCC, *Pacific National/Linfox—Proposed acquisitions of intermodal assets from Aurizon*, Statement of Issues, March 2018, p. 13, para. 78.

¹⁵⁵ Although road transport is not immune to disruptions: the Bruce Highway is a one-lane highway (in each direction) throughout most of Queensland and often runs parallel to the North Coast Line. Unplanned incidents (such as weather events or traffic incidents) are also likely to cause sections of the Bruce Highway to be inaccessible for periods of time.

¹⁵⁶ Queensland Rail, sub. 33, pp. 27–30, paras 144–46.

BITRE also made it clear that road transport has only a very minor role in bulk freight transport:

Australia's bulk freight task is dominated by rail (48 per cent) and shipping (36 per cent) ... Road carries only a small proportion of bulk freight, with most of this aggregate building supplies, such as sand and gravel, generally carted short distances (less than 20 kilometres) ... Transporting many of these commodities [bulk freight] by road would be vastly more expensive, and so there is effectively no competition from road transport.¹⁵⁸

Rail enjoys an inherent cost advantage in the transport of bulk freight. This is due to its ability to exploit economies of scale (e.g. by attaching more wagons to a locomotive, a greater volume of goods can be carried for the one locomotive and one train path). Therefore, rail is able to transport large volumes of bulk freight at lower average cost than road transport, as road transport does not have economies of scale for large volumes (e.g. one truck can only carry a defined volume).

In summary, road transport is generally more advantageous for transporting non-bulk goods such as food and consumer products, which may be fragile or time-sensitive, or require flexibility in their delivery. This is because rail transport is subject to the availability of train paths (which are usually pre-planned), whereas road transport can typically depart as soon as the cargo is loaded. Additionally, transit times by rail are often slower than transit times by road, as the condition of Queensland Rail's below-rail infrastructure, as well as constraints from train path planning, mean that trains on the Queensland Rail network may be constrained by speed restrictions which are below the speed restrictions on the equivalent highways (i.e. between the same origin-destination pair as the train path).¹⁵⁹

Rail transport is generally more advantageous for transporting bulk goods such as minerals or grain (which tend to be heavy), and non-perishable or long-shelf-life goods in large volumes, which generally require repeated deliveries to and from predefined points (e.g. a mine or grain silos to an export port). Figure 5 provides a graphical representation of the road and rail usage for the transportation of goods by product type.



Figure 5 Road and rail transportation by types of goods

¹⁵⁷ ACCC, *Pacific National/Linfox—Proposed acquisitions of intermodal assets from Aurizon*, Statement of Issues, March 2018, p. 13, para. 79.

¹⁵⁸ BITRE, *Road and rail freight: competitors or complements?,* information sheet 34, Australian Government, July 2009, p. 3.

¹⁵⁹ Deloitte Access Economics, *Establishing the need for the last mile. Making the case for a dedicated freight rail link from Acacia Ridge to the Port of Brisbane*, final report, prepared for the Port of Brisbane Pty Ltd, n.d., https://www.portbris.com.au/getmedia/b793e8b5-edee-4945-850f-6feec8835720/DAE-Connecting-Inland-Rail-to-the-Port-of-Brisbane.pdf.

The effect of distance

Outside these specific instances where road or rail have inherent advantages, market evidence suggests that price is the key determinant of modal choice for freight that can technically be carried by either road or rail.¹⁶⁰

BITRE noted that in terms of Australian average freight costs, the average road freight cost per kilometre is more or less constant with respect to distance, whereas average rail costs decline with increasing freight volumes and distances, such that rail is cheaper for door-to-door freight hauls (i.e. rail freight tasks including door-to-door pick-up and delivery) above 1,000 km.¹⁶¹

Similarly, the ACCC noted that there is a 'tipping point' in terms of distance travelled, at which transport by rail is significantly cheaper than road:

Market participants have referred to a 'tipping point (varying between 600-1000km, or roughly the distance between Brisbane to Rockhampton or Mackay) at which rail is significantly cheaper than road ... Examples provided to the ACCC indicate that for freight carried beyond this tipping point, from Brisbane to Townsville/Cairns, road has been quoted as significantly more expensive than rail, with examples ranging from 15% more expensive to 300% more expensive. The ACCC has received tender information from a variety of market participants which shows that requests for tender for freight tasks from Brisbane to Cairns or Townsville are often only provided to or received by rail-based solutions. Market participants have provided the ACCC with a consistent message that unless their freight particularly requires road transport (for example, because it has a shorter shelf life and this is easily damaged), they generally do not consider road as an option past this tipping point.¹⁶²

Ranbury also noted that 'rail competes strongly on price to the long-haul destinations from Brisbane to Mackay and north':¹⁶³

Rail cannot compete on short-haul legs (e.g. Gladstone/Rockhampton), other than in the industrial market where the heavier weight of product is more of a consideration ...

The Mid North Coast region, up to Mackay, is acknowledged as being a road captive corridor. Rail becomes competitive on cost above Mackay and has a reasonable differential once you reach Townsville ...

The large customers typically use rail for long haul interstate corridors such as Melbourne – Brisbane and long haul intrastate corridors such as Brisbane – Far North Queensland, while using road for all other medium and short haul corridors (e.g. Brisbane – Mid North Coast).¹⁶⁴

...

Rail competes strongly on price to the long haul destinations from Brisbane to Mackay and north. $^{\rm 165}$

Rail transport does compete with road transport for containerised (intermodal) freight on the North Coast corridor

On the basis of the evidence discussed above, the QCA considers that the freight task on the North Coast corridor is segmented. Some customers have specific freight that is most

¹⁶⁰ Queensland Rail, sub. 33, attachment C, p. 73; ACCC, *Pacific National/Linfox—Proposed acquisitions of intermodal assets from Aurizon*, Statement of Issues, March 2018, p. 13, para. 80.

¹⁶¹ BITRE, *Road and rail freight: competitors or complements?*, information sheet 34, Australian Government, July 2009, p. 8.

¹⁶² ACCC, Pacific National/Linfox—Proposed acquisitions of intermodal assets from Aurizon, Statement of Issues, March 2018, pp. 13–14, paras 80–81.

¹⁶³ Queensland Rail, sub. 33, attachment C, p. 11.

¹⁶⁴ Queensland Rail, sub. 33, attachment C, p. 100–102.

¹⁶⁵ Queensland Rail, sub. 33, attachment C, p. 11.

appropriately transported by road (e.g. fragile or time-sensitive goods), for which the freight task is road-captive; equally, some customers have freight that is most appropriately transported by rail (e.g. bulk products).

However, the new range of evidence from stakeholders suggests that containerised (intermodal) freight on the North Coast corridor can, in most instances, technically be transported by either road or rail.¹⁶⁶ In these instances, the QCA considers that rail transport does compete with road transport for this freight task, and competes primarily on price.

As Ranbury noted:

A significant freight task is undertaken on the Queensland north coast corridor. This corridor is serviced by all the transport modes with the majority of the freight on road and a minor share on rail. However not all of the freight task on the north coast corridor is contestable between road and rail. Bulk freight is predominantly on rail and intra-regional general freight is 100% on road. The contestable component of the NCL freight market is the general freight intermodal line-haul market segment that travels principally between the origin-destination pairs of SEQ [South East Queensland] and major North Queensland and Central Queensland regional cities include Cairns, Townsville, Mackay, Rockhampton and Gladstone.¹⁶⁷

The QCA recognises that there is some variation in the evidence that has been reviewed on the precise distance at which transport by rail becomes more cost-effective than transport by road. Based on the evidence before it, the QCA finds:

- For non-bulk products travelling short distances (less than 600 km), road is the preferred transport mode.
- For non-bulk products travelling medium distances (between 600 and 1,000 km), road and rail compete for the freight task, although rail has been losing market share to road freight in recent years.¹⁶⁸
- For non-bulk products travelling long distances (greater than 1,000 km), rail is the preferred transport mode (although rail has also been losing market share to road freight in this market segment in recent years).¹⁶⁹
- For bulk products, particularly mining products, rail is the preferred transport mode, almost irrespective of distance travelled (unless for very small distances, such as less than 20 km).

Figure 6 depicts the segmented nature of the freight task on the North Coast corridor.

¹⁶⁶ For example, the transport of non-bulk freight that is neither fragile nor time-sensitive.

¹⁶⁷ Queensland Rail, sub. 33, attachment C, p. 104.

¹⁶⁸ Queensland Rail, sub. 33, p. 31, para. 149.1; sub. 33, attachment C, p. 10.

¹⁶⁹ Queensland Rail, sub. 33, attachment C, p. 10.

Figure 6 Choice of transport modes for the North Coast corridor

North Coast corridor non-bulk goods freight task



* Note: bulk goods are usually carried by rail regardless of distance travelled

Does competition from road transport constrain Queensland Rail's ability and incentive to exercise market power?

The QCA considers that rail transport does compete with road transport for the transport of containerised (intermodal) freight on the North Coast corridor. The issue is then whether competition from road transport operators provides an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power in the dependent above-rail freight haulage market.

Queensland Rail repeatedly argued that it is materially constrained in the provision of below-rail services by competition from road:

Queensland Rail is materially constrained in the provision of below rail services to freight operators. Most significantly, for all freight other than some bulk commodities being transported over long distances (such as coal on the West Moreton System), Queensland Rail faces intense and increasing competition from road operators. Road transportation offers an effective substitute service to rail, which has a significant and direct downward impact on the prices that Queensland Rail negotiates with access seekers.¹⁷⁰

Parties requiring freight transportation services can readily shift to moving freight by road rather than rail in the event of an increase in access price and/or decline in quality of services provided.¹⁷¹

Queensland Rail's argument is considered in detail below, with the assistance of the diagram in Figure 7.

¹⁷⁰ Queensland Rail, sub. 8, p. 1, para. 3; sub. 33, p. 22, para. 122.

¹⁷¹ Queensland Rail, sub. 8, p. 4, para. 25(a)(ii)(a).



Figure 7 Road and rail price composition on the North Coast corridor for the transport of containerised (intermodal) freight

Diagram for illustrative purposes only. Not drawn to scale.

The road price

Figure 7 depicts the prices charged by road and rail operators on the North Coast corridor for the transportation of containerised (intermodal) freight. The road price represents the final price paid by beneficial freight owners (or forwarders) to transport containerised freight by road¹⁷², for a particular transport task (origin–destination pair) on the North Coast corridor.¹⁷³ There is no 'below-road' price in Queensland—that is, trucks accessing major road networks in Queensland, such as the Bruce Highway, typically do not pay an access fee. Therefore, the road price is expected to reflect largely the costs of the trucking operator (for example, labour costs, vehicle costs and fuel costs, as well as a profit margin). The road freight industry in Australia is highly competitive¹⁷⁴, and as such, this road price may be the market price for the transport of that particular task by road on the North Coast corridor.¹⁷⁵

The rail price

The rail price represents the final price paid by beneficial freight owners (or forwarders) to transport containerised freight by rail, for the same particular transport task (origin–destination pair) on the North Coast Route.¹⁷⁶ The total rail price consists of two components—an above-rail component, which represents the charges of the above-rail operator, and a below-rail component, which represents the charges of the below-rail operator (Queensland Rail).

Queensland Rail's consultant, HoustonKemp, stated:

¹⁷² 'Containerised freight' refers to the type of goods that would be transported in containers by rail haulage (see section 5.4.3). The precise packaging of these goods for transport (e.g. the use of rail containers) may differ between road and rail transport.

¹⁷³ For example, from Brisbane to Mackay via the Bruce Highway.

¹⁷⁴ See for example, BITRE, *Road and rail freight: competitors or complements?,* information sheet 34, Australian Government, July 2009, p. 9.

¹⁷⁵ However, the precise dollar amount of the road price, or the nature of the market for the transport of containerised freight by road on the North Coast corridor, is not critical to support this analysis and is therefore not discussed further.

¹⁷⁶ For example, from Brisbane (Acacia Ridge) to Mackay via the North Coast Route.

Unlike other systems, Queensland Rail does not have a direct relationship with end customers on the North Coast Line.¹⁷⁷

The reference to 'end customers' appears to refer to the beneficial freight owners (or forwarders) who ultimately acquire rail transport services to transport their goods. In the QCA's analysis of the dependent above-rail freight haulage market, these 'end customers' are customers in the above-rail freight haulage market (see Figure 3).

The QCA understands that on the North Coast Route, above-rail freight haulage operators (such as Pacific National and Linfox) contract with Queensland Rail directly for access, and negotiate a below-rail access price.¹⁷⁸ Above-rail haulage operators then add their above-rail charges (for example, labour costs, rollingstock costs and fuel costs, as well as a profit margin) to the below-rail access price to produce a final total rail price, which is the price offered to the 'end customers', the beneficial freight owners (or forwarders).

The total rail price is represented by the blue bar labelled 'Rail price 1' in Figure 7. This bar has been split into a below-rail and above-rail component to represent the analysis above (note the diagram is for illustrative purposes only and does not seek to represent the actual share of the total price between the above-rail and below-rail operators).

Queensland Rail's ability and incentive to exercise market power is constrained with respect to customers in the above-rail haulage market (end customers) on the North Coast corridor

Based on the evidence submitted by stakeholders, the QCA considers that rail transport competes with road transport for the transport of containerised (intermodal) freight on the North Coast corridor, and competes primarily on price.

For this contestable freight task, the road price (\$X in Figure 7) acts as a constraint on the rail price. This is because these goods can technically be transported by either road or rail. Therefore, if the total rail price exceeds the road price, end customers (e.g. beneficial freight owners) can choose to transport their goods by road instead, and rail operators will lose market share to road operators for the transport of this freight.

As a result, the above-rail price and below-rail price are collectively constrained by the road price. It is open for the above-rail operator and below-rail operator to seek to increase their *share* of the total rail price; however, the absolute *amount* of this total rail price cannot rise above the constraint of the road price. If it does, end customers may switch from using rail transport to using road transport, and this will in the long run negatively affect both the above-rail and below-rail operators.

Queensland Rail emphasised this point:

The alternative offered by road means that end customers have alternative avenues open to them other than using Queensland Rail's systems. This provides end customers with significant countervailing power, as the option of road transport means that they can make a credible threat to withdraw from negotiations with Queensland Rail (or an above rail provider) and switch to using road if a competitive price and reasonable terms and conditions are not offered by Queensland Rail for the below rail service.¹⁷⁹

¹⁷⁷ Queensland Rail, sub. 33, attachment B, p. 21.

¹⁷⁸ Currently, under Queensland Rail's 2016 Access Undertaking, there are no reference tariffs that apply to services on the North Coast Line.

¹⁷⁹ Queensland Rail, sub. 33, p. 26, para. 139.

The QCA considers that Queensland Rail is constrained in its ability and incentive to exercise market power with respect to the *end customers* on the North Coast corridor by competition from road transport.

In the QCA's analysis of the dependent above-rail freight haulage market, these 'end customers' are the customers in that market (see Figure 3). This is because if Queensland Rail sought to raise the below-rail price, and the above-rail operator is assumed to pass on this increase in full¹⁸⁰, the total rail price may exceed the road price, and if it does, end customers may simply switch from rail transport to road transport to transport their goods.

Queensland Rail's ability and incentive to exercise market power is not constrained with respect to above-rail operators in the above-rail freight haulage market

The QCA considers that Queensland Rail *is not* constrained in its ability and incentive to exercise market power against the above-rail freight haulage operators on the North Coast Line by competition from road transport. These above-rail haulage operators are the suppliers in the dependent above-rail freight haulage market, but customers in the primary market for the service (see Figure 3).

Given that rail competes with road for the transport of containerised (intermodal) freight on the North Coast corridor, the above-rail operator and the below-rail operator together face a collective constraint on the total rail price that can be charged—that constraint is the road price for the equivalent transport task (origin–destination pair).

In a future without declaration, Queensland Rail is likely to seek to exercise its market power, where it is possible to do so, in order to maximise its profits. It may seek to exercise this market power by raising the below-rail access charge that it charges the above-rail operator on the North Coast Route, and seek to claim a larger share of the total rail price (this scenario is illustrated as 'Rail price 2' in Figure 7).

In this scenario, the above-rail operator is unlikely to be able to switch to using road infrastructure, in response to the increase in the below-rail access charge, as rollingstock cannot be converted for use on roads. The above-rail operator will have made significant sunk investments into long-life assets, such as locomotives and wagons, which are often specifically configured for use for the transport task on the North Coast Route (e.g. narrow gauge, specific tonne axle loads) and which are unlikely to be easily redeployed elsewhere. As Pacific National noted:

PN has sunk investment into long-lived infrastructure, the use of which relies on access to QR's below-rail infrastructure. Where customers require haulage along routes served by the QR infrastructure, and in order for PN to utilise its sunk assets, PN has no option but to acquire rail track network access from QR. This places QR in a position where it can exercise market power, and creates a hold-up risk ...

This is the case even where PN faces competition in haulage markets. Due to the sunk assetspecific and long-lived nature of its investment, there is no real prospect of PN switching away from the use of the QR infrastructure ... PN cannot credibly make a threat to withdraw from negotiations with QR, regardless of the degree of competition from road operators. To illustrate, consider the credibility of a threat by PN to stop using QR's below-rail infrastructure, and to redeploy its rolling stock to offer freight services by road – such a threat clearly would not be

¹⁸⁰ The situation where the above-rail operator 'absorbs' the below-rail price increase (i.e. does not pass on the below-rail price increase in full) is considered below.

credible, given the sunk, asset-specific and long-lived nature of PN's investment in rolling stock and other above-rail infrastructure.¹⁸¹

The QCA considers that in this case, if Queensland Rail raised the below-rail access charge, the above-rail operator cannot make a credible threat to exit the market or switch to road transport in response. Furthermore, the above-rail operator faces the collective constraint (the road price), and cannot simply pass through the below-rail access charge increase. If the total rail price exceeds the road price, the above-rail operator is likely to lose end customers (who may switch to road transport to transport their goods).

For an existing above-rail operator, arguably this scenario can be regarded as a transfer of wealth between the above-rail operator and the below-rail operator, with little direct impact on competition in the above-rail market. However, the QCA considers that the critical issue is that a potential above-rail operator seeking to enter the above-rail market, or an existing operator seeking to reinvest in the market, can foresee this risk—that any future (sunk) investments it makes may be exposed to the risk of expropriation by the below-rail operator in a future without declaration.¹⁸² Queensland Rail may have an incentive to solve this problem before the investment, however, Queensland Rail's ability to solve this 'hold-up' problem is limited (this is discussed in detail in section 5.6.3).

Ranbury also discussed this risk in its report on the North Coast Line capacity improvement study:

Likewise, a lack of strategy to invest in a meaningful way in the [North Coast] corridor [by Queensland Rail], will signal to Rail Operators as well as current and future rail freight customers that there is little incentive for them to invest in their business for a rail intermodal freight future. There is a high stranding risk in the above-rail business for the contestable freight market, which is not protected by the long-term take-or-pay contracting arrangements that apply in the bulk haul market. The contestable intermodal market involves only short term contracting arrangements with even the major customers, which is a significant impediment to investment in long life rail assets.¹⁸³

Therefore, the QCA does not consider that competition from road will be an effective constraint on Queensland Rail's ability and incentive to exercise market power against the above-rail operators in the above-rail freight haulage market. The QCA considers that above-rail operators in the above-rail freight haulage market may be exposed to the risk of hold-up in a future without declaration. The hold-up issue is discussed in detail in section 5.6.2.

5.5.3 Other constraints on Queensland Rail's ability and incentive to exercise market power

Queensland Rail argued (in addition to the issues discussed at sections 5.5.1 and 5.5.2 above) that it has no ability or incentive to exercise market power to adversely affect competition in any dependent market because it is materially constrained by a range of other factors. These include:

- Queensland Rail's statutory obligations and position as a statutory authority and recipient of Transport Services Contract (TSC) payments
- access arrangements

¹⁸¹ Pacific National, sub. 37, p. 10.

¹⁸² A discussion of the sunk nature of investments in rollingstock on the North Coast Route is in sections 5.4.4 and 5.6.2.

¹⁸³ Queensland Rail, sub. 33, attachment C, pp. 186–87.

- the threat of regulation or declaration
- dependent markets are already effectively competitive.

Each of these factors will be considered in turn.

Queensland Rail's statutory obligations and position as a statutory authority

Queensland Rail submitted that its statutory obligations, its position as a statutory authority, and its obligations under the TSC would constrain it in the provision of below-rail services:

As a statutory authority, Queensland Rail is one of the avenues through which the Queensland Government achieves its rail policy objectives. One such objective is to facilitate the efficient movement of freight through expanding the use of rail. Increasing access charges would be inconsistent with this objective.

In the event of conduct by Queensland Rail that compromised the Queensland Government's objectives (for example, limiting access to its network), the responsible Ministers have powers including in particular the powers under the QRTA Act [*Queensland Rail Transit Authority Act 2013* (Qld)] to control strategic and operational plans (with which Queensland Rail must comply); and issue written directions to Queensland Rail.¹⁸⁴

Pacific National disagreed, stating:

Ongoing regulation is required to ensure QR operates and provides access to its rail track network infrastructure efficiently – the disciplines currently imposed by regulation (such as non-discrimination and ringfencing) would not arise simply by virtue of QR's position as a statutory authority, the fact that some downstream customers may have more limited ability to pay, or the existence of spare capacity in some parts of the system.¹⁸⁵

The QCA recognises Queensland Rail's position as a statutory authority, and its obligations under the *Queensland Rail Transit Authority Act 2013* (Qld). However, the QCA also notes that under section 10 of the *Queensland Rail Transit Authority Act 2013* (Qld), Queensland Rail is required to carry out its functions as a commercial enterprise (except its community service obligations).

The QCA does not have visibility over the government rail policies and objectives to which Queensland Rail refers. Generally, the QCA notes that government policies are subject to change and ministerial powers may be discretionary. The QCA considers that Queensland Rail's general obligations under the statute are not an effective long-term constraint on its ability and incentive to exercise market power in a future without declaration.¹⁸⁶

Access arrangements

The QCA has given consideration to whether the access arrangements that are to apply in a future without declaration would constrain Queensland Rail's ability to exercise market power. In doing so, the QCA has given consideration to the access agreements applied by Queensland Rail, including its deed poll and access framework.

As discussed in section 5.4.5, Queensland Rail's access agreements are not evergreen in nature. That is, they do not provide access seekers with the option to continue to access the service

¹⁸⁴ Queensland Rail, sub. 8, p. 4, para. 25(a)(ii)(c); sub. 33, p. 32, paras 152–54.

¹⁸⁵ Pacific National, sub. 37, p. 14.

¹⁸⁶ In contrast, in relation to the Tablelands system, the QCA considered that specific provisions of the *Transport Infrastructure Act 1994* (Qld), which imposed specific obligations on Queensland Rail in relation to passenger services, were likely to be an effective constraint on Queensland Rail's ability to exercise market power in relation to the above-rail passenger market on that system (see section 9.5).

based on the terms of access set out in the initial agreements. Users are required to renegotiate the terms of access at the time contractual arrangements expire, generally under the same principles applicable to an access seeker acquiring a service for the first time (subject to specific provisions in the access framework). Additionally, due to the fact that negotiations between the parties are bilateral, Queensland Rail's behaviour during negotiations is not transparent to other access seekers.

As outlined in Part B, Chapter 4, the QCA considers that the access framework is not an effective constraint on Queensland Rail's ability to expropriate rents¹⁸⁷ from access seekers at the time of renegotiating access contracts. The QCA considers that, in a future without declaration, the risk of hold-up will adversely affect the environment for competition in dependent markets.

The threat of regulation or declaration

Queensland Rail considered that the threat of regulation, with its associated compliance and regulatory costs, would deter it from exercising any market power to hinder competition in dependent markets.¹⁸⁸ Queensland Rail submitted that in a future without declaration, it would be materially constrained in the provision of below-rail services by 'the threat of regulation or declaration under Parts 3 or 5 of the QCA Act'.

In contrast, Pacific National, Glencore and the South West Producers considered that the threat of declaration or regulation does not provide an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power in a future without declaration.

Pacific National stated:

In sectors where there is a threat of regulation only, monopoly service providers have been found to be exploiting their position undeterred by the threat of future intervention. The threat of regulation also has a very long lead time so is not enough in itself to constrain damaging behaviour in the short to medium term.¹⁸⁹

The South West Producers stated:

As the South West Producers continue to note, the threat of redeclaration is not a real one. That is because in this hypothetical scenario the QCA would have already determined during the current declaration review process that, despite QR having no effective constraints, the service should not be declared applying the very same declaration criteria which would apply in any future application for declaration.¹⁹⁰

While Queensland Rail did not specifically elaborate on how Part 3 of the QCA Act may apply to its activities, the South West Producers disagreed with Queensland Rail's submission, stating:

The potential application of a Part 3 price monitoring and pricing principles investigation is very similar to the IPART price monitoring regime about which the NCC stated in the Newcastle shipping channel revocation draft decision that while 'these requirements may provide some very limited constraint of PNO's pricing practices by promoting transparency' they were 'not a substitute for the type of access regulation contemplated by the National Access Regime' and were not a 'direct regulatory constraint that acts to set or limit the prices that PNO may charge'.¹⁹¹

¹⁸⁷ In the case of the above-rail freight haulage market, this may be the possibility of expropriation of the firms' sunk costs, instead of rents. However, the hold-up problem also arises in these markets.

¹⁸⁸ Queensland Rail, sub. 8, p. 4, para. 25(a)(ii)(f); sub. 33, pp. 32–33, para. 155.

¹⁸⁹ Pacific National, sub. 37, p. 12.

¹⁹⁰ South West Producers, sub. 40, p. 28.

¹⁹¹ South West Producers, sub. 40, pp. 28–29.

Glencore agreed with the South West Producers' submission.¹⁹²

Examples of contrasting decisions in considering the effectiveness of the threat of declaration in constraining the ability and incentive of a service provider to exercise market power were provided by Queensland Rail and Pacific National; these decisions were made in the context of the regulation of gas pipelines in Australia.¹⁹³ It appears that whether the threat of declaration is considered to be an effective constraint in these decisions depended upon a case-by-case assessment of the particular characteristics of the market under consideration and the behaviour of the entity in question.

The QCA notes that Part 3 and Part 5 of the QCA Act serve different purposes. The access regime in Part 5 of the QCA Act is a comprehensive framework specifically designed to facilitate access to monopoly infrastructure services (where the service meets the access criteria). In contrast, Part 3 of the QCA Act provides for the declaration of monopoly business activities and the investigation and reporting by the QCA about pricing practices relating to monopoly business activities.¹⁹⁴

The QCA does not consider that, on its own, the threat of declaration would constrain Queensland Rail from exercising market power.

If the threat of declaration could be relied upon to curtail the exercise of market power by Queensland Rail, the QCA considers that this would have manifested in Queensland Rail's deed poll and access framework, offered at a time when the threat of declaration is clear.

In relation to the service supplied by DBCT Management, the QCA has found that the response of DBCT Management to the present threat of declaration indicates that the threat of declaration is a factor impacting upon DBCT Management's conduct (Part C, section 3.3.5). Further, the QCA has found that the threat of declaration will provide a constraint when coupled with DBCT Management's deed poll and access framework, which allows the QCA to be satisfied that access as a result of declaration would not promote a material increase in competition in dependent markets (Part C, sections 3.3.7 and 3.4).

Central to this conclusion is the fact that DBCT Management, following the QCA's draft recommendation, modified its deed poll and access framework to enshrine the key elements of its pricing methodology in the deed poll (i.e. they cannot be modified during the term of the deed poll). This demonstrates a responsiveness to the threat of declaration that has caused the QCA to give greater weight to this factor in deciding whether criterion (a) is satisfied in relation to DBCT.

In contrast, Queensland Rail has maintained its pricing methodology in its access framework, where it can be unilaterally modified by Queensland Rail within the limits set out in the deed poll. In any event, the form of the deed poll and access framework are not an effective constraint on Queensland Rail's ability to exercise monopoly power (section 4.4). This indicates that the possibility of declaration cannot be relied upon to constrain Queensland Rail in its dealings with access seekers as effectively as for DBCT Management.

The QCA considers that the threat of declaration, on its own, would not mitigate the risk of 'hold-up' for access seekers at the time of contract renewal. The risk of hold-up for an access seeker at the time of contract renewal is unlikely to be alleviated by the knowledge that it can apply for declaration should Queensland Rail seek to propose price increases, within the

¹⁹² Glencore, sub. 41, pp. 19–20.

¹⁹³ See Queensland Rail, sub. 33, p. 32, footnote 113; Pacific National, sub. 37, p. 12, footnote 11.

¹⁹⁴ See s. 13A of the QCA Act.

boundaries set out in the access framework. The risk of hold-up for an access seeker remains, with the access seeker deterred from utilising Queensland Rail's services.¹⁹⁵

In the case of DBCT Management, the QCA has found that the threat of declaration combined with the deed poll and access framework is sufficient to constrain DBCT Management from exercising market power in a manner that would materially impact competitive conditions in a dependent market. The QCA is not similarly satisfied in relation to Queensland Rail, as the QCA has found that the DBCT Management deed poll and access framework deliver a degree of certainty around access charges which is lacking in the Queensland Rail deed poll and access framework.

Both of these conclusions involve judgments about the extent to which the service provider is likely to be constrained by the threat of declaration in the context of the access arrangements they have put in place, and the extent to which this can be expected to curtail its ability and incentive to exercise market power. Taking into account the differences between the deed polls and access frameworks offered by DBCT Management and Queensland Rail, and the circumstances in which they were provided, the QCA is satisfied that the threat of declaration can be relied upon to constrain DBCT Management more effectively than it can for Queensland Rail.

Dependent markets are already effectively competitive

Queensland Rail submitted that key relevant dependent markets are already competitive, and that it is well established that if a dependent market is effectively competitive, access as a result of declaration is unlikely to promote a material increase in competition:

As noted in paragraph 103 above, it is well accepted that criterion (a) has no application to a dependent market that is workably or effectively competitive ...

Although Queensland Rail does not consider that it is necessary to definitively define dependent markets for the purposes of criterion (a) in this instance, Queensland Rail notes that key relevant dependent markets are effectively competitive, and would be with and without declaration such that there can be no material increase in competition as a result of declaration. For example, it is accepted that coal is an internationally traded commodity with prices set by reference to international spot prices. Similarly, sugar is an internationally traded product, with the returns to sugar producers in Australia (predominantly in Queensland Rail's services cannot be said to have affected the level of competition in those markets to date and thus the accepted approach of concluding that criterion (a) cannot be met where dependent markets are effectively competitive should be applied.¹⁹⁶

In response to Queensland Rail's reference to global coal markets, the South West Producers said:

As discussed in their last submission, the South West Producers continue to consider it is clear that (consistent with market realities, and judicial and regulatory precedent discussed in detail in previous submissions) there are in fact a number of other dependent markets distinct from such coal markets [global thermal coal markets].

¹⁹⁵ Any impacts on investment decisions in dependent markets as a result of the risk of hold-up will materialise prior to Queensland Rail potentially engaging in conduct that warrants a declaration application.

¹⁹⁶ Queensland Rail, sub. 8, p. 5, paras 28, 32; sub. 33, pp. 46–47, paras 224–25.

Critically, the question which falls to be answered under criterion (a) is not whether declaration would promote a material increase in competition in the thermal coal market, but whether it would promote a material increase in competition in *any* dependent market.¹⁹⁷

The QCA notes that there may be a number of dependent markets in which access (or increased access) to the service as a result of declaration may not promote a material increase in competition. However, criterion (a) is satisfied if access (or increased access) to the service as a result of declaration would promote a material increase in competition in *at least one market*, other than the market for the service.

5.5.4 Conclusions on Queensland Rail's ability and incentive to exercise market power

The QCA considers that Queensland Rail has the ability and incentive to exercise market power in a way that may adversely affect competition in the dependent above-rail freight haulage market in a future without declaration.

This prompts the question of whether competition would be adversely affected if Queensland Rail exercised this market power. For criterion (a) to be satisfied, the QCA must be satisfied that access (or increased access), on reasonable terms and conditions, as a result of a declaration of the service would promote a material increase in competition in at least one dependent market.

Therefore, the following sections contain a detailed analysis of the state of competition in the above-rail freight haulage market in a future with and without declaration.

5.6 Competition in the above-rail freight haulage market in a future with and without declaration

5.6.1 A future with declaration

The QCA considers that a future with declaration will continue to provide for access to the rail network on reasonable terms and conditions due to ongoing regulatory oversight. The regulatory regime under Part 5 of the QCA Act is intended to take into account the interests of the access provider and access seekers/users, while providing an effective constraint on Queensland Rail's ability and incentive to exercise market power in a way that may adversely affect competition in dependent markets.

The QCA considers that the protections given by the regulatory framework are an effective long-term constraint on Queensland Rail's market power. In particular, the regulatory regime in a future with declaration would be able to provide assurance and certainty to access seekers and users that Queensland Rail's services will be provided on reasonable terms and conditions and there is a statutory process within which terms and conditions of access can be negotiated (including to address sunk investments). Additionally, declaration can maintain an appropriate balance between the legitimate interests of the service provider and access seekers/users in the presence of sunk investments, and mitigate the risk of hold-up for access seekers. Mitigating the risk of hold-up for access seekers is likely to be a critical factor in supporting efficient entry to and efficient participation in the above-rail freight haulage market, thereby materially promoting competition (the QCA's approach to the concept of materiality is discussed in section 5.6.4 below).¹⁹⁸

¹⁹⁷ South West Producers, sub. 40, p. 16.

¹⁹⁸ Efficient participation in the market includes actions undertaken by incumbent market participants, such as investing in operational efficiencies and innovations, as well as re-investment into the market at the time of contract renewal.

5.6.2 A future without declaration: the hold-up problem

Overview of the hold-up problem

As a business, Queensland Rail has an incentive to maximise profits. In a future with declaration, its ability and incentive to exercise its market power in order to maximise profits will be constrained by the regulatory regime. In a future without declaration, the QCA considers that Queensland Rail will not face any effective long-term constraints on its ability and incentive to exercise market power in order to maximise profits. In particular, the QCA considers that the access arrangements applied by Queensland Rail will not act as an effective constraint.

It is in this environment that market participants will face decisions to enter or operate in the above-rail freight haulage market on the North Coast Route in a future without declaration. In particular, a new entrant to the above-rail freight haulage market will have to incur significant sunk costs. These include investments in physical assets, such as long-life rollingstock assets (which typically have a 20- to 25-year life)¹⁹⁹ and the associated maintenance and provisioning facilities, as well as ongoing investments in innovation to improve the competitiveness of the service offering of the firm against other competitors in the market.²⁰⁰ Given the specific operational characteristics of the North Coast Route, such as a narrow-gauge track with specific tonne axle loads, such assets cannot readily be switched to alternative uses elsewhere. The presence of sunk, transaction-specific investments gives rise to the 'hold-up problem', commonly described in the economics literature on natural monopoly regulation:

The basic story is as follows: the users of a monopoly firm routinely have the opportunity to take some irreversible action which will significantly increase the value of or demand for the monopolist's products or services. The users or consumers, however, fear that once they have taken that action and incurred the associated sunk cost, the monopolist will engage in "ex post opportunism" – raising the price for the monopolist service, expropriating the additional benefit or value achieved. Fearing this expropriation, the users or consumers are reluctant to put themselves in a position where they can be exploited by the monopolist. As a result, they fail to take socially efficient actions, or they take other actions which are less socially beneficial, but with lower risk of expropriation. The failure to take efficient actions results in a material economic welfare loss.²⁰¹

The QCA has identified that the most likely circumstance for Queensland Rail to exercise market power after an investment has been made is at the contract renewal stage. Specifically, if an access seeker decides to enter (or to expand its operations in) the market, it will incur significant sunk costs through the need to invest in long-life rollingstock assets. The 20–25-year useful life of rollingstock can be contrasted with the typical length of a below-rail access agreement of around 10 years.²⁰² Therefore, at some point during the useful life of the rollingstock, it would be expected that the below-rail access agreement will be due for renewal. Below-rail access agreements with Queensland Rail have historically not contained evergreen

¹⁹⁹ See, for example, Glencore, sub. 5, p. 13; South West Producers, sub. 16, p. 14; QRC, sub. 7, p. 19; Watco, sub. 48, p. 4; GrainCorp, sub. 52, p. 6.

²⁰⁰ Pacific National (sub. 37, p. 5) gives the example of investments in innovations by competing above-rail operators, including the introduction of electronically controlled pneumatic braking, increased locomotive power, safer operation of over-length trains and innovation in the design and configuration of train consists.

²⁰¹ D Biggar, 'Is protecting sunk investments by consumers a key rationale for natural monopoly regulation?', *Review* of Network Economics, vol. 8, no. 2, 2009, p. 13.

²⁰² See for example, QRC, sub. 7, p. 19; Watco, sub. 49, p. 1.

renewal clauses (see section 5.4.5).²⁰³ This means that any terms contained in the original access agreement (entered into in the first period) may not necessarily be replicated in the new access agreement. Therefore, when the below-rail access agreement is due for renewal, in subsequent periods after the above-rail operator has entered the market, the above-rail operator would be in a less favourable bargaining position relative to Queensland Rail, as it has made significant sunk investments in rollingstock assets that are not readily used elsewhere.²⁰⁴

In these subsequent periods, an exercise of market power by Queensland Rail against an aboverail operator may arguably be regarded as a transfer of wealth between the parties, with little impact on competition. However, the QCA considers that the critical issue is that in the first period, the above-rail operator can foresee this risk that any sunk investments it makes in the first period will be exposed to the risk of expropriation by the monopolist in subsequent periods. The QCA considers that this risk is sufficiently material that an efficient potential entrant will likely be deterred from entering the market in the first place. In this context, the Queensland Resources Council stated:

[O]ne needs to understand that, in order to produce or extract a commodity like coal, this requires a major sunk investment in mining equipment and infrastructure. These sunk investments give rise to what are known as "quasi-rent" which are subject to the threat of hold-up. The threat of expropriation of rents by a monopoly service provider in such a situation would only in extreme circumstances result in a pure transfer. More likely, even the threat of such expropriation can limit future investment and innovation by the upstream firms'.²⁰⁵

The prospect of Queensland Rail exercising its market power in future contracting periods creates a significant degree of uncertainty for potential market participants at the time they are considering investment, raising the hurdle rate required to justify the investment and potentially deterring efficient entry, investment and participation in the market. Furthermore, all market participants are exposed to this risk in a future without declaration: incumbent operators also face increased risk and uncertainty at the time of their contract renewals, due to the absence of evergreen renewal rights. This may undermine incentives for future efficient actions by those operators compared to the situation with declaration.²⁰⁶ These circumstances impact the environment in which access seekers will make an investment.

The presence of this risk of hold-up means that socially optimal investments will not proceed, or there will be an underinvestment. Queensland Rail may have an incentive to solve this hold-up problem ex ante—for example, it may be profit maximising for Queensland Rail to sell unused network capacity to new or renewing users, assuming it is not constrained to charging a uniform price.²⁰⁷ However, as will be discussed below, the QCA's view is that it will be difficult for Queensland Rail to credibly commit ex ante to solve the hold-up problem (for example through

²⁰³ In the absence of information to the contrary, the QCA has proceeded on the basis that this practice is likely to continue in the future, that is, that below-rail access agreements in the future are likely to also not contain evergreen renewal clauses.

²⁰⁴ An argument may be made that Queensland Rail would also have made sunk investments, for example in improvements or expansions to the below-rail infrastructure to accommodate the above-rail operator's needs. However, the QCA considers that Queensland Rail may be able to offer use of its below-rail infrastructure to another above-rail operator (e.g. a competitor operator), whereas the above-rail operator may face greater difficulties in attempting to find an alternative use for its narrow-gauge rollingstock.

²⁰⁵ QRC, sub. 20, p. 14.

²⁰⁶ Pacific National, sub. 9, pp. 6–7.

²⁰⁷ Under the 2016 access undertaking, Queensland Rail is not required to charge a uniform price on the North Coast Line or any other system except for West Moreton, where there is a reference tariff for coal trains. On the non-West Moreton systems, prices are negotiated between Queensland Rail and the customers seeking below-rail access.

a long-term contract). For instance, events could develop in the future where the benefits to Queensland Rail of expropriating the value of the investment at that later time exceeds the benefits of continuing to abide by status quo arrangements. The QCA considers that it is this risk—that significant sunk investments in rollingstock made by the above-rail operator will be expropriated—that will lead to a material adverse effect on competition in the above-rail haulage market in a future without declaration.

In contrast, the QCA considers that the access regime that would apply in a future with declaration is an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power. Access (or increased access), on reasonable terms and conditions, as a result of declaration would materially improve the environment for competition by encouraging efficient entry and actions (through a stable and predictable environment²⁰⁸). Such an environment would in turn promote a material increase in competition in the above-rail freight haulage market on the North Coast Route (the QCA's approach to materiality is discussed below in section 5.6.4).

A further discussion of the economic theory of hold-up is in Appendix A.

Stakeholder submissions

Users of the Queensland Rail service generally supported the QCA's analysis in the draft recommendation of the hold-up problem. In relation to the North Coast Route service, Pacific National stated:

QR's submissions on criterion (a) largely ignore the risk of hold-up due to the large, long-lived and sunk nature of users' investments. This is a key reason why ongoing declaration and regulation of access will promote competition ...

The risk of hold-up is illustrated by the circumstances facing PN. PN has invested well over \$1 billion in above rail infrastructure in Queensland. Much of this infrastructure has very long lives and only recently deployed ...

Rail investment is largely sunk, most of the infrastructure could not be deployed elsewhere if PN were to cease operating in Queensland. This is primarily because rollingstock is configured for electric infrastructure (which is unique to Queensland) and a narrow gauge track which is not widely deployed in Australia. Consequently, PN is in a very weak bargaining position when seeking to re-negotiate the terms of access with a monopoly supplier such as QR.²⁰⁹

Queensland Rail disagreed with the QCA's analysis in the draft recommendation of the hold-up problem, and argued that 'there is no two-period hold-up problem arising in respect of Queensland Rail's services provided using the North Coast Line, Mount Isa Line and West Moreton System'.²¹⁰ These submissions are considered in detail below.

5.6.3 Queensland Rail's submission that there is no hold-up problem

In response to the QCA's analysis of the hold-up problem in the draft recommendation, Queensland Rail argued that there is no hold-up problem arising in respect of Queensland Rail's services:

The HoustonKemp Expert Report describes why there is no two-period hold-up problem arising in respect of Queensland Rail's services provided using the North Coast Line, Mount Isa Line and West Moreton System. The reasons include that:

²⁰⁸ For example, the access regime in the QCA Act is transparent and can only be modified by parliament.

²⁰⁹ Pacific National, sub. 37, pp. 8–9.

²¹⁰ Queensland Rail, sub. 33, pp. 38–39, paras 190–92.

- (1) users can (and do) negotiate longer term contracts where required; and
- (2) Queensland Rail does not have an incentive to take advantage of users in the second round of negotiations given Queensland Rail is involved in multiple rounds of negotiations with multiple users. Taking advantage of users in second round negotiations would give rise to reputational damage, which would in turn erode Queensland Rail's financial viability.

As outlined by HoustonKemp, if the two-period hold-up problem described by the QCA did exist, it would arise under the current regulatory arrangements. Given the revenues resulting from Queensland Rail's access prices are well below the ceiling limits, Queensland Rail could, if it had the ability and incentive to do so, increase prices in the second round under the current regulatory arrangements. It does not do so.²¹¹

The HoustonKemp report discussed the points raised by Queensland Rail under three categories:

In summary:

- contracting is a solution to the hold-up problem Queensland Rail has a strong financial incentive to negotiate contracts that are acceptable to access seekers;
- Queensland Rail is in a 'multi-round' negotiation extorting access seekers/holders would likely damage Queensland Rail's reputation, thereby reducing its long run financial viability; and
- even if the hold-up problem does exist, existing regulation provides limited protection to consumers and it is not clear if price regulation is an effective or efficient solution.²¹²

The QCA considers each of these points below.

Contractual solutions to hold-up

Queensland Rail submissions

HoustonKemp argued that there is a mutual incentive for Queensland Rail and access seekers to agree to a contract:

One obvious solution to the hold-up problem would be to sign contracts with terms and conditions that meet the needs of the access seeker, e.g. duration of the contract and options for renewal.

As discussed above, Queensland Rail has a strong incentive to maximise throughput, and so increase the volume on its network, given that it has spare capacity. Put another way, Queensland Rail has an incentive to avoid the hold-up problem. Similarly, access seekers that believe there could be a hold-up problem also have an incentive to mitigate the hold-up problem.

It follows that there is a mutual incentive for Queensland Rail and the access holder to negotiate an access agreement that is acceptable to both parties, regardless of Queensland Rail's declaration status.²¹³

Long-term contracts

Queensland Rail argued that, in a future without declaration, the hold-up problem could be alleviated by the signing of long-term contracts between itself and an access seeker. For example, given that the useful life of rollingstock is typically 20 years (or more), an above-rail operator could enter into a 20-year contract, which matches the useful life of the rollingstock.

²¹¹ Queensland Rail, sub. 33, p. 39, paras 191–92.

²¹² Queensland Rail, sub. 33, attachment B, p. 14.

²¹³ Queensland Rail, sub. 33, attachment B, p. 13.

The QCA considers that while such contracts may be suitable in theory, long-term contracts (i.e. with terms that match the asset lives) are not commonly observed in practice, as it is difficult to write a complete contract for such a long period of time. For example, Biggar noted:

Finally, a monopolist might seek to prevent the hold-up problem by committing itself through a long-term contract ... But long-term contracts have their own problems. To begin with, negotiating a long-term contract is costly, so the transactions costs are high, particularly when there are a large number of buyers. In the long-run, the costs and demand facing the monopolist may vary significantly, according to factors which cannot be foreseen at the time the contract was signed. It is impossible to negotiate and specify actions to be taken in every possible future contingency – long-term contracts are inevitably incomplete.²¹⁴

In addition, Crocker and Masten noted:

Although parties will design contracts to balance the need for adaptation with the cost of effecting adjustments, the ability to define precise obligations in response to changing events in ways that can be enforced at low cost means that contracts will, on the one hand, tend to be inflexible and, on the other, leave considerable opportunity to cheat on the agreement or to attempt to evade performance.²¹⁵

The QCA considers that in practice, long-term contracts (of the type described above) are unlikely to be an effective solution to the hold-up problem due to their incompleteness and complexity. Specifically, complete contracting is unlikely, due to the inherent uncertainty about the circumstances that may prevail after the contract is entered into. In a world with uncertainty, writing a complete contract would require extensive consideration of all relevant contingencies, as well as detailed negotiations with the transacting party about the distribution of value for each contingency. As a result, writing such a complete contract would likely be prohibitively costly.²¹⁶ Moreover, due to the very low probability of most contingencies, such an exercise would likely not be cost-effective.

For example, the Australian Freight Rail Operators' Group, in a submission to the Productivity Commission, noted the implications of long-term contracts for an above-rail operator in seeking access to rail infrastructure:

One option would be to lock in infrastructure charges through a very long-term contract. However, there are a number of obvious problems with this approach. To begin with, devising such a contract when the long term prospects for rail are so uncertain poses substantial difficulties. Additionally, the above-rail operator entering into such a contract would be exposed to the risk that better terms would be offered to its competitors in later periods, a risk that MFN [most favoured nation] clauses can help address, but not fully offset. Finally, the experience of above-rail operators is that it is difficult to secure such contracts on reasonable terms with government-owned entities.²¹⁷

In addition, the South West Producers noted:

Rail access with long term take or pay components is not something coal producers can commit to for a term that would prevent the hold-up problem – which would effectively need to be a

²¹⁴ D Biggar, 'Is protecting sunk investments by consumers a key rationale for natural monopoly regulation?', *Review* of Network Economics, vol. 8, no. 2, 2009, p. 19.

²¹⁵ Crocker, KJ & Masten, SE, 'Regulation and administered contracts revisited: Lessons from transaction-cost economics for public utility regulation', *Journal of Regulatory Economics*, vol. 9, no. 1, 1996, pp. 5–39.

²¹⁶ If a contract is 'complete', then it will never require revision, and enforcement is always possible. In particular, the contract specifies all possible contingencies. As a complete contract contains no gaps or missing contingencies, it would perfectly govern the exchange between parties as circumstances unfolded over time.

²¹⁷ Freight Rail Operators' Group, submission to the Productivity Commission, *Road and Rail Freight Infrastructure Pricing*, 1 November 2006, accessed 9 August 2019,

https://www.pc.gov.au/inquiries/completed/freight/submissions/freight_rail_operators_group_/subdd086.pdf.

whole of mine life contract. Issues as varied as natural disasters, thermal coal prices, foreign exchange rates, Australia or international climate policies, trade or tariff policies of key Asian export destinations, Australian environmental, taxation, royalties or industrial relations mean that contracts for more than 10 year terms impose significant risk on the coal producer either:

- (a) having a shorter than anticipated mine life so that the producer would have a substantial 'take or pay tail' of liability trailing after any decision to cease production; or
- (b) having a longer than anticipated mine life so that the producer would be exposed to the two period economic hold-up problem anyway despite trying to resolve it with a longer term contract.

QR also has no incentives to provide renewal rights (other than at significant cost to the user) as by doing so it is thereby quarantining capacity which could otherwise be contracted by other users.²¹⁸

The QCA considers that a long-term contract could not anticipate every possible contingency, nor is it necessarily efficient for it to do so. Fundamentally, there is a tension between devising a contract structure that enables parties to adapt to uncertainties that eventuate in order to realise potential gains ('flexibility'), and devising one that at the same time reduces the scope for rent-dissipating efforts to redistribute existing value ('opportunism'). The longer the contract required, as in the case with long-lived sunk investments, the greater the need to allow for adaptation and adjustment in the face of changing market conditions. Relevantly, in commercial practice, there are few examples of long-term contracts being agreed between suppliers and users, and these tend to reflect special circumstances.²¹⁹ In particular, it is difficult to entirely eliminate the need for contract renegotiation in the context of a very long-term contract.

In the current situation, if the agreed (incomplete) contract needs to be renegotiated at a time in the future without declaration, the QCA considers that Queensland Rail will have the superior bargaining position, and the above-rail operator may be exposed to the risk of expropriation. While Queensland Rail could choose not to exercise this bargaining power ex post, it does not seem possible for Queensland Rail to credibly commit ex ante that it will not do so at that later time. For example, it does not seem that Queensland Rail would be able to credibly commit ex ante (e.g. at the time of signing a 20-year contract) that firstly, such a long-term contract will never need to be renegotiated during its term²²⁰; and secondly, that if such a renegotiation occurs, it would not expropriate investment value from the other negotiating party at that future time.

The problem is that events could develop in the future where the benefits to Queensland Rail of expropriating the value of an investment at that later time exceed the benefits of continuing to abide by the status quo arrangements. The QCA considers that it is this risk—that significant sunk investments in rollingstock made by the above-rail operator will be expropriated—that will lead to a material adverse effect on competition in the above-rail freight haulage market in a future without declaration. For example, existing customers or potential entrants into a market might either delay, or forgo, new investment that would otherwise be economically efficient.

²¹⁸ South West Producers, sub. 40, pp. 21–22.

²¹⁹ Coal supply contracts between mine-mouth electric utilities in the United States and adjacent coal mines tend to have long terms (e.g. 35 years), but they are complex contracts and some of them have renegotiation provisions. See Joskow, P, 'Vertical integration and long-term contracts: the case of coal-burning electric generating plants', *Journal of Law, Economics and Organization*, vol. 1, no. 1, 1985, pp. 33–80.

²²⁰ Noting that renegotiation may occur at the request of either, or both parties.

Additionally, stakeholders might expend considerable time and costs engaging in otherwise wasteful and inefficient activities in trying to protect their share of investment value.

In a future with declaration, the supporting regulatory structure would enable independent regulatory oversight in relation to material price and non-price terms. The QCA considers that this oversight would be sufficient to constrain Queensland Rail's ability and incentive to exercise market power. These protections offered in a future with declaration would materially improve the environment for competition by encouraging efficient entry and actions (through a stable and predictable environment), which would in turn promote a material increase in competition in the above-rail freight haulage market on the North Coast Route.

A further discussion of the economic theory of long-term contracts in the context of the hold-up issue is in Appendix A.

Reputational damage

Queensland Rail submissions

HoustonKemp argued that Queensland Rail does not have an incentive to take advantage of market power in the subsequent rounds of negotiations due to the risk of reputational damage:

The notion that expropriating sunk costs can damage an access provider financially is supported by the Productivity Commission (PC) in its inquiry report into the Electricity Network Regulatory Framework ...

In other words, by developing a 'bad brand image' through expropriating the sunk costs of small users, a network business risks losing future revenue. As a result, it is in the best interests of a network business to maintain a 'positive image' and not seek to expropriate sunk costs.

This logic can be applied to the situation faced by Queensland Rail. If Queensland Rail chose to expropriate the sunk costs of access seekers or access holders, its reputation would be damaged and thus future access seekers or users would be less willing to sign a contract with Queensland Rail. Given the long-lived nature of Queensland Rail's network, this could result in stranded assets where access revenue no longer covers the incremental cost of keeping the system open.²²¹

In contrast, Pacific National argued that:

PN does not consider that a monopolist should be left unregulated, on the basis that it might be expected to self-regulate its behaviour to avoid reputational damage. Economic theory and experience tell us the risk of reputational damage will not constrain a monopolist's behaviour. Rather, a monopolist will act on its incentive and ability to exploit its market power.²²²

HoustonKemp contended that in a future without declaration, Queensland Rail's incentives to exercise its market power to hold up its customers would be constrained by potential, adverse reputational effects. Given Queensland Rail would likely be entering into negotiations with multiple access seekers over time, such negotiating tactics would damage its long-run financial viability. The flow-on effects would likely undermine the investment incentives of existing users and reduce the entry of new users.²²³

The QCA's view is that there are two broad issues to be discussed in considering the effectiveness of the reputational mechanism (i.e. a firm's desire to maintain a good reputation) in constraining Queensland Rail's ability and incentive to exercise market power. These issues are:

²²¹ Queensland Rail, sub. 33, attachment B, p. 14.

²²² Pacific National, sub. 37, p. 12.

²²³ Queensland Rail, sub. 33, attachment B, p. 13.

- a firm's incentive to maintain a good reputation depends on a cost-benefit analysis—this analysis is a dynamic exercise, which may yield different incentives over time
- the operation of the reputation mechanism, which depends on the availability of information.

The benefits and costs of maintaining a good reputation

At the time of renewal of an access agreement in a future without declaration, Queensland Rail may face an opportunity to maximise its profits by exercising its market power to expropriate the value of a customer's investment. As with many commercial decisions, whether such a decision will be made is likely to depend on an analysis of the benefits of undertaking the expropriation compared to the costs of doing so.

The risk of reputational damage is one of a range of costs that a firm is likely to consider in making a decision.²²⁴ In general, the cost of reputational damage may be characterised as consisting of two parts:

- *future losses from termination of the specific relationship*—given the transaction-specific capital in a relationship, terminating the relationship means the potential loss of the discounted value of the (future) quasi-rents from the investments already made²²⁵
- future losses from general foregone business due to reputational damage—the transactor perpetrating hold-up will confront higher costs of doing business in the future as future transacting partners will demand more favourable and explicit contract terms, or a future transactor may not enter the market altogether.²²⁶

A firm would consider these costs in the context of the benefits that may be gained from undertaking an expropriation (e.g. the additional profits to be gained). If the overall benefits of expropriating the investment exceed the overall costs at that time, including the costs of reputational damage, then the firm would have an incentive to expropriate value from the investment.

Moreover, this cost-benefit analysis is a dynamic exercise. Over time, as long as the expected costs of the hold-up (including the costs of reputational damage) exceed the expected benefits (e.g. profits), Queensland Rail is unlikely to seek to expropriate a user's investment. However, changes in market conditions and firm-specific factors may arise in the future, which could cause the expected benefits of hold-up to exceed its costs (including the costs of reputational damage). Expropriation of the customer's investments may then be the appropriate business decision.

In particular, in the case of the above-rail operators accessing Queensland Rail's service, the significant sunk investment in long-life rollingstock means that their commercial relationship with Queensland Rail is likely to be long-term (e.g. 20 years or more). Such long time frames result in inherent uncertainties over future developments, and increases the chance that circumstances will develop that may result in the benefits of hold-up for Queensland Rail exceeding its costs. Given the long time frames involved in the above-rail freight haulage market, the QCA considers that Queensland Rail cannot credibly commit ex ante that the risk of

²²⁴ Other costs may include, for example, legal costs associated with settling a dispute raised by the customer.

²²⁵ That is, the customer whose investment is being expropriated may decide to exit the market and terminate the relationship altogether. Any future benefits from this specific relationship are thus foregone.

²²⁶ That is, future customers may be reluctant to enter the market if they see that other customers were held up in the past.

damage to its reputation will always serve as a sufficient constraint on its ability to exercise market power to expropriate any investment ex post—that is, that the costs of reputational damage will always be greater than the benefits.

Events that are not foreseeable at present may develop in the future, so that the benefits from expropriation exceed the costs (including the costs of reputational damage). The QCA considers that it is this risk that significant sunk investments in rollingstock made by the above-rail operator will be expropriated—a risk foreseeable to potential market entrants—that will lead to a material adverse effect on competition in the above-rail freight haulage market in a future without declaration.

In contrast, in a future with declaration, the QCA considers that the supporting regulatory structure would be sufficient to constrain Queensland Rail's ability and incentive to exercise market power to expropriate the value of a customer's investment. In turn, this will provide assurance to customers seeking to enter (or re-invest in) the market that their investment will not be exposed to the risk of hold-up, thereby materially improving the environment for competition by encouraging efficient market entry and investment.

The availability of information

A well-functioning and effective reputation mechanism depends on sufficient and available information on the firm's performance. Such mechanisms are most commonly associated with markets that feature repeated and frequent transactions. However, transactions between an above-rail operator and Queensland Rail (i.e. the signing of a contract for access) are infrequent and are spread across time. As a result, the 'repeat' purchase feature that is characteristic of markets with effective reputation mechanisms is not present.

Additionally, given these transactions are not contemporaneous (i.e. it is rare that two aboverail operators would be seeking to sign/renew their contract with Queensland Rail at the same time), it is less likely that firms learn from other firms about 'bad behaviour' on the part of Queensland Rail.

In a future without declaration, access seekers are unlikely to have transparency of certain access terms provided to other access seekers, including pricing terms. Rather, Queensland Rail would bilaterally negotiate with individual customers within the terms of the access framework. The QCA's view is that in such circumstances it would not be in the commercial interests of Queensland Rail to share information, particularly if it has exercised or is intending to exercise market power against a customer, as to share such information would be damaging to its reputation. Indeed, it may be that the publication of certain information, such as commercial terms, is contractually prohibited. In circumstances where there is limited availability of information, the QCA considers that the threat of reputational damage is not an effective constraint on Queensland Rail's ability and incentive to exercise market power.

In any case, notwithstanding the availability of information in a future without declaration, the QCA nevertheless considers that the cost-benefit analysis outlined above applies. That is, circumstances could arise where the cost-benefit analysis results in Queensland Rail deciding to expropriate the investment of a customer, whether information-sharing was available (where the act will result in reputational damage) or not (where the act will not necessarily result in reputational damage).

In contrast, the regulatory regime under Part 5 of the QCA Act provides an effective negotiate– arbitrate framework that requires and facilitates the sharing of information. Additionally, this regulatory regime would be sufficient to constrain Queensland Rail's ability and incentive to exercise market power to expropriate the value of a customer's investment. The QCA considers that these protections and certainties of access offered in a future with declaration would materially improve the environment for competition by encouraging efficient entry and actions (through a stable and predictable environment), which would in turn promote a material increase in competition in the above-rail freight haulage market on the North Coast Route.

A further discussion of the economic theory of reputation effects in the context of the hold-up issue is in Appendix A.

Existing regulation provides limited protections against hold-up

Queensland Rail submissions

Queensland Rail's consultant HoustonKemp argued that in any case, existing regulatory arrangements provide limited protection against the hold-up problem:

We also note that existing regulatory arrangements provide limited protection against the holdup problem, since Queensland Rail is currently allowed to increase prices in the 'second round' because its revenue is below the regulatory ceiling limit. Put another way, Queensland Rail can already impose significant increases in access charges under current arrangements, since the revenue it currently collects is far below the cost of providing rail services.

The PC inquiry [Productivity Commission inquiry into the Electricity Network Regulatory Framework] points out that economic regulation is not a definitively effective means to prevent the hold-up problem ... Put another way, even if there is a hold-up problem, it is not clear that price regulation would be an efficient or effective means of resolving it.²²⁷

Regulation as the second-best solution

The QCA notes that no revenue ceiling has been set, or verified by the QCA, for Queensland Rail's railway systems, with the exception of the West Moreton system.²²⁸ In a future with declaration, there is no requirement under the QCA Act for there to be an approved reference tariff (or ceiling price) in relation to any of Queensland Rail's rail systems. Under the 2016 access undertaking, only coal users accessing the West Moreton system and Metropolitan system are subject to a reference tariff approved by the QCA. Users on all other systems, including the North Coast Line, gain access under the negotiate–arbitrate regime in the QCA Act, where access prices and non-price terms are negotiated with Queensland Rail. Thus, the existing regulatory arrangements do not rely solely on ceiling limits to constrain Queensland Rail's ability and incentive to exercise market power.

The QCA considers that access seekers will be in a less favourable negotiating position with Queensland Rail at the time of renegotiating access contracts in a future without declaration. That is, an access seeker will be susceptible to hold-up after it has committed to entering the market and has incurred considerable sunk costs.

The QCA Act expressly provides a means of addressing this negotiating power imbalance and includes overarching obligations, established by Part 5 of the QCA Act, which apply to Queensland Rail in a future with declaration. Relevantly, the presence of regulation does not necessarily mean direct intervention by the QCA to set prices. Rather, it provides an incentive for Queensland Rail to offer reasonable terms and conditions in order to avoid the process of arbitration, and it seeks to more appropriately balance the bargaining power of access seekers with the bargaining power of Queensland Rail, through the requirements for Queensland Rail to provide information and follow transparent processes. Moreover, it is open for the QCA to approve an access undertaking that outlines terms of access to address any negotiating power

²²⁷ Queensland Rail, sub. 33, attachment B, p. 14.

²²⁸ Queensland Rail, sub. 33, attachment B, p. 10, figure 3.1.

imbalance that exists between Queensland Rail and access seekers. Additionally, where an access seeker and Queensland Rail cannot agree on terms, either party may have recourse to a transparent arbitration regime established by the QCA Act. The QCA considers that the negotiate–arbitrate regime in the QCA Act is able to credibly constrain Queensland Rail's ability and incentive to exercise market power in a future with declaration, and thus support efficient entry to and efficient participation in the above-rail freight haulage market, thereby materially promoting competition.

The QCA analysis of the benefits of declaration is supported by stakeholders. For example, Glencore stated:

By contrast, Glencore strongly supports the QCA's view that a third party access regime under Part 5 of the QCA Act would, in a future with declaration, provide a credible constraint on QR's [Queensland Rail's] use of market power.

While there is no tariff for [the] Mount Isa Rail Access Service, that does not mean that declaration provides users of that service with no pricing protection (as QR and Houston Kemp appear to assert).

To the contrary:

- (a) access seekers have a right to have the QCA arbitrate access disputes (including as to pricing); and
- (b) at each access undertaking renewal, users of the Mount Isa Rail Access [Service] have the potential to seek a reference tariff. Although Glencore has never sought a reference tariff, it has considered doing so in the past and strongly considers that the mere presence of that option provides a constraint on QR's behaviour.

The right to have the QCA arbitrate access dispute[s] means that users have certainty that they have a right to obtain a reasonable and appropriate price where negotiations fail.²²⁹

The South West Producers agreed with Glencore, and said:

Similarly, through the undertaking process, the terms of access QR offers (in its standard access agreement), and terms of the undertaking have become more favourable and appropriate than those proposed by QR. They provide a guaranteed reasonable position for obtaining access to all new entrants.

Even for non-reference services, or disputes over non-pricing terms, the ability to have the QCA arbitrate access disputes is a critical constraint that removes the potential for exercise of monopoly power.²³⁰

Similarly, Watco stated:

Declaration promotes long term access certainty on fair and reasonable terms, a right to have QCA arbitrate access disputes, efficient access pricing and the potential to have reference tariffs, which all promote competition in dependent markets, particularly in regional Queensland agricultural freight markets and above rail operations.²³¹

The QCA considers that regulation is a second-best solution in circumstances where the conditions for effective competition are absent. Queensland Rail's status as a natural monopoly means that there are currently no credible market-based constraints on its ability and incentive to exercise market power. In these circumstances, appropriate regulation is needed to act as a constraint on its market power, in relation to both pricing and non-pricing terms.

²²⁹ Glencore, sub. 41, p. 24.

²³⁰ South West Producers, sub. 40, p. 39.

²³¹ Watco, sub. 48, p. 4.

HoustonKemp also noted that the Productivity Commission concluded that economic regulation is not a definitively effective means of preventing the hold-up problem in its 2013 report into electricity network regulation. The Productivity Commission suggested alternative solutions to the hold-up problem: ex post action based on common law breaches of the duty of good faith and fair dealing; specific competition laws; and ex ante oversight of particular long-term contracts that risk anti-competitive outcomes.²³²

The first two options appear to rely on ex post legal remedies. While these alternatives might provide some relief ex post, the problem here is that the 'damage is already done' with respect to dynamic efficiency. That is, once hold-up occurs, any adverse impacts on dynamic efficiency (e.g. delayed or foregone investment) would already have been realised. Therefore, these two options do not satisfactorily address possible dynamic efficiency concerns. The third option involving ex ante oversight of long-term contracts presumably involves (at least) regulatory monitoring of some type. However, it is not clear how a regulator would have visibility of private contracts in order to fulfil an effective monitoring role.

The QCA's view is that declaration, and the operation of the associated regulatory regime, is likely to be the more effective way to address hold-up concerns in Queensland Rail's circumstances.

5.6.4 Promote a material increase in competition

Materiality

Criterion (a) requires that access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a *material increase* in competition in at least one market (emphasis added). In Queensland, the words 'material increase' were first introduced into criterion (a) by the *Motor Accident Insurance and Other Legislation Amendment Act 2010* (Qld). The explanatory notes to that Act state that the purpose of the amendment to criterion (a) was to:

amend section 76(2)(a) to clarify that access (or increased access) to the service should be expected to promote a material increase in competition in order for this criterion to be satisfied. This will prevent the declaration of services where only a trivial increase in competition is expected to result ...

Therefore, the QCA considers that the threshold of materiality will require a non-trivial increase in competition.

The QCA considers that the concept of promoting a material increase in competition involves an improvement in the opportunities and environment for competition, such that competitive outcomes are materially more likely to occur in a future with declaration, compared to a future without declaration. Promoting a material increase in competition is not necessarily equivalent to promoting the greatest number of competitors in the market—strong competition may exist between a few firms. Rather, it involves the possibility that efficient entry and efficient participation by firms would be promoted in a future with declaration, compared to a future without declaration. If efficient entry is likely to be promoted in a future with declaration (compared to a future without declaration), the QCA considers that this would indicate that access as a result of declaration would promote an increase in competition that is material. A

 ²³² Productivity Commission, *Electricity Network Regulatory Frameworks*, inquiry report no. 62, 2013, Appendix B, p. 6, https://www.pc.gov.au/inquiries/completed/electricity/report/27-electricity-appendixb.pdf.
detailed discussion of the literature and jurisprudence of criterion (a), as well as the QCA's approach to criterion (a), is in Overview—Chapter 2.

In the case of the North Coast Route service, the QCA considers that access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a material increase in competition in the dependent above-rail freight haulage market.

A decision to enter (or re-invest in) the above-rail freight haulage market will involve substantial sunk investments. In a future without declaration, the presence of sunk investments gives rise to the hold-up problem. The QCA considers that the risk of hold-up in the presence of substantial sunk investments is sufficiently material that it is likely to discourage efficient firms from entering the market. In contrast, declaration and the associated access regime are able to credibly constrain Queensland Rail's ability and incentive to exercise market power and credibly address the hold-up risk.

The QCA considers that the credible constraint on the risk of hold-up in the presence of substantial sunk investments will promote a non-trivial, material improvement in the environment for competition in the above-rail freight haulage market on the North Coast Route. The environment for competition in a future with declaration is likely to promote efficient entry (and efficient investment) by all market participants, such that competitive outcomes in the above-rail freight haulage market are materially more likely to occur.

If efficient entry is likely to be promoted in a future with declaration (compared to a future without declaration), this would indicate that access as a result of declaration would promote an increase in competition that is material. In this way, the QCA is satisfied that access as a result of declaration would promote a material increase in competition in the above-rail freight haulage market on the North Coast Route.

Conclusion on the hold-up problem in the above-rail freight haulage market on the North Coast Route

In the case of the North Coast Route service, the QCA considers that the uncertainties facing market participants in a future without declaration would affect all participants across the market, including more efficient firms. Conversely, the certainties and protections offered by the access regime in a future with declaration would promote efficient entry and efficient participation in the dependent above-rail freight haulage market.

Several stakeholders made submissions to the effect that participants in the above-rail freight haulage market would face considerable uncertainties as to the terms of access in a future without declaration.²³³ For instance, Pacific National warned that in a future without declaration:

Queensland Rail would be incentivised to remove any pricing transparency around the cost of access to the network, increase access charges and reduce network service performance standards in a way that would be damaging to competition in downstream markets.

In addition ... access dispute mechanisms in Part 5 of the QCA Act [will no longer apply] and could result in QR using its monopoly position to negotiate pricing, terms and conditions.²³⁴

Pacific National also highlighted the benefits of declaration in its experience:

²³³ For example, South West Producers, sub. 4, pp. 31–32; Glencore, sub. 5, p. 13; Pacific National, sub. 9, p. 12; Aurizon Coal, sub. 21, p. 2; Pacific National, sub. 37, pp. 8–9; Watco, sub. 48, pp. 2, 4–5; Linfox, sub. 50, para. 2.1; GrainCorp, sub. 52, pp. 6–7.

²³⁴ Pacific National, sub. 9, p. 12.

In this context, it will be important for the QCA to consider the benefits that declaration and the resultant competition has delivered. These extend well beyond simply constraining the exercise of market power and ensuring that a balanced risk profile underpins the setting of terms and conditions for access. Declaration has facilitated the introduction of important structural and behavioural constraints ... and has underpinned the growth of competition in related upstream and downstream markets ...²³⁵

The experience of PN in Queensland over the last decade [in entering the Queensland above rail market] clearly demonstrates how declaration and regulation by the QCA has been effective in creating an environment in which the scope for rail freight competition can develop and grow. Put simply, PN's ability to grow its business in Queensland has been critically dependent on the stable operation of Queensland's regulatory framework under Part 5 of the QCA Act.²³⁶

The QCA considers that in a future without declaration, Queensland Rail will not face any effective long-term constraints on its ability and incentive to exercise market power in relation to access terms. In a future without declaration, there will be an imbalance of negotiating power between Queensland Rail and access seekers/users in the presence of sunk investments. The QCA acknowledges that commercial firms face a range of risks and uncertainties in decision-making on a daily basis. However, an imbalance in bargaining power could inhibit the ability of access seekers/users to effectively manage risks, including the risk of hold-up, which have a significant effect on the expected profitability of entry into (and operations within) the market. The presence of these risks, and an imbalance in the ability of access seekers/users to address these risks in a future without declaration, are likely to deter efficient entry or efficient investments by market participants.

In contrast, the QCA considers that a future with declaration provides a transparent statutory process under the QCA Act²³⁷ within which terms and conditions of access can be negotiated. This process provides market participants with greater certainty that access will be provided on reasonable terms and conditions, including to address sunk investments and mitigate the risk of hold-up for access seekers. As such, the QCA considers that the protections offered by the access regime in a future with declaration will lead to a material improvement in the environment for competition in the above-rail freight haulage market, compared to a future without declaration.

5.7 Conclusion

The QCA considers that access (or increased access) to the North Coast Route service, on reasonable terms and conditions, as a result of declaration, would promote a material increase in competition in the dependent above-rail freight haulage market on the North Coast Route. This is because the opportunities and environment for competition in the above-rail freight haulage market will be materially enhanced in a future with declaration, given the constraints declaration imposes on Queensland Rail's ability and incentive to exercise market power, compared to a future without declaration.

The QCA considers that criterion (a) is satisfied in respect of the North Coast Route service, in relation to the dependent above-rail freight haulage market.

²³⁵ Pacific National, sub. 9, p. 4; sub. 37, pp. 3–4.

²³⁶ Pacific National, sub. 9, p. 6; sub. 37, p. 5.

²³⁷ The provisions of the QCA Act can only be changed by parliament.

6 CRITERION (A)—THE MOUNT ISA ROUTE SERVICE

6.1 Part of the existing declared service and the dependent markets

The QCA has assessed the following part of the existing declared service and the following dependent market:

Dependent market	Part of the existing declared service upon which the market is dependent	Facility for the relevant part of the service
The North West Queensland minerals tenements market	Mount Isa Route service, that is use of the Mount Isa Route	Mount Isa Line Those parts of the North Coast Line that interconnect the Mount Isa Line and the Port of Townsville (together, the Mount Isa Route)

Table 4 The Mount Isa Route service and the dependent market

6.2 Geographical description of the Mount Isa Route

The Mount Isa Line extends from Mount Isa east to Stuart (approximately 10 km south of Townsville), and includes the Flynn to Phosphate Hill branch line. From Stuart, the Mount Isa Line joins the North Coast Line.²³⁸ Access to the North Coast Line between Stuart and Townsville (an approximately 10 km section of track) allows users to access the Port of Townsville, from which most of the goods transported on the Mount Isa Line are exported.

6.3 Dependent markets

The QCA considers that relevant dependent markets of the Mount Isa Route service include:

- the North West Queensland minerals tenements market
- the above-rail freight haulage market on the Mount Isa Route
- the market for mining inputs in the North West Queensland minerals region.²³⁹

The QCA is satisfied that each of these relevant dependent markets are separate from the market for the service.

Queensland Rail did not comment on the identification of the North West Queensland minerals tenements market as a relevant dependent market, in its submission in response to the QCA's draft recommendation. Glencore supported the QCA's identification of the North West Queensland minerals tenements market as a relevant dependent market, saying:

QR has not contested the existence of the North West Queensland minerals tenements market in any of its submission [sic].

²³⁸ Queensland Rail, Mount Isa System Information Pack, October 2016, p. 5, https://www.queenslandrail.com.au/business/acccess/Documents/Mount%20Isa%20system%20information%20p ack%20-%20Issue%203%20-%20October%202016.pdf.

²³⁹ See also Glencore, sub. 5, p. 9; sub. 41, p. 13.

However, for completeness Glencore reconfirms that it considers the QCA is correct there is clearly a separate market for North West Queensland minerals tenements, being tenements for non-coal minerals in what is commonly referred to as the North West Queensland minerals province ...²⁴⁰

The QCA considers that one major relevant market dependent on the Mount Isa Route service is the North West Queensland minerals tenements market. Consequently, the analysis in this chapter focuses on this market. It may be the case that criterion (a) is also satisfied in relation to the other two dependent markets identified by the QCA. However, given the QCA's conclusions in relation to the North West Queensland minerals tenements market, detailed analyses are not included of other possible dependent markets in which criterion (a) may (or may not) be satisfied.

6.4 North West Queensland minerals tenements market

6.4.1 The market

The North West Minerals Province (NWMP) encompasses an area centred around Mount Isa in Queensland's north-west region, covering a land area of approximately 375,000 square kilometres (Figure 8).

²⁴⁰ Glencore, sub. 41, p. 12.



Figure 8 The North West Minerals Province

Source: Department of State Development, Manufacturing, Infrastructure and Planning, A Strategic Blueprint for Queensland's North West Minerals Province, 2018.

The NWMP contains approximately 75 per cent of Queensland's base metal and minerals endowment, including copper, lead, zinc, silver, gold and phosphate deposits, and is recognised as a prospective area with the potential for further discoveries across a range of commodities.²⁴¹

²⁴¹ Department of State Development, Manufacturing, Infrastructure and Planning, A Strategic Blueprint for Queensland's North West Minerals Province, Queensland Government, 2018, p. 5, https://www.statedevelopment.qld.gov.au/resources/plan/nwmp/nwmp-strategic-blueprint.pdf; Department of Natural Resources and Mines, Annual exploration program 2017–18, Queensland Government, 2017, p. 10,

A tenement is the right to carry out prospecting, exploration or mining activity in respect of a specific piece of land—a right created through licence issued by the state. Tenements are limited in time and area, with constraints on the ability of the tenement holder to tie up a tenement that it has no intention to develop. The tenement holder may also choose to sell part or all of its rights in the tenement to another party, including the rights to mine any deposits.

Within Queensland, three types of mining licences can be granted over any particular minerals tenement:

- an exploration permit for minerals, granted for up to five years—which permits the holder to use advanced exploration methods to determine the quantity and quality of minerals present
- a mineral development licence, granted for up to five years—which permits the holder to conduct geoscientific programs (e.g. drilling), mining feasibility studies and metallurgical testing to evaluate the development potential of the defined resource
- a mining lease, issued for a period depending on the identified reserves and projected mine life—which permits the holder to conduct larger scale mining operations.²⁴²

The QCA considers that there is a market for minerals tenements in the North West Queensland region encompassing the NWMP. The sellers in this market are the Queensland Government (through tender processes), and existing tenement holders who wish to sell their tenements. The buyers in this market are explorers, developers and producers of minerals who seek to acquire such tenements (these may be large established mining firms or smaller 'junior' miners and investors).²⁴³

Conceptually, the existence of a market for mining tenements has been recognised in the literature.²⁴⁴ Stakeholders to this review have identified the existence of a market for minerals tenements in the North West Queensland region.²⁴⁵ The QCA notes that in its concurrent review of the declaration of the DBCT service, the QCA has separately considered a market for exploration tenements and a market for development tenements (see Part C, Chapter 4). This was due to the detailed data provided to the QCA by stakeholders in relation to the DBCT service.

For the purposes of this analysis of the Queensland Rail service, the QCA does not consider that it is necessary to explore the distinction between the functional dimensions of the North West Queensland minerals tenement market. Moreover, stakeholders have not provided the detailed information in relation to the North West Queensland minerals tenements market that would enable the QCA to conduct such an analysis. For the purposes of this analysis, the QCA considers that the functional dimension of the North West Queensland minerals tenement market

https://www.dnrm.qld.gov.au/__data/assets/pdf_file/0008/1280474/annual-exploration-program-report-2017-18.pdf.

²⁴² Business Queensland, Mineral and coal authorities, 2018, https://www.business.qld.gov.au/industries/miningenergy-water/resources/minerals-coal/authorities-permits/applying/authorities.

²⁴³ Glencore, sub. 5, p. 9; sub. 41, p. 13. For an example of tenement releases in the NWMP, see Department of Natural Resources and Mines, *Annual exploration program 2017–18*, 2017, pp. 10–11, accessed 2 July 2019, https://www.dnrme.qld.gov.au/__data/assets/pdf_file/0008/1280474/annual-exploration-program-report-2017-18.pdf.

²⁴⁴ See for example *In the matter of Fortescue Metals Group Limited* [2010] ACompT 2; NCC, *Revocation of the declaration of the shipping channel service at the Port of Newcastle*, final recommendation, July 2019.

²⁴⁵ Glencore, sub. 5, p. 9.

includes the market for all three types of mining authorities. Stakeholders have also adopted this approach of defining the market for mineral tenements.²⁴⁶

6.4.2 Entry into the market

The decision of a buyer to enter the North West Queensland minerals tenement market (e.g. to buy a tenement) is heavily dependent on their valuation modelling for the tenement, with regard to three primary factors:

- anticipated revenue
- mine operating costs
- infrastructure and logistics costs.²⁴⁷

Due to the relative remoteness of the NWMP region to the nearest export port (a distance of approximately 1,000 km from Mount Isa to the Port of Townsville), freight costs are likely to be high.²⁴⁸ In particular, for a prospective buyer of a minerals tenement, the relative proportion of freight costs (both for mining inputs as well as mineral outputs) in comparison to likely revenue from the tenement is likely to be an important component of the overall decision-making process for acquiring the tenement.²⁴⁹

6.4.3 Relevant features of the current Mount Isa Route service

Two above-rail operators

The QCA understands that Pacific National and Aurizon Operations currently provide above-rail freight haulage services on the Mount Isa Route, accessing the below-rail service provided by Queensland Rail to haul end customers' products. Glencore's copper, zinc and lead businesses are the largest end users by volume on the Mount Isa Line, and Glencore's haulage task is undertaken by Pacific National.²⁵⁰

Structure of below-rail access agreements

In the Queensland bulk haulage industry, miners can (and do) enter directly into below-rail access agreements with the below-rail service provider (Queensland Rail).²⁵¹ The rights under those contracts are then allocated to that miner's nominated above-rail access provider.²⁵² The QCA understands that this contract structure may be suitable for miners with large volumes or a

²⁴⁶ See Glencore, sub. 5, p. 9 (in relation to tenements for minerals) and South West Producers, sub. 4, pp. 20–21 (in relation to tenements for coal). Based on these submissions, the QCA is satisfied that there are clearly different markets for minerals tenements and coal tenements in Queensland. For the Mount Isa Route, the relevant dependent market is clearly the market for minerals tenements, as defined above.

²⁴⁷ Glencore, sub. 5, p. 10.

²⁴⁸ Under the 2016 access undertaking, there is no reference tariff for services on the Mount Isa Route. Below-rail access charges are negotiated between the access seeker and Queensland Rail. There is no publicly available data on the current below-rail access charges on the Mount Isa Route. Therefore, in the absence of stakeholder submissions, total freight costs for any particular user cannot be precisely quantified.

²⁴⁹ Glencore, sub. 5, p. 10.

²⁵⁰ Glencore, sub. 5, p. 4; Pacific National, sub. 37, p. 5; Glencore, submission to the QCA, *Queensland Rail's 2015 Draft Access Undertaking, draft decision*, December 2015, p. 1, https://www.qca.org.au/wpcontent/uploads/2019/05/30170 Glencore-sub-on-2015-DAU-DD-Public Redacted-1.pdf.

²⁵¹ A majority of third party access agreements on the CQCN are held by miners; the access agreements on the West Moreton system are also directly held by the two mines, Yancoal and New Hope. See QRC, sub. 7, p. 31; Aurizon Coal, sub. 21, p. 1.

²⁵² Contractually, the miners have the right to run trains on the below-rail network, but in a practical sense, an aboverail provider is engaged to run those trains on the miner's behalf. See QRC, sub. 7, p. 31.

portfolio of mines, who require certainty of train paths and can meet the volumes required consistently on an on-going basis.

Alternatively, an above-rail haulage provider can directly enter into access agreements with the below-rail service provider for the use of the below-rail service.²⁵³ The above-rail haulage operator is then responsible for contracting with end users (e.g. miners) to haul their product. The QCA understands that this contract structure may be more suitable for miners with smaller or more variable volumes, who cannot guarantee haulage volumes (and train path usage) on a long-term basis.

In the absence of submissions, the QCA lacks information on the types of current contractual arrangements between miners and Queensland Rail on the Mount Isa Route. However, Glencore previously indicated that it negotiates directly with Queensland Rail regarding the below-rail price for its tonnages on the Mount Isa Route.²⁵⁴ In the analysis below, where possible, the QCA considers both types of contract structures.

No evergreen renewal rights in access agreements

Typically, below-rail access agreements are for a 10-year period and mine life is approximately 25–30 years.²⁵⁵ The QCA understands that below-rail access agreements with Queensland Rail on the Mount Isa Route do not include 'evergreen' renewal clauses. On expiry of existing agreements, a train operator or end customer would have to negotiate new terms of access within the bounds of an approved access undertaking. Therefore, existing terms under these agreements (e.g. in relation to pricing, capacity allocation or usage of facilities) will progressively expire, and existing terms will not necessarily be replicated in future agreements.

Under the current access regime (primarily regulated under Queensland Rail's Access Undertaking 1, 2016), use of the below-rail services on the Mount Isa Line are not subject to a QCA-approved reference tariff. Prices are individually negotiated with each user of the below-rail service, within the guidance of the pricing rules outlined in the 2016 Access Undertaking. Furthermore, the QCA Act allows all access seekers (as well as the access provider) to refer access disputes (including disputes regarding pricing) to the QCA for arbitration.²⁵⁶ The access regime also provides for the approval of a reference tariff. Where approved, a reference tariff can facilitate access negotiations by providing a basis for the negotiation of access charges.

Spare capacity exists on the Mount Isa Line and the North Coast Line

Queensland Rail data shows that there is currently existing spare capacity on the Mount Isa Line and North Coast Line, and stakeholders commented that 'all of the projections [of future foreseeable demand] could be met by the existing capacity of the Mount Isa Line, specifically noting that the Mount Isa Line has sufficient capacity to meet the high demand forecast'.²⁵⁷ The QCA considers the effect of spare capacity on Queensland Rail's ability and incentive to exercise market power in section 5.5.1.

²⁵³ Pacific National, sub. 37, p. 5.

²⁵⁴ Glencore, sub. 41, p. 17; Glencore, submission to the QCA, *Queensland Rail's 2015 Draft Access Undertaking*, 5 June 2015, p. 3, https://www.qca.org.au/wp-content/uploads/2019/05/28099_Glencore-Submission-on-QR-2015-DAU-June-15-1.pdf.

²⁵⁵ Glencore, sub. 5, p. 15; QRC, sub. 7, p. 19.

²⁵⁶ See QCA Act, Part 5, division 5 (Access disputes about declared services) and s. 101.

²⁵⁷ Glencore, sub. 5, p. 15; Queensland Rail, sub. 33, attachment B, p. 18; and Queensland Rail, sub. 33, attachment A, pp. 9–11 in respect of spare capacity on the North Coast Line.

6.5 Queensland Rail's ability and incentive to exercise market power

Whether access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote competition in the North West Queensland minerals tenements market depends firstly on whether Queensland Rail has market power that could be used to adversely affect competition in the dependent market; and secondly on whether Queensland Rail has an ability and incentive to exercise that market power, in a future without declaration.²⁵⁸

The QCA considers that as a business, Queensland Rail has an incentive to maximise profits. The QCA considers that Queensland Rail does have market power that could be used to adversely affect competition in markets dependent on the Mount Isa Route service, including the North West Queensland minerals tenements market. This is because Queensland Rail is the natural monopoly provider of a service that the market participants in the North West Queensland minerals tenements market participants in the North West Queensland minerals tenements market participants in the North West Queensland minerals tenements market rely upon to realise the value of their tenements.

Glencore agreed with the QCA's analysis in the draft recommendation:

Glencore agrees with the QCA's analysis that consideration of criterion (a) requires an analysis of whether QR has market power and whether it has the ability or incentive to exercise such market power.

It is absolutely clear from the QCA Draft Decision that QR does have market power in respect of the Mount Isa Rail Access Service. QR is a monopoly supplier of that service and is clearly not constrained given that there are no viable substitutable services (including for the transport of bulk commodities from Mount Isa to the Port of Townsville).

Consequently, the critical question is whether QR has the ability or incentive to exercise such market power with or without declaration $...^{259}$

Queensland Rail argued that in a future without declaration, it would be constrained in its ability and incentive to exercise that market power to adversely affect competition in any way. It argued, similar to the arguments it raised for the North Coast Line, that it was materially constrained in the provision of below-rail services for the purposes of transporting freight on the Mount Isa Line, including by:

- Competition by road operators, which provides a substitute service in respect of the transportation of freight other than some bulk commodities over long distances. Parties requiring freight transportation services can readily shift to moving freight by road rather than rail in the event of an increase in access price and/or decline in quality of service provided.
- Queensland Rail's statutory obligations and position as a statutory authority, including obligations to have approved and comply with strategic and operational plans.
- The threat of regulation or declaration under Parts 3 or 5 of the QCA Act.

The nature of these constraints are largely as discussed above in relation to the North Coast Line. Further, as highlighted by HoustonKemp, Queensland Rail is constrained by customers' ability to pay and countervailing power on the Mount Isa Line.²⁶⁰

These arguments are considered under three broad categories, and will be discussed in detail below:

²⁵⁸ For example, NCC, Declaration of Services, A guide to declaration under Part IIIA of the Competition and Consumer Act 2010 (Cth), April 2018 edn, p. 33, para. 3.26; Duke Eastern Gas Pipeline [2001] ACompT 2 at [116]; Queensland Rail, sub. 33, p. 19, para. 100; South West Producers, sub. 40, p. 19; Glencore, sub. 41, p. 14.

²⁵⁹ Glencore, sub. 41, p. 14.

²⁶⁰ Queensland Rail, sub. 33, p. 33, paras 158.1–158.3.

- (1) Queensland Rail is materially constrained in the provision of below rail services by road freight operators.
- (2) Queensland Rail is constrained by customers' ability to pay and countervailing power on the Mount Isa Route.
- (3) Queensland Rail is constrained by other factors, such as:
 - (a) its statutory obligations and position as a statutory authority
 - (b) the threat of regulation or declaration.

The operation of access arrangements in a future without declaration as a possible constraint on Queensland Rail's ability and incentive to exercise market power is considered in section 5.5.3.

6.5.1 Competition between road freight and rail freight transport

Queensland Rail argued that it has no ability or incentive to exercise market power to adversely affect competition in any dependent market that relies on the Mount Isa Line, because it is materially constrained in the provision of below-rail services by road freight operators. Queensland Rail highlighted the 'challenges of rail specific to the freight carried on the Mount Isa Line', and provided examples of 'increasing competition from road operators on the Mount Isa Line'.²⁶¹

The nature of the freight task on the Mount Isa Route

Queensland Rail provided data for the main types of traffic carried on the Mount Isa Line in 2016–17 as follows:

- Minerals, including refined copper and lead, and lead and zinc concentrates formed 25 per cent of total gtk.
- Other bulk freight, including acid, fertiliser and cement formed 47 per cent of total gtk.
- Intermodal, primarily industrial products used as mining inputs formed 25 per cent of total gtk.
- Agriculture, long distance passenger services and other freight together formed 3 per cent of total gtk.²⁶²

A breakdown of the types of traffic transported on the Mount Isa Line in 2016–17 is shown in Figure 9.

²⁶¹ Queensland Rail, sub. 33, pp. 33–34, paras 162–66.

 ²⁶² Queensland Rail, 2016–17 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2017, p. 18.



Figure 9 Mount Isa Line freight volumes by commodity (and passengers), 2016–17 (million gtk)

Source: Adapted from Queensland Rail, 2016–17 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2017, p. 18.

For the purposes of this analysis, the QCA has applied this data to the Mount Isa Route.

The Mount Isa Route is primarily used to transport mining inputs from the Port of Townsville to the NWMP, and to transport minerals products (e.g. ores and concentrates) from the mines of the NWMP to the Port of Townsville for export. In addition, bulk granular fertiliser is railed from Phosphate Hill to Townsville, and sulphuric acid and bulk sulphur are railed from Townsville and Mount Isa to Phosphate Hill as an input into fertiliser production.

Bulk and non-bulk freight carried on the Mount Isa Route

The QCA considers that the majority of freight transported on the Mount Isa Route can be classified as 'bulk freight', which generally involves large volumes of homogenous product, typically liquid or crushed material (e.g. minerals, acid, fertiliser), transported in mass quantities, without packaging, which tend to be relatively non-perishable and non-fragile. Applying the above data, this bulk freight includes mineral ores, concentrates, acid and fertiliser, and makes up 72 per cent of total gtk carried.

Road transport does not effectively compete with rail transport for bulk freight on the Mount Isa Route

The QCA considers that for bulk products on the Mount Isa Route, rail is the preferred transport mode (discussed in section 5.5.2). This is particularly the case where large volumes of product need to be transported.

In the case of the Mount Isa Route, the nature of the goods produced by the tenements, being high-volume bulk minerals, combined with a substantial distance to port, means that rail transport offers a cost advantage over road transport. The average per kilometre cost of road freight is approximately constant with respect to distance, whereas rail transport is associated with significant economies of scale, with costs decreasing with increasing freight volumes and

distances.²⁶³ The North West Queensland minerals tenement market is a substantial distance away from the nearest export port—the distance from Mount Isa to the Port of Townsville is over 1,000 km. For freight travelling greater than 600–1,000 km, rail transport is significantly cheaper than road.²⁶⁴

This is supported by Glencore, which stated:

For all of the bulk minerals services contracted by Glencore, rail transport is the only economic mode of transport ... road haulage does not provide any competitive constraint on rail costs for bulk minerals $...^{265}$

The natural cost advantage of rail for transporting freight (both bulk and non-bulk) over the long distances on the Mount Isa Route means that users of the Mount Isa Route service depend upon rail transport as the primary mode of transporting their freight, both for moving mining inputs into a tenement and for transporting minerals output out to port.

Rail transport may compete with road transport for some intermodal freight on the Mount Isa Route

Some intermodal freight on the Mount Isa Route may face competition from road transport (as with intermodal freight on the North Coast Route, discussed in section 5.5.2). However, the evidence that stakeholders presented on this topic is mixed.

Intermodal freight on the Mount Isa Route typically consists of:

- industrial products used as mining inputs, such as cement, which can be palletised and transported on intermodal services
- liquid fuels transported in tank containers
- smaller volumes of minerals and metals, which can be transported in containerised form.²⁶⁶

Queensland Rail provided examples of lead, cement and fuel being transported by road, and noted that the final remaining bulk fuel rail service on the Mount Isa Line ceased in 2016–17.²⁶⁷ Glencore acknowledged that some non-minerals freight such as fuel and cement have been utilising road transport from the Townsville region to the NWMP:

Glencore does not dispute that there are some mining inputs which it is economic to transport by road. However, that clearly involves the acquisition of a different product for a different purpose and does not change the fact that for bulk minerals, road haulage is not a competitive constraint (i.e. cost savings or synergies arising from backhaul deliveries has not made road competitive for bulk minerals).²⁶⁸

Based on the information before it, the QCA considers that there is evidence to suggest that the transport of some mining inputs, particularly as intermodal freight, can switch between road and rail transport.

However, there is little publicly available evidence as to the magnitude (i.e. volumes carried) of the road task compared to the rail task in the transport of these goods, particularly given the long distances involved on the Mount Isa Route. For example, it may be that the majority of

²⁶³ Bureau of Infrastructure, Transport and Regional Economics (BITRE), *Road and rail freight: competitors or complements?*, information sheet 34, Australian Government, 2009, p. 8.

²⁶⁴ ACCC, Pacific National/Linfox—Proposed acquisitions of intermodal assets from Aurizon, Statement of Issues, March 2018, p. 13.

²⁶⁵ Glencore, sub. 17, p. 8; sub. 41, p. 7.

²⁶⁶ Queensland Rail, sub. 33, attachment A, pp. 3, 12.

²⁶⁷ Queensland Rail, sub. 33, pp. 34–35, paras 163–66.

²⁶⁸ Glencore, sub. 41, p. 9.

intermodal freight between Townsville and the NWMP is carried by rail, despite some freight (e.g. fuel and cement) carried by road (or vice versa). In the absence of evidence on this issue, the degree of competition between road and rail for the transport of mining inputs as intermodal freight is unclear.

The evidence in relation to switching between road and rail for the transport of minerals and metals also appears to be somewhat equivocal. Queensland Rail submitted that the mode in which some mineral freight is transported has shifted from bulk freight to intermodal freight:

There has been a trend towards miners moving minerals concentrates as containerised freight on intermodal trains, to avoid the costs of specific bulk wagons and loading/unloading facilities. However, moving minerals concentrates in containers is less efficient (and more costly and thus less competitive) than a bulk transport solution.²⁶⁹

Investment in the Port Access Road in Townsville allowed high capacity Type 2 Road Trains direct access to the port, which has facilitated mode shift of minerals products from the Mount Isa Line catchment. Also, smaller scale mines on the Mount Isa Line wanting to reduce upfront capital costs on train loading and unloading facilities may choose an intermodal logistics solution that road can compete on.²⁷⁰

Similarly, Queensland Rail's consultant HoustonKemp stated:

Road freight provides an increasing constraint on rail freight along the Mount Isa system. Although the constraint from road haulage is less likely to bite for heavier, bulky items for which rail is most suited, road freight is becoming a viable option for some bulk items. In particular:

 Several new, smaller scale mines along the Mount Isa system are opting for intermodal solutions such as half-height containers, reducing the up-front capital costs necessary for new mine sites to put in place transport and logistics arrangements – this is often a preferred solution, even where the total cost is lower under traditional, bulk rail wagons, and makes road a closer constraint for bulk items ...²⁷¹

Both submissions suggest that some bulk products have been transported as intermodal freight, but it is not clear whether this intermodal freight is being transported by rail or road. There is little publicly available evidence as to the degree to which road competes with rail for the transport of intermodal containerised minerals freight on the Mount Isa Line specifically, particularly given the large distances involved for this transport.

For example, intermodal freight on the Mount Isa Route largely travels the entire length of the line from Townsville to the NWMP (or vice versa); unlike the North Coast Route, there is very little stopping at intermediate locations between these two points. Additionally, the types of goods carried as intermodal freight differ—from mainly mining inputs on the Mount Isa Route to mainly retail, consumer and agricultural products on the North Coast Route. As such, the analysis of road and rail issues for intermodal freight on the North Coast Route cannot simply be applied by analogy to the containerised minerals freight on Mount Isa Route.

Therefore, the QCA considers that the transport of some intermodal freight on the Mount Isa Route, in particular mining inputs transported as containerised freight, may face competition from road transport.

²⁶⁹ Queensland Rail, sub. 33, p. 34, para. 161.

²⁷⁰ Queensland Rail, sub. 33, p. 35, para. 167.

²⁷¹ Queensland Rail, sub. 33, attachment B, p. 18.

Does competition from road transport constrain Queensland Rail's ability and incentive to exercise market power?

Criterion (a) requires an analysis of the effects of declaration on competition in a dependent market. A key issue is therefore whether competition from road transport operators provides an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power against market participants in the North West Queensland minerals tenements market.

Inputs and outputs of a minerals tenement

Participants in the North West Queensland minerals tenements market rely upon Queensland Rail's Mount Isa Route service for the transport of mining inputs into the tenement, and for the transport of mining products out of the tenement.

The QCA considers that mining inputs may be transported as bulk or intermodal freight from the Townsville region into the NWMP. While there is evidence of competition between road and rail for the transport of specific mining inputs (e.g. cement and fuel), there is little publicly available evidence of the volumes of road freight relative to rail freight for these types of goods. In the absence of this evidence, and in accordance with the analysis in section 5.5.2, the QCA considers that because the Townsville region is a substantial distance from the NWMP, it is likely that there is a cost advantage in transporting mining inputs by rail rather than by road. This advantage is bigger when transporting large quantities/volumes of mining inputs.

The majority of mining outputs from the North West Queensland minerals tenements market is likely to be minerals in bulk form (e.g. refined copper and lead, and lead and zinc concentrates).²⁷² The QCA considers that road transport does not effectively compete with rail transport for bulk freight on the Mount Isa Route, and it is likely that the majority of this transport task is performed by rail.

Queensland Rail's ability and incentive to exercise market power is not constrained with respect to participants in the North West Queensland minerals tenements market

The QCA considers that Queensland Rail is not constrained in its ability and incentive to exercise market power against the participants in the North West Queensland minerals tenements market.

Figure 10 is a graphical representation of possible rail prices charged for the movement of freight on the Mount Isa Route in a future without declaration.

²⁷² While smaller volumes of some bulk minerals may be transported by road, this would not be sufficient to act as a competitive constraint to transportation by rail.



Figure 10 Rail prices on the Mount Isa Route in a future without declaration

Diagram for illustrative purposes only. Not drawn to scale.

The rail price in the figure represents the final price paid by market participants (e.g. miners) in the North West Queensland minerals tenements market. The total rail price consists of two components—an above-rail component, which represents the charges of the above-rail operator, and a below-rail component, which represents the charges of the below-rail operator (Queensland Rail).²⁷³

Assume that the current state of the market is 'Rail price 1'. The below-rail price and the aboverail price together add to \$X, the price paid by a miner to haul freight on the Mount Isa Route. In a future without declaration, Queensland Rail is likely to seek to exercise its market power, where it is possible to do so, in order to maximise its profits. It may seek to exercise this market power by raising the below-rail access charge. The QCA considers that two possible scenarios may then arise, depending upon the nature of the contracts between the miners and Queensland Rail; however, the scenarios lead to similar outcomes.

In the first scenario, miners have access agreements directly with Queensland Rail, and access rights under those contracts are allocated to that miner's nominated above-rail operator (see section 6.4.3). Therefore, the miner pays a below-rail price directly to Queensland Rail, and an above-rail price directly to the above-rail operator. In this case, if Queensland Rail raises its below-rail price, the higher price will be paid directly by the miner, and not necessarily result in an increase in the above-rail price. This scenario is illustrated by 'Rail price 2' in Figure 10, with an increased total rail price of \$Y. In this case, the increase in the access charge (represented by area A) is attributable to (i.e. considered as a part of) the below-rail price.

In the second scenario, an above-rail operator may have an access agreement with Queensland Rail, and then offer their haulage services to the miner (see section 6.4.3). In this case, if Queensland Rail raises its below-rail price, it is likely that the above-rail operator will have an incentive to, and is likely able to, pass through the increase in price in full.²⁷⁴ In this case, the increase in the access charge (area A) is attributable to (i.e. considered as a part of) the above-

²⁷³ The diagram is for illustrative purposes only and does not represent the actual share of the total price between the above-rail and below-rail operators.

²⁷⁴ This is because as businesses, above-rail operators are also profit maximisers. Therefore, to the extent that an above-rail operator can pass on a below-price increase to the end user (miner), the QCA considers it will do so.

rail price. However, it nevertheless leads to a total rail price of \$Y, illustrated by 'Rail price 2' in Figure 10.

Additionally, there may be competition between above-rail operators such that one operator may be prepared to 'absorb' the increase in access charge (i.e. not pass on the increase in full) in order to increase its market share. In this case, the size of area A will vary such that \$Y is greater than or equal to \$X. In the case where \$Y is equal to \$X, the above-rail operators are constrained by competition in the above-rail market from passing on any of the increase in access charges. In this case, the scenario of the collective constraint discussed in section 5.5.2 applies.²⁷⁵

Where \$Y is greater than \$X, the following analysis applies. In the scenario represented by 'Rail price 2', a miner in the North West Queensland minerals tenements market is unlikely to be able to credibly threaten to switch the majority of its transport task (for both mining inputs and outputs) to road in response to the increase in the total rail price. This is because:

- Mining outputs are largely transported in bulk form.
- Mining inputs and outputs travel large distances on the Mount Isa Route.
- Mining inputs and outputs may need to be transported in large volumes.

In addition, a miner in the North West Queensland minerals tenements market would likely have made significant sunk investments into long-life assets, such as the mine itself and associated production facilities and equipment. These investments are unlikely to be easily deployed elsewhere.

The QCA considers that if, in this case, Queensland Rail raised the below-rail charge, and access charges increased to \$Y (as in Figure 10), a miner in the North West Queensland minerals tenements market is unlikely to be able to credibly threaten to switch the majority of its transport task to road, or credibly threaten to exit the market, given its sunk investments.

In theory, the level of \$Y may be a value that could extract the value of the sunk investments of the miner. For an existing miner, this scenario may be regarded as a transfer of rents between the miner, the above-rail operator or the below-rail operator (as the case may be), with little impact on competition in the North West Queensland minerals tenements market. However, the QCA considers that the critical issue is that a potential miner seeking to enter the market, or an existing miner seeking to reinvest in the market, can foresee this risk—that any sunk investments it makes may be exposed to the risk of expropriation by Queensland Rail in a future without declaration. This is the hold-up risk, which is discussed in detail in section 6.6.2.

In summary, the ability of Queensland Rail to exercise market power in future contracting periods, in a future without declaration, creates a significant degree of uncertainty for potential market participants at the time they are considering investment, raising the hurdle rate required to justify the investment, and thereby potentially preventing efficient entry and efficient participation in the North West Queensland minerals tenements market.

²⁷⁵ The scenario is not discussed where an above-rail operator raises its charges, where Queensland Rail does not increase the below-rail access charge. This is because competition exists in the above-rail market. If an above-rail operator threatened to unilaterally raise its price or is otherwise not competitive in its offering, miners have the option of approaching other competitor above-rail operator. An example is on the Mount Isa Route, Glencore once switched its above-rail haulage operator from Aurizon to Pacific National. Glencore said (sub. 41, p. 8) that '[t]his was an anticipated short term cost of switching between competing rail providers, which Glencore considered was justified by the longer term improvement in price and service achieved for rail haulage'.

Therefore, the QCA considers that competition from road is not an effective constraint on Queensland Rail's ability and incentive to exercise market power against the market participants in the North West Queensland minerals tenements market. The QCA considers that market participants in this market may be exposed to the risk of hold-up in a future without declaration. The hold-up issue is discussed in detail in section 6.6.2.

6.5.2 Customers' ability to pay and countervailing power

Queensland Rail argued that it is constrained in the provision of below-rail services by customers' ability to pay and countervailing power on the Mount Isa Line.²⁷⁶ Queensland Rail did not further elaborate upon this point, but referred to the HoustonKemp report, which states:

Revenue collected from access prices on the Mount Isa system (\$74 million in 2017–8) is significantly below the revenue ceiling limit (\$181 million in the same year), and prices will not change materially if Queensland Rail became undeclared.²⁷⁷ This is because the current regulatory arrangements do not prevent Queensland Rail from increasing access prices.

It follows that the binding constraints on Queensland Rail's price setting are non-regulatory factors such as competition from road, end consumer's ability to pay and countervailing power. These factors will not change with removal of declaration and thus removing declaration, and it [sic] associated regulatory pricing constraint, would not lead to access price changes.²⁷⁸

HoustonKemp did not further elaborate upon these points in its report.

The QCA does not consider that the Queensland Rail submission or the HoustonKemp report demonstrated a customer's ability (or inability) to pay for the Mount Isa Route service, or the presence of countervailing power from users of the Mount Isa Route service.

HoustonKemp's view that 'current regulatory arrangements do not prevent Queensland Rail from increasing access prices' is discussed in detail in section 5.6.3.

Firstly, it is not clear how the ceiling revenue limit of \$181 million put forward by Queensland Rail has been calculated. The QCA does not currently (and has not in the past) set a ceiling revenue limit for the Mount Isa Line.²⁷⁹ In any case, whether any ceiling revenue limit is (or is not) achieved is not necessarily an indication of the absence of market power, nor an indication of the presence of countervailing power—this is because firms act to maximise profits, not revenue.

Secondly, the QCA considers that the existing regulatory regime imposes a range of constraints (in the form of both obligations on the access provider and rights to the access seeker) that prevent the service provider from exercising its market power. These requirements (e.g. in the QCA Act or the undertaking) cannot be unilaterally changed by Queensland Rail. These constraints extend beyond the setting of a ceiling revenue limit. As Glencore noted:

However, the main pricing protection that users of QR's services have is the right to refer access disputes (including as to pricing) to the QCA for an arbitrated access determination ... Glencore

²⁷⁶ Queensland Rail, sub. 33, p. 33, para. 159.

²⁷⁷ This is estimated access revenue and excludes TSC and other revenue. The ceiling limit value is estimated through the application of a modified DORC valuation and is calculated using revenue and expense forecasts from Queensland Rail below-rail product forecasts, which reflect Queensland Rail's 2017–18 Corporate Plan estimates. These values are generated by Queensland Rail.

²⁷⁸ Queensland Rail, sub. 33, attachment B, p. 19.

²⁷⁹ The 2016 access undertaking contains pricing rules for the determination of a ceiling revenue limit, however, the QCA does not currently (and has not in the past) set a ceiling revenue limit for any of Queensland Rail's rail systems, except for the West Moreton system.

confirms that it takes the likely QCA arbitrated outcome into account when negotiating access prices with QR and has raised the likely QCA outcome with QR as part of commercial negotiations ...

Glencore also notes that users of the Mount Isa Rail Access Service can raise the potential for a Mount Isa Rail Access Service reference tariff in QR access undertaking processes, which they would be anticipated to begin doing if QR was to consistently seek to engage in monopoly pricing while declaration exists.²⁸⁰

In addition, the QCA does not consider that any evidence or reasoning has been given to support the statements that 'prices will not change materially if Queensland Rail became undeclared'.²⁸¹ For example, the QCA approves access undertakings from time to time and the 2016 Queensland Rail undertaking has a range of pricing rules for developing access charges, which constrains the manner in which Queensland Rail negotiates access prices for the Mount Isa Route service.²⁸²

In relation to countervailing power, the QCA does not consider that customers on the Mount Isa Route have equal bargaining power with Queensland Rail, particularly at contract renewal stage.²⁸³ This is discussed in more detail in the context of the hold-up issue in section 6.6.2.

The QCA therefore considers that Queensland Rail is not constrained in its ability or incentive to exercise market power by customers' ability to pay or countervailing power.

6.5.3 Other constraints on Queensland Rail's ability and incentive to exercise market power

Queensland Rail argued that it is materially constrained in the provision of below rail services by:

- Queensland Rail's statutory obligations and position as a statutory authority, including obligations to have approved and comply with strategic and operational plans.
- The threat of regulation or declaration under Parts 3 or 5 of the QCA Act.²⁸⁴

Similar arguments were raised by Queensland Rail in relation to the North Coast Line. The QCA considers that its analysis of these issues in relation to the North Coast Route service is also applicable to the Mount Isa Route service. The detailed discussion of this issue is in section 5.5.3.

6.5.4 Conclusions on Queensland Rail's ability and incentive to exercise market power

Based on the analysis above, the QCA considers that Queensland Rail has the ability and incentive to exercise market power in the dependent North West Queensland minerals tenements market in a future without declaration.

The following sections will consider the state of competition in the North West Queensland minerals tenements market in a future with and without declaration.

²⁸⁰ Glencore, sub. 41, p. 18.

²⁸¹ Queensland Rail, sub. 33, attachment B, p. 19.

²⁸² For further information on the pricing rules, refer to QCA, *Queensland Rail's Draft Access Undertaking*, decision, June 2016, Chapter 3, https://www.qca.org.au/wp-content/uploads/2019/05/30680_Secondary-Undertaking-Notice-attachment-QCA-Decision-1.pdf.

²⁸³ That is, given that mines and rollingstock typically have useful lives of 25+ years, and access agreements with Queensland Rail have typically been for 10 years, at some stage during the useful life of the asset, the access agreement with Queensland Rail (presumably signed when the mine or rollingstock was first acquired) will be due for renewal.

²⁸⁴ Queensland Rail, sub. 33, p. 33, paras 158.2–158.3.

6.6 Competition in the North West Queensland minerals tenement market in a future with and without declaration

6.6.1 A future with declaration

The QCA considers that a future with declaration will entail a continuation of the existing regulatory regime, whereby participants in the North West Queensland minerals tenement market are able to gain access to below-rail services for both the transportation of mining inputs and the export of mine outputs, on reasonable terms and conditions, due to ongoing regulatory oversight and the protections provided by the access regime in Part 5 of the QCA Act.

The regulatory framework in a future with declaration is an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power. This will provide certainty to a potential new market entrant (or an existing tenement holder approaching renewal of its existing rail agreements or mining authorities) that access to the service will be provided on reasonable terms and conditions in the future with declaration.²⁸⁵

Freight costs, as well as non-pricing terms for the carriage of freight, are relevant to the overall decision-making process for a firm seeking to enter the market, or for an existing firm facing renewal of its rail agreement or mining authority. This is because of the relative remoteness of the NWMP region to the nearest export port (a distance of approximately 1,000 km from Mount Isa to the Port of Townsville) and the lack of effective competition from road for the transport of bulk materials, particularly large volumes over long distances.

HoustonKemp disagreed with the QCA's view in the draft recommendation that freight costs are likely to be a material component of the overall decision-making process, saying:

Estimates provided by Queensland Rail suggests that the importance of below rail costs varies depending on commodity. For example, Queensland Rail's analysis suggests that in 2017-18, below rail costs represent around:

- 0.3 per cent of estimated commodity price for cooper [sic];
- 0.8 per cent of estimated commodity price for zinc;
- 0.9 per cent of estimated price for lead; and
- 5 per cent of estimated commodity price for fertiliser.

In summary, below rail costs are an immaterial input costs for many of the bulk products on the Mount Isa system.

Our conclusion is that with or without declaration, the volumes and access prices on the Mount Isa system will be the same. This is because Queensland Rail has the incentive to maximise volume due to spare capacity and that the access pries [sic] are not constrained by regulation, and as such would not be expected to change without regulation.²⁸⁶

In contrast, Glencore argued:

Glencore's strong impression is that QR has the (misconceived) view that its pricing does not impact on the development of projects – such that it does not seek to set pricing in a way that incentivises or facilitates additional investment or demand.

Consequently, QR's submissions amount to pure speculation that it will not be incentivised to engage in monopoly pricing, and Glencore urges great caution in determining the likely

²⁸⁵ As declaration provides a process under the QCA Act and access undertakings for the setting of reasonable terms and conditions of access.

²⁸⁶ Queensland Rail, sub. 33, attachment B, p. 20.

outcomes with and without declaration in dependent markets on the basis of such speculation, when all past conduct of QR contradicts the very theoretical arguments QR and Houston Kemp raise.²⁸⁷

HoustonKemp does not describe how the below-rail prices it quotes are calculated. As prices on the Mount Isa Route are negotiated individually with access users, the QCA does not have visibility over these prices, nor Queensland Rail's calculations. However, the existing prices in 2017–18 represent prices negotiated under declaration, and thus provide some guidance on possible prices in a future with declaration as the framework and process within which prices are set will remain.²⁸⁸ In contrast, existing prices may not provide guidance as to prices in a future without declaration, where Queensland Rail can exercise market power. HoustonKemp did not provide reasoning or evidence to support its argument that 'volumes and access prices on the Mount Isa system will be the same' with or without declaration. The QCA's analysis of a future without declaration is discussed below.

6.6.2 A future without declaration: the hold-up problem

As a business, Queensland Rail has an incentive to maximise profits. In a future with declaration, its ability and incentive to exercise its market power in order to maximise profits will be constrained by the regulatory regime. The QCA considers that in a future without declaration, however, Queensland Rail will not face any effective long-term constraints on its ability and incentive to exercise market power in order to maximise profits. In particular, the QCA considers that access arrangements applied by Queensland Rail will not act as an effective constraint.

It is in this environment that market participants will face decisions to enter or operate in the North West Queensland minerals tenements market in a future without declaration. In particular, a new entrant to the North West Queensland minerals tenements market will have to incur significant sunk costs. Sunk costs include the costs of exploration and preparatory activities prior to developing a mine (e.g. feasibility studies), which are site-specific. Sunk costs also include the costs of developing the mine itself—the underlying value of the mine, once developed, resides in its potential output, and is site-specific. The presence of sunk investments gives rise to the 'hold-up problem' commonly described in the economics literature.

The hold-up problem is illustrated below with respect to two types of possible entrants in the North West Queensland minerals tenements market: entrants intending to develop the tenement, and entrants intending to resell the tenement. A discussion of the economic theory of hold-up is provided in Appendix A.

Entrants intending to develop the tenement

If a potential entrant commits to entering the market and developing a tenement, it will incur considerable sunk costs in developing the mine. These costs are site-specific sunk costs, in the sense that once the investment is incurred, the costs are 'attached' to the mine. According to Glencore, the typical duration of mining operations in the NWMP is in the vicinity of 10 to 30 years, depending on the operation.²⁸⁹ In contrast, the typical length of a rail haulage contract is approximately 10 years.²⁹⁰ Therefore, sometime during the useful life of the mining tenement, it can be expected that the below-rail access agreement will be due for renewal. As noted in

²⁸⁷ Glencore, sub. 41, p. 16.

²⁸⁸ The QCA notes that an access undertaking can be amended from time to time.

²⁸⁹ Glencore, sub. 5, p. 15.

²⁹⁰ QRC, sub. 7, p. 19.

section 6.4.3 above, below-rail access agreements with Queensland Rail have historically not contained evergreen renewal clauses. This means that any terms contained in the original access agreement may not be necessarily replicated in any new access agreement.

Therefore, when the below-rail access agreement is due for renewal, in subsequent periods after the miner has entered the market (or committed to an expansion within it), the miner would be in a less favourable bargaining position with Queensland Rail. The miner has already made significant sunk investments in developing the mining tenement, and will continue to largely rely on access to the Mount Isa Route service in order to import mining inputs and to export mine outputs. Without continued access to the Mount Isa Route service, a miner faces the risk that its asset (the tenement) will be stranded.

In a future without declaration, the QCA considers that Queensland Rail will not face any effective long-term constraints on its ability and incentive to exercise market power in these subsequent periods when the access agreement is due for renewal. For example, Queensland Rail may raise access charges, or offer less favourable non-price access terms (e.g. train path allocations). This is a point argued by Glencore, which said:

[I]n the absence of declaration there will be:

- (a) a significant increase in prices to existing users;
- (b) a much higher likelihood of differential pricing with a high prospect of favouring some users over others not justified on the basis of efficiency, but rather based on commercial negotiations and/or the uncertainty of different arbitral outcomes
- (c) a dramatic chilling effect on investment in mines or industrial facilities reliant on the use of the rail. In particular, it is difficult to see why an investor would incur considerable amounts in exploration and development (and obtaining related regulatory approvals), if the investor is ultimately faced with an access negotiation where QR is economically incentivised to charge the producer an access price which would leave the producer only covering marginal costs (and not being able to recover the sunk costs expended to that point).²⁹¹

In these subsequent periods, an exercise of market power by Queensland Rail against a miner may arguably be regarded as a transfer of rents between the parties, with little effect on competition. However, the QCA considers that the critical issue is that in the first period, a miner can foresee this risk that any sunk investments it makes will be exposed to the risk of expropriation by the monopolist in the subsequent periods.

The presence of this risk of hold-up means that socially optimal investments will not proceed, or there will be an underinvestment. Queensland Rail may have an incentive to solve this hold-up problem ex ante—for example, it may be profit maximising for Queensland Rail to sell unused network capacity to new or renewing users, assuming it is not constrained to charging a uniform price.²⁹² However, as discussed below, the QCA's view is that it will be difficult for Queensland Rail to credibly commit ex ante to solve the hold-up problem (for example through a long-term contract). The problem is that events could develop in the future where the benefits to Queensland Rail of expropriating the value of the investment at that later time exceed the benefits of continuing to abide by status quo arrangements. The QCA considers that it is this risk—that significant sunk investments made by miners into their mining tenements will be

²⁹¹ Glencore, sub. 17, p. 14.

²⁹² For instance, under the 2016 access undertaking, Queensland Rail is not required to charge a uniform price on the North Coast Line or any other system except for the West Moreton system, where there is a reference tariff for coal trains. On the non-West Moreton systems, prices are negotiated between Queensland Rail and the customers seeking below-rail access.

expropriated—that will lead to a material adverse effect on competition in the North West Queensland minerals tenements market in a future without declaration (the QCA's approach to the concept of materiality is discussed below).

The ability of Queensland Rail to exercise market power in future contracting periods creates a significant degree of uncertainty around material terms (such as pricing and terms of access) for potential market participants at the time they are considering investment. This is likely to adversely affect the value of the tenements in this market, and raise the hurdle rate required to justify an investment in a tenement. The QCA considers that this risk is sufficiently material that a miner may be deterred from entering the market in the first place. As Glencore said:

[I]t is impossible to see how investors would incur costs in exploration and development when there is such limited certainty of costs [on the Mount Isa Route service] and the knowledge that they can be held hostage to monopoly pricing at the time of seeking access.

It is highly likely that the prospect of new entry will be eliminated.

Even if Glencore was incentivised to continue to participate in the market due to its existing portfolio of Mount Isa mines and existing take or pay rail haulage or port commitments, that will forever entrench a position of there being few possible acquirers in the market (being the existing incumbents). This is particularly so as most (if not all) other producers in the region are producers of a single project such that once the life of those projects had expired (or prices had been increased to such a point that producers could not feasibly operate a single project in the region) they would not be incentivised to reinvest further in further tenements.²⁹³

The QCA considers that in a future without declaration, the hold-up problem will also apply to existing tenement holders at the time of their contract renewal (of the below-rail contract). Therefore, the QCA is concerned that in a future without declaration, existing tenement holders may begin to delay undertaking efficient actions in their existing tenements in anticipation of the possibility of such investments being held up.

Thus, the QCA is concerned that in a future without declaration, all market participants in the North West Queensland minerals tenement market will face uncertainties relating to material price and non-price terms for access to below-rail services on the Mount Isa Route, particularly at the time of contract renewal, and that these uncertainties will deter efficient entry and efficient participation across the North West Queensland minerals tenement market.

Entrants intending to resell the tenement

An argument may be made that an entrant to the North West Queensland minerals tenement market who is intending to simply hold the tenement and resell it at a later date will not incur the same sunk costs as an entrant intending to develop the tenement, and thus will not face the hold-up problem.

The QCA considers that this argument fails to appreciate the risks borne by the potential reseller. Even if a potential entrant is only seeking to acquire a tenement for future re-sale, it can foresee that any future buyer would need to incur considerable sunk costs in order to develop the tenement, and that the need to incur sunk costs would be a material part of that buyer's decision-making process. Thus, by entering the market, a reseller assumes the risk that Queensland Rail may act in an unconstrained manner in a future without declaration. If future potential entrants are deterred from entering the market due to concerns about the hold-up issue, this is likely to adversely affect the value of the reseller's investment.

²⁹³ Glencore, sub. 17, p. 14.

The QCA considers that a potential reseller would be able to foresee this risk at the time of entry, and would likely be deterred from entering the market in the first place. In this sense, the uncertainties in relation to access terms in a future without declaration are likely to also deter efficient entry and efficient participation by resellers in the North West Queensland minerals tenement market.

Queensland Rail's submission that there is no hold-up problem

In response to the QCA's analysis of the hold-up problem in the draft recommendation, Queensland Rail argued that there is no hold-up problem arising in respect of the services provided by the North Coast Line, Mount Isa Line and West Moreton systems.²⁹⁴ Queensland Rail's submissions on this issue are summarised in section 5.6.3. Glencore disagreed with each of the points raised by Queensland Rail, and argued that they did not address the hold-up problem.²⁹⁵

Queensland Rail's arguments are not specific to the Mount Isa Line, but instead apply to each of the services provided by the North Coast Line, Mount Isa Line and West Moreton systems. As such, the QCA refers to its detailed consideration of the arguments raised by Queensland Rail in section 5.6.3 above. In summary, the QCA considers that the issues raised by Queensland Rail are unlikely to sufficiently address the risk of hold-up in a future without declaration. A further discussion of the economic literature relating to hold-up problem is provided in Appendix A.

6.6.3 Promote a material increase in competition

Materiality

In the case of the Mount Isa Route service, the QCA considers that access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a material increase in competition in the North West Queensland minerals tenements market. The QCA's approach to the concept of materiality is discussed in section 5.6.4.

A decision to enter (or re-invest in) the North West Queensland minerals tenements market will involve substantial sunk investments. In a future without declaration, the presence of sunk investments gives rise to the hold-up problem. The QCA considers that the risk of hold-up in the presence of substantial sunk investments is sufficiently material that it is likely to discourage efficient firms from entering the market. In contrast, the QCA considers that declaration, and the associated access regime, is able to credibly constrain Queensland Rail's ability and incentive to exercise market power and address the hold-up risk.

The QCA considers that the credible constraint on the risk of hold-up in the presence of substantial sunk investments will promote a non-trivial, material improvement in the environment for competition in the North West Queensland minerals tenements market. The environment for competition in a future with declaration is likely to promote efficient entry (and efficient investment) by all market participants, such that competitive outcomes in the North West Queensland minerals tenements market are materially more likely to occur.

If efficient entry is likely to be promoted in a future with declaration (compared to a future without declaration), the QCA considers that this would indicate that access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote an increase in competition that is material. In this way, the QCA is satisfied that access

²⁹⁴ Queensland Rail, sub. 33, p. 39, paras 191–92, sub. 33, attachment B, pp. 13–14.

²⁹⁵ Glencore, sub. 41, pp. 16–18.

as a result of declaration would promote a material increase in competition in the North West Queensland minerals tenements market.

Conclusion on the hold-up problem in the North West Queensland minerals tenements market

In the case of the Mount Isa Route service, the QCA considers that the uncertainties facing market participants in a future without declaration would affect all participants across the market, including more efficient firms. Conversely, the QCA considers that the certainties and protections offered by the access regime under declaration would promote efficient entry and efficient participation in the North West Queensland minerals tenements market.

The QCA considers that in a future without declaration, Queensland Rail will not face any effective long-term constraints on its ability and incentive to exercise market power in relation to access terms. In a future without declaration, there will be an imbalance of negotiating power between Queensland Rail and access seekers/users in the presence of sunk investments. The QCA acknowledges that commercial firms face a range of risks and uncertainties in decision-making on a daily basis. However, an imbalance in bargaining power could inhibit the ability of access seekers/users to effectively manage risks, including the risk of hold-up, which have a significant effect on the expected profitability of entry into (and operations within) the market. The presence of these risks, and an imbalance in the ability of access seekers/users to address these risks in a future without declaration, are likely to deter efficient entry or efficient investments by market participants.

In contrast, the QCA considers that a future with declaration provides a transparent statutory process under the QCA Act²⁹⁶ within which terms and conditions of access can be negotiated. This process provides market participants with greater certainty that access will be provided on reasonable terms and conditions, including to address sunk investments and mitigate the risk of hold-up for access seekers. As such, the QCA considers that the protections offered by the access regime in a future with declaration will lead to a material improvement in the environment for competition in the North West Queensland minerals tenements market, compared to a future without declaration.

6.7 Conclusion

The QCA considers that access (or increased access), on reasonable terms and conditions, as a result of a declaration of the Mount Isa Route service, would promote a material increase in competition in the North West Queensland minerals tenements market. This is because the opportunities and environment for competition in that market will be materially enhanced in a future with declaration, given the constraints declaration imposes on Queensland Rail's ability and incentive to exercise market power, compared to a future without declaration.

The QCA considers that criterion (a) is satisfied with respect to the Mount Isa Route service, in relation to the dependent North West Queensland minerals tenements market.

²⁹⁶ The provisions of the QCA Act can only be changed by parliament.

7 CRITERION (A)—THE WEST MORETON ROUTE SERVICE

7.1 Part of the existing declared service and dependent markets

The QCA has assessed the following part of the existing declared service and the following dependent market:

Table 5 The West Moreton Route service and the dependent market

Dependent market	Part of the existing declared service upon which the market is dependent	Facility for the relevant part of the service
The West Moreton region coal tenements market	West Moreton Route service, that is use of the West Moreton Route	West Moreton system Metropolitan system (together, the West Moreton Route)

7.2 Geographical description of the West Moreton Route

The West Moreton system extends between Rosewood in the east and Miles in the west, and is over 314 km in length. It adjoins the Metropolitan system at Rosewood, and adjoins the Western system at Miles. The South Western system adjoins the West Moreton system at Toowoomba, and various branch lines of the Western system extend off the West Moreton system.²⁹⁷

Goods transported on the West Moreton system typically travel east, where access to the Metropolitan system allows users of the West Moreton system to access the Port of Brisbane, from which most of the goods transported on the West Moreton system are exported.

7.3 Dependent markets

The QCA considers that relevant dependent markets of the West Moreton Route service include:

- the West Moreton region coal tenements market
- the above-rail freight haulage market on the West Moreton Route
- the Port of Brisbane coal handling services market.

The QCA is satisfied that each of these relevant dependent markets are separate from the market for the service.

Queensland Rail did not, in its submission in response to the QCA's draft recommendation, disagree with the identification of the West Moreton region coal tenements market as a relevant dependent market. The South West Producers supported the QCA's identification of relevant dependent markets in the draft recommendation.²⁹⁸

²⁹⁷ Queensland Rail, West Moreton System Information Pack, October 2016, pp. 12–16, https://www.queenslandrail.com.au/business/acccess/Documents/West%20Moreton%20System%20Information %20Pack%20-%20Issue%203.1%20-%20October%202016.pdf.

²⁹⁸ South West Producers, sub. 40, p. 16.

The QCA considers that one major relevant dependent market on the West Moreton Route is the West Moreton region coal tenements market. Consequently, the analysis in this chapter focuses on this market. Criterion (a) may also be satisfied in relation to the other dependent markets identified by the QCA. However, given the QCA's conclusions in relation to the West Moreton region coal tenements market, the QCA has not included detailed analyses of other possible dependent markets in which criterion (a) may (or may not) be satisfied.

7.4 West Moreton region coal tenements market

7.4.1 The market

The South West Producers highlighted the existence of a tenements market:

As noted in the FMG Tribunal Decision, a tenements market (distinct from coal markets) exists as there is evidence of supply and acquisition of tenements (of non-operational projects), including evidence of companies which acquire tenements with a view to future divestment for a profit rather than development.²⁹⁹

The QCA considers that there is a market for coal tenements in the West Moreton region. In Queensland, a distinction is made between tenements for coal (such as in the West Moreton region) and tenements for minerals (such as in the Mount Isa region). However, as is the case for the Mount Isa region, three types of mining licences are relevant to the West Moreton region: an exploration permit for coal, a mineral development licence for coal, and a mining lease (see section 6.4.1).

The South West Producers said it is likely there are two separate markets in relation to coal tenements—one in respect of exploration and development tenements, and another in respect of production tenements. This is because there are differences between the risk profile and valuation of such tenements, resulting in them not being close substitutes.³⁰⁰

The QCA accepts that there are different functional dimensions to the market for coal tenements. The QCA notes that in its concurrent review of the declaration of the DBCT service, the QCA has separately considered a market for exploration tenements, a market for development tenements, and a market for operating mines (production tenements) (see Part C, Chapter 4). This was due to the detailed data provided to the QCA by stakeholders in relation to the DBCT service.

In the present case, stakeholders have not provided the detailed information³⁰¹ in relation to the West Moreton region coal tenements market that would enable the QCA to conduct an analysis into the functional dimensions of this market. In any case, for the purposes of this analysis of the Queensland Rail service, the QCA does not consider that it is necessary to explore the distinction between the functional dimensions of the West Moreton region coal tenements market.

For the purposes of this analysis, the QCA considers that the functional dimension of the West Moreton region coal tenements market includes all three types of mining authorities. The sellers in the West Moreton region coal tenements market are the Queensland Government and existing tenement holders. The buyers in this market are explorers, developers, and

²⁹⁹ South West Producers, sub. 4, p. 20. Although not directly referenced by the South West Producers in its submission, the FMG Tribunal Decision referred to in the quote is presumably a reference to *In the matter of Fortescue Metals Group Limited* [2010] ACompT 2.

³⁰⁰ South West Producers, sub. 31, p. 12; sub. 40, p. 5.

³⁰¹ That is, submissions received within the period for making submissions stated by the QCA.

producers of coal who seek to acquire such tenements (these may be large established mining firms, or smaller junior miners and investors).

The QCA considers the geographic dimension of the West Moreton region coal tenements market is the area for which the most efficient and lowest-cost point of export is via the West Moreton Route service to the Port of Brisbane—that is, the area surrounding the West Moreton Route service as seen in the map below (Figure 11).



Figure 11 Map of the West Moreton coal tenements region

Source: South West Producers, sub. 4, p. 8.

Stakeholders accepted this as the appropriate geographic dimension of the market, including its inclusion of tenements that use, or can be expected to use, the West Moreton Route service. On this, the South West Producers said the geographic scope of the market is clear and a feature resulting from:

- the unique rail infrastructure constraints applicable to the West Moreton and Metropolitan rail systems (lower axle loads and smaller passing loops resulting in significantly lower payload rolling stock relative to other coal systems, reserved paths for non-coal services and passenger priority);
- (2) the significantly different infrastructure costs per tonne;
- (3) lack of connections to other coal rail systems or coal terminals;
- (4) the vessel constraints at the Port of Brisbane; and
- (5) coal quality differences with the thermal coals produced by the mines in this region being 'harder' than most other coals. ³⁰²

The QCA notes Queensland Rail did not dispute any particular aspects of the market definition, as Queensland Rail considered precise definitions would not impact the competition analysis (of the future with or without declaration) required.³⁰³

³⁰² South West Producers, sub. 4, p. 20; sub. 31, p. 12.

7.4.2 Entry into the market

The decision of a buyer to enter the West Moreton region coal tenements market (e.g. to buy a tenement) depends heavily on their valuation modelling for the tenement. The South West Producers described the method they use to value tenements:

Firstly, the South West Producers, as acquirers in various tenement markets, can confirm that they value tenements using financial modelling, principally reflecting a discounted cash flow model. The critical parts of that cash flow model are:

- (a) expected revenue principally determined by coal prices and US\$/A\$ exchange rates;
- (b) mine operating costs; and
- (c) infrastructure / logistics costs.

The last of these is critically important for Port of Brisbane coal tenements.³⁰⁴

Such a valuation method is in line with market practice.

Participants in the West Moreton region coal tenements market face unique infrastructure constraints because of the market's location relative to the Brisbane metropolitan region. These constraints primarily relate to requiring access to the Metropolitan system to access the export port (e.g. there are train path limits for coal services through the Metropolitan system).

The South West Producers mentioned the importance of infrastructure costs:

[A]s the QCA is aware from consideration of the Western system coal tariffs in the last undertaking process, the infrastructure costs for this coal supply chain are higher than any other coal supply chain in Australia. This is exacerbated by the smaller vessel size which can be loaded at the Port of Brisbane.

 \dots the South West Producers both confirm \dots infrastructure cost considerations being critically important to future investment decisions under consideration in respect of New Acland and Cameby Downs. 305

Infrastructure and logistics costs are therefore likely to be a material consideration for a potential market participant in the West Moreton region coal tenements market (the materiality of costs are considered in more detail in section 7.6.2).

7.4.3 Relevant features of the current West Moreton Route service

Above-rail operators and exporting mines

Aurizon Operations (Aurizon Coal) currently operates above-rail coal haulage services on the West Moreton Route, accessing the below-rail service provided by Queensland Rail to haul end customers' coal. The QCA understands that from late 2019, the above-rail haulage operator Watco will also be traversing the West Moreton Route as it carries agricultural products from the South Western and Western systems (which connect to the West Moreton Route).³⁰⁶

Two mining companies are currently producing and exporting thermal coal from the West Moreton region—New Hope (which owns the New Acland and Jeebropilly mines) and Yancoal (which owns Cameby Downs mine). All coal currently transported on the West Moreton Route to the Port of Brisbane originates from these three mines. There are other mines in the region,

³⁰³ Queensland Rail, sub. 33, p. 20, para. 107.

³⁰⁴ South West Producers, sub. 4, pp. 20–21.

³⁰⁵ South West Producers, sub. 4, p. 21; sub. 31, p. 13.

³⁰⁶ Watco, sub. 48, p. 2.

but these are either co-located with (and exclusively supply) a nearby power station, or have closed.³⁰⁷

Structure of below-rail access agreements

The QCA understands that existing mining companies in the West Moreton region contract directly with Queensland Rail for access rights to below-rail infrastructure on the West Moreton Route. The mines separately contract with Aurizon Coal as their nominated rail operator.³⁰⁸ Thus, in the analysis below, reference is made to the miners requiring access to this service through negotiating directly with Queensland Rail on below-rail access terms.

No evergreen renewal rights in access agreements

Typically, below-rail access agreements are for a 10-year period, compared to the typical life of a coal mine, which is around 10 to 30 years.³⁰⁹ Below-rail access agreements do not include 'evergreen' renewal clauses. On expiry of existing agreements, miners would have to negotiate new terms of access within the bounds of an approved undertaking.³¹⁰ Therefore, existing terms under these agreements (e.g. in relation to pricing, capacity allocation or usage of facilities) will progressively expire, and existing terms will not necessarily be replicated in future agreements.

Under the 2016 access undertaking, a QCA-approved reference tariff applied for coal-carrying train services operating on the West Moreton system and the Metropolitan system.³¹¹ A reference tariff can facilitate access negotiations by providing a basis for negotiation of the below-rail access charges.

Spare capacity and utilisation of the service

According to Queensland Rail, the current capacity of the West Moreton system is around 9.5 mtpa—and it is not operating at full capacity:³¹²

The capacity limitation for the West Moreton System is the available capacity on the Toowoomba Range, where West Moreton, South Western and Western Systems traffic converges. There is currently spare capacity on the Toowoomba Range and through the Metropolitan System to the Port of Brisbane.

Coal services used 68% of available paths in 2016-17. The New Acland mine is nearing exhaustion and there is uncertainty as to whether it will continue operations beyond 2020.

There is capacity for additional freight services on the West Moreton System.

Overall 70% of available paths on the Toowoomba Range were used (coal and non-coal) in 2016-17, however, this was due to a record grain season. The four year average utilisation is [sic] for non-coal freight was 46% of the weekly 16 return preserved paths.³¹³

Queensland Rail submitted that there is excess capacity on its network, which is one of the reasons that demonstrate it does not have the ability or incentive to exercise market power.³¹⁴ The QCA considers these issues below.

³⁰⁷ South West Producers, sub. 4, pp. 35–37.

³⁰⁸ Contractually, the miners have the right to run trains on the West Moreton system and the Metropolitan system, but in practice, Aurizon Coal is engaged to run trains on the miners' behalf: South West Producers, sub. 4, p. 14; Aurizon Coal, sub. 21, p. 1; QRC, sub. 7, p. 31.

³⁰⁹ South West Producers, sub. 4, p. 35; QRC, sub. 7, p. 19.

³¹⁰ South West Producers, sub. 4, p. 28.

³¹¹ See Queensland Rail's Access Undertaking 1, 2016, section 3.5 and schedule D.

³¹² Queensland Rail, sub. 33, attachment A, p. 6, and attachment B, p. 25.

³¹³ Queensland Rail, sub. 33, attachment A, pp. 3–4.

7.5 Queensland Rail's ability and incentive to exercise market power

Whether access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a material increase in competition in the West Moreton region coal tenements market depends firstly on whether Queensland Rail has market power that could be used to adversely affect competition in the dependent market; and secondly on whether Queensland Rail has an ability and incentive to exercise that market power, in a future without declaration.³¹⁵

Queensland Rail submitted it has no ability or incentive to exercise market power in any dependent markets because it is no longer vertically integrated, there is excess capacity on the network and there are material market and other constraints on Queensland Rail other than regulation.³¹⁶

In relation to the West Moreton system, Queensland Rail said:

In the low tonnage scenario, it is likely that the level of access charge required to cover the costs of providing Queensland Rail's services would exceed the ability of the remaining mine to pay... As described by HoustonKemp, in these circumstances, and given the West Moreton System would be underutilised, Queensland Rail would have very strong incentives to negotiate a price with the remaining mine that maximises utilisation and to promote competition in dependent markets so as to maximise demand for services on the West Moreton System (including by facilitating access to the system).

In the high tonnage scenario, for the reasons outlined in the HoustonKemp Expert Report, Queensland Rail still has a strong incentive to increase mining output and it is not clear that the use of the West Moreton System satisfies criterion (a).³¹⁷

The South West Producers said it is clear that Queensland Rail does have market power and would be incentivised to increase prices to realise monopoly profits, reduce service levels and seek more favourable terms in a future without declaration. It responded to the constraints Queensland Rail raised and provided reasons why it did not consider any of these would be effective, for example:

It is absolutely clear that QR does have market power. In respect of the West Moreton corridor coal rail access service they are a monopoly supplier in the market, and there are (on QR's own admission), no viable substitute services.³¹⁸

The South West Producers also said:

In circumstances where it is clear that QR is not competitively constrained, and in the absence of declaration does not face regulatory constraints, and QR is otherwise motivated by commercial profit maximising incentives, it is clear that QR would have both the incentive and ability to exercise its market power in the absence of declaration.³¹⁹

The QCA considers that as a business, Queensland Rail has an incentive to maximise profits. Additionally, Queensland Rail is the natural monopoly provider of a service that the market participants in the West Moreton region coal tenements market rely upon to realise the value

³¹⁴ Queensland Rail, sub. 33, p. 15, para. 79.2; sub. 8, p. 4, para. 25.

³¹⁵ For example, NCC, Declaration of Services, A guide to declaration under Part IIIA of the Competition and Consumer Act 2010 (Cth), April 2018 edn, p. 33, para. 3.26; Duke Eastern Gas Pipeline [2001] ACompT 2 at [116]; Queensland Rail, sub. 33, p. 19, para. 100; South West Producers, sub. 40, p. 19; Glencore, sub. 41, p. 14.

³¹⁶ Queensland Rail, sub. 33, p. 21, para. 109.

³¹⁷ Queensland Rail, sub. 33, p. 35, paras 169–71.

³¹⁸ South West Producers, sub. 40, p. 19.

³¹⁹ South West Producers, sub. 31, p. 14.

of their tenements. The possible constraints on Queensland Rail's ability and incentive to exercise market power are considered below under three broad categories:

- (1) competition from road freight operators
- (2) market factors, such as the uncertainty around future volumes of coal that will be transported on the West Moreton Route
- (3) other factors, such as:
 - (a) Queensland Rail's statutory obligations and position as a statutory authority
 - (b) the threat of regulation or declaration.³²⁰

The operation of access arrangements in a future without declaration as a possible constraint on Queensland Rail's ability and incentive to exercise market power is considered in section 5.5.3.

7.5.1 Competition between road freight and rail freight transport

Queensland Rail submitted that it is materially constrained in the provision of below-rail services on its network by competition from road operators. However, it acknowledged this was not the case for users of the West Moreton system:

Queensland Rail is materially constrained in the provision of below rail services to freight operators. Most significantly, for all freight other than some bulk commodities being transported over long distances (such as coal on the West Moreton System), Queensland Rail faces intense and increasing competition from road operators.³²¹

This view is consistent with the position the South West Producers put forward:

[R]oad haulage does not provide an effective constraint on the terms QR could offer in the absence of declaration due to:

- (1) The significant price constraints for haulage of bulk products including coal; and
- (2) The unique non-price constraints on utilising road haulage to transport coal to the Port of Brisbane (Port of Brisbane leasing arrangements government policy, environmental, safety and social licence to operate issues arising from transport through the Metropolitan region).³²²

The nature of the freight task on the West Moreton Route

Queensland Rail provided data for the main types of traffic carried on the West Moreton system in 2017–18 as follows:

- Coal formed 98 per cent of total gtk carried.
- Agriculture, long distance passenger services and other freight together formed 2 per cent of total gtk carried.³²³

A breakdown of the types of traffic transported on the West Moreton system in 2017–18 is shown in Figure 12. For the purposes of this analysis, the QCA has applied this data to the West Moreton Route.

³²⁰ Queensland Rail, sub. 33, pp. 35–36, paras 169–174.

³²¹ Queensland Rail, sub. 33, p. 22, para. 122.

³²² South West Producers, sub. 31, pp. 13–14.

 ³²³ Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p.
7, https://www.queenslandrail.com.au/business/acccess/Compliance%20and%20reporting/2017 18%20QCA%20Annual%20Performance%20Report.pdf.



Figure 12 West Moreton system freight volumes by commodity, 2017–18 (million gtk)

Source: Adapted from Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p. 7.

Competition from road transportation is not an effective constraint

The Queensland Rail data shows that the significant majority of freight carried on the West Moreton Route is coal (98 per cent of total gtk carried), which is transported in bulk form to the Port of Brisbane for export. The QCA considers that for bulk products, rail is the preferred transport mode (section 5.5.2). This is particularly the case where large volumes of product need to be transported—approximately 6.5 million net tonnes of coal were transported on the West Moreton Route service in 2017–18.³²⁴

In addition to rail having a natural cost advantage to transport bulk goods, coal miners in the West Moreton region are entirely dependent on the use of rail haulage, in particular the use of the West Moreton Route, to transport coal to the Port of Brisbane. This is due to the unique non-price constraints prohibiting the use of road haulage to transport coal (particularly large volumes of coal) through the Brisbane metropolitan region. For example, the South West Producers noted:

In actual fact there are of course numerous non-price constraints on utilising road haulage including:

- (a) government policy, environmental, safety and social licence to operate issues which would make a large volume of trucking practically impossible; and
- (b) it is a condition of the QBH coal terminal's lease at the Port of Brisbane that it is prohibited from receiving coal by road haulage delivery without the Port of Brisbane's consent (which the South West Producers understand from previous discussions is highly unlikely to be given).³²⁵

Therefore, the QCA considers that participants in the West Moreton region coal tenements market rely upon access to the West Moreton Route service for the transport of coal products out of the tenements. Based on the evidence before it, the QCA considers that road transport cannot (and does not) compete with rail transport for the haulage of coal from the West

 ³²⁴ Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p.
7.

³²⁵ South West Producers, sub. 4, p. 15.

Moreton region coal tenements market. Therefore, competition from road transport operators does not provide an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power against market participants in the West Moreton region coal tenements market.

7.5.2 Market factors and other constraints

Queensland Rail submitted that it is 'materially constrained by market and other factors in the provision of below rail services'.³²⁶ These can be considered under two broad categories:

- (1) whether market factors, including the volume uncertainty and customers' ability to pay, will constrain Queensland Rail's ability and incentive to exert market power
- (2) whether Queensland Rail is constrained by other factors, such as:
 - (a) its statutory obligations and position as a statutory authority
 - (b) the threat of regulation or declaration.

Market factors

Stakeholder submissions

Queensland Rail submitted that, as outlined in its expert report, there is considerable uncertainty around future volumes of coal that will be transported on the West Moreton system, with the most likely scenarios within the next five years being:

- an increase to 9 mtpa across two mines with the development of the New Acland mine (high tonnage scenario), or
- a decrease to 2 mtpa with the closure of the New Acland mine (low tonnage scenario).³²⁷

If the high tonnage scenario eventuates, Queensland Rail said, for the reasons outlined in the HoustonKemp Expert Report, it still has a strong incentive to increase mining output. If the low tonnage scenario eventuates, Queensland Rail said 'ability to pay' constraints would suppress the access prices that could be imposed, and it would be incentivised to maximise demand:

As described by HoustonKemp, in these circumstances, and given the West Moreton System would be underutilised, Queensland Rail would have very strong incentives to negotiate a price with the remaining mine that maximises utilisation and to promote competition in dependent markets so as to maximise demand for services on the West Moreton System ...³²⁸

The South West Producers said there is no credible evidence that Queensland Rail has incentives to increase volumes, or is likely in the future to conduct itself in a way that incentivises volumes. For instance, the South West Producers said that in the past (at the time of the Wilkie Creek mine closure), Queensland Rail did not approach access pricing in this way:

QR's response was to increase tariffs at the next available opportunity to seek to recover the same revenue from the remaining users. That is the opposite pricing response to what would be expected from a service provider incentivised to increase volume, who would presumably reduce tariffs to seek to attract and restore demand.³²⁹

³²⁶ Queensland Rail, sub. 33, p. 22, para. 122.

³²⁷ Queensland Rail, sub. 33, p. 35, para. 169.

³²⁸ Queensland Rail, sub. 33, p. 35, paras 169–70.

³²⁹ South West Producers, sub. 40, pp. 4, 24.

The South West Producers considered that 'it is theoretically true that the South West Producers' ability to pay means there is a limit to the prices which QR can charge', but they then noted that:

the South West Producers experience is that when this constraint has actually arisen in the past, it has been largely ignored by QR, including at the cost of a previous West Moreton producer closing and noting that access pricing was a material factor in that decision.

That either indicates that QR is not actually constrained in its behavior by this issue, or is simply unable to accurately determine the price point at which this constraint exists.³³⁰

As such, the South West Producers said Queensland Rail's incentives for most (if not all) of the declaration period will be to engage in monopoly pricing, not to increase volume. They argued that even if it is assumed that Queensland Rail has incentives to increase volumes, that initial incentive (once volume has been increased) will be replaced with an incentive to use the next contracting period to expropriate monopoly profits.³³¹

QCA analysis

The QCA notes the current uncertainty regarding future volumes of coal to be transported on the West Moreton Route service.³³² Queensland Rail contended that in a future without declaration, where the 'low tonnage scenario' eventuates:

- it has strong incentives to maximise volumes, including 'to negotiate a price with the remaining mine that maximises utilisation and to promote competition in dependent markets so as to maximise demand for services on the West Moreton System'³³³
- it is constrained by customers' ability to pay under the low tonnage scenario, which 'suppress the access prices that can be imposed by Queensland Rail'.³³⁴

Queensland Rail's arguments regarding the low tonnage scenario on the West Moreton Route are similar to its statements regarding the presence of excess capacity on the rest of its network, for example:

Queensland Rail has excess capacity on its network. A non-vertically integrated service provider with excess capacity has strong economic incentives to maximise utilisation on its network (so as to recover some proportion of its fixed costs) and thus has an incentive to promote (rather than limit) competition in downstream markets.

As detailed in the HoustonKemp Expert Report, Queensland Rail has significant spare capacity on each of its systems. HoustonKemp observes that a consequence of spare capacity (in combination with the constraint imposed by market factors and, in the case of the West Moreton System under the low tonnage scenario, ability to pay constraints) is that an access seeker that can be charged any positive margin over the incremental cost of providing the service represents a contribution to Queensland Rail's substantial fixed cost base. As such, Queensland Rail has an incentive to maximise demand for its services (rather than price).³³⁵

The issue of excess capacity as a possible constraint on Queensland Rail's behaviour is discussed in detail in section 5.5.1. In particular, the QCA notes that in the low tonnage scenario on the

³³⁰ South West Producers, sub. 16, p. 7.

³³¹ South West Producers, sub. 40, pp. 4, 25.

³³² See for example, Queensland Rail's submission in the 2020 draft access undertaking process: Queensland Rail, DAU2 West Moreton System low volume coal reference tariff, submission, November 2019, pp. 5, 29, https://www.qca.org.au/wp-content/uploads/2019/05/queensland-rail-revised-west-moreton-coal-pricingproposal.pdf.

³³³ Queensland Rail, sub. 33, p. 35, para. 170.

³³⁴ Queensland Rail, sub. 33, p. 22, para. 122.

³³⁵ Queensland Rail, sub. 33, p. 22, paras 118–19.

West Moreton Route service, the presence of spare capacity does not imply that Queensland Rail would not behave in a profit-maximising manner. This is particularly the case under the hold-up problem (discussed below in section 7.6.2). The nature of the sunk investments required to enter the West Moreton region coal tenements market means that once the investment is sunk, the miner's willingness to pay increases significantly, relative to that prior to its investment. This is because miners in the West Moreton region coal tenements market rely upon access to the West Moreton Route service for the export of their coal, and to realise the value of their investment. As such, a customer's ability to pay is not necessarily a constraint on Queensland Rail's ability and incentive to exercise market power under the low tonnage scenario. The dynamic nature of the hold-up problem means that the risk of hold-up arises even in the presence of excess capacity.

The QCA therefore considers that Queensland Rail would not be constrained in its ability or incentive to exercise market power by customers' ability to pay or market factors such as the presence of excess capacity under the low tonnage scenario.

Other factors

The QCA has considered whether Queensland Rail is constrained by other factors, including its statutory obligations and the threat of declaration. These issues were raised by Queensland Rail broadly and have been discussed previously by the QCA in section 5.5.3:

- Queensland Rail's statutory obligations and position as a statutory authority
- the threat of regulation or declaration.

To the extent that these matters do not raise additional or different concerns specific to the West Moreton Route service, the QCA considers the same reasoning and conclusions apply here.

As such, the QCA's view is these factors would not effectively constrain Queensland Rail from exercising market power in a way that could adversely affect competition in dependent markets, including the West Moreton region coal tenements market.

7.5.3 Conclusion

Based on the analysis above, the QCA considers that Queensland Rail does have the ability and incentive to exercise market power in the dependent West Moreton region coal tenements market in a future without declaration.

7.6 Competition in the West Moreton region coal tenements with and without declaration

7.6.1 A future with declaration

The QCA considers that in a future with declaration, participants in the West Moreton region coal tenements market would be able to gain access to the below-rail service on reasonable terms and conditions, due to ongoing regulatory oversight and the protections provided by the access regime in Part 5 of the QCA Act.

The South West Producers referred to a range of issues that they described as 'protections which assist with promoting and providing the opportunity for competition which currently exist as a result of declaration':

(a) the QCA is responsible for setting [West Moreton] system coal reference tariffs and has a clearly established methodology for setting those tariffs at an efficient level...

- (b) the QCA is responsible for setting reasonable standard terms of access as per the standard access agreement terms;
- (c) a transparent queueing process which provides an even playing field for all access seekers;
- (d) a more transparent operating regime including through the operating requirements manual;
- (e) the disclosure and reporting regime which provides transparency and accountability and should assist in improving performance and informing access negotiations;
- (f) the QCA Act and undertaking provide rights to bring access disputes where an access seeker cannot reach agreement with QR on obtaining access to the QR Network;
- (g) other protections that the QCA Act provides for declared services generally, such as obligations to:
 - (i) negotiate access requests in good faith
 - (ii) try to meet the reasonable requirements of users; and
 - (iii) the prohibition against preventing or hindering access or use of the services.³³⁶

The South West Producers said these protections are critical to provide a credible constraint on Queensland Rail's use of market power and provide a guaranteed reasonable position for obtaining access to all new entrants. Even for non-reference services, or disputes over non-pricing terms, the ability to have the QCA arbitrate access disputes is a critical constraint that removes the potential for the exercise of monopoly power.³³⁷

The QCA considers the regulatory framework in a future with declaration is an effective longterm constraint on Queensland Rail's ability and incentive to exercise market power. These protections will provide certainty to a potential new market entrant (or an existing tenement holder approaching renewal of its existing rail agreements or mining authorities) that access to the service would be provided on reasonable terms and conditions now and into the future with declaration. In particular, the regulatory regime in a future with declaration establishes a transparent statutory process within which terms and conditions of access can be negotiated (including to address sunk investments and mitigate the risk of hold-up for access seekers). Additionally, declaration can maintain an appropriate balance between the legitimate interests of the service provider and access seekers/users in the presence of sunk investments. Mitigating the risk of hold-up is likely to be a critical factor in supporting efficient entry to and efficient participation in the West Moreton region coal tenements market, thereby materially promoting competition.

7.6.2 A future without declaration: the hold-up problem

As a business, Queensland Rail has an incentive to maximise profits. In a future with declaration, its ability and incentive to exercise its market power in order to maximise profits will be constrained by the regulatory regime. The QCA considers that in a future without declaration Queensland Rail would not face any effective long-term constraints on its ability and incentive to exercise market power in order to maximise profits. In particular, the QCA considers that access arrangements applied by Queensland Rail will not act as an effective constraint.

It is in this environment that market participants will face decisions to enter or operate in the West Moreton region coal tenements market in a future without declaration. In particular, a

³³⁶ South West Producers, sub. 4, p. 28.

³³⁷ South West Producers, sub. 40, p. 39.
new entrant to the West Moreton region coal tenements market will have to incur significant sunk costs. Sunk costs include the costs of exploration and preparatory activities prior to developing or expanding a mine, which are site-specific to the tenement. Sunk costs also include the costs of developing the mine itself—the underlying value of the mine, once developed, resides in its potential output, and is site-specific. The presence of sunk investments gives rise to the 'hold-up problem' commonly described in the economics literature.

Participating in the coal tenements market requires considerable investment in exploration, development and production over time, given mines are long-life projects that require significant upfront development costs. A typical coal mine has a 10- to 30-year useful life, compared with the 10-year typical term of a rail access agreement; as a result, at some point during the useful life of the coal tenement, it can be expected that the below-rail access agreement will be due for renewal.³³⁸ Therefore, when a potential new market entrant is considering investment, it will face the prospect that, having incurred these sunk costs, Queensland Rail will have the ability to exercise market power in setting access prices and other terms and conditions of access at the time of contract renewal.

The costs of accessing below-rail infrastructure, along with expected mine revenue and operating costs, are critical to the valuation of coal tenements and to the decision to invest, or continue to invest, in this market (at all stages, including exploration, development and production). Transport costs are particularly important for coal mines from the West Moreton region, as coal miners in the West Moreton region are entirely dependent on the use of rail haulage to transport coal to the Port of Brisbane due to unique non-price constraints prohibiting the use of road haulage to transport coal through the Brisbane metropolitan region. For example, based on the West Moreton reference tariff from Queensland Rail's 2016 access undertaking, the costs of below-rail access accounts for as much as 10 to 20 per cent of the final free-on-board price of export thermal coal.³³⁹ Therefore, below-rail costs are a material component of the decision-making process for entry into or operations in the market.

At the time of contract renewal, an exercise of market power by Queensland Rail against a miner may arguably be regarded as a transfer of rents between parties, with little effect on competition. However, the critical issue is that in the first period, a miner can foresee the risk that any sunk investments it makes will be exposed to the risk of expropriation by the monopolist in subsequent periods.

The presence of this risk of hold-up means that socially optimal investments will not proceed, or there will be an underinvestment. Queensland Rail may have an incentive to solve this hold-up problem ex ante—for example, it may be profit maximising for Queensland Rail to sell unused network capacity to new or renewing users, assuming it is not constrained to charging a uniform price.³⁴⁰ However, as will be discussed below, the QCA's view is that it will be difficult for Queensland Rail to credibly commit ex ante to solve the hold-up problem (for example through a long-term contract). The problem is that events could develop in the future where the

³³⁸ South West Producers, sub. 4, p. 35; sub. 31, p. 14.

³³⁹ This calculation is based on the reference tariffs applicable from 19 July 2018 to 30 June 2019, for a reference train service departing Cameby Downs mine, travelling fully loaded on the 'down' train path and empty on the return 'up' train path: see

https://www.queenslandrail.com.au/business/acccess/Tariffs/4.%20Ref%20Tariffs%202019%20RE%201.0%20(Que ensland%20Rail).pdf.

³⁴⁰ For instance, under the 2016 access undertaking, Queensland Rail is not required to charge a uniform price on its systems, except for the West Moreton system, where there is a reference tariff for coal trains. On the non-West Moreton systems, prices are negotiated between Queensland Rail and the customers seeking below-rail access.

benefits to Queensland Rail of expropriating the value of the investment at that later time exceed the benefits of continuing to abide by status quo arrangements. The QCA considers that it is this risk—that significant sunk investments made by miners into their mining tenements will be expropriated—that will lead to a material adverse effect on competition in the West Moreton region coal tenements market in a future without declaration (the QCA's approach to the concept of materiality is discussed below).

The ability of Queensland Rail to exercise market power in future contracting periods creates a significant degree of uncertainty around material terms (such as pricing and terms of access) for potential market participants at the time they are considering investment. This is likely to adversely affect the value of the tenements in this market, and raise the hurdle rate required to justify an investment in a tenement. The QCA considers that this risk is sufficiently material that the miner may be deterred from entering the market in the first place. This view is supported by the South West Producers who submitted that the uncertainty around terms of access in a future without declaration means that:

it is impossible to see how producers would incur costs in exploration and development [of a tenement] in the face of such limited certainty of costs of the West Moreton coal rail access service and the knowledge that they can be held hostage to monopoly pricing at the time of seeking access. It is highly likely that the prospect of new entry would be eliminated.³⁴¹

Existing tenement holders would also face higher risks in a future without declaration, with greater uncertainty around future terms of access on the expiry of existing contracts, given that there are no evergreen renewal clauses in existing contracts. Therefore, the QCA is concerned that in a future without declaration, existing tenement holders may begin to delay taking efficient actions in their existing tenements in anticipation of the possibility of such investments may be expropriated.

Thus, the QCA is concerned that in a future without declaration, all participants in the West Moreton region coal tenements market will face uncertainties relating to material price and non-price terms for access to below-rail services on the West Moreton Route, particularly at the time of contract renewal, and that these uncertainties will deter efficient entry and efficient participation across the West Moreton region coal tenements market.

Queensland Rail's submission that there is no hold-up problem

In response to the QCA's analysis of the hold-up problem in the QCA draft recommendation, Queensland Rail argued that no hold-up problem arises in respect of Queensland Rail's services provided using the North Coast Line, Mount Isa Line and West Moreton system.³⁴² Queensland Rail's submissions on this issue are discussed in the North Coast Route service analysis (section 5.6.3). The South West Producers disagreed with each of the points raised by Queensland Rail, and argued that they did not address the hold-up problem.³⁴³

Queensland Rail's arguments are not specific to the West Moreton Route service, and instead apply to each of the services provided using the North Coast Route, Mount Isa Route and West Moreton Route. As such, to avoid repetition, the QCA refers to its detailed consideration of the arguments raised by Queensland Rail in section 5.6.3. In summary, the QCA considers that the issues raised by Queensland Rail are unlikely to be sufficient to address the risk of hold-up in a

³⁴¹ South West Producers, sub. 16, p. 14.

³⁴² Queensland Rail, sub. 33, p. 39, para. 191–92; sub. 33, attachment B, pp. 13–14.

³⁴³ South West Producers, sub. 40, pp. 20–25.

future without declaration. A further discussion of the economic literature relating to the holdup problem is provided in Appendix A.

7.6.3 Promote a material increase in competition

Materiality

In the case of the West Moreton Route service, the QCA considers that access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a material increase in competition in the West Moreton region coal tenements market. The QCA's approach to the concept of materiality is discussed in section 5.6.4.

A decision to enter (or re-invest in) the West Moreton region coal tenements market will involve substantial sunk investments. In a future without declaration, the presence of sunk investments gives rise to the hold-up problem. The QCA considers that the risk of hold-up in the presence of substantial sunk investments is sufficiently material that it is likely to discourage efficient firms from entering the market. In contrast, the QCA considers that declaration, and the associated access regime, is able to credibly constrain Queensland Rail's ability and incentive to exercise market power and address the hold-up risk.

The QCA considers that the credible constraint on the risk of hold-up in the presence of substantial sunk investments will promote a non-trivial, material improvement in the environment for competition in the West Moreton region coal tenements market. The environment for competition in a future with declaration is likely to promote efficient entry (and efficient investment) by all market participants, such that competitive outcomes in the West Moreton region coal tenements market are materially more likely to occur.

If efficient entry is likely to be promoted in a future with declaration (compared to a future without declaration), the QCA considers that this would indicate that access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote an increase in competition that is material. In this way, the QCA is satisfied that access as a result of declaration would promote a material increase in competition in the West Moreton region coal tenements market.

Conclusion on the hold-up problem in the West Moreton region coal tenements market

In the case of the West Moreton Route service, the QCA considers that the uncertainties facing market participants in a future without declaration would affect all participants across the market, including more efficient firms. Conversely, the QCA considers that the certainties and protections offered by the access regime in a future with declaration would promote efficient entry and efficient participation in the West Moreton region coal tenements market.

The QCA considers that in a future without declaration, Queensland Rail will not face any effective long-term constraints on its ability and incentive to exercise market power in relation to access terms. In a future without declaration, there will be an imbalance of negotiating power between Queensland Rail and access seekers/users in the presence of sunk investments. The QCA acknowledges that commercial firms face a range of risks and uncertainties in decision-making on a daily basis. However, an imbalance in bargaining power could inhibit the ability of access seekers/users to effectively manage risks, including the risk of hold-up, which have a significant effect on the expected profitability of entry into (and operations within) the market. The presence of these risks, and an imbalance in the ability of access seekers/users to address these risks in a future without declaration, are likely to deter efficient entry or efficient investments by market participants.

In contrast, the QCA considers that a future with declaration provides a transparent statutory process under the QCA Act³⁴⁴ within which terms and conditions of access can be negotiated. This process provides market participants with greater certainty that access will be provided on reasonable terms and conditions, including to address sunk investments and mitigate the risk of hold-up for access seekers. As such, the QCA considers that the protections offered by the access regime in a future with declaration will lead to a material improvement in the environment for competition in the West Moreton region coal tenements market, compared to a future without declaration.

7.7 Conclusion

The QCA considers that access (or increased access), on reasonable terms and conditions, as a result of a declaration of the West Moreton Route service, would promote a material increase in competition in the West Moreton region coal tenements market. This is because the opportunities and environment for competition in that market will be materially enhanced in a future with declaration, given the constraints declaration imposes on Queensland Rail's ability and incentive to exercise market power, compared to a future without declaration.

The QCA considers that criterion (a) is satisfied with respect to the West Moreton Route service, in relation to the dependent West Moreton region coal tenements market.

³⁴⁴ The provisions of the QCA Act can only be changed by parliament.

8 CRITERION (A)—THE SOUTH WESTERN ROUTE SERVICE, THE WESTERN ROUTE SERVICE AND THE CENTRAL WESTERN ROUTE SERVICE

8.1 Parts of the existing declared service and the dependent markets

The QCA has assessed the following three parts of the existing declared service and the following dependent markets in respect of each:

Dependent market	Part of the existing declared service upon which the market is dependent	Facility for the relevant part of the service
The above-rail freight haulage market on the Central Western Route	Central Western Route service, that is use of the Central Western Route	Central Western system North Coast Line that interconnects the Central Western system and the Metropolitan system and the Port of Mackay Metropolitan system (together, the Central Western Route)
The above-rail freight haulage market on the Western Route	Western Route service, that is use of the Western Route	Western system West Moreton system that interconnects the Western system and Metropolitan system Metropolitan system (together, the Western Route
The above-rail freight haulage market on the South Western Route	South Western Route service, that is use of the South Western Route	South Western system West Moreton system that interconnects the South Western system and Metropolitan system (together, the South Western Route) (together, the Central Western Route, the Western Route and the South Western Route are referred to as 'the agricultural systems')

Table 6The South Western, Western and Central Western Route services and thedependent markets

The QCA has identified and assessed three separate dependent markets, as shown in the table above. Similar issues arise with respect to each of the three dependent markets (e.g. the type of goods carried as freight, the presence of above- and below-rail subsidies). To reduce repetition, the QCA has set out in this chapter a single analysis that applies to each of these dependent markets. Accordingly, the Central Western Route, the Western Route and the South Western

Route are referred to together as the 'agricultural systems' in this chapter, however the analysis should be understood to apply to each service in the context of its specific dependent market.

In this final recommendation, the QCA has revised its analysis of criterion (a) in relation to the agricultural systems. In the draft recommendation, the QCA noted that there was a lack of stakeholder submissions in relation to criterion (a) for the agricultural systems, and based its analysis on the limited publicly available information. In response to the QCA's draft recommendation, several users of the agricultural systems made submissions to the QCA. In addition, in its submission in response to the QCA's draft recommendation, Queensland Rail has substantially increased both the volume and detail of its arguments. In considering this substantial volume of new information, the QCA has revised its analysis of criterion (a) relating to the agricultural systems.

8.2 Geographical description of the South Western, Western and Central Western Routes

The South Western system extends south-west from Toowoomba to Thallon via Warwick. It includes the branch lines Wyreema to Millmerran, and Warwick to Wallangarra. The South Western system adjoins the West Moreton system at Toowoomba.³⁴⁵ From Toowoomba, users of the South Western system traverse the West Moreton system and the Metropolitan system to access the Port of Brisbane, from which goods transported on the South Western system may be exported.

The Western system extends west from Miles to Quilpie, and includes the branch lines Westgate to Cunnamulla, Dalby to Meandarra (Glenmorgan), Miles to Wandoan and Tycanba to Jandowae. The Western system adjoins the West Moreton system at Miles. From Miles, users of the Western system may traverse the West Moreton system and the Metropolitan system to access the Port of Brisbane, or (in the case of livestock) to processors based in the greater Brisbane region.³⁴⁶

The Central Western system extends west from Nogoa to Winton via Emerald, and includes the Emerald to Clermont branch line.³⁴⁷ From Nogoa, users of the Central Western system may traverse Aurizon Network's Blackwater system to connect onto the North Coast Line (and further onto the Metropolitan system), where users may access various export ports on Queensland's eastern coastline.

8.3 Dependent markets

The QCA considers that a relevant dependent market on each of the Central Western Route, the Western Route and the South Western Route (together, the agricultural systems) is the aboverail freight haulage market on that respective route—that is, the market for the transportation of freight by rail on each identified route.

³⁴⁵ Queensland Rail, South Western System Information Pack, October 2016, p. 5, https://www.queenslandrail.com.au/business/acccess/Documents/South%20Western%20System%20Information %20Pack%20-%20Issue%203%20-%20October%202016.pdf.

³⁴⁶ Queensland Rail, Western System Information Pack, October 2016, p. 5, https://www.queenslandrail.com.au/business/acccess/Documents/Western%20System%20Information%20Pack% 20-%20Issue%203%20-%20October%202016.pdf.

³⁴⁷ Queensland Rail, Central Western System Information Pack, October 2016, p. 5, https://www.queenslandrail.com.au/business/acccess/Documents/Central%20West%20System%20Information%2 0Pack%20-%20Issue%203%20-%20October%202016.pdf.

Watco and GrainCorp considered that an above-rail freight haulage market is a dependent market of the services on the South Western, Western and Central Western systems.³⁴⁸ Queensland Rail and other stakeholders did not comment on possible dependent markets of these systems.

GrainCorp identified other possible dependent markets on these systems, including 'markets for storage, containerisation and trading of grain'.³⁴⁹ However, GrainCorp focused its submission on the 'above-rail markets' and did not discuss these other markets in detail.

Other dependent markets include the downstream end product markets for the goods hauled on the agricultural systems; however, these markets are varied, given the diverse range of freight carried on these systems. The QCA has not examined these markets in detail due to limited publicly available information on these markets and the absence of specific stakeholder submissions.

Notwithstanding the existence of other possible markets, the QCA considers that a major relevant dependent market on each of the agricultural systems is the above-rail freight haulage market on each respective route. Consequently, the analysis in this chapter focuses on the above-rail freight haulage markets.³⁵⁰

8.4 Above-rail freight haulage markets on the agricultural systems

8.4.1 The markets

The market for above-rail freight haulage is the market in which primary producers or freight forwarders/logistics handlers contract with operators of rollingstock to haul freight via rail from an origin to a destination point along each of the agricultural systems.

Figure 13 illustrates the above-rail freight haulage market on each of the agricultural systems.

Figure 13 Above-rail freight haulage market on each of the agricultural systems



In the above-rail freight haulage markets on the agricultural systems, the suppliers are the operators of rollingstock. Aurizon Operations and Linfox currently provide above-rail freight haulage services on the agricultural systems, and a new entrant, Watco, has commenced above-rail freight haulage services on the agricultural systems from late 2019.³⁵¹

The customers in the above-rail freight haulage markets on the agricultural systems include primary producers (e.g. farmers and graziers); however, in practice, the customers are typically

³⁴⁸ Watco, sub. 48, p. 4; GrainCorp, sub. 52, p. 6.

³⁴⁹ GrainCorp, sub. 52, p. 7.

³⁵⁰ In doing so, the QCA has focused on traffic with origins or destinations on the Central Western, Western or South Western systems (as the case may be).

³⁵¹ GrainCorp, sub. 52, p. 5; Watco, sub. 48, p. 4. For more detail, see section 8.4.4.

freight forwarders/logistics handlers (e.g. GrainCorp is a bulk grain handler) who are able to consolidate and coordinate produce from various primary producers in order to meet the quantity of product required to fill a train-load. The Queensland Parliament Transport, Housing and Local Government Committee noted the complexity of agricultural supply chains:

Agricultural and livestock supply chains are characteristically complex with multiple participants who have indiscrete roles. Some examples of this complexity are provided below:

- Large meat processors act as both the freight forwarder (consigning trainloads of cattle for livestock producers) and large scale freight generators (packaged meat) and therefore above rail customers
- Large grain traders act as freight forwarders, booking rail space from other grain traders or merchants and some operate their own terminals.³⁵²

8.4.2 The nature of the traffic on the agricultural systems

The agricultural systems are primarily used to transport agricultural products, particularly grain and livestock.

South Western Route traffic

The South Western system primarily carries grain. No freight is carried on the Wallangarra branch (from Warwick to Wallangarra).³⁵³ Figure 14 shows a map of GrainCorp's grain operations on Queensland Rail's systems, including the South Western, Western and Central Western systems.³⁵⁴

³⁵² Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 82.

³⁵³ Queensland Rail, sub. 33, attachment A, p. 4.

³⁵⁴ Indications of the rail systems have been overlaid by the QCA onto the original GrainCorp map.



Figure 14 GrainCorp's Queensland operations

Source: Adapted from GrainCorp, sub. 52, p. 4.

Grain on the South Western system is typically transported to the bulk export terminals at the Port of Brisbane (Fisherman Islands) and Pinkenba (also located in Brisbane) for export.³⁵⁵ GrainCorp noted that:

GrainCorp relies on access to the QR and Aurizon rail networks to transport grain to its export terminals. Grain for domestic consumption is not transported over these rail networks.³⁵⁶

Figure 15 shows the freight volumes on the South Western system annually from 2013–14 to 2017–18.³⁵⁷

³⁵⁵ GrainCorp, sub. 52, p. 4.

³⁵⁶ GrainCorp, sub. 52, p. 5.

³⁵⁷ Queensland Rail reports disaggregated volume data for various sections of the South Western system (e.g. the Toowoomba to Warwick section, the Warwick to Goondiwindi section etc). To avoid double-counting of volumes, the data shown represents the maximum throughput on any individual section of the system. For the South Western system, this is the data for the Toowoomba to Warwick section, as due to the nature of the South Western Route, all traffic must travel through the Toowoomba to Warwick section of the system to/from the Port of Brisbane.



Figure 15 South Western system freight volumes by commodity and year ('000 tonnes)

Adapted from Queensland Rail, sub. 33, attachment A, p. 7.

There is substantial variation in grain freight volumes on the South Western system from year to year. This is due to seasonal variations in agricultural production, including a combination of supply-side factors (e.g. weather) and demand-side factors (e.g. export prices), which determine the volume of grain produced and exported each year:

Grain export volumes vary from year to year depending on weather and market conditions, but on average around 2 million tonnes per annum are exported out of Queensland, with approximately 40% of this transported by rail to export facilities.³⁵⁸

Additionally, the freight volumes for 2013–14 and 2014–15 may be attributable to containerised cotton shipments for export, which have ceased:

There has been significant change in the transport market in the South West, with all cotton movements from the South West switching from rail to road from 2014-15.³⁵⁹

Western Route traffic

The Western system carries agricultural products, primarily grain and livestock.³⁶⁰ Figure 16 shows the freight volumes on the Western system annually from 2013–14 to 2017–18.³⁶¹

³⁵⁸ GrainCorp, sub. 52, p. 11.

³⁵⁹ Queensland Rail, sub. 33, p. 37, para. 180.4; Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 18.

³⁶⁰ Queensland Rail, sub. 33, attachment A, p. 4.

³⁶¹ Queensland Rail reports disaggregated volume data for various sections of the Western system (e.g. the Miles to Roma section and the Roma to Charleville section). To avoid double-counting of volumes, the data shown represents the maximum throughput on any individual section of the system. For the Western system, this is the data for the Miles to Roma section.



Figure 16 Western system freight volumes by commodity and year ('000 tonnes)

Adapted from Queensland Rail, sub. 33, attachment A, p. 8.

As with the South Western system, it can be seen that there is substantial variation in grain volumes on the Western system from year to year. The QCA understands that grain on the Western system is typically transported to the Port of Brisbane (Fisherman Islands) for export.³⁶²

Additionally, livestock rail services operate on the Western system. These services are operated by Aurizon Operations through the Livestock Transport Services Contract with the Queensland Government. According to the Department of Transport and Main Roads (TMR):

These rail services transport large herds of cattle from regional hubs across the State to processing on Queensland's eastern coast. The contract supports non-commercial livestock services and ensures a minimum standard is offered to the beef industry.

Livestock rail services operate in:

• south west region - 27 services annually ...³⁶³

Livestock on the Western system are typically transported to the abattoirs at Dinmore (west of Brisbane) and Beenleigh (south of Brisbane).³⁶⁴ The operation of the Transport Services Contract subsidies on the agricultural systems in the context of criterion (a) is discussed in section 8.4.3.

Central Western Route traffic

The Central Western system carries grain from the Clermont branch, cattle from Winton, Longreach and Clermont, and small volumes of intermodal freight (food and groceries, retail products, industrial products and agricultural inputs) into Emerald, Alpha, Barcaldine and Longreach.³⁶⁵

³⁶² GrainCorp, sub. 52, p. 4.

³⁶³ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019, accessed 25 July 2019, https://www.tmr.qld.gov.au/business-industry/Transport-sectors/Rail-services-andinfrastructure/Rail-Transport-Contracts-and-Agreements.

³⁶⁴ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 72.

³⁶⁵ Queensland Rail, sub. 33, attachment A, p. 5; Linfox, sub. 50, para. 3.5.

Figure 17 shows the freight volumes on the Central Western system annually from 2013–14 to 2017–18.³⁶⁶



Figure 17 Central Western system freight volumes by commodity and year ('000 tonnes)

Adapted from Queensland Rail, sub. 33, attachment A, p. 13.

As with the South Western and Western systems, it can be seen that there is substantial variation in grain volumes on the Central Western system from year to year. The QCA understands that the grain on the Central Western system is typically transported to the Port of Gladstone and/or the Port of Mackay for export.³⁶⁷

Livestock rail services also operate on the Central Western system. As with the Western system, these livestock services are operated by Aurizon Operations through the Livestock Transport Services Contract with the Queensland Government. According to TMR:

Livestock rail services operate in:

- south west region 27 services annually
- central west region 184 services annually
- north west region 114 services annually.³⁶⁸

Livestock on the Central Western system may be transported to the abattoirs in Brisbane or in Rockhampton for processing.³⁶⁹ Queensland Rail announced a Yeppoon (Rockhampton area) rail line upgrade in its 2017–18 Annual and Financial Report:

Restoring the rail line will strengthen Rockhampton's beef industry and create an additional processing plant in the region, broadening options for regional producers and providing an

³⁶⁶ Queensland Rail reports disaggregated volume data for various sections of the Central Western system (e.g. the Nogoa to Emerald, Emerald to Longreach, and Longreach to Winton sections). To avoid double-counting of volumes, the data shown represents the maximum throughput on any individual section of the system. For the Central Western system, this is the data for the Nogoa to Emerald section.

³⁶⁷ GrainCorp, sub. 52, p. 4.

³⁶⁸ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019.

³⁶⁹ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 72.

opportunity to attract new producers. The upgraded rail line will unlock access to the abattoir, allowing livestock to be transported on rail directly from the north west and central west regions of Queensland to Rockhampton for processing.³⁷⁰

Additionally, general freight services operate on the Central Western system. These general freight services are operated by Linfox through the Regional Freight Transport Services Contract with the Queensland Government. According to TMR:

This contract ensures regional communities can access freight services at subsidised freight rates and ensures a minimum standard of service is delivered.

The contract supports about 2,400 road and rail freight services each year in regional Queensland:

- south west region: approximately 1,560 road freight services each year
- central west region: 260 road freight services and 208 rail freight services on the Central West Line each year
- north west region: 364 road freight services each year.

TMR transferred the contract to Linfox in early-2019, after the previous operator, Aurizon, sold Linfox its Queensland intermodal business, which delivers contracted regional freight services.³⁷¹

Linfox describes the types of goods it transports on the Central Western system:

Linfox transports food & grocery, liquor, retail products, industrial products and agricultural inputs for its customers on its train services into this region [central west Queensland] to locations such as Longreach, Barcaldine, Emerald and Alpha.³⁷²

The QCA understands this general freight task on the Central Western system is typically carried as intermodal containerised freight, similar to that on the North Coast Line (section 5.5.2).³⁷³

8.4.3 Transport Service Contract subsidies

The Queensland Government provides subsidies for the delivery of both above-rail services and below-rail services in Queensland. These subsidies are provided through the TSCs, which are managed on behalf of the Queensland Government by TMR.³⁷⁴

For the purposes of the QCA's analysis, the subsidies can be generally classified into four categories:

- (1) delivery of below-rail services by Queensland Rail
- (2) delivery of above-rail passenger services by Queensland Rail
- (3) provision of above-rail general freight services by an above-rail operator (currently Linfox—post-2019; formerly Aurizon Operations—pre-2019)
- (4) provision of above-rail livestock (cattle) services by an above-rail operator (currently Aurizon Operations).

³⁷⁰ Queensland Rail, *Annual and Financial Report 2017–18*, September 2018, p. 41, accessed 25 July 2019, https://www.queenslandrail.com.au/about%20us/Documents/Queensland%20Rail%20-%20Annual%20and%20Financial%20-%20Report%20-%202017-18.pdf.

³⁷¹ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019

³⁷² Linfox, sub. 50, para. 3.5.

³⁷³ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, pp. 71–72.

³⁷⁴ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, pp. 71–72.

(1) Subsidies for the delivery of below-rail services by Queensland Rail

Queensland Rail receives revenue under the Rail Transport Service Contract for the delivery of below-rail services and the maintenance of the network infrastructure across the Queensland Rail network.³⁷⁵

Below-rail services are the activities associated with the provision and management of rail infrastructure, including the construction, maintenance and renewal of rail infrastructure assets, and the network management services required for the safe operation of train services, including train control services and the implementation of safe working procedures.³⁷⁶

The Queensland Parliament Transport, Housing and Local Government Committee describes this subsidy as an infrastructure subsidy:

Queensland Rail receives funds from the State Government through a Rail Transport Services Contract for investment and maintenance in below-rail assets. The key purpose of the Contract is to ensure that the state of the infrastructure network is fit for service for above-rail operators. The subsidy is provided for the gap between access fees and the cost of maintenance.³⁷⁷

Queensland Rail is required under its 2016 access undertaking to publish financial statements annually for the below-rail services provided by Queensland Rail.³⁷⁸ The latest below-rail financial statement shows that in 2017–18 Queensland Rail received approximately \$567 million in TSC subsidies for the delivery of below-rail services on the Queensland Rail network.³⁷⁹ These subsidies were applied for below-rail services on each of Queensland Rail's systems (except for the Mount Isa Line), with approximately 0.13 per cent of the subsidy allocated to below-rail services on the West Moreton system, 27 per cent for the North Coast Line and 73 per cent for the rest of the network.³⁸⁰ The Mount Isa Line is regarded as commercial, so it does not receive any below-rail subsidy—the maintenance and upgrades on the Mount Isa Line are solely funded from the access fees paid by above-rail operators/miners who use that service.³⁸¹

(2) Subsidies for the delivery of above-rail passenger services by Queensland Rail

Queensland Rail also receives revenue under the Rail Transport Service Contract for the delivery of above-rail passenger services, including commuter Citytrain services on the Metropolitan

³⁷⁵ Under the Rail Transport Service Contract, below-rail services provided on the regional infrastructure are accounted for separately from below-rail services provided on the Brisbane Metropolitan network; however, this distinction is not critical to the QCA's analysis.

³⁷⁶ Queensland Rail, Financial Statements for the Year Ended 30 June 2018 - Below Rail Services Provided by Queensland Rail, December 2018, p. 3, accessed 30 July 2019, https://www.queenslandrail.com.au/business/acccess/Compliance%20and%20reporting/2017-18%20Below%20Rail%20Financial%20Statements.pdf.

³⁷⁷ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 43.

³⁷⁸ Queensland Rail, *Access Undertaking 1*, approved by the QCA 11 October 2016, p. 46, https://www.queenslandrail.com.au/business/access/Access%20Undertaking%20and%20related%20documents/ Queensland%20Rail%20Access%20Undertaking%201%202016.pdf.

³⁷⁹ Queensland Rail, *Financial Statements for the Year Ended 30 June 2018—Below Rail Services Provided by Queensland Rail*, December 2018, p. 4.

³⁸⁰ The 'rest of the network' includes below-rail services on the Metropolitan, South Western, Western, Central Western and Tablelands systems.

³⁸¹ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 43.

system, as well as long-distance travel and tourism passenger trains that operate across the regional network.³⁸²

Queensland Rail's latest annual report shows that in 2017–18 Queensland Rail received approximately \$1.1 billion for the delivery of above-rail passenger services.³⁸³ According to Queensland Rail, approximately 54 million trips were taken on the Metropolitan system in 2017–18, and more than 750,000 customers travelled on the regional travel and tourism passenger services (with approximately 55% of these customers travelling on the Kuranda Scenic Railway tourist service, which runs 34 km between Cairns and Kuranda on the Tablelands system).³⁸⁴

(3) Subsidies for the provision of above-rail general freight services by an above-rail operator

The Queensland Government subsidises an above-rail operator to provide rail freight services in regional Queensland through its Regional Freight Transport Service Contract.³⁸⁵ According to TMR:

This contract ensures regional communities can access freight services at subsidised freight rates and ensures a minimum standard of service is delivered.

The contract supports about 2,400 road and rail freight services each year in regional Queensland:

- south west region: approximately 1,560 road freight services each year
- central west region: 260 road freight services and 208 rail freight services on the Central West Line each year
- north west region: 364 road freight services each year.³⁸⁶

Linfox is the current above-rail (and road) operator providing regional freight services under this subsidy. TMR transferred the contract to Linfox in early 2019, after the previous operator, Aurizon Operations, sold its Queensland intermodal business to Linfox.³⁸⁷

The Regional Freight TSC subsidises the transport of general freight to regional Queensland communities, both producing and receiving freight. The goods moved include food and grocery products, retail products, manufactured products, industrial products and agricultural inputs.³⁸⁸

The value of the contract is commercial-in-confidence between TMR and the above-rail operator. The Queensland Parliament Transport, Housing and Local Government Committee estimated that the value of the Regional Freight TSC was approximately \$140 million in 2012–13. However, that amount was based on a TSC that facilitated 3,224 rail services and 2,756 road

³⁸² For a list of these regional passenger services, see the Queensland Rail Travel website at https://www.queenslandrailtravel.com.au/.

³⁸³ Queensland Rail, Financial Statements for the Year Ended 30 June 2018 - Below Rail Services Provided by Queensland Rail, December 2018, p. 15; Queensland Rail, Annual and Financial Report 2017–18, September 2018, p. 73 (p. 7 of the Financial Report).

³⁸⁴ Queensland Rail, Annual and Financial Report 2017–18, September 2018, pp. 19, 42.

³⁸⁵ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019.

³⁸⁶ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019.

³⁸⁷ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019.

³⁸⁸ Linfox, sub. 50, para. 3.5; Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 71.

services at that time (compared to 2,400 road and rail services (together) under the current TSC).³⁸⁹ There is no publicly available data on the value of the current contract.

Queensland Rail cannot provide above-rail freight haulage services, as it is not permitted to do so under the Queensland Rail Limited Constitution.³⁹⁰ Therefore, Queensland Rail does not receive subsidies under the Regional Freight Transport Service Contract.

(4) Subsidies for the provision of above-rail livestock services by an above-rail operator

The Queensland Government subsidises an above-rail operator to provide rail freight services specifically for the transport of cattle in regional Queensland through its Livestock Transport Service Contract.³⁹¹ The Livestock TSC is exclusive to rail; that is, no cattle moved by road attract the subsidy.³⁹² According to TMR:

The Queensland Government offers 325 rail cattle services each year through the Livestock Transport Services Contract with rail operator Aurizon.

These rail services transport large herds of cattle from regional hubs across the State to processing facilities on Queensland's eastern coat. The contract supports non-commercial livestock services and ensures a minimum standard is offered to the beef industry.

Livestock rail services operate in:

- south west region 27 services annually
- central west region 184 services annually
- north west region 114 services annually.
- .

A procurement process for a future cattle rail contract is now underway.³⁹³

Aurizon Operations is the current above-rail operator providing livestock rail services under this subsidy. In relation to the transport of livestock on rail, Queensland Rail noted:

Queensland is the only State [in Australia] where cattle are still transported by rail and the traffic is subsidised under the Livestock Transport Services Contract between the Queensland Government and Aurizon. Without such subsidies, it would not be commercially viable for these services to be provided.³⁹⁴

³⁸⁹ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 71.

³⁹⁰ Queensland Rail, *Draft Queensland Rail Access Undertaking 1*, explanatory submission, March 2012, p. 15, accessed 30 July 2019, https://www.qca.org.au/wp-content/uploads/2019/06/8354_r-qrail-qrail2012dau-expdocs2012dau-0412.pdf.

³⁹¹ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019.

³⁹² Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 72.

³⁹³ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019, accessed 28 January 2020. At the time of writing, it appears from the TMR website that this procurement process is still ongoing.

³⁹⁴ Queensland Rail, sub. 33, p. 37, para. 180.2.

The Queensland Parliament Transport, Housing and Local Government Committee also noted that Queensland is the last state in Australia to transport livestock by rail, and that this is due to the large distances required to source and transport cattle for processing in Queensland.³⁹⁵

The value of the contract is commercial-in-confidence between TMR and the above-rail operator. The Queensland Parliament Transport, Housing and Local Government Committee estimated that the value of the Livestock TSC was approximately \$28 million in 2012–13, which provided 325 rail services in that year from regional locations through to the various abattoirs.³⁹⁶ There is no publicly available data on the value of the current contract.

Queensland Rail cannot provide above-rail freight haulage services, as it is not permitted to do so under its Constitution. Therefore, Queensland Rail does not receive subsidies under the Livestock Transport Service Contract.

Subsidies (3) and (4) are applicable in the above-rail freight haulage markets on the agricultural systems

Of the four categories of TSC subsidies described above, only the Regional Freight TSC (subsidy (3)) and Livestock TSC (subsidy (4)) are relevant for the QCA's analysis of the above-rail freight haulage markets on the agricultural systems. These two subsidies are provided to above-rail freight operators on the agricultural systems.

The operation of the TSC subsidies is an important part of the above-rail freight haulage markets on the agricultural systems.

8.4.4 The features of the existing above-rail freight haulage markets on the agricultural systems

Three above-rail operators on the agricultural systems

In the current markets for above-rail freight haulage services on the agricultural systems, the above-rail operators are:

- Linfox—from early 2019, Linfox has operated general freight services on the Central Western Route; these services are subsidised through the Regional Freight TSC with the Queensland Government³⁹⁷
- Aurizon Operations—currently, Aurizon Operations continues to operate livestock freight services on the Western and Central Western Routes (as well as the Mount Isa Route); these services are subsidised through the Livestock TSC with the Queensland Government, although the QCA understands that the TMR is currently undertaking a procurement process for a future Livestock Transport Services Contract³⁹⁸
- Watco—from late 2019, Watco has commenced operating grain freight services on the South Western, Western and Central Western Routes, taking over the grain rail task on these

³⁹⁵ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 20.

³⁹⁶ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 72.

³⁹⁷ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019; Linfox, sub. 50, paras 3.3–3.7.

³⁹⁸ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019, accessed 28 January 2020. At the time of writing, it appears from the TMR website that this procurement process is still ongoing.

systems, which was previously hauled by Aurizon Operations.³⁹⁹ The QCA understands that Watco will not receive a subsidy for operating its grain freight services.⁴⁰⁰

Before 2019, Aurizon Operations was the only above-rail freight haulage operator on the agricultural systems. During that time, Aurizon Operations operated both subsidised regional freight and livestock freight services (pursuant to the TSCs), as well as non-subsidised commercial freight services, such as for the haulage of grain and cotton on the agricultural systems.

No evergreen renewal rights in access agreements

The QCA understands that access agreements between Queensland Rail and above-rail operators for non-subsidised services on the agricultural systems (such as Watco's haulage of grain) are typically for a period of 10 years.⁴⁰¹ Existing access agreements with Queensland Rail typically do not provide evergreen renewal rights for the terms of access. Terms under these agreements (e.g. in relation to pricing, capacity allocation or usage of facilities) will progressively expire and must be renegotiated for new contracts within the bounds of an approved access undertaking.

The QCA has no visibility over access arrangements for subsidised services provided by aboverail operators pursuant to a TSC. Thus, it cannot be said with certainty whether access for these subsidised services are contracted for in a similar way to commercial services (e.g. a 10 year access agreement between the above-rail operator and Queensland Rail), or whether special arrangements apply for above-rail services provided pursuant to a TSC.

Spare capacity exists on each of the agricultural systems

Data provided by Queensland Rail shows that there is currently significant spare capacity on each of the South Western, Western and Central Western systems.⁴⁰² The effect of spare capacity on Queensland Rail's ability and incentive to exercise market power is discussed in section 5.5.1.

8.5 Queensland Rail's ability and incentive to exercise market power

Whether access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a material increase in competition in the dependent aboverail freight haulage markets on each of the agricultural systems depends firstly on whether Queensland Rail has market power that could be used to adversely affect competition in the dependent markets; and secondly on whether Queensland Rail has an ability and incentive to exercise that market power, in a future without declaration.⁴⁰³

In its submission in response to the QCA's draft recommendation, Queensland Rail has argued that it has no ability or incentive to exercise market power in any dependent market, including

³⁹⁹ Watco, sub. 48, pp. 1–2; GrainCorp, sub. 52, p. 5.

⁴⁰⁰ GrainCorp, sub. 52, p. 6. See also Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 71, where the Committee said: 'There is currently no subsidy of non-livestock agricultural freight (cotton, grain, sugar etc.) in Queensland'.

⁴⁰¹ Watco, sub. 49, p. 1; Queensland Rail, sub. 51, pp. 1–2.

⁴⁰² Queensland Rail, sub. 33, attachment A, pp. 4–5, 7–8, 13.

⁴⁰³ For example, NCC, Declaration of Services, A guide to declaration under Part IIIA of the Competition and Consumer Act 2010 (Cth), April 2018 edn, p. 33, para. 3.26; Duke Eastern Gas Pipeline [2001] ACompT 2 at [116]; Queensland Rail, sub. 33, p. 19, para. 100; South West Producers, sub. 40, p. 19; Glencore, sub. 41, p. 14.

that it is constrained by various factors such as competition from road transport.⁴⁰⁴ These issues are considered below in relation to the agricultural systems. The operation of access arrangements in a future without declaration as a possible constraint on Queensland Rail's ability and incentive to exercise market power is considered in section 5.5.3.

8.5.1 Competition between road freight and rail freight

Queensland Rail argued that it has no ability or incentive to exercise market power to adversely affect competition in any dependent market because it is materially constrained in the provision of below-rail services by road freight operators:

Only a very low volume of freight is transported on the Central Western, Western and South Western Systems. The freight transported is primarily grain and livestock. As outlined above in respect of the North Coast Line, such freight is highly contestable by road.⁴⁰⁵

The main types of freight carried on the agricultural systems are intermodal general freight, grain and livestock. The degree of competition between road and rail for each of these freight tasks is considered below.

Grain

In Queensland, grain intended for domestic consumption is exclusively carried by road to its destinations (e.g. food manufacturers for human consumption, feedlots and farms for animal consumption).⁴⁰⁶ The advantage of road transport in the domestic grain freight task is likely to be due to the ability of road transport to easily adapt to the demand of transporting different types and quality of grain (e.g. for human or animal consumption) to different customers, located in geographically diverse locations.

In contrast, grain intended for export is more suited to the rail task as it is typically transported in bulk form to one destination for export (e.g. the port). However, even for grain intended for export, road transport is used as a viable alternative (or complement) to rail transport across Australia. For example, GrainCorp viewed rail as providing 'baseload capacity' for grain transport, with road being used to provide 'peaking capacity' in times of high demand.⁴⁰⁷ According to GrainCorp figures:

Over the past decade, Queensland grain production has averaged approximately 3.5 million tonnes per annum (mtpa). Of this, around 44% is consumed in the domestic market, being for human consumption (flour etc) or for stockfeed and industrial uses. The remainder is exported.

Queensland exports are usually derived from stocks that are surplus to the domestic market or of higher value to the producer than the domestic market is willing to pay (such as high protein wheat destined for the Japanese noodle market). Exports are therefore quite variable, but on average around 2mtpa are exported out of Queensland ...⁴⁰⁸

From late 2019, the above-rail operator Watco has commenced providing grain haulage services under a seven-year contract with GrainCorp.⁴⁰⁹ Grain will be transported from designated GrainCorp receival and loading sites on the Queensland Rail South Western, Western and

⁴⁰⁴ Queensland Rail, sub. 33, p. 15, para. 79.

⁴⁰⁵ Queensland Rail, sub. 33, p. 36, para. 175.

⁴⁰⁶ GrainCorp, sub. 52, p. 5.

⁴⁰⁷ GrainCorp, sub. 52, p. 7.

⁴⁰⁸ GrainCorp, sub. 52, p. 4.

⁴⁰⁹ Watco Companies, Rail Services: Australia, accessed 7 February 2020, https://www.watcocompanies.com/services/rail/australia/; *Tank News International*, 'NRE to build Watco Australia narrow-gauge locomotives', 3 December 2018, accessed 31 July 2019, https://www.tanknewsinternational.com/nre-to-build-watco-australia-narrow-gauge-locomotives/.

Central Western systems⁴¹⁰ to the Ports of Mackay, Gladstone and Brisbane (Fisherman Islands) for export. Figure 14 above shows a map of GrainCorp's grain operations on the Queensland Rail (and Aurizon Network) systems. This task is estimated to be approximately 1 million tonnes per annum in total (across all rail systems).⁴¹¹

In recent times, there has been a shift away from the transport of grain as a bulk commodity to containerised lots.⁴¹² Containerisation allows producers to differentiate their product, for example in markets for organics or premium products, compared with bulk transport where grain from many producers are mixed together for transport as a homogenous good.⁴¹³ Publicly available evidence suggests that containerised grain is transported almost entirely by road—because of the low axle limits (15.75 tonne axle load and less) on the agricultural systems, containerised grain cannot be freighted by rail due to its weight.⁴¹⁴

Given the evidence discussed above, the QCA considers that rail does compete with road for the transport of grain, particularly bulk grain intended for export, and competes primarily on price. Based on the evidence, road appears to be the preferred transport mode for grain intended for domestic consumption, and grain transported in containerised form (whether for domestic consumption or export).

Cattle/livestock

Queensland is the only state in Australia where cattle are transported by rail.⁴¹⁵ According to the Queensland Parliament Transport, Housing and Local Government Committee:

This [the movement of cattle by rail] is due to the large distances required to source and transport cattle for processing in this State.⁴¹⁶ While the State Government provides a subsidy for transporting cattle to processors through a Transport Service Contract it is estimated that less than 10% of Queensland cattle are consigned to an abattoir by train.⁴¹⁷

Queensland Rail also contended that these livestock rail services are only viable due to the subsidies provided:

[T]he [rail] traffic is subsidised under the Livestock Transport Services Contract between the Queensland Government and Aurizon. Without such subsidies, it would not be commercially viable for these services to be provided.⁴¹⁸

 ⁴¹⁰ In addition, some grain will also be transported on Aurizon Network's Goonyella, Blackwater and Moura systems.
⁴¹¹ Watco, sub. 48, p. 1.

⁴¹² Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 91.

⁴¹³ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 91; Lamb, K, 'Tour grapples with pinch points and growth opportunities', *Rural Weekly*, 10 July 2019, accessed 5 August 2019, https://www.weeklytimesnow.com.au/news/rural-weekly/tour-grapples-with-pinch-points-and-growth-opportunities/news-story/cfed534b7155629e70835c5750c65af8.

⁴¹⁴ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 24.

⁴¹⁵ Queensland Rail, sub. 33, p. 37, para. 180.2; Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 20.

⁴¹⁶ Cattle train services operate on the Western and Central Western systems, with a distance of approximately 950 km from Quilpie (on the Western system) to the abattoirs in Brisbane, and a distance of approximately 870 km from Winton (on the Central Western system) to the abattoirs in Rockhampton.

⁴¹⁷ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 20.

⁴¹⁸ Queensland Rail, sub. 33, p. 37, para. 180.2.

Other movements of cattle in Queensland are almost entirely undertaken by road, including transporting livestock domestically and transporting livestock to the port for export. Transport tasks undertaken by road include:

- moving livestock from feedlots to abattoirs (the remaining 90% that is not carried by rail)
- moving livestock from paddock to paddock depending on the availability of feed
- moving livestock from paddocks to feedlots across the state
- moving livestock from paddocks and feedlots to ports for export.⁴¹⁹

Given the evidence discussed above, the QCA considers that rail does compete with road for the transport of livestock, particularly domestic movements of livestock from feedlots to abattoirs for processing, although it may be the case that rail is only competitive with road due to the subsidies provided. Based on the evidence, road appears to be the preferred transport mode for a significant majority of the livestock transport task in Queensland.

Containerised freight

Containerised freight on the agricultural systems is mainly carried on the Central Western Route. Queensland Rail described this regional freight task as follows:

Aurizon [now Linfox] provides subsidised freight services on the Central Western System pursuant to the Regional Freight Transport Services Contract with the Queensland Government. Without such subsidies, it would not be commercially viable for these services to be provided.⁴²⁰

The issue of competition between road and rail for the transport of containerised (intermodal) freight is discussed in detail in section 5.5.2.

Consistent with the evidence discussed in that section⁴²¹, the QCA considers that the transport of containerised (intermodal) freight on the Central Western Route (and other agricultural systems) is technically able to be carried by either road or rail. Rail does compete with road transport for the transport of containerised freight, and competes primarily on price. There is little available data on the volumes that are carried by road compared to the volumes carried by rail.

The presence of competition for this freight task was acknowledged by Linfox, the above-rail freight operator that transports containerised freight on the Central Western Route:

Linfox transports food & grocery, liquor, retail products, industrial products and agricultural inputs for its customers on its train services into this region to locations such as Longreach, Barcaldine, Emerald and Alpha. All of this freight could be transported by road and there is a body of evidence of switching between the transport modes in these regions.⁴²²

Additionally, Queensland Rail presented evidence of switching between rail and road for the transport of containerised cotton freight on the South Western Route in the past:

There has been significant change in the transport market in the South West, with all cotton movements from the South West switching from rail to road from 2014-15.⁴²³

⁴¹⁹ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 15.

⁴²⁰ Queensland Rail, sub. 33, p. 37, para. 180.3.

⁴²¹ For example in relation to the effect of product type and the effect of distance on the competitiveness of rail and road in transporting containerised freight.

⁴²² Linfox, sub. 50, para. 3.5.

⁴²³ Queensland Rail, sub. 33, p. 37, para. 180.4.

This switch to road for the transport of containerised freight on the South Western system is reflected in Queensland Rail data, which show the transport of containerised freight on the South Western system ceasing from 2014–15.

The presence of competition between road and rail for containerised freight across regional Queensland is also reflected in the terms of the Regional Freight TSC. For example, the Regional Freight TSC supports 260 road freight services in the central west region, as well as 208 rail freight services on the Central Western system. This suggests that regional freight in the central western region is carried by a mix of road and rail (with a subsidy provided for both types of transport in order to ensure a minimum standard of service).⁴²⁴

Given the evidence discussed above, the QCA considers that rail does compete with road for the transport of containerised freight on the agricultural systems, in particular the Central Western Route.

Does competition from road transport constrain Queensland Rail's ability and incentive to exercise market power?

The QCA considers that rail transport does compete with road transport for the transport of all three of the main freight tasks on the agricultural systems: the transport of grain, livestock and containerised freight.

Figure 18 is a graphical representation of the composition of prices charged by road and rail operators on the agricultural systems for the transportation of grain, livestock or containerised freight.

Figure 18 Road and rail prices on the agricultural systems for the transport of containerised freight, grain or livestock



Diagram for illustrative purposes only. Not drawn to scale.

The road price

The road price represents the final price paid by freight owners or freight forwarders (e.g. bulk grain handlers) to transport their goods by road, for a particular transport task (origin-destination pair) on the agricultural systems. There is no 'below-road' price in Queensland—that is, trucks accessing major regional road networks in Queensland typically do not pay an access

⁴²⁴ Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019.

fee. Therefore, the road price is expected to reflect largely the costs of the trucking operator (for example, labour costs, vehicle costs, fuel costs, as well as a profit margin).

The rail price

The rail price represents the final price paid by freight owners or freight forwarders to transport their goods by rail, for the same particular transport task (origin-destination pair) on the agricultural systems. The total rail price consists of two components—an above-rail component, which represents the charges of the above-rail operator, and a below-rail component, which represents the charges of the below-rail operator (Queensland Rail).

On the agricultural systems, above-rail operators (such as Watco) contract with Queensland Rail directly for access, and negotiate a train path allocation and below-rail access price. Above-rail haulage operators then add their above-rail charges (for example, labour costs, rollingstock costs, fuel costs, as well as a profit margin) to the below-rail access price to produce a final total rail price, which is the price offered to the 'end customers', for example freight forwarders and bulk product handlers (such as GrainCorp), or individual producers.⁴²⁵

Queensland Rail's ability and incentive to exercise market power is constrained with respect to end customers in the above-rail haulage markets on the agricultural systems

The QCA considers that rail transport competes with road transport for the transport of the three main types of freight tasks on the agricultural systems: the transport of grain, livestock and containerised freight.

For these contestable freight tasks, the road price (\$X in Figure 18) acts as a constraint on the rail price. This is because the main types of goods carried on the agricultural systems (grain, livestock and containerised freight) can be transported by road or rail. Therefore, if the total rail price exceeds the road price, end customers can choose to transport their goods by road instead, and rail operators will lose market share to road operators for the transport of this freight. These end customers may include:

- in the case of grain, bulk grain handlers such as GrainCorp
- in the case of livestock, meat processors or individual producers
- in the case of containerised freight, a variety of end customers, such as retailers, wholesalers and manufacturers.

As a result, the above-rail price and below-rail price are collectively constrained by the road price. It is open for the above-rail operator and below-rail operator to seek to increase their share of the total rail price; however, the absolute amount of this total rail price cannot rise above the constraint of the road price. If it does, end customers may switch from rail transport to road transport, and this will in the long run negatively affect both the above-rail and below-rail operators.

The QCA considers that Queensland Rail is constrained in its ability and incentive to exercise market power against the end customers on the agricultural systems by competition from road transport. If Queensland Rail sought to raise the below-rail price, and the above-rail operator is assumed to pass on this increase in full, the total rail price may exceed the road price, and if it does, end customers may switch from using rail transport to using road transport to transport

⁴²⁵ Watco, sub. 48, p. 2; GrainCorp, sub. 52, p. 5; Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, pp. 64, 74–76.

their goods. Therefore, Queensland Rail is constrained in its ability and incentive to exercise market power against these end customers.

Queensland Rail's ability and incentive to exercise market power is not constrained with respect to above-rail operators in the above-rail freight haulage markets

The QCA considers that Queensland Rail is not constrained in its ability and incentive to exercise market power against the above-rail haulage operators on the agricultural systems by competition from road transport.

As discussed above, given that rail competes with road for the transport of the main types of goods typically carried on the agricultural systems, the above-rail operator and the below-rail operator together face a collective constraint on the total rail price that can be charged—that constraint is the road price for the equivalent transport task (origin–destination pair).

In a future without declaration, Queensland Rail can exercise its market power, in order to maximise its profits. It may seek to exercise this market power by raising the below-rail access charge that it charges the above-rail operator(s) on the agricultural systems, and seek to claim a larger share of the total rail price (the 'Rail price 2' scenario in Figure 18).

In this scenario, the above-rail operator(s) cannot simply switch to using road transport in response to the increase in the below-rail access charge, as rollingstock cannot be easily converted for use on roads. The above-rail operator(s) will have made significant sunk investments into long-life assets, such as locomotives and wagons, often specifically configured for use for the transport task on the agricultural systems (e.g. narrow gauge, specific tonne axle loads), which cannot be easily redeployed elsewhere. As Watco noted:

As a new entrant into the Queensland above rail market, Watco is having to incur significant sunk costs through investing in long-life rollingstock assets (with a typical 25 year+ life) as well as the associated maintenance and provisioning facilities. Given the specific operational characteristics of the Central Western, Western and South Western systems, such as a narrow-gauge track with specific tonnage allowances, such assets are unable to be readily switched to alternative uses elsewhere.⁴²⁶

Publicly available industry information on Watco notes the significance of the sunk investments that need to be made in order for Watco to enter above-rail markets on the agricultural systems in Queensland:

Watco has ordered eight 1067mm-gauge [narrow gauge] diesel locomotives from National Railway Equipment Corporation (NREC), United States, for the contract, along with a new fleet of grain wagons, which will be manufactured in China. Because of the lead times required, start-up for the contract [with GrainCorp] is expected in the fourth quarter of 2019.⁴²⁷

NREC, the locomotive manufacturer, notes on its website the degree of asset specificity required to support Watco's entry, given the specific engineering requirements needed to be met in order to operate on the Queensland Rail agricultural systems, which are narrow-gauge, with specific permissible tonne axle loads and specific speed limitations:

Watco demanded the utilisation of a proven medium-speed diesel prime mover ... The Queensland Australia network requires a very specific and unique set of design parameters ... The locomotives must be narrow-gauge (1.067mm). They must have six axles, each able to carry

⁴²⁶ Watco, sub. 48, p. 4; GrainCorp, sub. 52, p. 6.

 ⁴²⁷ Carter, M, 'Watco expands Australian rail freight operations', *International Railway Journal*, 3 July 2018, accessed 6 August 2019, https://www.railjournal.com/regions/australia-nz/watco-expands-australian-rail-freight-operations/; VanBecelaere, T, 'Watco signs agreement to move Eastern Australia grain', *The Dispatch*, vol. 19, no. 7, July 2018, p. 2, http://www.watcocompanies.com/pdfs/Dispatch2018/07July2018Web.pdf.

15.75 metric tonnes, with new NRE bogies. They must have 2,250 traction horsepower with a minimum radius curve of 80mm. They must weigh no more than 94.5 metric tonnes with dimensions that do not exceed 3,900mm in height, 2,850mm in width and 19,018 in length.

Requirements also include a 12-645E3B diesel engine, an AR10/D14 main alternator, companion, 761 traction motors, a WBO compressor. The locomotive must have at least 27 percent adhesion available during traction mode with 99 percent reliability in all weather conditions and must demonstrate 250kN of adhesion-limited tractive effort.⁴²⁸

Given the significance of the sunk investments that the above-rail operator(s) would have made in entering the above-rail markets on the agricultural systems, the QCA considers that if Queensland Rail raised the below-rail access charge on these systems, the above-rail operator cannot make a credible threat to exit the market or to switch to road transport in response. Furthermore, the above-rail operator faces the collective constraint (the road price), and cannot simply pass through the below-rail access charge increase, because if the total rail price exceeds the road price, the above-rail operator is likely to begin losing end customers (who may switch to road transport to transport their goods).

For an existing above-rail operator, arguably this scenario can be regarded as a transfer of wealth between the above-rail operator and the below-rail operator, with little impact on competition in the above-rail market. However, the QCA considers that the critical issue is that a potential above-rail operator seeking to enter the above-rail market, or an existing operator seeking to reinvest in the market, can foresee this risk—that any sunk investments it makes may be exposed to the risk of expropriation by the below-rail operator in a future without declaration. This is the hold-up risk (discussed in detail in section 8.6.2).

In summary, the ability of Queensland Rail to exercise market power in future contracting periods, in a future without declaration, creates a significant degree of uncertainty for potential market participants at the time they are considering investment, raising the hurdle rate required to justify the investment, and thereby potentially preventing efficient entry and efficient participation in the above-rail freight haulage market.

Therefore, the QCA does not consider that competition from road will be an effective constraint on Queensland Rail's ability and incentive to exercise market power against the above-rail operators in the above-rail haulage markets on the agricultural systems. The QCA considers that these above-rail operators will be exposed to the risk of hold-up in a future without declaration.

8.5.2 Transport Service Contract subsidies

Subsidised above-rail services

The Regional Freight TSC and the Livestock TSC subsidies are provided to some above-rail operators for the haulage of specific goods (general freight and cattle respectively) on some parts of the agricultural systems. These subsidies add a layer of complexity when considering Queensland Rail's ability and incentive to exercise market power in the above-rail freight haulage markets on the agricultural systems.

In particular, the complexity arises because of a lack of public information relating to the terms of the Regional Freight and Livestock TSCs. This is further complicated by the fact that Queensland Rail also receives a TSC (the Rail TSC), which is separate from the Regional Freight and Livestock TSCs provided to above-rail freight operators.

⁴²⁸ *Tank News International*, 'NRE to build Watco Australia narrow-gauge locomotives', 3 December 2018, accessed 6 August 2019, https://www.tanknewsinternational.com/nre-to-build-watco-australia-narrow-gauge-locomotives/.

Some stakeholders expressed their views on the effect of the TSC subsidies on Queensland Rail's ability and incentive to exercise market power. For example, Linfox said:

Linfox submits that conditions associated with the TSC subsidies do not provide an effective long-term constraint on Queensland Rail's ability to exercise monopoly market power and adversely affect competition in dependent markets. Government policies routinely change and the TSC funding and its conditionality, could be removed, reduced or adjusted at any time.⁴²⁹

GrainCorp said:

While subsidies may be provided for transportation of livestock, so far as GrainCorp is aware, no subsidies are provided for grain haulage. This means that there is real potential for growth in competition for grain haulage, including in the South Western, Western and Central Western systems ...

The fact that some level of subsidy is currently offered to the incumbent above rail operator for some freight tasks does not preclude new entry, nor does it mean that competition would not be promoted by ensuring that access to the below rail service is provided on reasonable terms.⁴³⁰

However, no further details on the terms or operation of the TSC subsidies are provided by stakeholders (this may be due to reasons such as commercial confidentiality).

Given this lack of publicly available information on the terms and the operation of the various TSCs (including both the below-rail and above-rail subsidies), and the interactions (if any) between them, it is unclear whether the subsidies provide any constraint on Queensland Rail's ability and incentive to exercise market power against the above-rail operators who receive these subsidies in the above-rail freight haulage markets on the agricultural systems.

Non-subsidised above-rail services

The TSC subsidies are provided to only some above-rail operators, for the haulage of specific goods, on some parts of the agricultural systems. As a result, the same above-rail operator, or another above-rail operator, may operate services on the agricultural systems that are not subsidised. Based on the information before the QCA, it appears that above-rail services for the haulage of grain on the agricultural systems currently do not receive TSC subsidies (and historically have not received subsidies), and operate on a commercial basis.⁴³¹

The QCA considers that the presence of TSC subsidies does not affect Queensland Rail's ability and incentive to exercise market power against non-subsidised above-rail operators (or aboverail operators operating non-subsidised services) on the agricultural systems. That is, the QCA considers that Queensland Rail does have the ability and incentive to exercise market power with respect to non-subsidised above-rail operators (see section 8.5.1 above).

8.5.3 Other arguments previously raised in relation to other Queensland Rail systems

Queensland Rail submitted that there are other constraints on its ability and incentive to exercise market power to adversely affect competition in the above-rail freight haulage markets on the agricultural systems:

The other material constraints on Queensland Rail in the provision of below rail services for the purposes of transporting freight on the Other Systems include those discussed above in respect of the North Coast Line, namely:

⁴²⁹ Linfox, sub. 50, para. 3.4.

⁴³⁰ GrainCorp, sub. 52, p. 6.

⁴³¹ GrainCorp, sub. 52, p. 6; Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 71.

- (1) Queensland Rail's statutory obligations and position as a statutory authority, including obligations to have approved and comply with strategic and operational plans.
- (2) The threat of regulation or declaration under Parts 3 or 5 of the QCA Act.⁴³²

These issues have been previously raised with respect to the North Coast Line. The QCA has considered these issues in detail in the preceding sections of this report, and considers that analysis can also be applied with respect to the agricultural systems. In particular, the QCA has considered the following points raised by Queensland Rail in sections 5.5.1 and 5.5.3:

- the presence of excess capacity on the rail systems
- Queensland Rail's statutory obligations
- the threat of regulation or declaration.

8.5.4 Conclusions on Queensland Rail's ability and incentive to exercise market power

Based on the analysis above, the QCA considers that Queensland Rail has the ability and incentive to exercise market power in a way that may affect competition in the dependent above-rail freight haulage markets in a future without declaration.

This prompts the question of whether competition would be adversely affected if Queensland Rail exercised this market power. For criterion (a) to be satisfied, it must be established that access (or increased access) to the service, on reasonable terms and conditions, as a result of a declaration of the service would promote a material increase in competition in at least one dependent market. The following sections contain a detailed analysis of the state of competition in the above-rail freight haulage markets on the agricultural systems in a future with and without declaration.

8.6 Competition in the above-rail freight haulage markets in a future with and without declaration

8.6.1 A future with declaration

The QCA considers that a future with declaration will continue to provide for access to the rail network on reasonable terms and conditions due to on-going regulatory oversight. The regulatory regime under Part 5 of the QCA Act will provide an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power in a way that may adversely affect competition in dependent markets. For example, Watco highlighted the benefits of declaration in its view:

Declaration promotes long term access certainty on fair and reasonable terms, a right to have QCA arbitrate access disputes, efficient access pricing and the potential to have reference tariffs, which all promote competition in dependant markets, particularly in regional Queensland agricultural freight markets and above rail operations ...

On the basis of the current track access regulatory regime, Watco has witnessed the success of above rail competition in Queensland and the positive impact this has had for freight customers, the economy and the broader community. The ability for [sic] new above rail entrant like Watco to enter the Queensland market is critically dependant on the stable operation of Queensland's regulatory regime under Part 5 of the QCA Act.⁴³³

⁴³² Queensland Rail, sub. 33, p. 37, para. 181.

⁴³³ Watco, sub. 48, pp. 4–5.

The QCA considers that the regulatory framework in a future with declaration will provide certainty to a potential new market entrant (or renewing access holder) that access to the service will be provided on reasonable terms and conditions. Declaration, and the associated access regime, provides a transparent statutory process within which terms and conditions of access can be negotiated (including to address sunk investments and mitigate the risk of hold-up for access seekers). Additionally, declaration can maintain an appropriate balance between the legitimate interests of the service provider and access seekers/users in the presence of sunk investments. Mitigating the risk of hold-up for access seekers is likely to be a critical factor in supporting efficient entry to and efficient participation in the above-rail freight haulage market, thereby materially promoting competition (the QCA's approach to the concept of materiality is discussed below).⁴³⁴

8.6.2 A future without declaration: the hold-up problem

The QCA's analysis of the hold-up problem—overview

As a business, Queensland Rail has an incentive to maximise profits.⁴³⁵ In a future with declaration, its ability and incentive to exercise its market power in order to maximise profits will be constrained by the regulatory regime. In a future without declaration, the QCA considers that Queensland Rail will not face any effective long-term constraints on its ability and incentive to exercise market power in order to maximise profits.⁴³⁶ In particular, the QCA considers that access arrangements applied by Queensland Rail will not act as an effective constraint—the access framework is considered in detail in Part B, Chapter 4.

It is in this environment that market participants will face decisions to enter or operate in the above-rail markets on the agricultural systems in a future without declaration. In particular, a new entrant to the above-rail haulage markets on the agricultural systems will have to incur significant sunk costs. These include investments in physical assets, which typically have a 20–25-year life, and the associated maintenance and provisioning facilities.⁴³⁷ For example, Watco, a new above-rail operator entering the above-rail markets on the agricultural systems, has had to incur significant sunk investments to facilitate its entry:

Watco has ordered eight 1067mm-gauge [narrow gauge] diesel locomotives from National Railway Equipment Corporation (NREC), United States, for the contract, along with a new fleet of grain wagons, which will be manufactured in China. Because of the lead times required, start-up for the contract [with GrainCorp] is expected in the fourth quarter of 2019.⁴³⁸

⁴³⁴ Efficient participation in the market includes actions undertaken by incumbent market participants, such as investing in operational efficiencies and innovations, as well as reinvestment into the market at the time of contract renewal.

⁴³⁵ Or minimise losses.

⁴³⁶ The presence of TSC subsidies for some above-rail freight haulage services on the agricultural systems may affect Queensland Rail's ability and incentive to exercise market power, but this does not affect Queensland Rail's ability and incentive to exercise market power against the non-subsidised services on the agricultural system—see section 8.5.2.

⁴³⁷ Watco, sub. 48, p. 4; GrainCorp, sub. 52, p. 6.

⁴³⁸ Carter, M, 'Watco expands Australian rail freight operations', *International Railway Journal*, 3 July 2018, accessed 6 August 2019, https://www.railjournal.com/regions/australia-nz/watco-expands-australian-rail-freightoperations/; VanBecelaere, T, 'Watco signs agreement to move Eastern Australia grain', *The Dispatch*, July 2018, accessed 6 August 2019, http://www.watcocompanies.com/pdfs/Dispatch2018/07July2018Web.pdf.

In addition, Watco has committed to building a rail transport facility/depot in Warwick (southern Queensland) for rollingstock maintenance and storage.⁴³⁹

Given the specific operational characteristics of the agricultural systems, such as a narrowgauge track with specific tonne axle loads, assets such as rollingstock are unable to be readily switched to alternative uses elsewhere. In particular, the below-rail infrastructure on the agricultural systems have specific infrastructure limitations, which affect the characteristics of the rollingstock able to operate on them, including:

- narrow-gauge lines, which limit the type of locomotives and rollingstock that can be used
- low axle limits (15.75 tonne axle loads (TAL) or less), compared to 20–22 TAL on the North Coast Line and Mount Isa Line, or 22–30 TAL on the Central Queensland Coal Network⁴⁴⁰
- restricted tunnel heights, particularly on the Toowoomba Range Crossing (for South Western system and Western system traffic), which affect the permissible height of wagons and containers
- old railway bridges—which restrict speeds and limits the tonne axle loads of rollingstock.⁴⁴¹

The presence of sunk investments in assets specific to the Queensland Rail network (e.g. rollingstock and associated maintenance facilities) gives rise to the 'hold-up problem' commonly described in the economics literature.

Specifically, if an access seeker decides to enter (or increase its operations in) the above-rail freight haulage markets on the agricultural systems, it will incur significant sunk costs through the need to invest in long-life rollingstock assets. The 20–25-year useful life of rollingstock can be contrasted with the typical length of a below-rail access agreement, which is around 10 years.⁴⁴² Therefore, at some point in the middle of the useful life of the rollingstock, it might be expected that the below-rail access agreement will be due for renewal. As noted in section 8.4.4, below-rail access agreements with Queensland Rail have historically not contained evergreen renewal clauses⁴⁴³—this means that any terms contained in the original access agreement. Therefore, when the below-rail access agreement is due for renewal, in subsequent periods after the above-rail operator has entered the market, the above-rail operator would be in a less favourable bargaining position relative to Queensland Rail, as it has made significant sunk investments in rollingstock assets that are not able to be readily used elsewhere.⁴⁴⁴

The QCA considers that in a future without declaration, Queensland Rail will not face any effective long-term constraints on its ability to exercise market power. In these subsequent periods, an exercise of market power by Queensland Rail against an above-rail operator may

⁴⁴² Watco, sub. 49, p. 1; Queensland Rail, sub. 51, pp. 1–2.

⁴³⁹ Southern Downs Regional Council, *Major rail service company Watco invests in Southern Downs*, news release, 20 September 2018, accessed 7 August 2019, https://www.sdrc.qld.gov.au/council/alerts-news-notices/2018news/major-rail-service-company-watco-invests-in-southern-downs.

⁴⁴⁰ Queensland Rail, Western System Information Pack, October 2016, p. 21, accessed 7 August 2019, https://www.queenslandrail.com.au/business/acccess/Documents/Western%20System%20Information%20Pack% 20-%20Issue%203%20-%20October%202016.pdf.

⁴⁴¹ Transport, Housing and Local Government Committee, *Rail freight use by the agriculture and livestock industries*, report no. 45, Queensland Parliament, June 2014, p. 15.

⁴⁴³ In the absence of information to the contrary, the QCA has proceeded on the basis that this practice will continue in the future—that is, that below-rail access agreements in the future are likely to also not contain evergreen renewal clauses.

⁴⁴⁴ Watco, sub. 48, p. 4; GrainCorp, sub. 52, p. 7.

arguably be regarded as a transfer of wealth between the parties, with little impact on competition. However, the QCA considers that the critical issue is that in the first period, the above-rail operator can foresee this risk that any sunk investments it makes in the first period will be exposed to the risk of expropriation by the monopolist in the subsequent periods. The QCA considers that this risk is sufficiently material that a potential more efficient entrant will likely be deterred from entering the market in the first place. For example, Watco noted:

Watco's entry into the Queensland above rail market is predicated on its proposed significant investment in new above rail rollingstock, which given the potential change in access arrangements and subsequent impact on investment risk profile, has the potential to seriously alter service provision.⁴⁴⁵

The presence of this risk of hold-up means that socially optimal investments will not proceed, or there will be an underinvestment. Queensland Rail may have an incentive to solve this hold-up problem ex ante—for example, it may be profit maximising for Queensland Rail to sell unused network capacity to new or renewing users, assuming it is not constrained to charging a uniform price.⁴⁴⁶ However, as discussed below, the QCA's view is that it would be very difficult for Queensland Rail to credibly commit ex ante to solve the hold-up problem (for example through a long-term contract). The problem is that events could develop in the future where the benefits to Queensland Rail of expropriating the value of the investment at that later time exceed the benefits of continuing to abide by status quo arrangements. The QCA considers that it is this risk—that significant sunk investments in rollingstock made by the above-rail operator will be expropriated—that will lead to a material adverse effect on competition in the above-rail freight haulage markets on the agricultural systems in a future without declaration.

The ability of Queensland Rail to exercise market power in future contracting periods creates a significant degree of uncertainty for potential market participants at the time they are considering investment, raising the hurdle rate required to justify the investment and potentially preventing efficient entry and efficient participation in the market. Furthermore, all market participants are exposed to this risk in a future without declaration: incumbent operators also face increased risk and uncertainty at the time of their contract renewals, due to the absence of evergreen renewal rights. This may undermine incentives for future efficient actions by those operators compared to the situation with declaration.⁴⁴⁷

In contrast, the QCA considers that the access regime that would apply in a future with declaration is an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power. Access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would materially improve the environment for competition by encouraging efficient entry and actions (through a stable and predictable environment). Such an environment would in turn promote a material increase in competition in the above-rail freight haulage markets on the agricultural systems (the QCA's approach to materiality is discussed below).

⁴⁴⁵ Watco, sub. 48, p. 4.

⁴⁴⁶ Under the 2016 access undertaking, Queensland Rail is only required to charge a uniform price (i.e. the reference tariff) for coal users on the West Moreton system and Metropolitan system. For all the other systems (including non-coal users of the West Moreton system and Metropolitan system), prices are negotiated between Queensland Rail and the customers seeking below-rail access.

⁴⁴⁷ Pacific National, sub. 9, pp. 6–7.

Queensland Rail's submission that there is no hold-up problem

In response to the QCA's analysis of the hold-up problem in the QCA draft recommendation, Queensland Rail argued that there is no hold-up problem arising in respect of Queensland Rail's services provided using the North Coast Line, Mount Isa Line and West Moreton system.⁴⁴⁸ Although Queensland Rail's submissions do not purport to address the hold-up issue in relation to the above-rail markets on the agricultural systems, the QCA considers that its analysis in response to these arguments applies equally to the agricultural systems.

As such, Queensland Rail's submissions on this issue, as well as the QCA's detailed consideration of them, are discussed in the North Coast Line analysis (section 5.6.3). In summary, the QCA considers that the issues raised by Queensland Rail are unlikely to be sufficient to address the risk of hold-up in a future without declaration. A further discussion of the economic literature relating to the hold-up problem is provided in Appendix A.

8.6.3 Promote a material increase in competition

Materiality

In the case of the South Western Route service, the Western Route service and the Central Western Route service, the QCA considers that access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a material increase in competition in the above-rail freight haulage markets on the agricultural systems. The QCA's approach to the concept of materiality is discussed in section 5.6.4.

A decision to enter (or re-invest in) the above-rail freight haulage markets will involve substantial sunk investments. In a future without declaration, the presence of sunk investments gives rise to the hold-up problem. The QCA considers that the risk of hold-up in the presence of substantial sunk investments is sufficiently material that it is likely to discourage efficient firms from entering the market. In contrast, the QCA considers that declaration, and the associated access regime, is able to credibly constrain Queensland Rail's ability and incentive to exercise market power and credibly address the hold-up risk.

The QCA considers that the credible constraint on the risk of hold-up in the presence of substantial sunk investments will promote a non-trivial, material improvement in the environment for competition in the above-rail freight haulage markets on the agricultural systems. The environment for competition in a future with declaration is likely to promote efficient entry (and efficient re-investment) by all market participants, such that competitive outcomes in the above-rail freight haulage markets are materially more likely to occur.

If efficient entry is likely to be promoted in a future with declaration (compared to a future without declaration), the QCA considers that this would indicate that access as a result of declaration would promote an increase in competition that is material. In this way, the QCA is satisfied that access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration would promote a material increase in competition in the above-rail freight haulage markets on the agricultural systems.

Conclusion on the hold-up problem in the above-rail freight haulage markets on the agricultural systems

In the case of the South Western Route service, the Western Route service and the Central Western Route service, the QCA considers that the uncertainties facing market participants in a

⁴⁴⁸ Queensland Rail, sub. 33, p. 39, para. 191–92; sub. 33, attachment B, pp. 13–14.

future without declaration would affect all participants across the market, including more efficient firms. Conversely, the QCA considers that the certainties and protections offered by the access regime in a future with declaration would promote efficient entry and efficient participation in the dependent above-rail freight haulage markets.

The QCA considers that in a future without declaration Queensland Rail will not face any effective long-term constraints on its ability and incentive to exercise market power in relation to access terms. In a future without declaration, there will be an imbalance of negotiating power between Queensland Rail and access seekers/users in the presence of sunk investments. The QCA acknowledges that commercial firms face a range of uncertainties in decision-making on a daily basis. However, an imbalance in bargaining power could inhibit the ability of access seekers/users to effectively manage risks, including the risk of hold-up, which have a significant effect on the expected profitability of entry into (and operations within) the market. The presence of these risks, and an imbalance in the ability of access seekers/users to address these risks in a future without declaration, are likely to deter efficient entry or efficient investments by market participants.

In contrast, the QCA considers that a future with declaration provides a transparent statutory process under the QCA Act⁴⁴⁹ within which terms and conditions of access can be negotiated. This process provides market participants with greater certainty that access will be provided on reasonable terms and conditions, including to address sunk investments and mitigate the risk of hold-up for access seekers. As such, the QCA considers that the protections offered by the access regime in a future with declaration will lead to a material improvement in the environment for competition in the above-rail freight haulage markets, compared to a future without declaration.

8.7 Conclusion

The QCA considers that access (or increased access) to the South Western Route service, the Western Route service and the Central Western Route service, on reasonable terms and conditions, as a result of declaration, would promote a material increase in competition in the above-rail freight haulage markets dependent on each respective service. This is because the opportunities and environment for competition in the above-rail freight haulage markets will be materially enhanced in a future with declaration, given the constraints declaration imposes on Queensland Rail's ability and incentive to exercise market power, compared to a future without declaration.

The QCA considers that criterion (a) is satisfied in respect of the South Western Route service, the Western Route service and the Central Western Route service.

⁴⁴⁹ The provisions of the QCA Act can only be changed by parliament.

9 CRITERION (A)—THE TABLELANDS SYSTEM SERVICE

9.1 Part of the existing declared service and dependent markets

The QCA has assessed the following part of the existing declared service and the following dependent market:

Table 7	The Tablelands sys	tem service and the dependent market

Dependent market	Part of the existing declared service upon which the market is dependent	Facility for the relevant part of the service
The above-rail passenger market on the Tablelands system	Tablelands system service, that is use of the Tablelands system	Tablelands system

9.2 Geographical description of the Tablelands system

The Tablelands system consists of two distinct, unconnected sections of railway. The first railway extends from Cairns south-west to Forsayth, and the second railway extends from Croydon north-west to Normanton.⁴⁵⁰ Both systems carry tourist passenger services only.⁴⁵¹

9.3 Dependent markets

No freight is transported on the Tablelands system, and three passenger services operate on various parts of the system.⁴⁵² According to Queensland Rail:

No freight is transported on the [Tablelands] system as a consequence of severe limitations of the rail infrastructure and the lack of a freight market scale to warrant upgrade of the system [to] operate regular freight services.

The Kuranda Scenic Railway operates four one-way services per day, 28 per week between Cairns to Kuranda, and the Savannahlander (operated by Cairns Kuranda Steam) runs a four day return service weekly between March and December between Cairns and Forsayth. The Gulflander operates between Normanton and Croydon (generally two one-way seasonal scheduled services per week from mid-February to mid-December, plus charter services between April and May).⁴⁵³

The QCA considers that a relevant dependent market is the above-rail passenger market on the Tablelands system—which is the market for the transportation of passengers on the Tablelands system. The QCA is satisfied that this dependent market is separate from the market for the below-rail service.

Other dependent markets may include the downstream tourism markets in the regions covered by the Tablelands system. However, there is a lack of publicly available information on which to base an analysis of such markets, and therefore the QCA has not formed a view regarding these markets.

⁴⁵⁰ Queensland Rail, Tablelands System Information Pack, October 2016, p. 5,

https://www.queenslandrail.com.au/business/acccess/Documents/Tablelands%20System%20Information%20Pack %20-%20Issue%203%20-%20October%202016.pdf.

⁴⁵¹ Queensland Rail, sub. 33, attachment A, p. 14.

⁴⁵² Queensland Rail, sub. 33, attachment A, p. 5.

⁴⁵³ Queensland Rail, sub. 33, attachment A, p. 5.

No stakeholders made submissions that sought to identify markets dependent on the use of the Tablelands system.

9.4 Above-rail passenger market on the Tablelands system

9.4.1 The market

The three passenger services that operate on the Tablelands system for tourism purposes are:

- the Kuranda Scenic Railway passenger service
- the Gulflander passenger service
- the Savannahlander passenger service.

The Kuranda Scenic Railway service operates between Cairns and Kuranda, with approximately 1,456 one-way services per year.⁴⁵⁴ The Gulflander service operates between Normanton and Croydon, with approximately 78 one-way services per year.⁴⁵⁵ Both the Kuranda Scenic Railway and the Gulflander passenger services are operated by Queensland Rail.

Additionally, the Savannahlander service operates on the Tablelands system between Cairns and Forsayth, with approximately 88 one-way services per year.⁴⁵⁶ The Savannahlander is operated by a private company, Cairns Kuranda Steam, under a Transport Service Contract (TSC) subsidy from the Queensland Government.⁴⁵⁷

On the Tablelands system, Queensland Rail provides both the below-rail service and the aboverail passenger services (with the exception of the Savannahlander, which is operated by a private company). Queensland Rail operates almost all above-rail passenger services in Queensland⁴⁵⁸, including the Brisbane metropolitan commuter passenger services as well as regional travel and tourism passenger services across the Queensland Rail network. Under its Constitution, Queensland Rail is authorised to provide 'rail passenger transport services in Queensland for reward'. Importantly, Queensland Rail is not permitted to provide above-rail freight services under its Constitution; Queensland Rail's above-rail operations extend to the provision of passenger services only.⁴⁵⁹

9.4.2 Transport Service Contract subsidies

The Queensland Government provides subsidies for both the delivery of above-rail and belowrail services in Queensland. These subsidies are provided through the TSCs, which are managed on behalf of the Queensland Government by the Department of Transport and Main Roads (TMR).

⁴⁵⁴ Four one way services operate daily all year, except Christmas Day—see Queensland Rail Travel, *Kuranda Scenic Railway Timetables*, accessed 13 August 2019, https://www.ksr.com.au/Tourpackages/Pages/Timetables.aspx.

⁴⁵⁵ Two one-way services operate per week between February and December each year—see Queensland Rail Travel, Gulflander Travel Information, accessed 13 August 2019,

https://www.gulflander.com.au/Pages/TimetableandFares.aspx.

⁴⁵⁶ The Savannahlander travels a return trip between Cairns and Forsayth once a week between March and December—see: Cairns Kuranda Steam, *Savannahlander General Information*, accessed 13 August 2019. http://www.savannahlander.com.au/general-information/.

⁴⁵⁷ Queensland Rail, sub. 33, p. 38, para. 187.2.

⁴⁵⁸ With the exception of some small local private tourism operators.

⁴⁵⁹ Queensland Rail, *Draft Queensland Rail Access Undertaking 1*, explanatory submission, March 2012, p. 15, accessed 30 July 2019, https://www.qca.org.au/wp-content/uploads/2019/06/8354_r-qrail-qrail2012dau-expdocs2012dau-0412.pdf.

The QCA provides a detailed discussion of TSCs in section 8.4.3. Three categories of TSC subsidies are applicable to the Tablelands system:

- (1) subsidies for the delivery of below-services by Queensland Rail on the Tablelands system
- (2) subsidies for the delivery of above-rail passenger services by Queensland Rail, in particular the Kuranda Scenic Railway and the Gulflander passenger services
- (3) subsidies for the delivery of the above-rail passenger service called the Savannahlander by a private above-rail operator, Cairns Kuranda Steam.

Queensland Rail receives revenue under the 'Rail TSC'. In 2017–18, Queensland Rail received approximately \$567 million in TSC subsidies for the delivery of below-rail services across its whole network, and \$1.1 billion in TSC subsidies for the delivery of above-rail passenger services across its whole network.⁴⁶⁰ According to Queensland Rail, more than 750,000 customers travelled on its regional travel and tourism passenger services in 2017–18, with approximately 55 per cent of these customers travelling on the Kuranda Scenic Railway tourist service on the Tablelands system.⁴⁶¹

Additionally, the 'Savannahlander TSC' is provided exclusively to support the Savannahlander passenger service on the Tablelands system. According to TMR:

The Savannahlander tourist train service operates in Far North Queensland between Cairns and Forsayth between March and December each year. The service is operated by Cairns Kuranda Steam Partnership.

TMR subsidises the Savannahlander tourist train service through the Savannahlander Transport Service Contract. $^{\rm 462}$

The value of the Savannahlander TSC is commercial-in-confidence between TMR and the private operator, Cairns Kuranda Steam.

9.4.3 Statutory obligations in relation to passenger services

Special legislative provisions apply to the operation of passenger train services in Queensland. Queensland Rail has previously described these legislative provisions:

Queensland Rail operates almost all passenger train services in Queensland. Passenger train services receive special treatment under the TI Act [Transport Infrastructure Act 1994 (Qld)] as follows:

- section 265 of the TI Act sets out an obligation for a railway manager [i.e. Queensland Rail] to bring a delayed passenger train service back to its scheduled running time (including where this may result in a freight train service being delayed);
- section 266 of the TI Act sets out a right for the Director General of the DTMR [Queensland Department of Transport and Main Roads] to identify the requirements for regularly scheduled passenger train services (e.g. identify the capacity requirements) and obliges railway managers to allocate rail capacity that is available or will become available to meet those requirements; and

⁴⁶⁰ Queensland Rail does not report disaggregated data on the amount of TSC subsidies attributable to the Tablelands system—see Queensland Rail, *Financial Statements for the Year Ended 30 June 2018—Below Rail Services Provided by Queensland Rail*, December 2018, pp. 4, 15; Queensland Rail, *Annual and Financial Report 2017–18*, September 2018, p. 73 (p. 7 of the Financial Report).

⁴⁶¹ Queensland Rail, Annual and Financial Report 2017–18, September 2018, pp. 19, 42.

⁴⁶² Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2019, accessed 13 August 2019, https://www.tmr.qld.gov.au/business-industry/Transport-sectors/Rail-servicesand-infrastructure/Rail-Transport-Contracts-and-Agreements.

section 266A of the TI Act – provides for the preservation of train paths for regularly scheduled passenger train services and for non-coal freight services.⁴⁶³

Queensland Rail stated that all of its passenger train services use either preserved train paths or train paths that are the subject of the Director General of DTMR's passenger train service requirements.464

9.5 Queensland Rail's ability and incentive to exercise market power

Queensland Rail contended that it has no ability or incentive to use any market power to affect passenger service markets.⁴⁶⁵ These issues are considered below in relation to the above-rail passenger market on the Tablelands system.

9.5.1 Above-rail passenger services operated by Queensland Rail

On the Tablelands system, Queensland Rail provides both the below-rail service and some of the above-rail passenger services, including the Kuranda Scenic Railway service. In this way, Queensland Rail can be characterised as being vertically integrated into the above-rail passenger market on the Tablelands system.

As the provider of both the below-rail service and the above-rail passenger services, it is somewhat nonsensical to inquire whether Queensland Rail would have the ability or incentive to exercise market power against itself.

9.5.2 Above-rail passenger services operated by private operators

On the Tablelands system, a private operator (Cairns Kuranda Steam) operates the Savannahlander tourist passenger service. This service does not operate in direct competition with either of the above-rail services provided by Queensland Rail.⁴⁶⁶ The QCA considers that despite being vertically integrated into the above-rail passenger market, it is unlikely that Queensland Rail will have the ability and incentive to exercise market power in the above-rail passenger market on the Tablelands system, with respect to above-rail passenger services operated by private operators, for reasons set out below.

Queensland Rail may be constrained in its ability to exercise market power in the above-rail passenger market on the Tablelands system

Special legislative provisions apply to the operation of passenger train services in Queensland. The QCA considers that it is likely that Queensland Rail's ability to exercise market power in a way that may adversely affect competition in the above-rail passenger market on the Tablelands system is constrained by these legislative provisions.

For example, s. 266A of the Transport Infrastructure Act 1994 (Qld) (TI Act) identifies preserved train paths, including a train path allocated to a regularly scheduled passenger service on the commencement of the section. If a preserved train path for a regularly scheduled passenger

⁴⁶³ Queensland Rail, Draft Queensland Rail Access Undertaking 1, explanatory submission, March 2012, p. 18.

⁴⁶⁴ Queensland Rail, *Draft Queensland Rail Access Undertaking* 1, explanatory submission, March 2012, p. 18. ⁴⁶⁵ Queensland Rail, sub. 33, p. 38, paras 186–89.

⁴⁶⁶ Specifically, the Savannahlander operates between Cairns and Forsayth, whereas the Kuranda Scenic Railway service (operated by Queensland Rail) operates between Cairns and Kuranda only, and the Gulflander service (operated by Queensland Rail) operates between Normanton and Croydon. To the extent that the path of the Kuranda Scenic Railway overlaps with that of the Savannahlander (i.e. the Cairns to Kuranda portion), the QCA considers that these services are not operating in competition with each other, as each service is operated as a particular tourism package, rather than as a mode of transportation between two locations.
service becomes available for Queensland Rail to reallocate, s. 266A(2) of the TI Act prohibits Queensland Rail from allocating that train path to a different type of service (e.g. a freight service), unless at least two months written notice is given to the chief executive and the chief executive consents to the proposed reallocation. Queensland Rail is still permitted to allow the 'use' of a preserved train path for a different type of service when that train path is not being used for the service for which it is allocated.⁴⁶⁷

In these circumstances, if a private above-rail passenger operator operated on a preserved train path, unless the private operator willingly disclaimed the use of its allocated train path (e.g. if there are not enough tourists to justify running a service), Queensland Rail is constrained in its ability to reallocate train paths at its discretion, as that train path is preserved for that passenger service and is not available to be allocated elsewhere at the discretion of Queensland Rail.

In another example, s. 266(6) of the TI Act provides that in charging for access to regularly scheduled passenger services, Queensland Rail must not differentiate between similar regularly scheduled passenger services operating or proposed to operate over the same route at different times of the day, or set an access charge for a train path that is greater than the access charge for similar train paths. Given this provision, the QCA considers that Queensland Rail is likely to be constrained in its ability to exercise market power with respect to the access charge that it may impose on a private operator of a regularly scheduled passenger service.

The QCA considers the provisions of the TI Act will likely continue to apply to existing and future private operators of above-rail passenger services in a future with or without declaration.

9.5.3 The reliance on government subsidies

Queensland Rail contended that the provision of above-rail passenger services in Queensland would not be commercially viable without the presence of government subsidies:

Each of Queensland Rail's passenger transport services are subsidised by the Queensland Government and would not be commercially viable without transport support payments under the TSC. Only one additional operator, Cairns Kuranda Steam, provides regular tourist passenger services on one section of the Tablelands System. Cairns Kuranda Steam also receives transport support payments from the Queensland Government and would not be commercially viable without these.

The majority of costs associated with operation of the passenger network are supported by subsidies through the TSC. Accordingly, passenger network revenues do not cover a small fraction of incremental costs let alone contribute towards Queensland Rail's substantial fixed assets.⁴⁶⁸

The QCA considers that regular above-rail passenger services on the Tablelands system, whether operated by Queensland Rail or a private operator, are unlikely to be commercially viable without the presence of subsidies. The QCA notes that almost all above-rail passenger services in Australia receive some form of government subsidy to support their operation.⁴⁶⁹ As such, it appears that a regular above-rail passenger service would be unlikely to be established without the presence of government subsidies.

⁴⁶⁷ Section 266A(6) of the TI Act.

⁴⁶⁸ Queensland Rail, sub. 33, p. 38. paras 187.2–187.3.

⁴⁶⁹ Centre for International Economics, Subsidies and the social costs and benefits of public transport, prepared for the Independent Pricing and Regulatory Tribunal of NSW, 2001, pp. 5–11, accessed 21 August 2019, https://www.ipart.nsw.gov.au/files/0395703f-9e11-4c78-9b99-9f7200c4f7a4/Report_Prepared_for_IPART_-_Subsidies_and_the_social_costs_and_benefits_of_public_transport_- CIE_-_March_2001.pdf.

9.5.4 An enduring lack of competition in above-rail passenger markets

Queensland Rail is vertically integrated into the above-rail passenger market on the Tablelands system (and across its network), and the presence of a vertically integrated monopolist may raise prima facie concerns that the monopolist will seek to favour its related party in the above-rail passenger market to the detriment of competitors (and the environment for competition). However, for the reasons discussed below, the QCA considers that the unique nature of the above-rail passenger market on the Tablelands system means that these incentives are unlikely to arise in this market.

Queensland Rail argued that it is very unlikely that it will operate in direct competition with a private above-rail passenger service provider on any part of its network:

There is no realistic prospect of the Queensland Government paying transport service payments to both Queensland Rail and a third party operator to establish competing passenger train services. Given the lack of commerciality of passenger train services in the absence of such payments, there will effectively be no competition as there will only be one operator for a passenger train service using a preserved train path ... whether Queensland Rail or a third party.⁴⁷⁰

The QCA considers that it is unlikely that a private operator would operate in direct competition with Queensland Rail for the provision of a regular above-rail passenger service on the Tablelands system (or any other rail system). Based on the evidence before the QCA, it appears that above-rail passenger services in Queensland have in the past either been operated by Queensland Rail (e.g. the Kuranda Scenic Railway service), or a private operator (e.g. the Savannahlander service); but there has rarely (if ever) been two above-rail passenger services that operated in direct competition with each other (i.e. for the same route) in Queensland.

It may be argued that as a vertically integrated above-rail passenger operator, Queensland Rail may have an incentive to deny access to a private above-rail operator, in order to operate that service itself and claim the related subsidy. However, the QCA considers that Queensland Rail is only likely to act in such a way if it can be sure that it will be a recipient of the relevant subsidy. Given that TSCs are likely to be entered into prior to the service commencing (such that Queensland Rail will know if it has been awarded the subsidy), the QCA considers that it is more likely that Queensland Rail will compete with a private operator to secure the TSC (e.g. at the time of tendering for the TSCs), rather than to deny access to a private above-rail passenger operator at a later stage in the hope that it (Queensland Rail) will secure the TSC. In addition, as discussed in section 9.5.2, the QCA considers that Queensland Rail may be constrained in its ability to exercise market power to deny access in this manner.

As a result, despite the fact that Queensland Rail is vertically integrated into the above-rail passenger market in Queensland, the QCA considers that there is no evidence to demonstrate that this vertical integration has had or will have an adverse effect on competition in the above-rail passenger market on the Tablelands system.

9.5.5 Queensland Rail may have no incentive to exercise market power to affect competition in the above-rail passenger market on the Tablelands system

The QCA considers that as a business, Queensland Rail has an incentive to maximise profits. However, the evidence before the QCA suggests that any private operator of above-rail passenger services would need to rely heavily on government subsidies in order to ensure that the service was commercially viable. This suggests that above-rail passenger services, such as

⁴⁷⁰ Queensland Rail, *Draft Queensland Rail Access Undertaking 1*, explanatory submission, March 2012, pp. 24–25.

the tourist services on the Tablelands system, are likely to be provided for public policy reasons (e.g. to promote the local tourism industry) rather than purely commercial reasons.

The QCA does not have visibility over the commercial-in-confidence terms of the above-rail subsidies (e.g. the Savannahlander TSC), so it has not conducted a detailed analysis of the effect of these subsidies on the incentives facing Queensland Rail and any private above-rail passenger operators in the above-rail passenger market on the Tablelands system. However, given the market characteristics, the QCA considers that broadly, two possibilities may arise.

The first is that Queensland Rail may have no incentive to exercise market power to affect competition in the above-rail passenger market on the Tablelands system. This may be, for example, because Queensland Rail receives a defined amount of access charges under the relevant TSC for supplying the below-rail service to the private above-rail passenger operator, or it is required to supply the below-rail service to that operator in any case, due to public policy reasons.

The second is that Queensland Rail may have an incentive to exercise market power to affect competition in the above-rail passenger market on the Tablelands system, because it perceives an opportunity to maximise profits. In this case, the relevant question then becomes whether the exercise of market power will have any effects on competition in that market. This is discussed below.

9.6 Competition in the above-rail passenger market on the Tablelands system

Declaration is unlikely to promote a material increase in competition in the aboverail passenger market

Even if it were the case that Queensland Rail has an ability and incentive to exercise market power, the QCA nevertheless considers that criterion (a) is not satisfied, because access (or increased access) to the service, on reasonable terms and conditions, as a result of declaration of the service would not promote a material increase in competition in the above-rail passenger market on the Tablelands system.

Based on the evidence before it, the QCA considers that in the above-rail passenger market on the Tablelands system, the prominent and decisive factor informing a private above-rail passenger operator's decision to enter or reinvest in the market is the provision of government subsidies. For example, Queensland Rail has reiterated its view that above-rail passenger services, whether operated by itself or a private operator, would not be commercially viable without government subsidies.⁴⁷¹

In the case of a subsidised private above-rail passenger operator on the Tablelands system, if Queensland Rail sought to exercise its market power against this operator, arguably this would be a transfer of wealth between the two parties, with no effect on competition. The QCA considers that the decision of a future potential above-rail passenger operator to enter the market depends critically on the availability and terms of the government subsidy it would receive, rather than the presence of an access regime as a result of declaration.

Therefore, the QCA is not satisfied that in the case of the above-rail passenger market on the Tablelands system, access (or increased access), on reasonable terms and conditions, as a result of a declaration of the service would promote a material increase in competition in that market.

⁴⁷¹ Queensland Rail, sub 33, p. 7, para. 27; p. 38, para. 187.

10 CONCLUSION FOR CRITERION (A)

The QCA is not satisfied that criterion (a) is met with respect to the service as a whole. The QCA does not consider that there exists a single above-rail market for the transportation of freight that is dependent on access to the whole of the Queensland Rail service.

The QCA has therefore assessed a range of markets dependent on access to parts of the Queensland Rail service, as set out in the following table. The definitions of each service and facility is in Appendix B.

Description of the relevant part of the declared service	Dependent market
North Coast Route service	The above-rail freight haulage market
Mount Isa Route service	The North West Queensland minerals tenements market
West Moreton Route service	The market for coal tenements in the West Moreton region
Central Western Route service	The above-rail freight haulage market
Western Route service	The above-rail freight haulage market
South Western Route service	The above-rail freight haulage market
Tablelands system service	The above-rail passenger market

 Table 8
 Parts of the service and the dependent market assessed in criterion (a)

The QCA is satisfied that each part of the service identified in the table above is itself a 'service' within the meaning of s. 72 of the QCA Act.

With the exception of the Tablelands system service, the QCA is satisfied that access, or increased access, on reasonable terms and conditions as a result of declaration of each service identified above, would promote a material increase in competition in each of the identified dependent markets (as set out in Table 8 above). As a result, with the exception of the Tablelands system service, the QCA is satisfied criterion (a) is met for each part of the service identified above.

The QCA is not satisfied that access (or increased access) on reasonable terms and conditions, as a result of declaration of the Tablelands system service, would promote a material increase in competition in any market. As such, the QCA considers that this part of the service does not satisfy criterion (a).

11 CRITERION (B)—MEET TOTAL FORESEEABLE DEMAND AT LEAST COST

11.1 Introduction

Section 76(2)(b) of the QCA Act is expressed as follows:

- that the facility for the service could meet the total foreseeable demand in the market-
- (i) over the period for which the service would be declared; and
- (ii) at the least cost compared to any 2 or more facilities (which could include the facility for the service)

Sections 76(3) and (4) of the QCA Act state:

(3) For subsection (2)(b), if the facility for the service is currently at capacity, and it is reasonably possible to expand that capacity, the authority and the Minister may have regard to the facility as if it had that expanded capacity.

(4) Without limiting subsection (2)(b), the cost referred to in subsection (2)(b)(ii) includes all costs associated with having multiple users of the facility for the service, including costs that would be incurred if the service were declared.

A summary of key matters raised by stakeholders with respect to criterion (b), as well as the QCA's final recommendations, are set out in Table 9.

Criterion (b)					
Issue	Queensland Rail	Other stakeholders	QCA final recommendation		
The service	See section 2.2	See section 2.2	(a) As per s. 250(1)(b)—that is, the existing declared service which is the 'service as a whole'		
			(b) Parts of the existing declared service, which are each a 'service', identified in the criterion (a) analysis (see Appendix B)		
			See section 2.2		
The facility	There are eight railway systems – each of which is a facility	The facility for the service is the below-rail infrastructure of the Queensland Rail network	(a) For the service as a whole, the facility is as per s.250(1)(b)		
			(b) For the parts of the service identified in the criterion (a) analysis, each facility is the relevant rail transport infrastructure used (see Appendix B)		
			See section 2.3		
The market	A key issue is whether the product dimension of the market for the relevant services includes road haulage services	Pacific National said the relevant market is the one in which Queensland Rail provides access to the below-rail services	The relevant market is the market for the provision of access to and use of rail infrastructure by above-rail operators. The QCA considers this is either:		

Table 9 Summary of key positions—s. 76(2)(b) of the QCA Act

Criterion (b)						
		The South West Producers and Glencore focused on definitions of the market for the West Moreton system and Mount Isa Line respectively	 (a) a single geographic market (b) smaller geographic markets corresponding to the parts of the service identified in the criterion (a) analysis See section 11.4 			
The period for assessing total foreseeable demand	The period of declaration should not be for more than 5 years	The South West Producers and Glencore said the period of declaration should be 15 years	The recommended declaration period is 15 years See section 11.5			
Total foreseeable demand over the period of declaration	Key issues include identifying foreseeable demand for each of the main freight tasks (e.g. bulk freight, intermodal freight, livestock and passengers), as well as demand for road services where road services are encompassed by the relevant market for the service	The South West Producers and Glencore said the total foreseeable demand on the West Moreton system and the Mount Isa Line respectively could be met by the existing facility	The facility for each service identified can meet the total foreseeable demand in the market for each service over the period of declaration See section 11.6			
At least cost compared to any two or more facilities	No information provided on costs	The South West Producers and Glencore said that given there was not another existing facility capable of providing the service, the existing facility would meet the demand at least cost compared to the cost of constructing a new facility	The facility for each service identified can meet the total foreseeable demand in the market for each service at the least cost compared to any two or more facilities See section 11.7 Criterion (b) is satisfied			

11.2 The QCA's approach to assessing criterion (b)

The QCA considers that only part of the existing declared service satisfies criterion (a) (see Part B, Chapter 10). As this is a recommendation to the Minister, the QCA considers it appropriate to make findings about criterion (b) with respect to both:

- the existing service taken to be declared under s. 250(1)(b) of the QCA Act and the facility for this service as identified in s. 250(1)(b) of the QCA Act. This is the existing declared service which is the service 'as a whole'. The facility identified in s. 250(1)(b) of the QCA Act is the 'rail transport infrastructure' if the infrastructure is used for operating a railway for which Queensland Rail Limited (or a successor, assign or subsidiary) is the railway manager
- the parts of the service set out in the criterion (a) analysis, which are each a 'service' within the meaning of s. 72 of the QCA Act, and the facility for each service (see Part B, Chapter 10 and Appendix B).

11.3 The structure of criterion (b) analysis

The analysis of the 'service' and the 'facility for the service' is discussed in Part B, Chapter 2. This chapter follows on from that analysis, and discusses:

- the market for the service
- the period for assessing total foreseeable demand
- the total foreseeable demand over the period of declaration
- the analysis of whether it is least cost to meet total foreseeable demand in the market using the facility for the service, compared to any two or more facilities.

11.4 The market

11.4.1 Stakeholder submissions

Queensland Rail maintained that the QCA's analysis should be undertaken consistent with the part services Queensland Rail identified (i.e. the eight individual rail systems) and the markets in which each service is provided. Queensland Rail also said it is evident from the geographically distinct network parts, and the provision of services for a range of different purposes, that each should be examined.⁴⁷²

Pacific National said:

There are no substitute facilities for the supply of the relevant service, being the 'use of rail transport infrastructure for providing transportation by rail'

•••

In the context of criterion (b), the relevant 'market' is the market in which QR provides access to below-rail services. As has been recognised on numerous occasions, this is separate to the market in which PN and other freight businesses provide haulage services ...⁴⁷³

Both Glencore and the South West Producers said the relevant market, for the purpose of criterion (b), will be the market in which the facility provides the declared service (or the relevant parts of the declared service).⁴⁷⁴ That is, the market is confined to rail infrastructure access for below-rail services.⁴⁷⁵

11.4.2 QCA analysis

Consistent with the approach outlined in Overview—Chapter 2, the QCA has considered the market (or markets) in which the relevant service is provided and other services (if any) that are able to be substituted for, or are otherwise competitive with, the relevant service.

The service is defined in s. 250(1)(b) of the QCA Act as:

the use of rail transport infrastructure for providing transportation by rail if the infrastructure is used for operating a railway for which Queensland Rail Limited, or a successor, assign or subsidiary of Queensland Rail Limited, is the railway manager.

The supplier of this service is Queensland Rail. The customers acquiring this service are generally the above-rail entities operating rollingstock on the rail infrastructure, and not the end users who require goods or people to be transported. However, access rights can be held by end users, and demand for the service is ultimately driven by the needs of end users.

⁴⁷² Queensland Rail, sub. 33, p. 50, para. 246.

⁴⁷³ Pacific National, sub. 9, pp. 12–13.

⁴⁷⁴ Glencore, sub. 5, p. 6; South West Producers, sub. 4, p. 13.

⁴⁷⁵ South West Producers, sub. 31, p. 4.

In Queensland, the existing above-rail freight operators are Pacific National, Aurizon Operations, Linfox and Watco.⁴⁷⁶ Queensland Rail operates above-rail passenger trains over its network, but not freight services. Figure 19 illustrates the relationship between Queensland Rail and users of the relevant service.



Figure 19 The relevant service and market for the service

The QCA considers that the market for the service is a market for access to and use of rail infrastructure. No competing rail access service is being provided and, from the information available, this is unlikely to change in the future. There are no substitutes for Queensland Rail's service for an above-rail operator. While Aurizon Network also owns and operates rail infrastructure in Queensland, there is very little geographic overlap between the rail systems.

Geographic boundary of this market

Queensland Rail said it provides eight services by means of eight distinct facilities. Queensland Rail's comments go to the nature of demand for use of the network and, in particular, the proposition that demand for above-rail services is, in each case, a derived demand.

The differing nature of each of Queensland Rail's services mean the relevant markets can only be properly assessed having regard to the derived demand specific to the service under consideration.⁴⁷⁷

In the context of criterion (a), the QCA considered possible markets that depend on access to the entire service and to different parts of the service, the relationship between these dependent markets and the market for the service, as well as the nature of demand and supply in each of these markets. While the QCA has not identified a market that is dependent on access to the entire service for the purpose of criterion (a), the QCA found a number of markets that are dependent on access to different parts of Queensland Rail's infrastructure.

The fact that there are different markets which are dependent on the use of Queensland Rail's infrastructure does not necessarily mean that access to this infrastructure is provided in separate markets. Moreover, it is not evident that other forms of infrastructure services are substitutable for the rail infrastructure service provided by Queensland Rail, other than another railway if it were constructed.

⁴⁷⁶ Watco commenced providing haulage services on the network in late 2019: Watco, sub. 48, p. 2; Watco Companies, Rail Services: Australia, accessed 7 February 2020,

https://www.watcocompanies.com/services/rail/australia/.

⁴⁷⁷ Queensland Rail, sub. 33, p. 52, para. 254.

Ultimately, the QCA does not consider it necessary to reach a conclusion on whether Queensland Rail provides access to its rail infrastructure in a single market, or in a number of smaller markets. The QCA has considered both possibilities and found that, in either case, criterion (b) is satisfied.

In summary, the market in which the Queensland Rail service is provided is a market for the provision of access to, and use of, rail infrastructure by above-rail operators. The QCA considers this is either a single geographic market, or a series of smaller geographic markets, corresponding to the parts of the service identified by the QCA in its assessment of criterion (a). There are no competitors to Queensland Rail in the relevant market or markets.

In this context, the question is whether:

- the Queensland Rail facility, as a whole, can satisfy total foreseeable demand in the single market for the below-rail service provided by Queensland Rail, or
- the facility for each part of the below-rail service provided by Queensland Rail identified in the criterion (a) analysis (see Part B, Chapter 10 and Appendix B), can satisfy total foreseeable demand in the market for each service.

11.5 Period for assessing total foreseeable demand

11.5.1 Stakeholder submissions

Queensland Rail said the period of declaration should be tailored to reflect changing market developments and dynamics relating to particular systems/services. In particular, it provided information for each of the North Coast Line, Mount Isa Line, West Moreton system and Metropolitan system, and concluded any period of declaration should not exceed five years. For instance, Queensland Rail discussed:

- road substitution—for the North Coast Line and Mount Isa Line, road freight is a strong substitute for the main commodities, and rail volumes have declined over time. Also, road competitiveness is likely to increase in future, including due to road developments and investments in higher-productivity vehicles (with performance schemes administered by the National Heavy Vehicle Regulator)
- potential rail developments—there are planned rail projects which may impact the competitive environment of the service provided by the West Moreton System, including the Inland Rail Project.⁴⁷⁸

Glencore and the South West Producers both said that the Queensland Rail service should be declared for a period of 15 years.

Glencore said:

In consideration of the duration of mining operations from exploration to rehabilitation being in the vicinity of 10 - 30 years depending upon the operation, Glencore considers the reasonable period over which foreseeable demand should be assessed – and for which the service should be declared, is a period of 15 years.⁴⁷⁹

The South West Producers said:

In considering ... the long term tenure of coal mine investments (usually around 10 to 30 years) and the long useful life of both above and below rail investments, the South West Producers

⁴⁷⁸ Queensland Rail, sub. 33, pp. 50–51, paras 247–250.

⁴⁷⁹ Glencore, sub. 5, p. 15.

consider that the relevant period [for declaration] ought to be the longest period for which the test of foreseeable demand at least cost (and all other access criteria) is met.

This is on the basis that an extended declaration period will better produce efficiencies, including capitalisation on assets (including coal mine infrastructure and both the below rail infrastructure and rolling stock), and investment certainty in dependent markets.

... The South West Producers consider that a reasonable period over which to declare the service is a period of 15 years $...^{480}$

11.5.2 QCA analysis

The QCA's view is that the period for assessing total foreseeable demand in each identified market should be 15 years.

In recommending this period, the QCA considers that the need for access seekers and holders to have certainty over the period of declaration must be balanced with the legitimate business interests of the infrastructure owner (Queensland Rail) to have its service declared for only as long as the service is considered to meet the access criteria.

Long-term certainty and asset lives

Previous declarations of services provided by railways have typically been for periods of 10 years or longer. For instance, the service provided by the Tasmanian Railway was declared for 10 years, while the service provided by the Goldsworthy Railway (Western Australia) was declared for 20 years.⁴⁸¹

In part, this reflects the long-lived nature of the sunk investments that are involved. For example, investments in rollingstock and mining operations typically have a useable life of 20 to 30 years.⁴⁸²

The QCA acknowledges that some investments will have a longer or shorter remaining life span.⁴⁸³ Assuming that at a particular point in time (e.g. at the time of declaration), asset lives across the industry would be partially life expired, a declaration period of 10 to 15 years could be appropriate. However, the QCA notes that given the absence of evergreen renewal rights, users may be disadvantaged to the extent that their asset lives extend beyond a proposed declaration period.

Certainty of demand over the foreseeable period

Forecasts are by their nature an inexact science and the QCA acknowledges that a high level of uncertainty in relation to future demand could necessitate a shorter declaration period. However, the QCA considers that there is sufficient evidence of spare capacity over the proposed 15-year declaration period to be satisfied that the facility for the parts of the Queensland Rail service recommended for declaration could meet total foreseeable demand. It is likely that the facility (or facilities) for the service will continue to meet demand in the

⁴⁸⁰ South West Producers, sub. 4, p. 35; sub. 31, p. 5.

⁴⁸¹ QRC, sub. 7, p. 10.

⁴⁸² For example, BITRE data shows that in 2018, locomotives aged between 0 and 30 years old accounted for around 80 per cent of all Australian narrow gauge locomotives in operation. See Bureau of Infrastructure, Transport and Regional Economics (BITRE), *Trainline 6*, statistical report, Australian Government, 2018, p. 63; South West Producers, sub. 4, p. 52; Glencore, sub. 5, pp. 13, 15.

⁴⁸³ For instance, South West Producers (sub. 40, p. 11) said that investments (e.g. New Acland Stage 3) are more likely to be made earlier in the declaration period.

relevant market without the need for expansion over the next 15 years. As such, a shorter period is not necessary.

Timing of future market changes

It is not evident that there would be substantial changes over the next 15 years in any of the markets in which the service or identified parts of the service are provided, which would affect whether the service or identified parts of the service provided by Queensland Rail would continue to satisfy criterion (b). For example, there are no foreseeable major developments of any of Queensland Rail's rail systems.

The exception is the Inland Rail project, which is expected to be operational in 2024–25.⁴⁸⁴ This project will connect Melbourne with Brisbane via an inland route. Queensland Rail said that if this project continues in its current form, it is likely to affect the competitive environment in respect of the West Moreton system. In particular:

the [ARTC] Inland Rail Business Case is predicated on picking up thermal coal volumes from the Moreton Basin currently transported on the West Moreton system, providing that 'up to 19.5 million tonnes of coal is expected to use Inland Rail ...'⁴⁸⁵

The South West Producers considered that the Inland Rail project will not provide a substitute for the West Moreton Route service, in part because it is intended as an interstate freight link for general freight haulage between Melbourne and Brisbane, and not as a dedicated track for the haulage of Queensland bulk freight (e.g. coal, agricultural products) from origin and destination points within Queensland. Also, the ARTC business case is based on a range of assumptions, and the charging regime is unknown.⁴⁸⁶

Queensland Rail also said there are proposals to develop a Surat Basin Rail link from the Wandoan project to the Port of Gladstone (together with a Central Surat Rail link) that if developed, would affect the West Moreton system's competitive environment.⁴⁸⁷

The QCA considers that there are difficulties in having regard to these proposals. The QCA considers there are a number of uncertainties in relation to the Inland Rail project, including uncertainties as to the final alignment (route), its operational characteristics (e.g. operating in conjunction with, or in competition with, Queensland Rail systems), the charging regime and the expected completion date of the project. Additionally, there is a lack of publicly available information to conclude that development of the Surat Basin Rail and Central Surat Rail links is actively progressing.⁴⁸⁸

To the extent these projects were completed and could be demonstrated to materially affect circumstances on these systems during the declaration period, if declared, it would be open for Queensland Rail to submit a revocation application for the relevant service at that time.

⁴⁸⁴ ARTC, Inland Rail, accessed 25 October 2018, https://inlandrail.artc.com.au/.

⁴⁸⁵ Queensland Rail, sub. 33, p. 51, para. 248.

⁴⁸⁶ South West Producers, sub. 4, p. 19; sub. 31, p. 5; sub. 40, p. 12.

⁴⁸⁷ Queensland Rail, sub. 33, p. 51, para. 248.

⁴⁸⁸ Queensland Government, Department of State Development, Manufacturing Infrastructure and Planning, Surat Basin Infrastructure Corridor State Development Area, accessed 13 November 2019, http://statedevelopment.qld.gov.au/coordinator-general/state-development-areas/current/surat-basininfrastructure-corridor-state-development-area.html.

Periodic review of declarations

The QCA considers that it is appropriate for any declaration to be periodically reviewed.⁴⁸⁹ The QCA considers that a 15-year declaration period appropriately provides for such a periodic review of the declaration of the service, by adequately balancing the legitimate business interests of Queensland Rail, while providing a period of certainty for access seekers and holders in the context of industries that require large sunk investments.

In balancing each of these factors, the QCA's view is that a 15-year period for assessing total foreseeable demand is appropriate for the service and each identified part of the service.

11.6 Total foreseeable demand over the declaration period

11.6.1 Stakeholder submissions

Queensland Rail submitted that 'the QCA will have to identify total foreseeable demand in the market that encompasses the services provided by each of the distinct rail facilities over the declaration period'.⁴⁹⁰ It said this involves identifying 'foreseeable demand for the derivative service of transport on each rail system for each of the main freight tasks (i.e. bulk freight, intermodal freight, livestock and passengers)' and identifying 'all of the foreseeable demand in the relevant market over the declaration period', including demand for road services where road services are also are also encompassed by the relevant market for the service.⁴⁹¹

Pacific National considered that:

it is clear that the QR infrastructure (either in its current form or as expanded) would be able to meet foreseeable demand in this market at lowest cost, compared to two or more facilities. The NCC has previously observed that railways typically exhibit natural monopoly properties due to high fixed costs and significant economies of scale. The QR rail infrastructure is no exception in this regard.⁴⁹²

11.6.2 QCA analysis

There is no evidence to suggest that total foreseeable demand for the below-rail service will exceed the capacity of the Queensland Rail network.

Queensland Rail confirmed that there is spare available capacity across the network⁴⁹³ and, on current (and expected) utilisation rates, it appears this is likely to continue in the foreseeable future. Further, even where different demand scenarios may eventuate—this was discussed, for example, in respect of the West Moreton Route service (in the context of a 'low' and 'high' tonnage scenario)—there is no information to show this demand cannot be met by the spare capacity available.⁴⁹⁴

Based on the available information, each of Queensland Rail's rail systems is currently operating below capacity and has historically operated below capacity.⁴⁹⁵ There is no information to suggest that total foreseeable demand over the proposed 15-year declaration period would at any time exceed the existing available capacity on any of these systems.

⁴⁸⁹ Section 87A of the QCA Act provides for a declaration to be reviewed at least 6 months, but not more than 12 months before the expiry date of a declaration.

⁴⁹⁰ Queensland Rail, sub. 8, p. 5, para. 36.

⁴⁹¹ Queensland Rail, sub. 33, pp. 52–53, para. 256.

⁴⁹² Pacific National, sub. 9, p. 13.

⁴⁹³ Queensland Rail, sub. 33, p. 4, para. 8.

⁴⁹⁴ South West Producers, sub. 40, p. 14.

⁴⁹⁵ Queensland Rail, sub. 33, attachment A.

As such, the QCA is satisfied that the facility for the service (and the facility for each part of the service identified by the QCA–see Appendix B) can meet the total foreseeable demand in the relevant market over the period for which the service (or each part of the service) would be declared.

11.7 At the least cost compared to any two or more facilities

11.7.1 Stakeholder submissions

Queensland Rail did not provide details regarding the costs across its rail systems in providing the service, or information regarding the consideration of 'any 2 or more facilities' in considering 'least cost'. However, it noted that its services are subsidised by TSC payments.⁴⁹⁶ There is limited publicly available data on such costs across Queensland Rail's systems.⁴⁹⁷

11.7.2 QCA analysis

The service provided by means of the Queensland Rail network is a rail access service (i.e. a service needed for the operation of trains). There is currently no other existing facility (i.e. rail transport infrastructure) that is capable of meeting any part of the foreseeable demand for the rail access service Queensland Rail provides, either in a single market or in a series of smaller markets. Put another way, there is no other railway network in Queensland that duplicates Queensland Rail's rail systems or extends across similar routes.

The QCA has considered whether the development of a potential alternative facility could meet part or all of the foreseeable demand in the relevant market or markets, and if so at what cost. It is clear that the development of such a potential alternative facility, whether for the whole service or an identified part of the service, would require extensive costs—including land acquisition, planning, design, development and construction. For the entire railway, approximately 6,000 km in length, this would cost billions of dollars. For any part of the service, the costs of duplication would obviously be less, but still sizable.

The NCC noted that it usually costs more to construct a new facility than to extend an existing one:

Railways generally exhibit natural monopoly characteristics and across a range of demand levels continued extension of an existing facility will usually involve less cost than constructing a new facility. As some of the major construction costs of a railway, such as earthworks, construction of bridges and signalling infrastructure are avoided or at least minimised when extending an existing railway, extension will almost always be cheaper than duplicating a facility. The Organisation for Economic Co-operation and Development (OECD), for example, has reported that expanding from single to double track roughly quadruples capacity at less than double the costs ...⁴⁹⁸

When the cost of providing the service using the existing Queensland Rail facility—which has spare capacity—is compared with the cost of providing the service using two or more facilities (including a potential alternative facility), and similarly for each relevant facility for each part of the service, it is clear that each existing Queensland Rail facility will meet total foreseeable demand in each market at the least cost. Total foreseeable demand is clearly not satisfied at

⁴⁹⁶ Queensland Rail, sub. 33, p. 24, para. 128.

⁴⁹⁷ An exception is the West Moreton system, which Queensland Rail (sub. 33, p. 68, para. 343) said generates around '\$44m per year in access revenue which is sufficient to cover operating costs'.

⁴⁹⁸ NCC, Central Queensland Coal Rail Network, Application for declaration of four services comprising the Central Queensland Coal Network under s. 44F(1) of the Trade Practices Act 1974 (Cth), draft recommendation, 14 September 2010, p. 28.

least cost by building another potential alternative facility to be used in addition to the existing Queensland Rail facility, in whole or in part, compared to simply using the existing Queensland Rail facility, in whole or in part, to satisfy total foreseeable demand.

Therefore, the QCA considers that the facility for the service (or the facilities for each identified part of the service) could meet the total foreseeable demand in the relevant market or markets at the least cost compared to any two or more facilities.

11.8 Conclusion

The QCA considers that the facility providing the Queensland Rail service as a whole satisfies criterion (b). That is, Queensland Rail's facility could meet total foreseeable demand in the market over the period for which the service would be declared, and at the least cost compared to any two or more facilities.

The QCA is also satisfied that the facility for each identified part of the service could meet total foreseeable demand in the market over the period for which the respective part of the service would be declared, and at the least cost compared to any two or more facilities. This is because, based on the evidence, each facility for each identified part of the service:

- has existing spare capacity and could meet total foreseeable demand in each market over the proposed declaration period
- could meet total foreseeable demand in the market at least cost compared to any two or more facilities.

12 CRITERION (C)—STATE SIGNIFICANCE

12.1 Introduction

Section 76(2)(c) of the QCA Act is expressed as follows:

that the facility for the service is significant, having regard to its size or its importance to the Queensland economy

A summary of key matters raised by stakeholders with respect to criterion (c), as well as the QCA's final recommendations, are set out in Table 10.

Criterion (c)						
lssue	Queensland Rail	Other stakeholders	QCA final recommendation			
Queensland Rail's service as a whole (i.e. as defined in s. 250(1)(b) of the QCA Act)	If the network is assessed as a whole, the entire network satisfies criterion (c) But the relevant inquiry is whether the eight facilities for the eight services (as identified by Queensland Rail) satisfy criterion (c)	Pacific National, Glencore, the South West Producers and GrainCorp each said that the Queensland Rail network as a whole satisfied criterion (c)	Criterion (c) is satisfied See section 12.6			
Each facility for each service identified by the QCA (see section 2.3)	North Coast Line, Metropolitan system, and Mount Isa Line each satisfy criterion (c) West Moreton system, Western system, South Western system, Central Western system and Tablelands system each do not satisfy criterion (c)	Glencore submitted that the Mount Isa Line (including the section of the North Coast Line that enables connection to the Port of Townsville) satisfies criterion (c) The South West Producers submitted that combinations of the West Moreton system and Metropolitan system satisfy criterion (c) Watco submitted that the South Western, Western and Central Western systems satisfy criterion (c) Pacific National and GrainCorp did not agree with Queensland Rail's approach but in any event considered each system is significant	Criterion (c) is satisfied in respect of each facility for each service identified by the QCA See section 12.7			

Table 10 Summary of key positions— s. 76(2)(c) of the QCA Act

12.2 The QCA's approach to assessing criterion (c)

12.2.1 Stakeholder submissions

Queensland Rail disagreed with the QCA's interpretation of criterion (c) in the draft recommendation that the word 'or' means only one of the two factors in s. 76(2)(c) of the QCA Act is required to be satisfied in order for criterion (c) to be satisfied.

Queensland Rail submitted that s. 76(2)(c) of the QCA Act, properly construed, requires a facility to be significant having regard to both its size and its importance to the Queensland economy.⁴⁹⁹ It argued that 'the interpretation of the meaning of "size" is often contextualised and significantly affected by the <u>mutually supportive</u> consideration of the economic "importance" of a facility' (original emphasis).⁵⁰⁰

Queensland Rail indicated that there was support for this approach 'apparent' in the National Competition Council's (NCC) assessment in practice under Part IIIA of the *Competition and Consumer Act 2010* (Cth) (CCA).⁵⁰¹ Queensland Rail further submitted that the NCC has not previously concluded that a facility is of national significance⁵⁰² solely because of its 'size' on physical dimensions alone, without supporting evidence pertaining to the economic importance of the facility.⁵⁰³

Accordingly, with regard to assessing the size of a facility, Queensland Rail submitted that:

first, the economic significance of the facility must be taken into account when assessing whether the facility is of sufficiently significant 'size' to be of 'significance', notwithstanding the use of the word 'or' in criterion (c); and

secondly, a rail system's length or geographic coverage should not, in and of itself, be determinative of whether the facility is of sufficient 'size' to be of 'significant' [sic].⁵⁰⁴

In relation to what is relevant in considering 'size', Queensland Rail submitted that the NCC's conclusions with respect to the 'size' of the Herbert River cane railway network are 'highly material' when assessing the systems operated by Queensland Rail.⁵⁰⁵ Queensland Rail also noted that the QCA's recognition that the throughput of goods or services using a facility can be relevant to determining size is grounded in jurisprudence⁵⁰⁶ and that by that measure rail systems with comparatively low levels of traffic are unlikely to satisfy the 'size' test in criterion (c).⁵⁰⁷

Queensland Rail further submitted that the principles for determining 'national significance' (under Part IIIA of the CCA) are of direct relevance for the QCA in applying criterion (c).⁵⁰⁸ Queensland Rail considered the following approach 'reflects an accurate application of the law, and the economic principles underpinning the law, for each relevant facility':

First, regarding 'size', the physical dimensions, whilst relevant, are to be informed by a consideration of the relevant economic activity facilitated by the system.

Secondly, regarding 'importance to the Queensland economy':

- 305.2.1 the total value and volume of throughput is highly relevant, especially with regard to the extent this throughput contributes to Queensland exports, movement of imports from Queensland ports, and regional development and employment in the Queensland freight industry;
- 305.2.2 the utilisation of the facility is highly relevant, reflecting the level of economic demand for the service provided by the facility; and

⁴⁹⁹ Queensland Rail, sub. 33, pp. 58–59, paras 286–90.

⁵⁰⁰ Queensland Rail, sub. 33, p. 58, para. 286.

⁵⁰¹ Queensland Rail, sub. 33, p. 58, para. 287.

⁵⁰² Applying the relevant test under Part IIIA of the *Competition and Consumer Act 2010* (Cth).

⁵⁰³ Queensland Rail, sub. 33, p. 59, para. 288.

⁵⁰⁴ Queensland Rail, sub. 33, p. 58, para. 280.

⁵⁰⁵ Queensland Rail, sub. 33, pp. 59–60, paras 293–294.

⁵⁰⁶ Queensland Rail, sub. 33, p. 60, paras 295–296.

⁵⁰⁷ Queensland Rail, sub. 33, p. 60, para. 297.

⁵⁰⁸ Queensland Rail, sub. 33, pp. 60–61, para. 301.

305.2.3 the commercial viability of the facility is relevant, reflecting the economic viability of the infrastructure.⁵⁰⁹

The South West Producers submitted that the QCA's interpretation of criterion (c) is consistent with the approach applied by the NCC.⁵¹⁰ They argued that Queensland Rail's reference to the NCC's assessment of the Herbert River cane railway under Part IIIA of the CCA is inappropriate, including because the Herbert River cane railway is a different railway and service, and it was being assessed for 'national significance'.⁵¹¹

GrainCorp submitted that Queensland Rail has conflated size and economic significance, which are separate considerations under the QCA Act. It argued that Queensland Rail's interpretation 'is manifestly at odds with the plain and ordinary meaning of the statute'.⁵¹²

12.2.2 QCA approach

As set out above, in response to the draft recommendation, Queensland Rail submitted that s. 76(2)(c), properly construed, requires a facility to be significant having regard to both its size *and* its importance to the Queensland economy.⁵¹³

The QCA approaches criterion (c) on the basis that it must have regard to both considerations in deciding whether the criterion is satisfied. However, the use of the word 'or' clearly indicates that the criterion may be satisfied based on only one of these considerations. For example, a facility that is relatively small in size may nevertheless be important to the Queensland economy. The statute also leaves room for the possibility that a 'large' facility may satisfy this criterion, even if its economic importance is debatable.

The QCA Act does not prescribe that the QCA must take into account specific indicators of 'size' or 'importance to the Queensland economy' in undertaking its assessment. While previous regulatory decisions in relation to the 'national significance' of other facilities under the CCA may be relevant, they are not determinative. Further, factors that were identified by the NCC as relevant to its assessment of the 'national significance' of other facilities, such as the Herbert River cane railway, may or may not be relevant to the assessment of the rail facilities before the QCA in the context of Part 5 of the QCA Act. The QCA acknowledges that depending on the facility being assessed, there may be indicators of 'size' that could be said to have an 'economic' flavour which are relevant to the assessment. The assessment of the significance of a facility—having regard to 'size' and 'importance to the Queensland economy'—and the weight that is attributed to any indicators, is a matter of judgement.

In *Pilbara Infrastructure Pty Ltd v Australian Competition Tribunal and Ors*⁵¹⁴ the High Court (by majority) observed that the equivalent provision of 'national significance' in Part IIIA of the CCA (which is broadly consistent with criterion (c), but not in identical terms)⁵¹⁵ 'may also direct attention to matters of broad judgment of a generally political kind'.⁵¹⁶ Accordingly, the QCA

⁵⁰⁹ Queensland Rail, sub. 33, p. 61, para. 305.

⁵¹⁰ South West Producers, sub. 31, p. 17; sub. 40, p. 42.

⁵¹¹ South West Producers, sub. 40, p. 42.

⁵¹² GrainCorp, sub. 52, p. 11.

⁵¹³ Queensland Rail, sub. 33, pp.58–59, paras 286–90.

^{514 (2012) 246} CLR 379.

 $^{^{515}}$ At the time of the High Court decision, this was s. 44G(2)(c) of the CCA; it is now s. 44CA(1)(c) of the CCA.

⁵¹⁶ (2012) 246 CLR 379 at [43]. This observation followed on from the majority's views in relation to the application of the equivalent of criterion (d) at [42].

has approached the assessment of significance as a matter of judgment rather than a matter of precise calculation.⁵¹⁷

Ultimately, whether this criterion is satisfied, and the basis on which this conclusion is reached, depends on the weight given by the decision-maker to the considerations prescribed in s. 76(2)(c).

12.3 The structure of criterion (c) analysis

The analysis of criterion (c) is set out as follows:

- a summary of stakeholder submissions on whether criterion (c) is satisfied
- a summary of data with respect to each of Queensland Rail's rail systems
- consideration of criterion (c) with respect to the facility for the service as a whole (i.e. the service as defined in s. 250(1)(b) of the QCA Act)
- consideration of criterion (c) with respect to the facilities for the parts of the service identified in the QCA's assessment of criterion (a) (see Appendix B).

12.4 Stakeholder submissions on criterion (c)

Queensland Rail accepted that if the network is assessed 'as a whole', it would satisfy criterion (c).⁵¹⁸ However, Queensland Rail considered that the relevant inquiry is whether the eight facilities for the eight services (as identified by Queensland Rail) that it provides, each individually satisfies criterion (c).

In its view:

- The North Coast Line satisfies criterion (c), due to its physical size and its strategic alignment, servicing four significant ports and major centres in Queensland.⁵¹⁹
- The Mount Isa Line satisfies criterion (c), due to its size and importance to the Queensland economy's transportation of bulk minerals and industrial products.⁵²⁰
- The Metropolitan system satisfies criterion (c), as it is of 'sufficient size' and clear importance to the Queensland economy.⁵²¹

The remaining systems, in Queensland Rail's view, do not satisfy criterion (c):

• The West Moreton system 'is not infrastructure that is of state significance with regard to its size or its importance to the Queensland economy (measured by contribution to exports and gross state product) to satisfy criterion (c)^{1,522} Further, it is not relevantly of state significance, due to the low volume of freight hauled compared to Queensland's total coal exports.⁵²³

⁵¹⁷ See also NCC, *Declaration of Services, A guide to declaration under Part IIIA of the Competition and Consumer Act 2010 (Cth),* April 2018 edn, p. 39, para. 5.4.

⁵¹⁸ Queensland Rail, sub. 33, p. 55, para. 262.

⁵¹⁹ Queensland Rail, sub. 33, pp. 63, 65, paras 309, 322.

⁵²⁰ Queensland Rail, sub. 33, pp. 63, 66, paras 309, 328.

⁵²¹ Queensland Rail, sub. 33, p. 63, paras 309, 314.

⁵²² Queensland Rail, sub. 33, p. 68, para. 344.

⁵²³ Queensland Rail, sub. 33, p. 63, para. 310.

 The Western, South Western, Central Western and Tablelands systems are not significant for the purpose of criterion (c), due to low volume/value of freight hauled with regard to contribution to, as appropriate, Queensland's exports, imports or domestic freight industry, low utilisation rates and the high degree of dependence on TSC revenue indicating a lack of commercial viability.⁵²⁴ According to Queensland Rail, while the QCA set out throughput volumes in the context of considering 'size' in the draft recommendation, these volumes are either not relied on by the QCA or do not support the QCA's findings in the draft recommendation.⁵²⁵

The South West Producers submitted that the Queensland Rail network as a whole satisfied criterion (c).⁵²⁶ They also considered that the West Moreton system and parts of the Metropolitan system, which provide the 'West Moreton corridor coal rail access service' (as defined by the South West Producers), satisfied criterion (c).⁵²⁷ Their view was the same with regard to the West Moreton and Metropolitan systems together.⁵²⁸

Pacific National submitted that the Queensland Rail network should be considered as a whole, and that as a whole it satisfied criterion (c). Pacific National argued that in any event, each part of the network identified by Queensland Rail is significant.⁵²⁹

Glencore said that the Queensland Rail network as a whole satisfied criterion (c).⁵³⁰ Glencore considered that criterion (c) is satisfied for the Mount Isa Line (including rail links to the Port of Townsville)⁵³¹, which is 'highly significant both having regard to its size (being approximately 1032 kilometres in length) and in its significance to the Queensland economy (as a major freight route for the North West minerals province, and an import [sic] regional route for livestock, agricultural products and regional passenger services)'.⁵³²

GrainCorp also submitted that Queensland Rail's network should be considered as a whole, and that as a whole it satisfied criterion (c). GrainCorp argued that in any event, each of the systems identified by Queensland Rail are also significant having regard to their size and importance to the Queensland economy.⁵³³ It commented that the South Western, Western and Central Western systems support the transfer of grain and the viability of the State's grain export industry.⁵³⁴

Watco submitted that the South Western, Western and Central Western systems satisfy criterion (c), having regard to size. It also considered that these rail systems are significant having regard to their importance to the Queensland economy. While Watco recognised that these systems do not transport large volumes of bulk freight, it stated that they 'do provide key transport links accessing regional Queensland agricultural markets and communities'.⁵³⁵

Linfox submitted that the Central West rail system (as it defined it) satisfied criterion (c).536

⁵²⁴ Queensland Rail, sub. 33, pp. 63, 70–74, paras 311, 359, 370, 379, 386.

⁵²⁵ Queensland Rail, sub. 33, p. 68, para. 347.

⁵²⁶ South West Producers, sub. 4, pp. 5, 44.

⁵²⁷ South West Producers, sub. 4, pp. 5, 6, 44–49; sub. 16, pp. 18–25.

⁵²⁸ South West Producers, sub. 31, pp. 17–18; sub. 40, pp. 42–45.

⁵²⁹ Pacific National, sub. 9, p. 12; sub. 37, p. 15.

⁵³⁰ Glencore, sub. 5, p. 16.

⁵³¹ Glencore, sub. 41, p. 2.

⁵³² Glencore, sub. 5, p. 3.

⁵³³ GrainCorp, sub. 52, p. 11.

⁵³⁴ GrainCorp, sub. 52, p. 11.

⁵³⁵ Watco, sub. 48, p. 5.

⁵³⁶ Linfox, sub. 50, para. 3.2.

12.5 Data on Queensland Rail's rail systems

This section contains information about the Queensland Rail network by reference to its systems (as widely described and understood by Queensland Rail and users).

12.5.1 North Coast Line

The North Coast Line extends from Nambour (near Brisbane) north along Queensland's eastern coastline to Cairns, consisting of approximately 1,428 km of track (excluding Parana to Rocklands and Kaili to Durroburra).⁵³⁷

The line connects and services the major population centres of Brisbane (via the Metropolitan system), Bundaberg, Gladstone, Rockhampton, Mackay, Townsville and Cairns. It also connects to various major ports, including the Ports of Townsville, Mackay, Gladstone and Brisbane.⁵³⁸ The North Coast Line interconnects with the Metropolitan system (at Nambour), the Mount Isa Line (at Stuart, near Townsville), and the Tablelands system (at Cairns). Aurizon Network's Central Queensland Coal Network (CQCN) also intersects with the North Coast Line at various points in the Central Queensland region.

The North Coast Line is the principal regional freight and passenger line within the Queensland Rail network.⁵³⁹ The system carries predominantly intermodal freight (comprising 86 per cent of total freight carried in gtk terms⁵⁴⁰ in 2017–18)⁵⁴¹ between Brisbane and major regional centres in Queensland, including Rockhampton, Mackay, Townsville and Cairns. It also carries:

- sugar traffic from sugar mills to the Ports of Mackay and Townsville
- minerals exports, mining inputs and industrial products between the Mount Isa Line and the Port of Townsville
- various agricultural products from the Mount Isa Line and Central Western system to various export ports, including the Port of Brisbane
- long-distance passenger and tourism services, including on the Spirit of Queensland (Brisbane to Cairns), the Spirit of the Outback (Brisbane to Longreach), the Tilt Train (Brisbane to Rockhampton) and The Westlander (Brisbane to Charleville).⁵⁴²

In 2017–18, the North Coast Line transported approximately 6,700 million gtk of freight and passengers.⁵⁴³ A breakdown of volumes carried by category is provided in Figure 20.

⁵³⁷ Queensland Rail, sub. 33, pp. 63–64, para. 315.

⁵³⁸ Queensland Rail, sub. 33, pp. 63–64, para. 315.

⁵³⁹ Queensland Rail, North Coast line system, https://www.queenslandrail.com.au/forbusiness/the-regionalnetwork/north-coast-line-system.

⁵⁴⁰ Gross tonne kilometres (gtk) is a measure of the level of operating activity on a particular rail system. It is the product of the total gross weight of the train (i.e. including the locomotives and wagons used, as well as the goods and passengers carried) and the distance (in kilometres) travelled by the train.

⁵⁴¹ Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p. 11, https://www.queenslandrail.com.au/business/acccess/Compliance%20and%20reporting/2017-18%20QCA%20Annual%20Performance%20Report.pdf.

⁵⁴² Queensland Rail, North Coast Line System North Information Pack, October 2016, p. 6, https://www.queenslandrail.com.au/business/acccess/Documents/North%20Coast%20Line%20North%20System% 20Information%20Pack%20-%20Issue%203%20-%20October%202016.pdf; Queensland Rail Travel, https://www.queenslandrailtravel.com.au/.

 ⁵⁴³ Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p. 11.



Figure 20 North Coast Line freight volumes by commodity in 2017–18 (million gtk)

Source: Adapted from Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, p. 11.

Queensland Rail submitted that in 2016–17 approximately 40 per cent of the state's sugar exports were carried on the North Coast Line, along with approximately 12 per cent of the state's total containerised freight.⁵⁴⁴ Linfox noted that the train services it operates on the North Coast Line (and Central Western system) 'deliver essential items to communities across Central, Western and Northern Queensland and provide a critical link for regional Queensland agricultural, manufactured and bulk commodity products to reach southern and export markets'.⁵⁴⁵

Queensland Rail data show that in 2017–18, Queensland Rail received approximately \$46.6 million in access charges for the below-rail service it provided on the North Coast Line.⁵⁴⁶

12.5.2 Mount Isa Line

The Mount Isa Line extends from Stuart (near Townsville) west to Mount Isa, including the Flynn to Phosphate Hill branch line. Taken altogether, the system is approximately 1,039 km in length, consisting of a non-electrified single track. The Mount Isa Line services a number of regional communities, and acts as a critical link from the North West Minerals Province to the Port of Townsville (via the North Coast Line between Stuart and Townsville). It also has a significant 'back-haul' freight task in delivering mining and industrial inputs to the Mount Isa region.⁵⁴⁷

⁵⁴⁴ Queensland Rail, sub. 33, p. 64, para. 319.

⁵⁴⁵ Linfox, sub. 50, para. 1.4.

⁵⁴⁶ Queensland Rail, *Financial Statements for the Year Ended 30 June 2018, Below Rail Services Provided by Queensland Rail*, p. 4,

https://www.queenslandrail.com.au/business/acccess/Compliance%20and%20reporting/2017-18%20Below%20Rail%20Financial%20Statements.pdf.

⁵⁴⁷ Queensland Rail, sub. 33, p. 65, para. 323; Queensland Rail, *Mount Isa System Information Pack*, February 2017, p.5,

https://www.queenslandrail.com.au/business/acccess/Documents/Mt%20Isa%20System%20Information%20Pack %20-%20Issue%203.1%20-%20February%202017.pdf.

The Mount Isa Line carries predominantly:

- industrial products and metals—including acid, fertiliser, intermodal freight (mining inputs) and metals, totalling 2,735 million gtk in 2017–18, or 62 per cent of total freight carried on the line
- minerals concentrate—including copper, lead, zinc, magnetite and sulphur, totalling 1,532 million gtk in 2017–18, or 35 per cent of total freight carried on the line
- agriculture (including livestock), passengers and other freight—including the long-distance
 'The Inlander' (Townsville to Mount Isa) passenger and tourism service.⁵⁴⁸

In 2017–18, the Mount Isa Line transported 4,377 million gtk of freight and passengers. A breakdown of volumes carried by commodity on the Mount Isa Line is provided in Figure 21.

Figure 21 Mount Isa Line freight volumes by commodity in 2017–18 (million gtk)



Agriculture includes livestock

Source: Adapted from Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, p. 11.

Queensland Rail previously described the Mount Isa Line as 'of particular national interest as it runs along some of the world's largest deposits of copper, lead, zinc, silver and phosphate rock', also stating that the region around the Mount Isa Line produces 75 per cent of Queensland's non-coal mineral output.⁵⁴⁹ According to Queensland Rail, the asset replacement value of the Mount Isa Line (as at 2012) was \$12.8 billion, with the regional mineral production using the line worth approximately \$6.67 billion.⁵⁵⁰

 ⁵⁴⁸ Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p. 11.

⁵⁴⁹ Queensland Rail, Mount Isa System Information Pack, February 2017, p. 5, https://www.queenslandrail.com.au/business/acccess/Documents/Mount%20Isa%20system%20information%20p ack%20-%20Issue%203%20-%20October%202016.pdf.

⁵⁵⁰ Queensland Rail, Mount Isa Line Rail Infrastructure Master Plan, 2012, p. 6, https://www.queenslandrail.com.au/business/acccess/Documents/Maps/QR4159.1%20Infrastructure%20Master %20Plan%202012_Updated_LR.pdf.

The Mount Isa Line is the only Queensland Rail system that operates on an entirely commercial basis—it does not receive or rely upon below-rail subsidies to remain commercially viable to operate.⁵⁵¹ In 2017–18, Queensland Rail received approximately \$74.3 million in access charges from users of the Mount Isa Line.⁵⁵² In June 2019, the Queensland Government announced \$20 million in annual funding for four years in incentives for commercial freight users on the Mount Isa Line, with rail payments paid to eligible users of the rail service.⁵⁵³

Glencore stated that '[t]he mineral deposits in the North West region are brought to market by miners such as Glencore using the Mount Isa Line as a critical supply chain—with the production brought to market making up 75 per cent of Queensland's non-coal mineral output'.⁵⁵⁴

Glencore also submitted that the Mount Isa Line is significant due to its importance to the Queensland economy, when having regard to the Mount Isa Line as a national key freight route and the significant economic contributions to Queensland from the mining, processing and other industries that depend on access to the relevant service.⁵⁵⁵ In 2016–17, contributions of the resources sector in the North West Queensland region included:

- \$1.3 billion in gross regional product (in the North West region)
- \$397 million in wages paid to 2,709 full time employees
- \$544 million in royalties
- \$354 million spent to the benefit of local businesses and community organisations.⁵⁵⁶

12.5.3 West Moreton system

The West Moreton system extends from Rosewood west to Miles, consisting of approximately 314 km of non-electrified, predominantly single track.⁵⁵⁷ At Rosewood (in the east), the West Moreton system connects with the Metropolitan system; at Miles (in the west), the West Moreton system connects with the Western system; the Western system's Dalby to Meandarra branch line also connects to the West Moreton system at Dalby. Queensland Rail submitted that the total haul distance from the furthest West Moreton system coal mine to the Port of Brisbane is 380 km.⁵⁵⁸

The West Moreton system (together with the Metropolitan system) serves three mines that export via the Port of Brisbane—Jeebropilly (New Hope), New Acland (New Hope) and Cameby Downs (Yancoal).⁵⁵⁹ There are other coal mines near the West Moreton system that supply

⁵⁵¹ Queensland Rail, sub. 8, p. 1, para. 4.

⁵⁵² Queensland Rail, *Financial Statements for the Year Ended 30 June 2018: Below Rail Services Provided by Queensland Rail*, 2018, p. 4.

⁵⁵³ A Palaszczuk, J Trad and M Bailey, More minerals freight to go on the fast track thanks to rail subsidy, media release, Queensland Government, 18 November 2019, http://statements.qld.gov.au/Statement/2019/11/18/more-minerals-freight-to-go-on-the-fast-track-thanks-to-railsubsidy.

⁵⁵⁴ Glencore, sub. 5, pp. 16–17. See also Glencore, sub. 17, p. 19.

⁵⁵⁵ Glencore, sub. 5, pp. 17–18; Glencore, sub. 17, p. 19.

⁵⁵⁶ Glencore, sub. 5, p. 17.

⁵⁵⁷ Queensland Rail, West Moreton System Information Pack, October 2016, pp. 12–16, https://www.queenslandrail.com.au/business/acccess/Documents/West%20Moreton%20System%20Information %20Pack%20-%20Issue%203.1%20-%20October%202016.pdf.

⁵⁵⁸ Queensland Rail, sub. 33, p. 66, para. 329.

⁵⁵⁹ South West Producers, sub. 4, p. 7.

nearby coal power stations⁵⁶⁰, thus the system currently serves five operating mines.⁵⁶¹ The West Moreton system has the capacity to deliver up to 10.8 million tonnes per annum (mtpa) of coal to the Port of Brisbane.⁵⁶² Additionally, various agricultural products from the Western and South Western systems travel via the West Moreton system (and Metropolitan system) to the Port of Brisbane.

Queensland Rail said that 7.17 million tonnes of coal was transported on the West Moreton system in 2017–18. Thermal coal accounts for 94 per cent of annual train paths and 98 per cent of tonnages transported on the system.⁵⁶³ Queensland Rail highlighted the uncertainty surrounding coal volumes, including the 2013 closure of Peabody's Macalister mine and the future of the New Acland Stage 3 development. If this development proceeds, Queensland Rail expects coal volumes will increase to 9 mtpa but if not, volumes will reduce to 2 mtpa.⁵⁶⁴

The West Moreton system transported approximately 2,494 million gtk of freight and passengers in 2017–18.⁵⁶⁵ A breakdown of volumes carried by commodity on the West Moreton system is provided in Figure 22.



Figure 22 West Moreton system freight volumes by commodity in 2017–18 (million gtk)

Source: Adapted from Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p. 7.

Queensland Rail submitted that coal volumes hauled on the West Moreton system accounted for 3.2 per cent of Queensland coal exports and 0.21 per cent of gross state product (GSP) in

⁵⁶⁰ BITRE, *Freightline 4—Australian coal freight transport*, Department of Infrastructure and Regional Development, Australian Government, 2016, p. 14, https://bitre.gov.au/publications/2016/files/Freightline_04.pdf.

⁵⁶¹ South West Producers, sub. 4, p. 45.

⁵⁶² BITRE, *Freightline 4—Australian coal freight transport,* Department of Infrastructure and Regional Development, Australian Government, 2016, p. 14.

⁵⁶³ Queensland Rail, sub. 33, p. 66, para. 332.

⁵⁶⁴ Queensland Rail, sub. 33, p. 66, paras 333–34.

 ⁵⁶⁵ Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, p.
 7.

2017–18.⁵⁶⁶ It also compared the approximately 7 mtpa of coal transported on the West Moreton system with the 220 mtpa hauled on the CQCN.⁵⁶⁷ It refuted the QCA's comparison of volumes hauled on the West Moreton and Mount Isa Lines, arguing that the Mount Isa Line transports approximately 72 per cent of the state's mineral and metal exports.⁵⁶⁸

The South West Producers submitted that the West Moreton system and the relevant parts of the Metropolitan systems are significant, having regard to their importance to the Queensland economy:

QR's rail network generally, and/or those parts of it which provide the West Moreton corridor coal rail access service are significant, having regard to:

... its importance to the Queensland economy, in the case of the West Moreton corridor through servicing coal, grain, livestock and passenger traffic and the economic contribution made by the coal mines, rail haulage, coal handling services, and resulting royalties and indirect economic benefits of the coal supply chain which the West Moreton system rail transport infrastructure forms part of.⁵⁶⁹

The South West Producers also said that the West Moreton rail corridor is significant, as it has been 'specifically recognised by the National Transport and Infrastructure Council as a national key freight route', and due to its economic contributions to the state.⁵⁷⁰

The Department of Infrastructure, Transport, Cities and Regional Development explains that identifying national key freight routes:

assist governments and industry to better understand, and plan for, critical freight flows, and will be maintained to inform a range of national transport infrastructure policy, regulatory, planning and operational issues. It provides a detailed picture of the road and rail routes connecting Australia's nationally significant places for freight.⁵⁷¹

In 2017–18, Queensland Rail received approximately \$42.8 million in coal access charges, and \$2.5 million in non-coal access charges, from providing its below-rail service on the West Moreton system.⁵⁷² In this period, the West Moreton system received approximately \$0.74 million in government subsidies.⁵⁷³ This suggests that, unlike most of Queensland Rail's network (with the exception of the Mount Isa Line), usage of the West Moreton system has been predominantly funded by its users on a commercial basis, and the West Moreton service has not relied heavily on government subsidies to operate.

The South West Producers provided data on the indirect contributions to the economy facilitated by access to the West Moreton rail corridor, saying that together, the 'economic

⁵⁶⁶ Queensland Rail, sub. 33, p. 67, paras 335, 337.

⁵⁶⁷ Queensland Rail, sub. 33, p. 67, para. 339.

⁵⁶⁸ Queensland Rail, sub. 33, p. 67, para. 341.

⁵⁶⁹ South West Producers, sub. 4, pp. 5–6.

⁵⁷⁰ South West Producers, sub. 4, pp. 46–47. See also Department of Infrastructure, Transport, Cities and Regional Development, Freight Network, Australian Government web page, http://maps.infrastructure.gov.au/KeyFreightRoute/index.html.

⁵⁷¹ Department of Infrastructure, Transport, Cities and Regional Development, Freight Network, Australian Government web page, https://www.infrastructure.gov.au/transport/freight/network.aspx.

⁵⁷² Queensland Rail, Financial Statements for the Year Ended 30 June 2018: Below Rail Services Provided by Queensland Rail, 2018, p. 4.

⁵⁷³ Based on publicly available data, it was not possible to determine the amount of subsidies received by the Metropolitan system alone as this number is not separately reported for the Metropolitan system, and is instead grouped in the reported number for the 'rest of the network'. See Queensland Rail, *Financial Statements for the Year Ended 30 June 2018: Below Rail Services Provided by Queensland Rail*, 2018, p. 4.

...

contributions made by the coal, freight and passenger services that travel the West Moreton coal rail access service are significant^{1,574} They said:

Coal from the West Moreton network constituted 22 per cent of the Port of Brisbane's throughput

In addition, QR, Aurizon and QBH employ a material number of people in connection with below rail and above rail operations on the West Moreton corridor, and the coal handling services at the Port of Brisbane, which form part of the West Moreton coal supply chain. ⁵⁷⁵

Further, the South West Producers provided data from the Queensland Resources Council (QRC) that shows that in 2016–17, the resources industry in the West Moreton region contributed approximately \$30 million in wages paid to 263 full time employees, \$7 million spent on regional communities, and \$3.8 billion in royalties paid to the state.⁵⁷⁶

The South West Producers also highlighted statements made by the Minister for Transport and Main Roads when announcing \$28 million of upgrades to the West Moreton system in February 2018 and mentioning the support it provides to the local tourism economy, agriculture and resource sectors:

The West Moreton system is a critical link for rail services from Brisbane to the west and south west communities of the state and is a major artery to the Darling Downs, which is predominantly used to transport thermal coal and grains.⁵⁷⁷

The South West Producers said that coal mining activities in the South West generate approximately \$700 million in annual revenue.⁵⁷⁸ They did not agree with Queensland Rail's comparison of the significance of the system to the CQCN, submitting that the threshold is whether the facility itself is significant, not in comparison to other facilities.⁵⁷⁹

12.5.4 Metropolitan system

The Metropolitan system radiates from Roma Street station in Brisbane's CBD, and extends north to Nambour, where it joins the North Coast Line; east via various branch lines, including the dedicated dual gauge freight and coal lines from Lytton Junction to reach the Port of Brisbane at the Fisherman Islands; south through the Gold Coast area to the Varsity Lakes station (where it terminates); west via Ipswich to Rosewood, where it connects with the West Moreton system; and south-west to the Acacia Ridge Terminal, where it connects with the interstate network to New South Wales. The entire system consists of approximately 612 km of track, with mostly duplicated, electrified lines in the metropolitan region.⁵⁸⁰

The Metropolitan system predominantly carries passenger services for metropolitan Brisbane, as well as freight services for agriculture, coal and intermodal customers. In order to reach the Port of Brisbane, freight from the West Moreton system (and the Western and South Western systems, which connect to the West Moreton system), as well as the North Coast Line must travel through the Metropolitan system. Additionally, all interstate rail services (travelling

⁵⁷⁴ South West Producers, sub. 4, p. 47.

⁵⁷⁵ South West Producers, sub. 4, p. 47.

⁵⁷⁶ South West Producers, sub. 4, p. 49.

⁵⁷⁷ Minister for Transport and Main Roads, cited in South West Producers, sub. 31, p. 18.

⁵⁷⁸ South West Producers, sub. 40, p. 44.

⁵⁷⁹ South West Producers, sub. 40, p. 44.

⁵⁸⁰ Queensland Rail, Brisbane Metropolitan System Information Pack, October 2016, https://www.queenslandrail.com.au/business/acccess/Documents/Brisbane%20Metropolitan%20System%20Infor mation%20Pack%20-%20Issue%203%20-%20October%202016.pdf.

to/from any of Queensland Rail's rail systems) must travel on the Metropolitan system in order to connect to the interstate network at Acacia Ridge (south-west of Brisbane).

In 2017–18, the Metropolitan system transported approximately 1,900 million gtk of coal, agricultural and intermodal traffic (excluding Citytrain).⁵⁸¹ In 2017–18, Queensland Rail received approximately \$19.3 million in access charges from coal trains accessing the below-rail service on the Metropolitan system.⁵⁸² A breakdown of volumes carried by commodity on the Metropolitan system, excluding Citytrain services, is provided in Figure 23.



Figure 23 Metropolitan system freight volumes by commodity in 2017–18 (million gtk)



In 2017–18, 53.66 million passenger trips were taken on the Metropolitan system (the Citytrain network)⁵⁸³, with regional passenger services⁵⁸⁴ also departing from Brisbane and travelling along the Metropolitan system.

12.5.5 South Western system

Queensland Rail submitted that the South Western system is approximately 617.5 km in length.⁵⁸⁵ The South Western system's primary corridor extends from Toowoomba to Thallon via Warwick, with branch lines from Warwick to Wallangarra and Wyreema to Millmerran. The South Western system connects to the West Moreton system at Toowoomba.⁵⁸⁶

 ⁵⁸¹ Queensland Rail, 2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2017, p.
 12.

⁵⁸² Queensland Rail, Financial Statements for the Year Ended 30 June 2018: Below Rail Services Provided by Queensland Rail, 2018, p. 4.

⁵⁸³ Queensland Rail, Annual and Financial Report 2017–18, 2018, p. 19.

⁵⁸⁴ Including the Spirit of Queensland (Brisbane to Cairns), the Spirit of the Outback (Brisbane to Longreach), the tilt train (Brisbane to Rockhampton) and the Westlander (Brisbane to Charleville).

⁵⁸⁵ Queensland Rail, sub. 33, p. 70, para. 360.

⁵⁸⁶ Queensland Rail, South Western System Information Pack, October 2016, p. 5, https://www.queenslandrail.com.au/business/acccess/Documents/South%20Western%20System%20Information %20Pack%20-%20Issue%203%20-%20October%202016.pdf.

The theoretical capacity limitation for the South Western system is the available capacity across the Toowoomba Range. Out of the approximately 112 return train paths across the Toowoomba Range per week, 14 paths are preserved for primary industry, with a further 19 uncontracted paths available for access by primary producers. However, Queensland Rail data indicates that actual utilisation across the Toowoomba Range has never reached this theoretical capacity (in terms of return train paths) in the last 10 years.⁵⁸⁷

Traditionally, grains, cotton lint and cotton seed have been the primary products hauled on this line, with the volumes of agricultural products varying seasonally. Queensland Rail submitted that in 2016–17, 329,000 tonnes of grain was transported on the South Western system (which is the maximum throughput on the Toowoomba to Warwick section of the system). According to Queensland Rail, this volume comprised 12 per cent of Queensland's grain exports.⁵⁸⁸

Queensland Rail also stated:

There has been significant change in the transport market from the South West, with all cotton movements switching from rail to road from 2014-15. As a result of increases to heavy vehicle mass limits, over 85% of grain is moved in containers by trucks from the South West to the Port of Brisbane.⁵⁸⁹

Queensland Rail stated that it has 90 per cent excess capacity on the South Western system, adding that:

- train path utilisation on the Warwick to Goondiwindi section was 8 per cent in 2016–17
- there were less than four return services per week between Goondiwindi and Toowoomba.⁵⁹⁰

12.5.6 Western system

The Western system consists of the corridor from Miles to Cunnamulla with branch lines— Westgate to Quilpie, Dalby to Glenmorgan, Miles to Wandoan and Tycanba to Jandowae. The Western system adjoins the western section of the West Moreton system at Miles, with Western system branch lines running directly off the West Moreton system.⁵⁹¹ Queensland Rail submitted that the Western system is over 1,082 km in length, although currently 312.8 km of this is non-operational.⁵⁹²

The Western system mainly carries agricultural products, including grain and livestock (cattle), general freight (e.g. pipes), and The Westlander passenger service between Brisbane and Charleville. Freight traffic from the Western System travels via the West Moreton and Metropolitan systems. Grain and cotton are exported, and other freight is imported, via the

⁵⁸⁷ Doyle, G, South Western Rail System, Queensland Rail, presentation to Rail Forum—Explore best utilisation of existing rail infrastructure on the South West line, Goondiwindi, 4 December 2015.

⁵⁸⁸ Queensland Rail, sub. 33, p. 71, para. 365.

⁵⁸⁹ Queensland Rail, sub. 33, p. 71, para. 363.

⁵⁹⁰ Queensland Rail, sub. 33, p. 71, para. 367.

⁵⁹¹ Queensland Rail, Western System Information Pack, October 2016, p. 5, https://www.queenslandrail.com.au/business/acccess/Documents/Western%20System%20Information%20Pack% 20-%20Issue%203%20-%20October%202016.pdf.

⁵⁹² Queensland Rail, sub. 33, p. 69, para. 352.

Port of Brisbane, with livestock transported to processors in Dinmore and Holmview (in the Brisbane region).⁵⁹³

Queensland Rail submitted that in 2016–17, 44,000 tonnes of grain was transported on the Western system (Miles to Roma, which joins the West Moreton system) and 7,000 tonnes of livestock (Miles to Roma). The Westlander passenger service operated four one-way services a week. It stated that the most heavily utilised section, being Miles to Roma, operated at 3 per cent of available capacity, with 42 return freight services operated during the year. The Roma to Charleville section recorded a total of 18 return services.⁵⁹⁴ Queensland Rail attributed this underutilisation to competition from road transport and comparatively small freight demand from this origin.

12.5.7 Central Western system

Queensland Rail submitted that the Central Western system is approximately 704 km in length.⁵⁹⁵ The Central Western system runs from Emerald west to Winton via Longreach, and includes the Emerald to Clermont and Blair Athol branch line. It connects to Aurizon Network's Blackwater system at Emerald/Nogoa.⁵⁹⁶ Freight traffic on the Central Western system typically travels via the Blackwater system from/to the North Coast Line, and from/to various ports including Rockhampton, Gladstone and Brisbane.⁵⁹⁷

The Central Western system mainly carries rural commodities (including grain and livestock), containerised freight, and the Spirit of the Outback passenger service between Brisbane and Longreach.⁵⁹⁸

Queensland Rail provided the following maximum throughput information for the Central Western system for 2016–17:

- Freight (Nogoa to Emerald): 15,000 tonnes
- Grain (Nogoa to Emerald): 110,000 tonnes (volumes transported constitute approximately 4 per cent of Queensland grain exports)
- Livestock (Nogoa to Emerald): 38,000 tonnes (volumes transported constitute approximately 3 per cent of Queensland livestock exports)
- Spirit of the Outback passenger service: four one-way services per week.⁵⁹⁹

It stated that 93 per cent of available train paths on the Emerald to Longreach section are unused, with 97 per cent unused on the Nogoa to Emerald section. If passenger services are excluded, there were fewer than five return services per week on the Nogoa to Emerald section.⁶⁰⁰

⁵⁹³ Queensland Rail, Western System Information Pack, October 2016, p. 5, https://www.queenslandrail.com.au/business/acccess/Documents/Western%20System%20Information%20Pack% 20-%20Issue%203%20-%20October%202016.pdf.

⁵⁹⁴ Queensland Rail, sub. 33, p. 70, para. 356.

⁵⁹⁵ Queensland Rail, sub. 33, p. 72, para. 371.

⁵⁹⁶ Queensland Rail, Central West system, accessed 15 November 2019, https://www.queenslandrail.com.au/forbusiness/the-regional-network/central-western-system.

⁵⁹⁷ See for example, Linfox, sub. 50, para. 1.3.

⁵⁹⁸ Queensland Rail, Central West system, accessed 15 November 2019, https://www.queenslandrail.com.au/forbusiness/the-regional-network/central-western-system.

⁵⁹⁹ Queensland Rail, sub. 33, attachment A, p. 13.

⁶⁰⁰ Queensland Rail, sub. 33, p. 73, para. 376.

12.5.8 Tablelands system

Queensland Rail indicated the Tablelands system is 575 km in length, comprising two separate rail corridors (which do not adjoin each other):

- Cairns to Forsayth—423 km
- Normanton to Croydon—151.8 km.⁶⁰¹

Historically, the Tablelands system has carried sugar and molasses for the many sugarcane farms around the Atherton Tablelands region. However, no freight traffic currently operates on the Tablelands system, and the system is used to carry three tourism (passenger) services: the Gulflander service from Normanton to Croydon and the Kuranda Scenic Railway service from Cairns to Kuranda (both are operated by Queensland Rail), as well as the Savannahlander service from Cairns to Forsayth (operated by a private company).⁶⁰²

In 2017–18, more than 411,000 passengers travelled on the Kuranda Scenic Railway service, accounting for approximately 60 per cent of all passengers travelling on the Queensland Rail regional tourism network that year.⁶⁰³

12.5.9 General information regarding the Metropolitan, South Western, Western, Central Western and Tablelands systems

Separate information is not available on the revenue and funding sources of the Metropolitan, South Western, Western, Central Western and Tablelands systems. This is because Queensland Rail groups these systems together as 'rest of network' for financial reporting purposes, and there is no publicly available disaggregated financial data reported for each of the systems.

Based on Queensland Rail's financial statements, the South Western, Western, Central Western, Tablelands and Metropolitan systems together received \$131.2 million in non-coal access charges in 2017–18.⁶⁰⁴ However, some proportion of these access charges may be attributable to 'internal charges' from within Queensland Rail—'internal charges treated as revenue are access charges, telecommunications charges and ancillary infrastructure charges'. Queensland Rail's financial statements show that in 2017–18, \$138.6 million of internal access charges were expensed.⁶⁰⁵ It is not clear what amount of internal charges is attributable to each of Queensland Rail's systems; however, Queensland Rail submitted that in 2017–18, only \$1.1 million was received as external access revenue from the South Western, Western and Central Western systems.⁶⁰⁶

In 2017–18, approximately \$413.6 million was received in Transport Service Contract (TSC) payments for the operation of the South Western, Western, Central Western, Tablelands and Metropolitan systems.⁶⁰⁷ That is, excluding revenue from coal access charges on the Metropolitan system, approximately 75 per cent of the revenue received on the South Western,

⁶⁰¹ Queensland Rail, sub. 33, p. 73, para. 380.

⁶⁰² Queensland Rail, sub. 33, p. 73, para. 384; sub. 33, attachment A, p. 14.

⁶⁰³ Queensland Rail, Annual and Financial Report 2017–18, p. 42.

⁶⁰⁴ Queensland Rail, *Financial Statements for the Year Ended 30 June 2018: Below Rail Services Provided by Queensland Rail*, 2018, p. 4.

⁶⁰⁵ For example, internal charges may be access charges paid by Queensland Rail's above-rail passenger services to its below-rail service for access: Queensland Rail, *Financial Statements for the Year Ended 30 June 2017*: *Below Rail Services Provided by Queensland Rail*, 2017, pp. 9, 15.

⁶⁰⁶ Queensland Rail, sub. 33, p. 69, para. 350.

⁶⁰⁷ Queensland Rail, *Financial Statements for the Year Ended 30 June 2018*: *Below Rail Services Provided by Queensland Rail*, 2018, p. 4.

Western, Central Western, Tablelands and Metropolitan systems was from government subsidies.

Consistent with this information, Queensland Rail submitted that these systems 'are heavily underutilised and are substantially subsidised by TSC payments for regional development and other public policy purposes'.⁶⁰⁸

GrainCorp submitted that it relies heavily on access to the South Western, Western and Central Western systems for transport of grain to export facilities. It said that while export volumes vary annually, on average around 2 mtpa are exported out of Queensland, with approximately 40 per cent of this transported by rail to export facilities. GrainCorp submitted that the sale of exported wheat and other grains generates revenue of around \$600 million per annum for the Queensland economy.⁶⁰⁹

Watco submitted that the South Western, Western and Central Western systems satisfy criterion (c), having regard to their size as well as their importance to the Queensland economy. While it recognised that these systems do not transport large volumes of bulk freight, 'these rail systems do provide key transport links accessing regional Queensland agricultural markets and communities'.⁶¹⁰

12.6 Criterion (c)—the facility for the service as a whole

The QCA considers that the Queensland Rail network as a whole—that is, the facility for the service described in s. 250(1)(b) of the QCA Act, which includes seven regional systems and the Metropolitan system—is significant, having regard to its size or its importance to the Queensland economy.

12.6.1 Size

Stakeholders provided slightly different estimates of the physical dimensions (total rail track distance) of the Queensland Rail network. For example, Queensland Rail stated that its rail network 'extends more than 6,600 km across Queensland'; Glencore stated that 'the whole of the QR network is clearly significant when regard is had to the 7,000 km of rail track and associated infrastructure that it includes, as well as the significant area of the state of Queensland that the network covers'; and New Hope and Yancoal said that 'the QR Network as a whole is clearly significant, consisting of approximately 8,000km of track which covers a significant proportion of the State'.⁶¹¹ Estimates of the total length of rail track may differ, based on factors such as whether yards, sidings, passing loops and duplicate tracks are included.

Moreover, the eight systems are interconnected (with the exception of the standalone Normanton to Croydon track on the Tablelands system), forming a network that extends across a significant area of the state.⁶¹²

Queensland Rail data show that approximately 15.4 billion gtk of long distance passengers and freight (including intermodal, coal, minerals, industrial products and agricultural products) were

⁶⁰⁸ Queensland Rail, sub. 33, p. 69, para. 349.

⁶⁰⁹ GrainCorp, sub. 52, p. 11.

⁶¹⁰ Watco, sub. 48, p. 5.

⁶¹¹ Queensland Rail, sub. 8, p. 2, para. 6; Glencore, sub. 5, p. 16; South West Producers, sub. 4, p. 44.

⁶¹² The Central Western system is interconnected with the North Coast Line through the Aurizon Network's Central Queensland Coal Network (CQCN), and freight and passenger traffic have historically been permitted access across the CQCN, for example the Spirit of the Outback passenger service from Brisbane to Longreach. See Aurizon Network, sub. 6, pp. 45–47.

carried on the Mount Isa Line, North Coast Line, West Moreton and Metropolitan systems in 2017–18.⁶¹³ The QCA understands that together, these systems account for approximately 97 per cent of all freight tonnage transported on Queensland Rail's network.⁶¹⁴

In terms of passengers carried, Queensland Rail estimated that 53.66 million passenger trips were taken in 2017–18 on the Citytrain network (on the Metropolitan system), and more than 750,000 passengers travelled on the Queensland Rail regional systems as part of the Queensland Rail travel and tourism network.⁶¹⁵

The QCA considers that Queensland Rail's network is of a substantial physical size (at least 6,600 km of rail track based on estimates provided), covering a large geographic area across the state. Further, substantial volumes of freight and numbers of passengers are transported on the network. The QCA therefore considers that the Queensland Rail network as a whole is significant having regard to its size.

12.6.2 Importance to the Queensland economy

The QCA considers that the Queensland Rail network as a whole is a vital component of the Queensland economy, as it facilitates the operation of various industries which depend upon access to the railway network, including:

- the freight haulage industry that transports goods by rail across Queensland
- the mining industries that depend upon the Mount Isa Line and the West Moreton system for the transportation of minerals and coal products for export
- agricultural and livestock industries across regional Queensland
- tourism across Queensland, particularly in regional communities that are connected to the rail network
- regional communities, which rely upon industries including agriculture, mining and tourism for local employment and economic growth
- the Brisbane Metropolitan system commuter service.

Queensland Rail's financial statements show that total below-rail access revenue received by Queensland Rail across its entire network totalled \$316.3 million in 2017–18.⁶¹⁶ Stakeholders provided estimates of the monetary contributions of various individual systems to the Queensland economy.

Queensland Rail emphasised that 'it is critical that the economic significance is not conflated with cultural, historical or societal significance when applying criterion (c)^{1.617} The QCA considers that importance to the Queensland economy does not merely refer to monetary contributions to the GSP, but may also include contributions to employment, regional development and economic growth and productivity.

⁶¹³ Queensland Rail does not publicly report data for gtk carried on the South Western, Western, Central Western and Tablelands systems. Queensland Rail, *2017–18 Annual Performance Report, Queensland Rail Access Undertaking 1*, December 2018, pp. 7, 11, 12.

⁶¹⁴ Queensland Rail, sub. 8, p. 2, para. 8.

⁶¹⁵ Queensland Rail, *Annual and Financial Report 2017–18*, pp. 19, 42, https://www.queenslandrail.com.au/about%20us/Documents/Queensland%20Rail%20-%20Annual%20and%20Financial%20-%20Report%20-%202017-18.pdf.

⁶¹⁶ Queensland Rail, *Financial Statements for the Year Ended 30 June 2018, Below Rail Services Provided by Queensland Rail*, p. 4.

⁶¹⁷ Queensland Rail, sub. 8, p. 6, para. 41.

The QCA also notes that the amount of government subsidies that Queensland Rail's facility (or facilities) attracts is a positive factor in determining its significance. Indeed, if the Queensland Government chooses to provide subsidies to ensure that a facility continues to provide a service to the community, it would be consistent with the view that facility is of state significance.

The QCA considers that access to the Queensland Rail network facilitates the development of various industries in Queensland, which contribute significantly to the Queensland economy through GSP as well as regional development and employment. Thus, the QCA is satisfied that the Queensland Rail network as a whole is significant having regard to its importance to the Queensland economy.

12.7 Criterion (c)—the facilities for each part of the service

The QCA considers that each of the facilities that comprise each part of the service identified in the criterion (a) analysis are significant, having regard to their size or importance to the Queensland economy.

12.7.1 North Coast Route

The North Coast Route comprises the North Coast Line and the Metropolitan system (Appendix B). Detailed information on the North Coast Line and the Metropolitan system is provided above in sections 12.5.1 and 12.5.4.

Queensland Rail considered that both the North Coast Line and the Metropolitan system satisfy criterion (c).⁶¹⁸ Linfox stated that all criteria are met in relation to the North Coast Rail System (as it defined it).⁶¹⁹

Based on the information before it, the QCA considers that the North Coast Route is significant, having regard to its size or its importance to the Queensland economy. The North Coast Route is a substantial rail system, extending across a significant area of the state. Substantial annual volumes of freight are carried on the North Coast Route, as well as a number of regional passenger services. The QCA therefore considers that the North Coast Route is significant, having regard to its size.

The North Coast Route is a crucial freight corridor on Queensland's eastern coastline. Both the North Coast Line and the Metropolitan system are critical pieces of rail infrastructure, not only connecting key coastal Queensland cities (including Brisbane), but also connecting freight traffic from other rail systems to various export ports on Queensland's eastern coastline (including freight from the Mount Isa Line, Central Western, Western, South Western and West Moreton systems). The North Coast Line and Metropolitan system play a critical role in supporting economic activity across a large area of the state. The QCA therefore considers that the North Coast Route is significant, having regard to its importance to the Queensland economy.

12.7.2 Mount Isa Route

The Mount Isa Route comprises the Mount Isa Line and a small part of the North Coast Line between Stuart and the Port of Townsville (Appendix B). Detailed information on the Mount Isa Line is provided above in section 12.5.2.

Queensland Rail considered the Mount Isa Line satisfies criterion (c).⁶²⁰

⁶¹⁸ Queensland Rail, sub. 33, pp. 63, 65, paras 314, 322.

⁶¹⁹ Linfox, sub. 50, para. 2.2.

⁶²⁰ Queensland Rail, sub. 33, p. 66, para. 328.

Based on the information before it, the QCA considers that the Mount Isa Route is significant, having regard to its size or its importance to the Queensland economy. The Mount Isa Route is of an extensive length and geographic spread (extending from Queensland's eastern coastline to almost its western border). Substantial annual volumes of freight are carried on the Mount Isa Route. The QCA therefore considers that the Mount Isa Route is significant, having regard to its size.

The Mount Isa Route makes significant contributions to the Queensland economy through its operation as a fully commercial line, contributing significant amounts in access charges. It also plays a substantial indirect role in supporting the development of the North West Minerals Province, a highly prospective mining region, which relies upon the line for both importing mining inputs and exporting mining products. The Mount Isa Route enables key regional industries that contribute substantially to the Queensland economy including through employment, local spending and royalties paid to the state.⁶²¹ In announcing the Mount Isa Line incentive scheme, the Queensland Government acknowledged 'how important the North West Minerals Province is to the resources sector and how important that sector is for regional employment, exports and economic growth'.⁶²² The QCA therefore considers that the Mount Isa Route is significant, having regard to its importance to the Queensland economy.

12.7.3 West Moreton Route

The West Moreton Route comprises the West Moreton system and the Metropolitan system (Appendix B). Detailed information on the West Moreton system and the Metropolitan system is provided above in sections 12.5.3 and 12.5.4.

Queensland Rail considered that 'the West Moreton System is not relevantly of state significance due to the low volume hauled compared to Queensland's total coal exports'⁶²³, noting that:

Queensland Rail considers that while these tonnages [of coal transported on the West Moreton system] are significant to Queensland Rail and the South West Producers operating in the West Moreton coal freight corridor, these coal volumes are dwarfed by the scale of the operations in the CQCN [Aurizon Network's Central Queensland Coal Network].⁶²⁴

The QCA considers that criterion (c) does not require it to directly compare the facility for the service in question with other infrastructure in Queensland for the purposes of determining whether the facility is significant. Criterion (c) is satisfied if the facility for the service is significant, having regard to its size or its importance to the Queensland economy.

Based on the information before it, the QCA considers that the West Moreton Route is significant, having regard to its size or importance to the Queensland economy. The West Moreton system is primarily used to transport coal and agricultural products to the Port of Brisbane for export, via the Metropolitan system. The West Moreton system and the Metropolitan systems are of substantial length, and cover a wide geographic scope extending across a large area of southern Queensland and the greater Brisbane region. The QCA considers

⁶²¹ Queensland Resources Council, *What are resources worth to the North West region?*, 2018–19 financial year, https://www.qrc.org.au/wp-content/uploads/2019/10/2019_NorthWest_Region.pdf.

⁶²² A Palaszczuk, J Trad and M Bailey, More minerals freight to go on the fast track thanks to rail subsidy, media release, Queensland Government, 18 November 2019, http://statements.qld.gov.au/Statement/2019/11/18/more-minerals-freight-to-go-on-the-fast-track-thanks-to-rail-

subsidy.

⁶²³ Queensland Rail, sub. 33, p. 63, para. 310.

⁶²⁴ Queensland Rail, sub. 33, p. 67, para. 339.

that the volume of freight carried annually by the West Moreton Route is substantial. For example, the West Moreton system carried approximately 6.6 million net tonnes of freight, including coal and agricultural products, in 2017–18, and approximately 6.4 million net tonnes of freight in 2018–19.⁶²⁵ The QCA therefore considers that the West Moreton Route is significant, having regard to its size.

Queensland Rail commented on the uncertainty in relation to future tonnages on the West Moreton system.⁶²⁶ At the time of writing, this uncertainty persists and it is not clear which of the two scenarios presented by Queensland Rail (or indeed a different scenario) may eventuate in the future.⁶²⁷ Given this uncertainty, the QCA has formed its views in this recommendation based on available data (including volume and revenue data from the 2017–18 and 2018–19 reporting period).

The QCA is satisfied that the West Moreton Route is significant, having regard to its importance to the Queensland economy. This is based on the substantial direct contributions to the economy in the form of access revenue, as well as the substantial indirect contributions to the economy that access to the West Moreton Route provides. In particular, access to this route facilitates the operation of specific industries (such as coal mining, rail haulage and agriculture), that contribute substantially to the Queensland economy, both in terms of direct revenue (GSP) as well as employment and regional development. It is also relevant that it has been identified by government as a national key freight route and hence an enabler of critical freight flows.

12.7.4 The South Western Route, Western Route and Central Western Route

The South Western Route, the Western Route and the Central Western Route are defined in Appendix B. Detailed information on the individual rail systems is provided above in sections 12.5.5 to 12.5.9.

Queensland Rail submitted that the Western, South Western and Central Western systems were not significant in terms of each system's size or importance to the Queensland economy.⁶²⁸ In contrast, GrainCorp submitted that each of Queensland Rail's systems is significant in its own right, having regard to both size and importance to the Queensland economy. It commented that the South Western, Western and Central Western systems support the transfer of grain and the viability of the state's grain export industry.⁶²⁹ Watco disagreed with the QCA's draft recommendation that these systems are not significant in terms of their importance to the Queensland economy. While it recognised that they do not transport large volumes, it argued that 'these rail systems do provide key transport links accessing regional Queensland agricultural markets and communities'.⁶³⁰

Based on the information before it, the QCA considers that the South Western Route, the Western Route and the Central Western Route are all significant having regard to size. This is because each of these routes comprises an extensive length of railway track and covers a substantial geographic area of the state.

⁶²⁵ Queensland Rail, 2017-18 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2018, pp. 7, 11; Queensland Rail, 2018-19 Annual Performance Report, Queensland Rail Access Undertaking 1, December 2019, p. 7.

⁶²⁶ See for example, Queensland Rail's submission in the 2020 draft access undertaking process: Queensland Rail, DAU2 West Moreton System low volume coal reference tariff, submission, November 2019, pp. 5, 29.

⁶²⁷ See Queensland Rail, sub. 33, p. 67, para. 338.

⁶²⁸ Queensland Rail, sub. 33, pp. 70–73, paras 359, 370, 379.

⁶²⁹ GrainCorp, sub. 52, p. 11.

⁶³⁰ Watco, sub. 48, p. 5.

The QCA considers that it is less clear whether each of the routes are significant having regard to their importance to the Queensland economy. Importance to the Queensland economy does not merely refer to monetary contributions to the gross state product, but may also include contributions to employment, contributions to regional development and contributions to economic growth and productivity. For example:

- The Central Western Route provides an option for livestock transport for Central Western Queensland, an option for freight and supplies to be transported into regional communities, and a source of tourism through Queensland Rail's Spirit of the Outback (Brisbane to Longreach) passenger service.⁶³¹
- The Western Route provides an option for livestock and grain transport for the southwestern Queensland regions, an option for freight and supplies to be transported into regional communities, and a source of tourism through Queensland Rail's The Westlander (Brisbane to Charleville) passenger service.
- The South Western Route provides an option for the transport of agricultural products such as grain from the productive agricultural regions of the Darling Downs.

Whether the Central Western Route, the Western Route and the South Western Route are each significant, having regard to each one's importance to the Queensland economy, is a matter of judgement. Based on the information before it, the QCA considers that it is less clear that each one of these facilities is significant, if regard is had only to each facility's importance to the Queensland economy. However, the QCA notes that the fact that these facilities attract government subsidies suggests that they are of economic importance in terms of the flow-on benefits they provide to regional communities. Indeed, the Queensland Government has, in the past, stated that the government transport service contracts were designed to 'facilitate regional development and employment'.⁶³²

On balance, the QCA considers that the South Western Route, Western Route and Central Western Route are each significant having regard to their size, though the importance of each individual facility to the Queensland economy is less clear.

On this basis, the QCA considers that each route satisfies criterion (c). Section 76(2)(c) does not require that a facility be significant having regard to both size and economic importance. In each case the 'size' of the facility, in the sense used in the QCA Act, is sufficient to satisfy this criterion.

12.7.5 Tablelands system

Queensland Rail considered that the Tablelands system 'is neither of sufficient size nor importance to the Queensland economy (measured by contribution to exports and GSP) to satisfy criterion (c)'.⁶³³ Detailed information on the Tablelands system is provided above in section 12.5.8.

Based on the information before it, the QCA considers that the Tablelands system is significant in terms of size, though it is less clear whether it is significant in terms of importance to the Queensland economy.

⁶³¹ Queensland Rail, Annual and Financial Report 2016–17, p. 29.

⁶³² Queensland Government, Department of Transport and Main Roads, *Moving Freight – A strategy for more efficient freight movement*, December 2013, p. 36.

⁶³³ Queensland Rail, sub. 33, p. 74, para. 386.
The Tablelands system consists of an extensive length of railway track, which covers a substantial geographic area of Queensland's northern region, and substantial passenger numbers are carried on the Tablelands system annually. Thus, the QCA considers that the Tablelands system is significant, having regard to its size.

It is less clear whether the Tablelands system is significant, having regard only to its importance to the Queensland economy. The Tablelands system is an important facilitator for regional tourism in North Queensland. For example, the Normanton to Croydon railway line commemorated 125 years of service in 2016, and the Kuranda Scenic Railway service from Cairns to Kuranda is a significant tourism attraction in its own right for visitors to the Cairns region.⁶³⁴ Queensland Rail stated that this service alone directly contributes approximately \$3 million in tourist services revenue per year.⁶³⁵ Very little data is publicly available on the benefits that the Tablelands system impart to the tourism industry, although the QCA notes that the provision of government subsidies may suggest the facility provides flow-on benefits to the regional community. Based on the information before it, the QCA is inclined to conclude that the Tablelands system is not significant, if regard is had only to its importance to the Queensland economy.

The QCA considers that the Tablelands system is significant having regard to its size, though the system's importance to the Queensland economy as a whole is less clear.

On this basis, the QCA considers that the system satisfies criterion (c). Section 76(2)(c) does not require that a facility be significant having regard to both size and economic importance. In this case the 'size' of the facility, in the sense used in the QCA Act, is sufficient to satisfy this criterion.

⁶³⁴ Queensland Rail, Annual and Financial Report 2016–17, p. 29.

⁶³⁵ Queensland Rail, sub. 33, p. 73, para. 384.

13 CRITERION (D)—PROMOTE THE PUBLIC INTEREST

13.1 Introduction

Section 76(2)(d) of the QCA Act is expressed as follows:

that access (or increased access) to the service, on reasonable terms and conditions, as a result of a declaration of the service would promote the public interest

Section 76(5) of the QCA Act further states:

In considering the access criterion mentioned in subsection (2)(d), the authority and the Minister must have regard to the following matters –

- (a) if the facility for the service extends outside Queensland⁶³⁶
 - (i) whether access to the service provided outside Queensland by means of the facility is regulated by another jurisdiction; and
 - (ii) the desirability of consistency in regulating access to the service;
- (b) the effect that declaring the service would have on investment in
 - (i) facilities; and
 - (ii) markets that depend on access to the service;
- (c) the administrative and compliance costs that would be incurred by the provider of the service if the service were declared;
- (d) any other matter the authority or Minister considers relevant.

A summary of the key matters raised by stakeholders with respect to criterion (d), as well as the QCA's final recommendations, are set out in Table 11.

Table 11 Summary of key positions—s. 76(2)(d) of the QCA Act

Criterion (d)				
Issue	Queensland Rail	Other stakeholders	QCA final recommendation	
Queensland Rail's service as a whole (i.e. as defined in s. 250(1)(b) of the QCA	None of the rail systems satisfies criterion (d)	Pacific National, Glencore, and the South West Producers (New Hope and Yancoal) each submitted that the Queensland Rail network as a whole satisfied criterion (d)	Not satisfied	
Act)		Pacific National submitted that all the access criteria are satisfied for all Queensland Rail systems, including the South Western, Western, Central Western and Tablelands systems		
		GrainCorp submitted declaration should be maintained across the entire Queensland Rail network		
North Coast Route service	Criterion (d) is not satisfied	Linfox—criterion (d) is satisfied for the 'North Coast Rail System'	Criterion (d) is satisfied	

⁶³⁶ As the Queensland Rail facility does not extend outside Queensland, the QCA has not considered s. 76(5)(a) any further.

Criterion (d)				
Mount Isa Route service	Criterion (d) is not satisfied	Glencore—the 'Mount Isa Rail Access Service' (which it defines as the rail access service for the Mount Isa Line including the rail links to the Port of Townsville via the Jetty branch line) satisfied criterion (d)	Criterion (d) is satisfied	
West Moreton Route service	Criterion (d) is not satisfied	The South West Producers (New Hope and Yancoal)—the 'West Moreton corridor coal rail access service' (as they defined it) satisfied criterion (d)	Criterion (d) is satisfied	
Central Western Route service Western Route service South Western Route service	Criterion (d) is not satisfied	Watco and GrainCorp—the South Western, Western and Central Western systems satisfied criterion (d) Linfox—criterion (d) is satisfied for the 'Central West Rail System'	Criterion (d) is satisfied	
Tablelands system service	Criterion (d) is not satisfied	No stakeholders made submissions specifically in relation to the Tablelands system, although some stakeholders (summarised above) considered that the access criteria are satisfied for all Queensland Rail systems, including the Tablelands system	Criterion (d) is not satisfied	

13.2 The QCA's approach to assessing criterion (d)

13.2.1 Stakeholder submissions

Queensland Rail agreed with the QCA's approach in the draft recommendation that there were compelling reasons to assess criterion (d) on a system-by-system basis, given the different characteristics, investment effects and dependent markets.⁶³⁷ The South West Producers also agreed with this.⁶³⁸ Pacific National, on the other hand, considered that the interconnected nature of the Queensland Rail network means that benefits from the North Coast Line, Mt Isa Line, West Moreton system and Metropolitan system services flow to other services.⁶³⁹

13.2.2 QCA approach

Criterion (d) requires satisfaction of a positive test—that access (or increased access) to the service, on reasonable terms and conditions, as a result of a declaration would promote the public interest. This chapter addresses each of the matters the QCA is required to have regard to under section 76(5) of the QCA Act, as well as other matters the QCA considers relevant⁶⁴⁰ to the assessment of criterion (d). The QCA will consider whether criterion (d) is satisfied for the service as a whole⁶⁴¹, as well as for each part of the service identified by the QCA (see Appendix B).

In the draft recommendation, the QCA considered the South Western, Western, Central Western and Tablelands systems together as the 'other systems' services, and noted the lack of

⁶³⁷ Queensland Rail, sub. 33, p. 75, para. 389.

⁶³⁸ South West Producers, sub. 31, p. 19.

⁶³⁹ Pacific National, sub. 28, p. 2.

⁶⁴⁰ Section 76(5)(d) of the QCA Act.

⁶⁴¹ The service as a whole is described in s. 250(1)(b) of the QCA Act.

stakeholder submissions addressing public interest factors in relation to these services. In response to the QCA's draft recommendation, several stakeholders provided new information in relation to the South Western, Western, Central Western and Tablelands systems. For example:

- Queensland Rail confirmed that the Tablelands system now exclusively provides tourist train services.⁶⁴²
- Watco, Linfox and GrainCorp confirmed that the South Western, Western and Central Western systems primarily service freight from the agricultural sector.⁶⁴³

In addition, in its submission in response to the QCA's draft recommendation, Queensland Rail has substantially increased both the volume and detail of its arguments. In considering this substantial volume of new information, the QCA has reviewed its analysis of criterion (d) in relation to the services provided on the South Western, Western, Central Western and Tablelands systems.

13.3 The structure of criterion (d) analysis

The analysis of criterion (d) is set out under the following headings:

- Investment in facilities
- Investment in dependent markets
- Administrative and compliance costs
- Other relevant matters.⁶⁴⁴

These headings reflect the matters expressed in s. 76(5) of the QCA Act. The QCA will first discuss general principles applicable to each matter, and will then apply these principles to the service as a whole, and each particular part of the service identified by the QCA (see Appendix B).

13.4 Investment in facilities

In considering criterion (d), the QCA is required to have regard to:

the effect that declaring the service would have on investment in facilities $^{\rm 645}$

Investment in facilities includes, but is not limited to, investment in the facility for the service (i.e. the rail transport infrastructure).⁶⁴⁶ Importantly, this investment can be made by users of the service, and does not necessarily need to be made by Queensland Rail.⁶⁴⁷ This section focuses on investment in below-rail facilities. The QCA discusses investment in other facilities, for example in mines or above-rail assets, as part of investment in markets that depend on access to the service, in section 13.5 below.

⁶⁴² Queensland Rail, sub. 33, p. 73, para. 384.

⁶⁴³ Watco, sub. 48, p. 6; Linfox, sub. 50, paras 3.8–3.12; GrainCorp, sub. 52, pp. 12–14.

⁶⁴⁴ These headings reflect the matters expressed in s. 76(5) of the QCA Act.

⁶⁴⁵ Section 76(5)(b)(i) of the QCA Act.

⁶⁴⁶ Within the meaning of s. 250 and schedule 2 of the QCA Act.

⁶⁴⁷ For example, Queensland Rail's Mount Isa Line Rail Infrastructure Master Plan envisages that capacity improvements on the Mount Isa Line may be paid for or recovered from users through increased access charges— Queensland Rail, Mount Isa Line Rail Infrastructure Master Plan, 2012, p. 10,

https://www.queenslandrail.com.au/business/acccess/Documents/Maps/QR4159.1%20Infrastructure%20Master %20Plan%202012_Updated_LR.pdf.

For the reasons discussed below, the QCA considers that declaration would likely have a positive, albeit small, effect on investment in facilities used in relation to the service as a whole, and the services on the North Coast Route, Mount Isa Route, West Moreton Route and the agricultural systems. The QCA considers that declaration is unlikely to have an effect on investment in facilities used in relation to the service on the Tablelands system.

13.4.1 Stakeholder submissions

Stakeholder submissions on the impact of declaration on investment in the facilities for the services (i.e. the rail transport infrastructure) were limited. Stakeholders who commented on the impact on investment in other facilities (such as above-rail facilities) considered this in the context of investment in dependent markets. The impact on investment on other facilities (such as above-rail facilities) is discussed in section 13.5.

Queensland Rail argued that declaration does not promote investment in its own facilities (i.e. the below-rail infrastructure), although it did not specifically address these impacts.⁶⁴⁸ Queensland Rail emphasised how it is incentivised to increase the utilisation of its systems in the absence of declaration, focusing on the impact on investment in dependent markets (including above-rail facilities).

The South West Producers submitted that 'declaration of the West Moreton coal rail access service facilitates ongoing investment in coal mining in the West Moreton region', including by promoting and facilitating investments in the below-rail facilities, as well as in coal mines in the West Moreton region.⁶⁴⁹ They further commented that investments made by Queensland Rail (and the state) in network upgrades (such as bridge replacements and re-sleepering) on the West Moreton system are seen as evidence that declaration is facilitating continued network investment.⁶⁵⁰

13.4.2 QCA analysis—general principles

Investment in below-rail facilities⁶⁵¹ may be made by Queensland Rail itself or may come from other investors such as users of the railway line or third party investors. It may also be undertaken by Queensland Rail in partnership with a third party.

Apart from ongoing investment required to maintain the network in a condition that will allow for the continued delivery of below-rail services (i.e. replacement expenditure), two factors that could trigger the need for investment in below-rail facilities are:

- an increase in demand for below-rail services where there is insufficient capacity in the system
- unforeseen circumstances (for example, the occurrence of natural disasters, the discovery of new minerals deposits, the announcement of new government policies or infrastructure projects), that necessitate investment in the below-rail facilities.

In either case, the key issue impacting investment incentives is the extent to which Queensland Rail (and/or third party investors) is confident that it will be able to recover the costs of that investment, including a return on and of capital. Under the existing regulatory regime in the

⁶⁴⁸ Queensland Rail, sub. 33, pp. 80–82, paras 407–14.

⁶⁴⁹ South West Producers, sub. 4, p. 50.

⁶⁵⁰ South West Producers, sub. 31, p. 20.

⁶⁵¹ In this section, the QCA uses the words 'below-rail facilities' to refer to the facilities for the service; that is, the rail transport infrastructure as defined in s. 250 and schedule 2 of the QCA Act.

QCA Act, Queensland Rail can seek to recover the costs of any investments it makes via access charges from users. However, the QCA must approve these investments as prudent and efficient before the costs can be recovered via access charges. In this context, the QCA notes the comments made by the South West Producers (in the context of the West Moreton Route) regarding the protections that the current regulatory regime provides to Queensland Rail to recover costs from users in the event of a fall in volumes.⁶⁵²

Almost all of Queensland Rail's rail systems (except the Mount Isa Line) rely heavily on Queensland Government subsidies under the Transport Services Contracts (TSCs) to operate (for more information, see section 12.5). Additionally, all of Queensland Rail's rail systems have substantial excess capacity. The QCA considers that under these circumstances, significant new investment is unlikely to be required to expand the network in the near future, although replacement expenditure will still be required to maintain the current level of service. On these subsidised systems, the QCA recognises that the business case for investment in below-rail facilities will depend on the extent to which the costs of that investment will be able to be recovered from a combination of access charges and government funding (and that in some cases there is a much heavier reliance on the latter).

However, not all investments in facilities are necessarily in response to a need to increase capacity on the existing system(s) (and hence tied to the amount of excess capacity in the system). Unforeseen circumstances may arise during the future period for declaration that may require investment in facilities (including investment in rail transport infrastructure)—for example, natural disasters, the discovery of new mining deposits, an unexpected increase in demand for a particular mineral, or the announcement of new government policies or infrastructure projects. Given this, investments may be made into improving the reliability of the rail system or its resilience to natural disasters.

In comparing a future with and a future without declaration, the QCA considers that Queensland Rail's incentives to invest in its below-rail network (whether in new infrastructure or in maintenance and replacement expenditure) will largely depend on its confidence in its ability to recover those costs. In a future with declaration, Queensland Rail may seek to recover the costs of this investment from its users via access charges, provided the investment has been approved by the QCA. In a future without declaration, Queensland Rail can similarly seek to recover the costs of its investments through an exercise of its market power over its users, and in this case, the investments do not have to be approved as prudent and efficient by the QCA. Therefore, the QCA considers that declaration is unlikely to significantly affect Queensland Rail's incentives to invest in the below-rail facility, as long as it can recover the costs from its users (and/or through the government subsidy). The QCA recognises the availability of Queensland Government subsidies could affect Queensland Rail's incentives for investment in certain systems; and these incentives are likely to be independent of whether or not the service is declared.

The QCA considers that in both a future with and without declaration, most of the investments in below-rail facilities would likely continue to be undertaken by Queensland Rail, as it is the railway manager. However, situations may arise where a user contributes directly to an investment in below-rail facilities through an upfront payment. For example, a user may pay for a proportion of the costs of a new passing loop in conjunction with Queensland Rail, or may pay the full cost of an investment into a private rail siding (e.g. to connect to grain silos) or a private rail yard. In these cases, the QCA considers that declaration will have a positive effect on

⁶⁵² South West Producers, sub. 40, p. 13.

promoting investments by users. This is because any investments in below-rail facilities are likely to be highly site-specific and sunk⁶⁵³; the presence of sunk investments leads to the risk of hold-up (for example, see section 5.6.2). Specifically, in a future without declaration, there is a risk that once the user has made the investment, Queensland Rail may subsequently exercise its market power to expropriate the value of that investment from the user. A user can foresee this risk in a future without declaration, and may be less inclined to contribute to an investment in the below-rail facilities, even if such an investment would otherwise be efficient.

In contrast, in a future with declaration, the QCA considers that the regulatory regime would impose an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power (for example, to hold-up a user). Therefore, the QCA considers that declaration would be likely to promote users of the Queensland Rail service to make investments in the below-rail facilities, given that users can have confidence that they will reap the benefits of their current investment in the future.

Having regard to its general assessment of the likely future with and without declaration, the QCA has also considered specific issues applicable to each service, as outlined below.

13.4.3 The service as a whole

Queensland Rail submitted that it has excess capacity across its network⁶⁵⁴, and it is not evident that additional below-rail investment would be required to accommodate new demand in the near future on the network.

As discussed in section 13.4.2 above, the QCA considers that declaration is unlikely to significantly affect Queensland Rail's incentives to invest in the below-rail network, however, declaration would likely have the effect of promoting investment by users in the network. On balance, the QCA considers that declaration of the service as a whole is likely to have a small positive effect on investment in the Queensland Rail below-rail network.

13.4.4 North Coast Route service

The QCA has concluded that the stable market environment created by declaration is likely to promote a material increase in competition in the above-rail haulage market on the North Coast Route. This could promote higher utilisation compared to the situation that may exist in the absence of declaration.

The North Coast Line and the Metropolitan Line currently have significant spare capacity.⁶⁵⁵ Further, Queensland Rail is more dependent on TSC funding for the North Coast Line (compared to the Mount Isa Line or West Moreton system); this was \$152.3 million in 2017–18, comprising approximately 75 per cent of its total revenue.⁶⁵⁶ While increased utilisation would increase the revenue that it can earn from access charges, Queensland Rail is still likely to remain heavily reliant on TSC funding for this line.

The QCA notes that Queensland Rail has undertaken recent investments on the North Coast Line, including restoring the Yeppoon branch line and repairing sections of the North Coast Line

⁶⁵³ For example, a passing loop on the West Moreton system has very little alternative use elsewhere.

⁶⁵⁴ Queensland Rail, sub. 33, p. 22, para. 118.

⁶⁵⁵ Queensland Rail, sub. 33, p. 77, para. 403.

⁶⁵⁶ Queensland Rail, Financial Statements for the Year Ended 30 June 2018, Below Rail Services Provided by Queensland Rail, p. 4,

https://www.queenslandrail.com.au/business/acccess/Compliance%20and%20reporting/2017-18%20Below%20Rail%20Financial%20Statements.pdf.

due to flooding.⁶⁵⁷ To the extent that users may be incentivised or required to invest in improvements to the North Coast Route infrastructure over the future period for declaration, the QCA considers that its analysis in section 13.4.2 applies in relation to the incentives facing a user to invest in the below-rail infrastructure on the North Coast Route.

On balance, the QCA considers that declaration is likely to have a positive, albeit small, effect on investment in below-rail facilities used in relation to the North Coast Route service.

13.4.5 Mount Isa Route service

The QCA understands that the Mount Isa Line is the only rail system in the Queensland Rail network that operates on a purely commercial basis; that is, it does not depend upon TSC payments to fund its operation, maintenance or expansions (if needed).⁶⁵⁸ Queensland Rail's Mount Isa Line Rail Infrastructure Master Plan notes that any future investments to increase capacity on the Mount Isa Line will be fully funded through access charges, on a user-pays basis. However, Queensland Rail also noted that this system currently has significant excess capacity 'with no portion of the system operating at over 60 per cent utilisation'.⁶⁵⁹

The QCA has concluded that the stable market environment created by declaration is likely to promote a material increase in competition in the North West minerals tenements market, which is dependent on the Mount Isa Route service. This could promote higher utilisation of the Mount Isa Route compared to the situation that may exist in the absence of declaration. However, given the excess capacity that is available, it is not evident that additional below-rail investment would be required to accommodate new demand in the near future.

The recent impacts to the Mount Isa Line as a result of flooding events has attracted state and federal government funding to support maintenance and upgrades on the line to ensure it will be better equipped to respond to severe and unexpected weather events.⁶⁶⁰ To the extent that users may be incentivised or required to invest in improvements to the Mount Isa Route infrastructure over the future period for declaration, the QCA considers that its analysis in section 13.4.2 above applies in relation to the incentives facing a user to invest in the below-rail infrastructure on the Mount Isa Route.

On balance, the QCA considers that declaration is likely to have a positive, albeit small, effect on investment in below-rail facilities used in relation to the Mount Isa Route service.

13.4.6 West Moreton Route service

Queensland Rail did not express any specific views in relation to the effect of declaration on investment in the West Moreton system or the Metropolitan system. As noted above, the South West Producers considered that declaration would promote, and has promoted, investment in below-rail facilities on these systems. Queensland Rail has indicated that the West Moreton

⁶⁵⁷ Queensland Rail, *Annual and Financial Report 2018–19*, pp. 30–33, https://www.queenslandrail.com.au/about%20us/Documents/Queensland%20Rail%20Annual%20and%20Financia l%20Report%202018-19.pdf.

⁶⁵⁸ Queensland Rail, sub. 8, p. 1, para. 4. See also the Department of Transport and Main Roads, Rail Transport Contracts and Agreements, Queensland Government, 2018, https://www.tmr.qld.gov.au/businessindustry/Transport-sectors/Rail-services-and-infrastructure/Rail-Transport-Contracts-and-Agreements.

⁶⁵⁹ Queensland Rail, sub. 33, p. 80, para. 409.1.

⁶⁶⁰ Bailey, M, *Mount Isa Line flood repairs put freight on the fast track*, media release, Queensland Government, 26 April 2019, http://statements.qld.gov.au/Statement/2019/4/26/mount-isa-line-flood-repairs-put-freight-on-the-fast-track.

system has 42 per cent spare capacity based on contracted paths⁶⁶¹, although it has also suggested that capacity constraints on the Toowoomba Range could trigger the need for an expansion, particularly if New Acland Stage 3 is developed.⁶⁶² While the case for expansion does not currently apply, it could be the case in future. In addition, the QCA notes that in 2017–18, only a very small proportion of West Moreton system revenue (less than 1 per cent) came from TSC funding.⁶⁶³

The West Moreton system is dependent on coal access charges for funding, which accounted for approximately 91 per cent of revenue for this system in 2017–18.⁶⁶⁴ A reference tariff is published in the 2016 Queensland Rail access undertaking for access to the coal service provided on the West Moreton and Metropolitan systems. This differs from the regulatory regime for the remainder of Queensland Rail's service, as no reference tariff applies to the other services.⁶⁶⁵ The QCA considers that this publicly available reference tariff provides additional pricing certainty for coal access seekers and access holders on the West Moreton and Metropolitan systems.

The South West Producers submitted that the access regime has in the past mitigated the effects of uncertainty around volumes on the West Moreton coal rail access service:

The application of a reference tariff under the existing declaration (and the resetting of tariffs at the time of QR's Access Undertaking 1) effectively immunised QR from the volume risk associated with the Wilkie Creek mine shutting, and the QCA's presence provides certainty to both QR and access holders / seekers that even with uncertainty around volumes it will be possible to resolve a tariff that mitigates the consequences of such uncertainty of volume.⁶⁶⁶

The QCA has concluded that the stable market environment created by declaration is likely to promote a material increase in competition in the coal tenements market in the West Moreton region, which is dependent on access to the West Moreton Route service. To the extent that this increases utilisation and the demand for below-rail services, it could also trigger investment in the network, although at least in the medium term, this may mainly be required to alleviate capacity constraints on the Toowoomba Range.

The QCA considers that on balance, declaration is likely to have a positive, albeit small, effect on investment in below-rail facilities used in relation to the West Moreton Route service.

13.4.7 South Western, Western and Central Western Route services

The QCA has concluded that the stable market environment created by declaration is likely to promote a material increase in competition in the above-rail freight haulage markets on the South Western, Western and Central Western Route services.

The QCA notes that the below-rail services provided on these agricultural systems are heavily dependent on TSC funding, and low volumes of freight are currently transported on these systems. However, as noted in section 13.2.2, the QCA has reviewed its analysis in the draft recommendation after considering the submissions received from stakeholders in response to

⁶⁶¹ Queensland Rail, sub. 33, p. 80, para. 409.1.

⁶⁶² This assumes that coal services are not limited by path constraints in the Metropolitan system.

⁶⁶³ Queensland Rail, Financial Statements for the Year Ended 30 June 2018, Below Rail Services Provided by Queensland Rail, p. 4.

⁶⁶⁴ Queensland Rail, *Financial Statements for the Year Ended 30 June 2018, Below Rail Services Provided by Queensland Rail*, p. 4.

⁶⁶⁵ Whether the QCA approves reference tariffs for other systems in a future with declaration is a matter for the applicable undertaking process and is outside the scope of the review. See s. 101 of the QCA Act.

⁶⁶⁶ South West Producers, sub. 31, p. 20.

the draft recommendation. On the basis of the material before it, the QCA considers that declaration is likely to have a small positive effect on investment in facilities used in relation to the agricultural systems. In particular, it is no longer the case, as the QCA considered in its draft recommendation, that the only services that operate on the agricultural systems are those provided under an above-rail subsidy. The presence of commercial (i.e. non-subsidised) above-rail operators on the agricultural systems enhances the possibility that a third party may make investments in facilities over the proposed declaration period, and the QCA considers that the effect of declaration would be to promote such investments being made, as users can have confidence that they will reap the benefits of their investments into the future.

Therefore, the QCA considers that on balance, declaration is likely to have a positive, albeit small, effect on investment in below-rail facilities used in relation to the South Western, Western and Central Western Route services.

13.4.8 Tablelands system service

The QCA has concluded that declaration is unlikely to promote a material increase in competition in any dependent markets of the Tablelands system service, including the above-rail passenger market.

The QCA notes that both below-rail and above-rail services on the Tablelands system are heavily dependent on TSC subsidies for their continued operation (see section 9.4.2). The QCA considers that the prominent and decisive factor informing a decision to invest in below-rail facilities in relation to the Tablelands system service, whether that decision is made by Queensland Rail or a third party, is the presence of government subsidies.

As such, the QCA considers that declaration would be unlikely to have a positive effect in promoting investment in below-rail facilities used in relation to the Tablelands system service.

13.5 Investment in markets that depend on access to the service

In considering criterion (d), the QCA is required to have regard to:

the effect that declaring the service would have on investment in markets that depend on access to the service $^{\rm 667}$

The QCA considers that declaration would likely have a positive effect on investment in dependent markets of the services on the North Coast Route, Mount Isa Route, West Moreton Route and the agricultural systems. The QCA considers that declaration is unlikely to have an effect on investment in dependent markets of the service on the Tablelands system.

In particular, the QCA has considered the relevant circumstances in each of the services, and considers that declaration will have a positive effect on investment in the following dependent markets:

- North Coast Route service—the above-rail haulage market
- Mount Isa Route—the North West Queensland minerals tenements market
- West Moreton Route service—the coal tenements market in the West Moreton region
- South Western, Western and Central Western Route services—the above-rail haulage markets dependent on each respective service.

⁶⁶⁷ Section 76(5)(b)(ii) of the QCA Act.

13.5.1 Stakeholder submissions

Queensland Rail submitted that it has a strong incentive to promote efficient investment and entry into related markets as it will increase the utilisation of its systems, noting that it has substantial spare capacity on its networks.⁶⁶⁸ It stated:

The ACCC has previously recognised that the private commercial incentives of rail infrastructure access holders align with the promotion of the public interest, as non-vertically integrated rail access holders have incentives to increase the efficient use and operation of the network, promote efficient investment in the network, and to increase rail volumes and asset utilisation.⁶⁶⁹

Queensland Rail considered that submissions made by Pacific National, Glencore and the South West Producers on the effect of declaration on incentives to invest in dependent markets of the North Coast Line, Mount Isa Line and West Moreton system are overstated, with investments likely to have still occurred in the absence of declaration.⁶⁷⁰

Queensland Rail argued that these submissions have failed to establish the required nexus between declaration and increased investment and ignored Queensland Rail's incentives to promote investment. Further, despite claims that declaration has promoted a 'stable regulatory environment' that is conducive to investment, Queensland Rail argued that its access framework will result in an environment and investment opportunities that are not materially different, when comparing the future with and without declaration.⁶⁷¹ It argued:

The immateriality of any difference between the future with and without is established by the basic fact that users, including Pacific National, Glencore and other users currently acquire access to the North Coast Line and Mount Isa Line pursuant to prices contained in negotiated agreements with Queensland Rail. There is no evidence to suggest that such commercial agreements will not be struck on substantively similar terms in the future without declaration under the Access Framework. Further, the beneficial terms of access already extracted by users reflect non-regulatory constraints which ultimately discipline Queensland Rail, including access prices well below the costs of providing the service, and long term agreements which often do not contain full take or pay requirements.⁶⁷²

A number of stakeholders refuted Queensland Rail's claims that it is incentivised to promote investment and entry into related markets. The South West Producers argued that even though it is not vertically integrated, Queensland Rail is still incentivised to maximise profits and does not face the constraints that it has claimed.⁶⁷³ It refuted Queensland Rail's claims that the access framework will still result in public benefits, as the framework is not seen as providing reasonable terms and conditions:

The flaws in the Deed Poll means that even if it is considered as part of the likely future without declaration (which the South West Producers contend it should not), it does not constrain QR's conduct or the adverse public interest impacts in the way QR asserts.⁶⁷⁴

The South West Producers were concerned that in the absence of declaration, Queensland Rail would significantly increase prices after producers have incurred sunk costs in coal

⁶⁶⁸ Queensland Rail, sub. 33, p. 80, para. 407.

⁶⁶⁹ Queensland Rail, sub. 33, p. 80, para. 408.

⁶⁷⁰ Queensland Rail, sub. 33, p. 80, para. 410.

⁶⁷¹ Queensland Rail, sub. 33, p. 81, para. 413.

⁶⁷² Queensland Rail, sub. 33, pp. 81–82, para. 414.

⁶⁷³ South West Producers, sub. 40, p. 46.

⁶⁷⁴ South West Producers, sub. 40, p. 47.

development. They considered that this 'economic holdup' would not exist where there are reference tariffs and the QCA can arbitrate access disputes.⁶⁷⁵

Glencore commented that while Queensland Rail may still provide access without declaration, such access may not be on reasonable terms and conditions, given that prices (and non-price terms) are effectively unconstrained 'and it is that potential for monopoly pricing and other monopolistic behaviour which impacts on the incentives to make investments in the minerals tenements market and a variety of other public interest factors'.⁶⁷⁶ It also noted that Queensland Rail remains incentivised to maximise profits, and in the absence of declaration it will not be constrained in exercising its market power and engaging in monopoly pricing. Glencore considered that the protections under the QCA Act, along with having the QCA as regulator, serve as substantial constraints on Queensland Rail's behaviour.

Watco expressed significant concerns with Queensland Rail's access framework and the uncertainty it creates, and observed the following, based on its experience in Western Australia:

Our experience with CBH in Western Australia where "what the market can bear" principles were used has led to significant closures of regional railway lines. These lines were considered by Watco as safe and profitable; but economic theory was applied, including 'hub and spoke' principles of roading to the nearest railhead with the principle of consolidation to create economic scale. Unfortunately, once the task has transferred from rail to road it tends to stay on that mode. This grain freight now arrives at the Kwinana Port via the metropolitan road network in Perth.⁶⁷⁷

Watco highlighted that the only avenue available to challenge amendments to Queensland Rail's access framework is legal proceedings 'which are cost and time prohibitive and generally ineffective (as experienced in the ongoing grain rail freight network access dispute in Western Australia between the track provider and access seeker)'.⁶⁷⁸ Watco considered that having the QCA being able to determine access disputes creates more certainty as to how such disputes will be assessed.

13.5.2 QCA analysis—general principles

It is evident from stakeholders' submissions that in making decisions to invest in dependent markets, long-term certainty of below-rail network access, and the terms and conditions of that access, are key factors. Queensland Rail submitted that it is incentivised to promote investment and entry into related markets and that it will continue to provide access on reasonable terms and conditions under its access framework. Stakeholders have generally disagreed with this, arguing that there will be no real constraints that would prevent Queensland Rail from misusing its market power to 'extract monopoly rents'.⁶⁷⁹ Queensland Rail's access framework is not seen by its users as providing any certainty, which could lead to time-consuming and costly disputes, including over any proposed amendments to the terms of the framework. The QCA's analysis of Queensland Rail's access framework is in Part B, Chapter 4.

In assessing the potential impact of declaration on investment in dependent markets, the QCA has considered the likely environment in a future with declaration compared to a future without declaration. There are inherent challenges in undertaking this analysis, recognising that a number of factors will influence behaviour and investment decision-making. Practically, this has

⁶⁷⁵ South West Producers, sub. 31, p. 20.

⁶⁷⁶ Glencore, sub. 41, p. 28.

⁶⁷⁷ Watco, sub. 48, p. 3.

⁶⁷⁸ Watco. sub. 48, p. 3.

⁶⁷⁹ See for example, South West Producers, sub. 40, p. 20.

also meant that the assessment of the extent to which declaration will impact investment in dependent markets (and hence promote the public interest) is linked to the analyses made in relation to criterion (a).

The QCA considers that in a future with declaration, the regulatory regime forms an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power, and will provide certainty to market participants in dependent markets that access to the service will be provided on reasonable terms and conditions. Market participants are thus more likely to make investments in dependent markets, given that they can have confidence that they will reap the benefits of their investment into the future. In contrast, the QCA considers that in a future without declaration, market participants are exposed to the risk of an exercise of market power by Queensland Rail, including the risk of hold-up (see for example, section 5.6.2). Stakeholders have submitted that the risk that the value of their investments could be expropriated ex post could undermine their investment decisions in dependent markets.

The QCA considers that in a future without declaration, Queensland Rail will not face any effective long-term constraints on its ability and incentive to exercise market power. Even if Queensland Rail's access framework is put in place in the absence of declaration (so as to govern access to services on the North Coast Line, Mount Isa Line, West Moreton system and Metropolitan system), the QCA has concluded that this will not be an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power (see Part B, Chapter 4). This could result in an environment that is characterised by protracted negotiations and costly disputes, which could in turn lead to reduced output and investment, as well as deter market entry.

Having regard to its general assessment of the likely future with and without declaration, the QCA has also considered specific issues applicable to each service, as outlined below.

13.5.3 The service as a whole

The service as a whole is defined in s. 250(1)(b) of the QCA Act. In the analysis of criterion (a), the QCA has not identified a market, which depends on access to the Queensland Rail service as a whole, that satisfied criterion (a). As such, the QCA could not be satisfied that access (or increased access) to the service as a whole, on reasonable terms and conditions, as a result of a declaration of the service as a whole, would promote a material increase in competition in one or more dependent markets (even though access as a result of a declaration of part of the service may do so).

In considering criterion (d), and having regard to the conclusion reached in criterion (a), the QCA has been unable to identify a market that depends on access to the service as a whole, in which the QCA considers that declaration would be likely to promote investment. As a result, the QCA considers that declaring the service as a whole would be unlikely to have an effect on investment in markets that depend on access to the service as a whole. However, the QCA notes that there are various markets that depend on access to various parts of the service. These effects are considered below.

13.5.4 North Coast Route service

Queensland Rail reiterated its incentives to promote utilisation of the 'heavily subsidised, underutilised' North Coast Line, submitting that it 'will set terms and conditions of access to promote efficient investment and promote entry, notwithstanding the exceedingly high barriers to entry into narrow-gauge intermodal rail markets which render entry unlikely with or without declaration.'680

Pacific National said that declaration has promoted investment in the above-rail haulage market:

PN's market entry and success in Queensland, across a number of varied freight tasks [i.e. for both haulage of non-bulk freight and bulk freight] relied on the stable operation of the economic regulatory framework ... PN considers that continued declaration ... under Part 5 of the QCA Act is critical to promoting competition and freight transport supply chain investment moving forward.⁶⁸¹

...

[The benefits of declaration] extend well beyond simply constraining the exercise of market power and ensuring that a balanced risk profile underpins the setting of terms and conditions of access. Declaration has facilitated the introduction of important structural and behavioural constraints ... and has underpinned the growth of competition in related upstream and downstream markets.⁶⁸²

Pacific National said that this growth of competition has in turn led to increased investment in the above-rail haulage market in terms of increased operational efficiency and increased innovation in above-rail rollingstock. It referred to the significant amounts of capital investment it had undertaken in both intermodal and bulk rollingstock, freight terminals and maintenance facilities, and said that this investment 'would not have been justified, absent a stable regulatory environment'.⁶⁸³

Queensland Rail refuted Pacific National's claims that its investment into the Queensland market was 'critically dependent' on the declaration of the North Coast Line.⁶⁸⁴ It argued that this was driven by the privatisation of government-owned above-rail operations and the sponsorship of Pacific National's entry into this market via Toll. It stated:

Following Pacific National's rapid increase in scale, Toll sponsored Pacific National's entry into the Queensland North Coast Freight corridor in March 2005, by shifting its customer volumes, said to account for 70% of Queensland Rail's above-rail freight volumes at the time, from Queensland Rail to Pacific National. At that time, Queensland Rail was vertically integrated, which is no longer the case today.⁶⁸⁵

The QCA considers that the fact that Pacific National made significant investments to enter the above-rail haulage market on the North Coast Route at a time when Queensland Rail was vertically integrated strongly suggests that declaration and the presence of the regulatory regime have been successful in promoting investment in the above-rail haulage market on the North Coast Route in the past.⁶⁸⁶ This is because, prima facie, a vertically integrated Queensland Rail had stronger incentives to exercise its market power in a way that favoured its related above-rail haulage provider (QR National at that time), to the detriment of an entrant competitor such as Pacific National, compared to a non-vertically integrated entity (such as the current Queensland Rail). As Pacific National submitted, its view was that 'declaration and

⁶⁸⁰ Queensland Rail, sub. 33, p. 82, para. 417.

⁶⁸¹ Pacific National, sub. 9, p. 3.

⁶⁸² Pacific National, sub. 9, p. 4.

⁶⁸³ Pacific National, sub. 9, p. 6.

⁶⁸⁴ Queensland Rail, sub. 33, p. 82, paras 415–16.

⁶⁸⁵ Queensland Rail, sub. 33, p. 82, para. 416.2.

⁶⁸⁶ The QCA, and the regulatory regime established under the QCA Act, was established in 1997.

effective regulation have facilitated entry and strong business growth for PN in Queensland'687 and that:

[t]he stable market environment created by declaration promotes efficient investment in the above-rail haulage market and promotes investment in below-rail infrastructure. PN's entry and expansion in Queensland was critically dependent on this stable regulatory environment.⁶⁸⁸

On the basis of the material before it, the QCA considers that in a future with declaration, the regulatory regime imposes an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power (for example, to hold up a user), and thus reduces the risk of undertaking an investment in the above-rail haulage market. The QCA considers that a future with declaration is likely to promote efficient investment in the above-rail haulage market on the North Coast Route, as above-rail operators can be more certain that they will reap the benefits of their investments (e.g. into rollingstock efficiency) into the future.

Therefore, on balance, the QCA considers that declaration is likely to have a positive effect on investment in the above-rail haulage market, a dependent market of the North Coast Route service.

13.5.5 Mount Isa Route service

Queensland Rail reiterated its incentives to promote investment in dependent markets given the significant spare capacity on the Mount Isa (and West Moreton) systems, stating that each system is only recovering sufficient revenue to cover operating (and not capital) costs.⁶⁸⁹ It also argued that there are more material factors affecting investment in dependent markets that are not related to declaration which, in the case of the Mount Isa Line, include:

- the risks associated with operating in global commodity markets
- the impact of Aurizon's exit from the intermodal market, affecting Pacific National's incentives to invest
- continued competition from road transport, necessitating continued investment in rail efficiency improvements.⁶⁹⁰

Queensland Rail also submitted that Glencore had overstated the relative significance of belowrail freight costs in assessing the risks and returns of mining investment, arguing that 'in respect of the Mount Isa Line, below-rail costs represent 0.3% of the copper price, 0.8% of the zinc price, 0.9% of the lead price and 5% of the fertiliser price.'⁶⁹¹

Glencore submitted that criterion (d) is clearly satisfied in respect of the Mount Isa Route service and that access as a result of declaration has facilitated investment in the North West Queensland mining industry, including dependent markets such as:

- the market for mining tenements
- the market for above-rail haulage services

⁶⁸⁷ Pacific National, sub. 37, p. 3.

⁶⁸⁸ Pacific National, sub. 37, p. 4.

⁶⁸⁹ Queensland Rail, sub. 33, p. 83, para. 419.

⁶⁹⁰ Queensland Rail, sub. 33, p. 83, para. 421.

⁶⁹¹ Queensland Rail, sub. 33, p. 83, para. 420.

 other indirect markets, such as the market for labour (employment) in the North West Queensland mining industry and the market for household goods and services in North West Queensland regional communities.⁶⁹²

Glencore gave specific examples of the beneficial effect that declaring the Mount Isa Route service would have on investment in dependent markets. For example, it said that access as a result of declaration has minimised barriers to entry especially for new and junior investors, thus facilitating investment in the mining tenements market.⁶⁹³ It viewed that declaration provides long-term assurance for access, and the certainty to support the 'significant sunk costs required in the exploration, establishment and development phases', thus supporting the continued investment in the North West Queensland mining industry.⁶⁹⁴

Glencore also considered that the QCA underestimated the impact of 'no declaration' on mining investment in the draft recommendation, when the QCA concluded that only additional investment in dependent markets would be promoted. It said the QCA 'does not fully appreciate the significant levels of risk inherent in logistics cost and how this would impact any investment in mining with or without declaration'⁶⁹⁵. In particular, Glencore noted:

As an entity that has actually considered and made investments in the North West Queensland minerals province, Glencore implores the QCA to appreciate that declaration is a material driver of investment in the mining industry. This is particularly the case when considering the:

- (a) nature of the products produced in the North West Queensland minerals province, being bulk commodities;
- (b) cost of logistics for a mining operation as a key investment decision and as a significant portion of costs once a mine is operational; and
- (c) very significant distance between the North West Queensland minerals province and the point of export at the Port of Townsville.

Glencore argued that all investment would be impacted, noting that declaration is a material driver of investment in the mining industry. It considered that in the absence of declaration, it would be an 'impossible expectation' for a new or existing miner to take on the 'enormous' levels of risk in exploring and developing a new tenement so far from the point of export, and negotiating access with no certainty as to whether the outcome of those negotiations will produce economically or commercially feasible terms.⁶⁹⁶

The QCA notes that the existence of declaration is one of a number of factors affecting investment decisions in the North West Queensland minerals tenements market, and that some level of investment would be expected in that market in any case. However, in considering the effect of declaration on investment in dependent markets, the QCA is satisfied that declaration will promote additional positive investment effects in the North West Queensland minerals tenements market, compared to a future without declaration.

The QCA notes submissions by Queensland Rail that past access charges on the Mount Isa Route service are a small proportion of the price of some of the commodities transported on the service. However, the QCA considers that these charges are reflective of an environment with declaration, and are not necessarily informative of a future without declaration. Additionally,

⁶⁹² Glencore, sub. 5, pp. 18–19.

⁶⁹³ Glencore, sub. 5, p. 19.

⁶⁹⁴ Glencore, sub. 5, p. 19.

⁶⁹⁵ Glencore, sub. 41, p. 28.

⁶⁹⁶ Glencore, sub. 41, p. 28.

access terms in relation to pricing is one of many considerations in comparing a future with and without declaration.

On the basis of the material before it, the QCA considers that in a future with declaration, the regulatory regime imposes an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power (for example, to hold up a user), and thus reduces the risk of undertaking an investment in the North West Queensland minerals tenements market.

Therefore, on balance, the QCA considers that declaration is likely to have a positive effect on investment in the North West Queensland minerals tenements market, a dependent market of the Mount Isa Route service.

13.5.6 West Moreton Route service

Queensland Rail highlighted the incentives that it has to promote investment in underutilised systems (including the West Moreton system), and named other material factors affecting investment in dependent markets that are not related to declaration, which in the case of the West Moreton system include:

- the risks associated with operating in global commodity markets
- the uncertainty as to New Acland Stage 3
- the potential impact of the Inland Rail and/or Surat Rail Basin projects, which could deter investment in the West Moreton system.⁶⁹⁷

Queensland Rail also submitted that the South West Producers had overstated below-rail costs relative to the risks and returns of undertaking mining investments.⁶⁹⁸

The South West Producers submitted that the long-term certainty and transparency of price and non-price terms have promoted, and will continue to promote, investment in the West Moreton region coal tenements market.⁶⁹⁹ They pointed to the decisions by Yancoal to increase production at the Cameby Downs mine (expanding from 1.8 mtpa initially to around 2–2.5 mtpa currently), and to the proposal by New Hope to extend the operation timeframe of the New Acland mine, as evidence that declaration has promoted investment in mining facilities.⁷⁰⁰ The South West Producers submitted that the investment in these mines were made, in part, in reliance on the certainty provided by the declaration and resulting access undertaking.⁷⁰¹

The South West Producers further contended that if declaration was retained in the Central Queensland Coal Network and Hunter Valley, but not for the West Moreton system and Metropolitan systems, the relative lack of pricing constraints and transparency would be taken into account by investors in determining whether to invest in coal tenements in the West Moreton region.⁷⁰² The South West Producers argued:

Given the importance of rail infrastructure costs for both coal investments and above rail investments, and the long term nature of such investments, the uncertainty of pricing (and access), will make it difficult if not impossible to provide any certainty to financiers or investors regarding the returns that are achievable. That, at a minimum significantly increases financing

⁶⁹⁷ Queensland Rail, sub. 33, p. 83, para. 421.

⁶⁹⁸ Queensland Rail, sub. 33, p. 83, para. 420.

⁶⁹⁹ South West Producers, sub. 31, pp. 19–20.

⁷⁰⁰ South West Producers, sub. 4, p. 51.

⁷⁰¹ South West Producers, sub. 4, p. 51.

⁷⁰² South West Producers, sub. 40, p. 47.

costs (thereby reducing profitability and, in turn, the prospects of investment), and more likely simply makes such projects completely unbankable.⁷⁰³

Once again, the QCA notes that the presence of declaration is one of a number of factors affecting investment decisions in the coal tenements market in the West Moreton region and that some level of investment would be expected in that market in any case. However, in considering the effect of declaration on investment in dependent markets, the QCA is satisfied that declaration will promote additional positive investment effects in the dependent market, compared to a future without declaration. On the basis of the material before it, the QCA considers that in a future with declaration, the regulatory regime imposes an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power (for example, to hold up a user), and thus reduces the risk of undertaking an investment in the coal tenements market in the West Moreton region.

Therefore, on balance, the QCA considers that declaration is likely to have a positive effect on investment in the coal tenements market in the West Moreton region, a dependent market of the West Moreton Route service.

13.5.7 South Western, Western and Central Western Route services

Watco submitted that declaration has supported 'significant investment and competitive entry', citing its entry into the grain freight market on the Central Western, Western and South Western systems.⁷⁰⁴ It argued that declaration promotes long-term certainty as to the rights of access and its availability on reasonable terms, which will facilitate investment. It also pointed to the opportunities that currently exist, stating that it is 'actively investigating' other freight opportunities in central and south-western Queensland. It highlighted that as part of its discussions:

Watco has been surprised with the resurgent interest in shifting freight from road to rail and identifying investment and innovation opportunities to enable this to occur. Watco's corporate history of similar investments in Western Australia and in North America support these potential opportunities in Queensland, with stable track access arrangements one of the keys to its success.⁷⁰⁵

Some examples of investments Watco has made to facilitate its entry into the above-rail freight haulage markets on the South Western, Western and Central Western Routes are discussed in section 8.6.2.

GrainCorp also submitted that declaration has supported investment and growth in competition in grain supply chains in Queensland, citing the decision by Watco to enter the relevant aboverail markets in Queensland to fulfil GrainCorp's grain haulage contract. It stated:

This competitive entry and investment has been underpinned by a stable regulatory framework which provides certainty around the terms of access to below rail infrastructure.⁷⁰⁶

GrainCorp highlighted that, in conjunction with its new arrangements with Watco, it is 'currently assessing business cases for proposed strategic investments in new sites and rail loading capability to achieve greater efficiencies to support the rail to port supply chain'.⁷⁰⁷ GrainCorp also said that significant investments in above-rail and supply chain infrastructure

⁷⁰³ South West Producers, sub. 31, p. 21.

⁷⁰⁴ Watco, sub. 48, p. 6.

⁷⁰⁵ Watco, sub. 48, p. 6.

⁷⁰⁶ GrainCorp, sub. 52, p. 3.

⁷⁰⁷ GrainCorp, sub. 52, p. 3.

have already been committed, which have been underpinned by the indicative access arrangements and pricing offered by Queensland Rail.⁷⁰⁸

GrainCorp noted that access charges are already higher than in other states and is concerned about a material increase in charges in the absence of declaration. It is concerned that in the absence of declaration, Queensland Rail 'would be free to exercise monopoly power in ways that could undermine the value of long-term investments', with the risk of hold-up potentially undermining the business case for future investment.⁷⁰⁹

GrainCorp also highlighted the importance of cost-effective rail transport to the agricultural sector, which will affect returns and ultimately investment in this sector, saying:

Queensland grain exports are dependent on access to below rail infrastructure on reasonable terms ... a significant proportion of grain exports must be carried by rail, given road capacity constraints. The terms of access to essential rail infrastructure directly impacts on the profitability of grain exports and returns to farmers.⁷¹⁰

It stated that supply chain costs are the single largest cost for grain farmers.

In the context of the Central Western system, Linfox submitted that essential items are transported into this region, some of which are used to produce agricultural products for domestic and international markets.⁷¹¹ If costs increase, which it considered likely in the absence of declaration, this will have a negative impact on dependent regional markets via higher input costs, lower productivity and reduced competitiveness. This in turn will have a detrimental impact on investment.

Queensland Rail did not make detailed submissions in relation to criterion (d) issues for the agricultural systems, noting only that it agreed with the QCA's draft recommendation that the services provided on the Western, South Western and Central Western systems would not promote the public interest:

It is evident that the services provided on these systems will be provided in the same manner, pursuant to TSC subsidies and for public policy reasons with or without declaration.⁷¹²

The QCA has reviewed its analysis in the draft recommendation after considering the submissions received from stakeholders in response to the draft recommendation. On the basis of the material before it, the QCA considers that declaration is likely to have a positive effect on investment in the above-rail haulage markets on the agricultural systems. In particular, it is no longer the case that, as the QCA considered in its draft recommendation, the only services that operate on the agricultural systems are those provided under an above-rail subsidy.

The QCA notes that the presence of declaration is one of a number of factors affecting investment decisions in the above-rail haulage market. In particular, it appears that, as was the case with Pacific National, the presence of a haulage contract underwriting the entry is a critical factor affecting an above-rail operator's decision to enter a market in which it has not previously operated. Therefore, it may be the case that some level of investment in the relevant above-rail markets would have occurred in any case. However, in considering the effect of declaration on investment in dependent markets, the QCA is satisfied that declaration will

⁷⁰⁸ GrainCorp, sub. 52, p. 12.

⁷⁰⁹ GrainCorp, sub. 52, p. 3.

⁷¹⁰ GrainCorp, sub. 52, p. 12.

⁷¹¹ Linfox, sub. 50, p. 3.

⁷¹² Queensland Rail, sub. 33, p. 91, para. 455.

promote additional positive investment effects in the dependent markets, compared to a future without declaration.

On the basis of the material before it, the QCA considers that in a future with declaration, the regulatory regime imposes an effective long-term constraint on Queensland Rail's ability and incentive to exercise market power (for example, to hold up a user), and thus reduces the risk of undertaking an investment in the above-rail haulage markets on the agricultural systems.

The QCA considers that a future with declaration is likely to promote efficient investment in the above-rail haulage markets on each of the South Western, Western and South Western Routes, as above-rail operators can be more certain that they will reap the benefits of their investments into the future.

Therefore, on balance, the QCA considers that declaration is likely to have a positive effect on investment in the above-rail haulage markets, which are dependent markets of the South Western, Western and Central Western Route services.

13.5.8 Tablelands system service

The QCA has concluded that declaration is unlikely to promote a material increase in competition in any dependent markets of the Tablelands system service, including the aboverail passenger market.

The QCA considers that both below- and above-rail services on the Tablelands system are heavily dependent on TSC subsidies for their continued operation (see section 9.4.2). The QCA considers that the prominent and decisive factor informing a decision to invest in the above-rail passenger market on the Tablelands system is the presence of government subsidies. In particular, based on the evidence before it, the QCA considers that both above-rail and below-rail services on the Tablelands system depend on TSC subsidies to fund investments to support the continued operation of these services. As such, the QCA considers that declaration would be unlikely to have a positive effect in promoting investment in the above-rail passenger market dependent on the Tablelands system service.

The QCA notes that other dependent markets may include the downstream tourism markets in the regions covered by the Tablelands system. However, there is a lack of publicly available information on which to base any analysis of such markets, and therefore the QCA has not formed a view regarding these markets.

13.6 Administrative and compliance costs incurred by the provider of the service

In considering criterion (d), the QCA is required to have regard to:

the administrative and compliance costs that would be incurred by the provider of the service if the service were declared $^{713}\,$

In this case, the provider of the service is Queensland Rail. The QCA considers that Queensland Rail would incur administrative and compliance costs if the service were declared. However, the QCA considers that Queensland Rail would be likely to incur some level of administrative and compliance costs in any case. That is, there are costs associated with dealing with multiple users, as long as Queensland Rail provides open access to its network, whether access is provided in a future with declaration or a future without declaration.

⁷¹³ Section 76(5)(c) of the QCA Act.

The QCA has considered administrative and compliance costs under three broad categories:

- The costs incurred by the QCA in regulating Queensland Rail, which are passed on to Queensland Rail through the QCA levy: Evidence suggests that these costs are passed through to users and are therefore not ultimately borne by Queensland Rail. In any case, they are not considered significant in the context of total access charges.
- The costs incurred by Queensland Rail in complying with the regulatory regime, including having a QCA-approved access undertaking: Queensland Rail has indicated that these costs are significant; however, it has not provided any estimates. Further, as Queensland Rail has submitted that its access framework is 'substantially the same' as its approved 2016 access undertaking, it is reasonable to expect that at least some of the costs associated with administering, managing and complying with the access framework would still be incurred.
- The costs incurred by Queensland Rail in dealing with multiple users: Queensland Rail has indicated that it will continue to provide third party access in a future without declaration (for example, through the access framework). It will therefore incur costs associated with third party access (such as negotiating access agreements), whether that access is provided in a future with or future without declaration. The QCA considers that these costs are unlikely to be materially different in either future scenario.

The QCA considers that no evidence has been provided to suggest that the costs incurred by Queensland Rail will be materially different in the absence of declaration.

13.6.1 Stakeholder submissions

Queensland Rail generally referred to the costs incurred by regulated entities, the QCA and other stakeholders, arguing that:

[d]eclaration gives rise to significant direct costs to Queensland Rail, the QCA and other stakeholders including costs of developing access undertakings, costs arising from the reference tariff process, and costs arising from the capital expenditure approval process.⁷¹⁴

Queensland Rail viewed that the 2016 access undertaking contains 'excessive prescription and unnecessary regulatory burden', including:

- the West Moreton reference tariff process
- the West Moreton capital expenditure process
- onerous reporting requirements, which it argued are disproportionate to any resulting benefits
- requirements to lodge draft amending undertakings to ensure ongoing 'workability' (e.g. in complying with national safety legislation)
- operational constraints prohibiting the provision of maintenance services to private railways
- operational constraints from committing to urgent infrastructure investment (e.g. in response to a safety issue) given the need to seek approval of that expenditure
- the ability to lodge a dispute under the Network Management Principles regarding changes to planned network possessions for maintenance up to the date of the possession.⁷¹⁵

⁷¹⁴ Queensland Rail, sub. 33, p. 84, para. 425.

⁷¹⁵ Queensland Rail, sub. 33, pp. 84–85, para. 428.

Queensland Rail argued for the costs and benefits arising from declaration to be weighed against each other:

Regardless of the incidence of these costs, it is critical to weight these costs against the benefits of declaring each system (which would not otherwise arise without declaration) to assess whether on balance, declaration of any system promotes the public interest.⁷¹⁶

Apart from its share of the QCA levy, Queensland Rail did not provide any estimates of the costs that it incurs internally as a result of declaration—for example, in preparing draft access undertakings to submit to the QCA.

West Moreton system

Queensland Rail highlighted in particular the administrative and compliance costs on the West Moreton system. The QCA currently approves reference tariffs only with respect to the coal train services accessing the West Moreton system and the Metropolitan system. Access charges for all other users across the Queensland Rail network are negotiated directly between Queensland Rail and the user, with no QCA oversight or participation unless an access dispute is referred to the QCA by either party.

Queensland Rail submitted that the West Moreton system incurs the most significant regulatory costs of all of its systems.⁷¹⁷ It stated that:

Queensland Rail respectfully disagrees with the QCA's view that the compliance costs are 'minor' relative to the total access revenue generated on the system as the QCA FY18 Levy collected was approximately 4% of the total access tariff revenue. Queensland Rail considers that given that access revenues are only sufficient to recover operating costs, let alone a return on the substantial fixed cost base of the system, 4% of access revenues devoted to the QCA levy is significant.⁷¹⁸

Queensland Rail stated that in comparison:

Aurizon coal network's expenditure on the QCA levy was less than 0.7% of its coal access revenue in 2017-18, demonstrating that regulatory costs are significant (by several orders of magnitude) in relation to the scale of operations of the West Moreton system.⁷¹⁹

Queensland Rail also contended that the costs associated with having an approved reference tariff for the West Moreton system outweigh the benefits, and stated its view that the approval process for capital expenditure is more prescriptive than the processes of other regulators.⁷²⁰

The South West Producers supported the QCA's views in the draft recommendation on administrative and compliance costs, arguing that:

- the QCA levy is a relatively minor cost, which is ultimately borne by users
- Queensland Rail's administrative and compliance costs are allowed for in reference tariffs
- the benefits of declaration outweigh the administrative and compliance costs.⁷²¹

They also referred to statements by the National Competition Council (NCC), that costs borne by users via access charges are unlikely to be relevant to the assessment of the public interest.

⁷¹⁶ Queensland Rail, sub. 33, p. 85, para. 429.

⁷¹⁷ Queensland Rail, sub. 33, p. 85, para. 430.

⁷¹⁸ Queensland Rail, sub. 33, p. 85, para. 432.

⁷¹⁹ Queensland Rail, sub. 33, p. 85, para. 432.

⁷²⁰ Queensland Rail, sub. 33, p. 86, box 3.

⁷²¹ South West Producers, sub. 31, p. 21.

The South West Producers also argued that it is Queensland Rail's approach to the regulatory process that is causing costs, citing 'ambit pricing claims' under regulation.⁷²² Further:

To the extent that QR wishes to decrease regulatory and compliance costs it is always welcome to provide draft amending access undertakings to the QCA seeking to remove or amend provisions that it considers imposes costs that are not justified by public benefit outcomes. The fact that QR has not done that, clearly raises questions about their assertions of undue regulatory burdens that are imposed as a result of declaration.⁷²³

The South West Producers submitted that as they remain willing to pay for the costs of regulation (via access charges) 'they must see real benefits from the regime in advance of the costs incurred.'⁷²⁴ Glencore expressed similar views.⁷²⁵

The South West Producers highlighted that the costs involved in managing a multi-user system will always exist with and without declaration. They raised an example from the stakeholder forum where Queensland Rail proposed that it will undertake an increased level of consultation in making amendments to the access framework, suggesting it would be a 'QCA like' approach.⁷²⁶

Agricultural systems

GrainCorp argued that Queensland Rail's administrative and compliance costs under declaration of the South Western, Western and Central Western systems would be low, because the QCA levy does not apply to services on those systems and, in its view, the ongoing compliance and administration needs of the QCA from Queensland Rail in relation to these systems are minimal.⁷²⁷ In any case, it considered that the costs would be outweighed by the benefits of declaration.

Watco agreed with GrainCorp, noting that it also considered that 'the costs of QCA administering the regulatory regime applicable to declared systems are small compared to the considerable benefits of declaration'.⁷²⁸ Similarly, Linfox considered that 'the incremental cost of maintaining the declared regulatory regime in relation to the Central West Rail System are small when compared to the benefits of declaration'.⁷²⁹

13.6.2 QCA analysis—general principles

Queensland Rail noted several examples of what it considered to be relevant regulatory costs:

Relevant regulatory costs include:

- (1) Costs borne by the QCA is [sic] regulating declared facilities including \$3.7 million in considering DAUs in 2014-15 and 2015-16, and resourcing costs borne by the QCA which are not publicly available.
- (2) Costs incurred by regulated entities pursuant to the QCA Levy. In 2015-16, the QCA generated \$13.9 million in fees received from regulated entities from the QCA levy, of which Queensland Rail spent \$760,452.
- (3) Costs incurred by regulated entities in developing Access Undertakings.

⁷²² South West Producers, sub. 40, p. 48.

⁷²³ South West Producers, sub. 40, p. 48.

⁷²⁴ South West Producers, sub. 40, p. 48.

⁷²⁵ Glencore, sub. 41, p. 29.

⁷²⁶ South West Producers, sub. 40, p. 48.

⁷²⁷ GrainCorp, sub. 52, p. 14.

⁷²⁸ Watco, sub. 48, p. 6.

⁷²⁹ Linfox, sub. 50, para. 3.12.

(4) Costs incurred by access seekers associated with the expense of the QCA process. For example, New Hope Corporation acknowledged the expense of the QCA process writing:

The development of a new undertaking for Queensland Rail (QR) has been costly and time consuming, and the resulting lack of regulatory certainty has caused a loss of confidence in the long term future of the Western System.⁷³⁰

Queensland Rail has provided examples of costs incurred by various parties due to the regulatory process. However, the QCA notes that s. 76(5)(c) of the QCA Act specifically directs the QCA to the costs that would be incurred by the *provider of the service* if the service were declared. As such, the QCA's view is that some of the costs referred to by Queensland Rail⁷³¹ should not be considered for the purposes of s. 76(5)(c), but may be considered as other relevant matters under s. 76(5)(d).

The costs that would be incurred by Queensland Rail if the service were declared can be discussed under three broad categories:

- the costs incurred by the QCA in regulating Queensland Rail, which are passed on to Queensland Rail through the QCA levy
- the costs incurred by Queensland Rail in complying with the regulatory regime (e.g. costs of preparing undertakings as well as ongoing compliance costs)
- the costs incurred by Queensland Rail in dealing with multiple users of its service (e.g. costs of negotiating access contracts, and train coordination and maintenance costs).

In comparing the future 'with and without' declaration, the QCA is interested in the extent to which a future with declaration is likely to result in Queensland Rail incurring higher (or lower) administrative and compliance costs compared to the costs that would be incurred in a future without declaration. Ultimately, any such costs are weighed against the benefits of declaration for the purpose of assessing whether declaration will promote the public interest.

Each of the three categories of cost will be discussed below.

Costs incurred by Queensland Rail through the QCA levy

The QCA charges Queensland Rail a fee for providing regulatory services (the QCA levy).⁷³² This fee includes all costs incurred by the QCA in regulating Queensland Rail, including costs in relation to the access undertaking approval process and ongoing compliance. Importantly, the regulatory regime (including the access undertaking) applies to the whole of the Queensland Rail service (as declared under s. 250(1)(b) of the QCA Act). However, the costs incurred by the QCA in providing regulatory services—as reflected through the QCA levy—are allocated amongst particular types of users, on particular systems, based on a 'beneficiary pays' principle. The QCA approves the QCA levy on an annual basis, following a transparent consultative process with stakeholders.

For 2018–19, the QCA levy is as follows:

• \$0.14152 per net tonne for the West Moreton system coal users

⁷³⁰ Queensland Rail, sub. 33, p. 84, para. 427.

⁷³¹ Such as the internal costs of the QCA, which are borne by the QCA itself and are not passed on to Queensland Rail; costs incurred by regulated entities other than Queensland Rail (e.g. as a general guide to the costs of regulation); or costs incurred by access seekers or users in participating in the regulatory process.

⁷³² Pursuant to s. 245(2) of the Queensland Competition Authority Act 1997 (Qld), s. 3 of the Queensland Competition Authority Regulation 2007 (Qld) (repealed 24 August 2018) and since 24 August 2018, s. 3 of the Queensland Competition Authority Regulation 2018 (Qld).

- \$0.07388 per thousand gross tonne kilometres for the Mount Isa Line freight and mineral users
- \$0.04416 per thousand gross tonne kilometres for the North Coast Line and West Moreton system freight and mineral users
- \$0.01561 per track kilometre for long distance passenger services.⁷³³

The QCA considers that these are comparatively minor costs.

In any case, under the current and previous access undertakings, Queensland Rail has been permitted, and has in practice, passed the entirety of the QCA levy through to its users. The coal users of the West Moreton system (the South West Producers) and the majority freight and minerals user of the Mount Isa Line (Glencore), who together bear up to 85.7 per cent of the total cost of the QCA levy in 2018–19⁷³⁴, have both indicated that they are willing to pay for the QCA levy as they see 'real benefits from the [regulatory] regime in advance of the costs incurred'.⁷³⁵

Additionally, the QCA notes the NCC took the following position in considering administrative and compliance costs:

Costs to a service provider that can be compensated for through access charges are unlikely to be relevant to the assessment of the public interest. ⁷³⁶

To the extent that these costs are passed through to users, this may still be a relevant matter the QCA considers under s. 76(5)(d) (see section 13.7.2).

Costs incurred by Queensland Rail in complying with the regulatory regime

Queensland Rail indicated that it incurs costs in complying with the regulatory regime, including costs in developing access undertakings and costs associated with the capital expenditure approvals process. Queensland Rail maintained that these costs are higher for the West Moreton system (compared to its other systems), because the West Moreton system is subject to more detailed regulation via the development and approval of reference tariffs for coal services.⁷³⁷

Queensland Rail did not provide estimates of the amount of the costs it incurs internally in complying with the regulatory regime. As such, the QCA is unable to form a view on the quantum of these costs.

Costs incurred by Queensland Rail in dealing with multiple users of its service

Queensland Rail indicated that it is likely to continue to provide third party access to its network in a future without declaration (for example, through its access framework).⁷³⁸ The QCA considers that the Queensland Rail network is likely to be a multi-user system in a future with or without declaration, particularly as Queensland Rail is a state-owned entity and is not vertically integrated with respect to freight services. Administration and compliance costs in managing a

⁷³⁵ South West Producers, sub. 40, p. 48; Glencore, sub. 41, p. 29.

⁷³⁷ Queensland Rail, sub. 33, p. 85, para. 430.

⁷³³ Queensland Competition Authority, *Queensland Rail's 2018–19 QCA levy proposal*, final decision, 2018, https://www.qca.org.au/wp-content/uploads/2019/05/34374_QCA-Final-decision-notice-QR-Levy-1.pdf. The 2018–19 figures have been used as the 2019–2020 QCA levy process was ongoing at the time of writing.

⁷³⁴ Queensland Competition Authority, *Queensland Rail's 2018–19 QCA levy proposal*, final decision, 2018, p. 2.

⁷³⁶ National Competition Council, *Declaration of Services: A guide to declaration of services under*

Part IIIA of the Competition and Consumer Act 2010 (Cth), Melbourne, 2018, pp. 44–45, para. 6.17.

⁷³⁸ Queensland Rail, sub. 33, p. 39, para. 193.

multi-user system will therefore likely exist in the foreseeable future with or without declaration. The nature and extent of these costs will vary depending on the number of users, the volume of traffic and the associated complexity in coordinating and managing operations.

The QCA has concluded that in a future with declaration, declaration will promote a material increase in competition in various markets dependent on Queensland Rail's service. As such, it may be that declaration promotes additional access seekers of Queensland Rail's service, and increases Queensland Rail's administration costs. However, this cost must be balanced against the benefits, including the benefits of competition in those dependent markets generally.

Queensland Rail submitted that in a future without declaration, its access framework is 'substantially the same' as its approved 2016 access undertaking.⁷³⁹ It did not discuss the level of costs it expects to incur under its access framework. The QCA considers that it may be expected that some of the costs incurred by Queensland Rail associated with administering, managing and complying with its access framework in a future without declaration will be similar to the costs Queensland Rail incurs in a future with declaration. This is because, in either scenario, Queensland Rail will need to consider access applications, negotiate access agreements and resolve disputes with its users, among the many other activities associated with providing a multi-user network service.

13.6.3 Consideration of the service as a whole and particular parts of the service

Given that the regulatory regime (including the access undertaking) applies to the whole of the Queensland Rail service (as declared under s. 250(1)(b) of the QCA Act), it is difficult to precisely attribute the various costs discussed above to either the service as a whole or particular part(s) of the service. As such, the QCA has undertaken a broad consideration of the administrative and compliance costs that would be incurred by Queensland Rail if the service were declared, except for where stakeholders have addressed particular parts of the service, which are considered below.

West Moreton Route service

Queensland Rail stated that 'the West Moreton System is subject to the most intrusive form of regulation and incurs the largest regulatory costs of Queensland Rail's eight systems'.⁷⁴⁰ These costs include costs related to the development of the West Moreton system reference tariffs and capital expenditure approval processes. Queensland Rail described the areas in which additional costs are incurred but did not provide any estimate of its internal costs in relation to the regulation of the West Moreton system.⁷⁴¹

The approved reference tariff for coal users on the West Moreton system and Metropolitan system includes a component reflecting Queensland Rail's operating expenses, which include Queensland Rail's internal 'corporate overhead' costs in complying with the regulatory regime. This means that for coal services on the West Moreton Route, the costs incurred by Queensland Rail internally in complying with the regulatory regime are recovered from coal users through the access charges. It remains possible that Queensland Rail is still bearing some costs that have not been passed on to users; however, the QCA has no information to confirm if this is the case and, if so, what the quantum of those costs are (or are expected to be in future).

The South West Producers said:

⁷³⁹ Queensland Rail, sub. 33, p. 44, para. 210.

⁷⁴⁰ Queensland Rail, sub. 33, p. 85, para. 430.

⁷⁴¹ Queensland Rail, sub. 33, pp. 85–86, paras 430–433.

[C]ustomers [coal users of the West Moreton System] bear their own costs of the regulatory process, and the prudent expenditure by QR on the regulatory process which is taken into account in the prices set by the QCA. Accordingly, if the customers effectively bear all of the costs arising from regulation and are still supportive of declaration continuing, it must be that the reasonable terms and conditions of access arising from declaration promote the public interest.⁷⁴²

This highlights that ultimately, in assessing whether declaration will promote the public interest the costs of declaration must be weighed against the benefits, which occurs by having regard to the various limbs of s. 76(5) of the QCA Act.

For completeness, the QCA has also considered non-coal users (e.g. agriculture) of the West Moreton Route service. The QCA understands that the volumes of non-coal services are seasonal and minor compared to the coal access services. In this respect, Queensland Rail has previously requested (and the QCA has approved) that freight, minerals and long distance passenger services on the West Moreton system attract a weighting that is significantly less than that of West Moreton coal traffic for the purposes of allocating the QCA levy.⁷⁴³ While these services are covered by the existing declaration and access regime, Queensland Rail has previously indicated that it does not consider that it incurs significant QCA-related costs for these services. Therefore, the costs incurred by Queensland Rail in relation to these services are likely to be very minor.

South Western, Western, Central Western Route services and the Tablelands system service

Currently, the QCA does not approve reference tariffs for the South Western, Western or Central Western Route services, nor the Tablelands system service. Under the 2016 access undertaking, access charges for users of these services are negotiated directly between the users and Queensland Rail under the existing regulatory framework.

Freight users of the South Western, Western or Central Western systems were not required to contribute to the QCA levy in 2018–19, and have historically not been required to contribute to the QCA levy.⁷⁴⁴

Queensland Rail did not provide any estimates of the costs that it incurs internally in dealing with the QCA in relation to the South Western, Western or Central Western Route services and the Tablelands system service, and there are no publicly available data on such costs. There is also no information on the interaction of such costs with the TSC subsidies that Queensland Rail would receive for below-rail services provided on these systems. As such, the QCA is unable to form a view on the quantum of these costs.

The QCA does not expect that the costs incurred in dealing with multiple users of these services will be materially different in a future with declaration, compared to a future without declaration.

13.6.4 Conclusions

With regard to the QCA levy, under the current and previous access undertakings, Queensland Rail has been permitted to pass the entirety of the QCA levy on to its users, and it has done so in practice. Consistent with the approach applied by the NCC, to the extent that the QCA levy

⁷⁴² South West Producers, sub. 16, p. 29.

⁷⁴³ Queensland Rail, 2017–18 Queensland Competition Authority Levy, Queensland Rail Access Undertaking 1, January 2018, https://www.qca.org.au/wp-content/uploads/2019/06/32627_queensland-rail-2017-18-qca-levy-report.pdf.

⁷⁴⁴ Queensland Competition Authority, *Queensland Rail's 2018–19 QCA levy proposal*, final decision, 2018.

continues in effect to be borne by the users of the Queensland Rail service, the QCA considers that the cost burden of the QCA levy on Queensland Rail is relatively minor. However, the costs borne by access seekers and holders may still be a relevant matter for the QCA to consider under s. 76(5)(d) (see section 13.7.2).

With regard to costs incurred by Queensland Rail in complying with the regulatory regime, Queensland Rail has not provided any estimates of these costs—for example, the additional costs it considers are associated with the regulation of the West Moreton system. The extent to which these costs are passed through to users via access charges is also not known, although for the West Moreton system, at least some of these costs are expected to be included in the approved operating expenditure allowance. The South West Producers and Glencore have indicated their willingness to continue to bear the costs associated with declaration (including their own costs as well as any costs passed through by Queensland Rail), which reflects their view that the benefits of declaration outweigh the costs.

With regard to costs incurred by Queensland Rail in dealing with multiple users of the service, Queensland Rail's stated intent is for its access framework to mirror the 2016 access undertaking. It is therefore reasonable to expect that at least some of the costs it currently incurs in managing the various processes—in dealing with multiple users of the service—will still be incurred in a future without declaration. The QCA does not expect that the costs of dealing with multiple users will be materially different in a future with or without declaration.

In conclusion, the QCA considers that Queensland Rail could be expected to incur additional administrative and compliance costs if the service were declared, compared to a future without declaration. However, the QCA notes that currently many of these costs (for example, the QCA levy) are ultimately borne by the users of the service, rather than by Queensland Rail itself.

Ultimately, these administrative and compliance costs are considered against the benefits of declaration, for the purpose of assessing whether declaration will promote the public interest.

13.7 Other relevant matters

13.7.1 Matters to be considered

There are a number of other matters that the QCA considers are relevant to the assessment of criterion (d) under s. 76(5)(d), including:

- costs of regulation that are borne by access seekers and holders
- economic and regional development issues, including employment and growth
- environmental and safety issues.

Stakeholder submissions

Queensland Rail considered that in addition to environmental and safety considerations, the following matters were also relevant to the QCA's assessment:

- the appropriateness of alternative regulatory arrangements applying in the future without declaration
- the extent to which private benefits resulting from declaration accrue to foreign owned entities.⁷⁴⁵ It argued that this has been extensively applied by the Federal Court and

⁷⁴⁵ Queensland Rail, sub. 33, p. 76, para. 398.

Australian Competition Tribunal when applying the net public benefit test in assessing authorisation applications under the *Competition and Consumer Act 2010* (CCA).⁷⁴⁶

Queensland Rail also maintained that any costs savings or other benefits accruing to it in the absence of declaration should be considered a public benefit, because it is 'a statutory entity under Government direction'.⁷⁴⁷

Glencore disagreed with Queensland Rail's claims that private benefits flowing to foreign owned companies should be discounted.⁷⁴⁸ It argued that Glencore pays taxes to the Commonwealth Government and royalties to the State Government, employs people in its Australian operations and invests in regional communities. It is therefore not appropriate to discount public benefits based on an element of foreign ownership.

QCA analysis

The QCA considers that Queensland Rail was referring to its access framework when mentioning 'alternative regulatory arrangements' applying in the future without declaration, which the QCA should have regard to. The presence of the access framework is relevant in attempting to predict the environment in a future without declaration. The QCA has had regard to Queensland Rail's access framework in assessing criteria (a) and (d).

In relation to Queensland Rail's comments about excluding private benefits to foreign owned companies, the assessment of authorisation applications under the CCA has quite a different purpose and context. For example, the *Qantas Airways* decision to which Queensland Rail referred⁷⁴⁹ involved an authorisation application for a strategic alliance between Qantas and a foreign entity (Air New Zealand). Also, the ACCC makes no reference in its 2019 *Guidelines for the Authorisation of Conduct (Non-Merger)* to excluding benefits flowing to foreign owned companies in assessing public benefits.⁷⁵⁰

In the QCA's view, in assessing an authorisation application between two businesses, one of which is domiciled overseas, the benefits that accrue to the Australian public as opposed to overseas beneficiaries will be a more relevant consideration. However, the declaration of a service for third party access is quite a different situation. This assessment does not look to identify and assess specific benefits flowing to one or more individual supply chain participants. Not only will different benefits accrue to different participants, but the number and identity of the relevant participants will change over the declaration period. In effect, such a company-by-company assessment would be necessary if the QCA were to somehow 'carve out' private benefits accruing to foreign companies as Queensland Rail is proposing. Such an approach is not feasible and is not appropriate, given the requirements of the QCA Act.

The QCA considers that in this context, the fact that some access seekers or holders of the Queensland Rail service may be owned by foreign entities is of limited relevance in assessing the public benefit (and costs) from declaration under criterion (d).

⁷⁴⁶ Queensland Rail, sub. 33, p. 89, para. 446.

⁷⁴⁷ Queensland Rail, sub. 33, p. 89, para. 447.

⁷⁴⁸ Glencore, sub. 41, p. 30.

⁷⁴⁹ *Qantas Airways Limited* [2004] ACompT 9; Queensland Rail, sub. 33, p. 89, para. 446.

⁷⁵⁰ Australian Competition and Consumer Commission, *Guidelines for the Authorisation of Conduct (Non-Merger)*, Commonwealth of Australia, March 2019.

13.7.2 Costs borne by access seekers and holders

Stakeholder submissions

The South West Producers expressed concerns that the costs to access seekers and holders will increase in the absence of declaration because costly and protracted negotiations are more likely.⁷⁵¹ The South West Producers discussed negotiation costs:

[D]eclaration has also provided an important role in reducing the costs of negotiation ... A standard access agreement and independently determined efficient pricing [under the current declaration] reduces the barriers to obtaining access, saves all parties (including QR) costs that would be otherwise associated with a drawn out negotiation.⁷⁵²

The South West Producers considered that the costs of the QCA's processes are likely to be significantly less than the costs that will be incurred by users if they need to rely on formal court dispute processes, including to prevent Queensland Rail from amending or breaching its access framework.

In respect of disputes regarding the access charge, the costs of those protracted arbitrations are very likely to result in users that are unable to fund such lengthy processes to settle early at a higher price, compared to users with deeper pockets that may have more ability to fund those disputes.⁷⁵³

The South West Producers highlighted the QCA's information gathering powers under the QCA Act, which will not be available in the absence of declaration, leaving users to engage in processes such as requests under the *Right to Information Act 2009* (Qld), with no assurance that this will be successful. Glencore made similar comments, and added:

QR has never answered the QCA's questions about how an Access Framework that is (on QR's view) supposed to produce a 'QCA like result' will reduce costs in the way that QR alleges.⁷⁵⁴

Watco commented that rail operators and customers bear their own costs of regulatory processes, while Queensland Rail's costs are reflected in approved access charges.⁷⁵⁵ It considered that the costs of administering the regulatory regime are small, compared to the benefits. Glencore also referred to users' willingness to pay, for the sake of the benefits of regulation:

The point also remains that if users, including Glencore, are willing to pay for the QCA levy and their costs of the regulatory system, they must see real benefits from the regime in advance of the costs incurred, and any remaining costs incurred by QR itself must be relatively minimal.⁷⁵⁶

QCA analysis

As noted previously, at least some of the costs incurred by Queensland Rail due to regulation by the QCA are passed through to access holders via access charges. This includes the QCA levy, and in the case of the West Moreton system, an allowance for Queensland Rail's operating expenditure, which is reflected in the reference tariff. Most of Queensland Rail's users who made submissions considered that the benefits of declaration outweigh the costs that they (the

⁷⁵¹ South West Producers, sub. 31, pp. 21–22.

⁷⁵² South West Producers, sub. 4, p. 53.

⁷⁵³ South West Producers, sub. 40, p. 48.

⁷⁵⁴ Glencore, sub. 41, p. 30.

⁷⁵⁵ Watco, sub. 48, p. 6.

⁷⁵⁶ Glencore, sub. 41, p. 29.

access users/seekers) are required to bear, and none suggested that these costs are excessive.⁷⁵⁷

Access seekers and holders bear their own costs in negotiating access with Queensland Rail and will continue to do so in a future with or future without declaration. Some stakeholders viewed that these costs would be higher in a future without declaration, as access seekers and holders could face more protracted and costly negotiation processes, and that in particular, the costs associated with disputes could be material.⁷⁵⁸ The South West Producers noted that transactional costs may be an issue for smaller access seekers, who may end up settling disputes or negotiations early at a higher price, which also increases their costs and could reduce their competitiveness.⁷⁵⁹

The QCA considers that to the extent that a standard access agreement continues to be approved by the QCA in a future with declaration, this has the potential to reduce administrative costs for access seekers and users (as well as Queensland Rail), by introducing a transparent standard 'starting position' for all parties to commence negotiations for access. This is because standard terms can avoid the need for new bespoke access agreements to be negotiated for each individual access request, and can help minimise the scope for access disputes and the costs associated with dispute resolution.

On balance, based on the material before it, the QCA considers that in a future with declaration, access seekers and holders are likely to incur administrative and compliance costs associated with the regulatory regime. However, these costs must be balanced against the benefits of the regulatory regime. Some costs in particular, such as the costs of negotiating access, are likely to be lower for access seekers and holders in a future with declaration, due to the availability of public information, and of documents with terms approved by the QCA (such documents include the access undertaking and the standard access agreement). This information that is available reduces to some degree the transactional costs associated with negotiating access, and sets a common 'starting point' for such negotiations to commence.

13.7.3 General economic issues, including employment and regional development

To the extent that declaration will promote competition in one or more dependent markets, this should promote increased efficiency and investment, compared to a future without declaration. This in turn will result in additional economic benefits, for example, increased employment and regional development. If competition is promoted, more new competitors may enter dependent markets—therefore, declaration may also increase the revenue that Queensland Rail will recover via access charges, which may reduce its reliance on TSC funding from the Queensland Government.

While the incremental benefit cannot be estimated with any certainty, the QCA considers that declaration is likely to result in net economic benefits in the regions serviced by the Queensland Rail service, to the extent that declaration promotes competition and investment in markets and regions dependent on the Queensland Rail service.

⁷⁵⁷ See for example, Pacific National, sub. 37, p. 16; South West Producers, sub. 40, p. 48; Glencore, sub. 41, p. 29; Watco, sub. 48, p. 6; Linfox, sub. 50, para. 3.12; GrainCorp, sub. 52, p. 14.

⁷⁵⁸ See for example, South West Producers, sub. 40, p. 48; Glencore, sub. 41, p. 30.

⁷⁵⁹ South West Producers, sub. 40, p. 48.

Stakeholder submissions

Queensland Rail argued that declaration 'does not result in any economic benefits which would not otherwise be promoted without declaration'.⁷⁶⁰

Glencore also submitted that declaration has promoted mining investments in the North West Queensland region, which has in turn brought broader public benefits to the North West Queensland regional economy in terms of employment and contribution to regional businesses and development.⁷⁶¹ Glencore submitted data from the Queensland Resources Council (QRC) that showed that in 2016–17 the resources sector in the North West Queensland region paid approximately \$397 million to 2,709 full-time employees, spent approximately \$354 million to the benefit of local businesses and community organisations, and contributed approximately \$1.3 billion in gross regional product, which was 19 per cent of North West Queensland's total gross regional product.⁷⁶² Glencore argued that such benefits arose directly as a result of the declaration of the Queensland Rail Network, including the Mount Isa Line.⁷⁶³ However, it did not address how these benefits might be affected in a future with and without declaration.

In the context of the West Moreton system, the South West Producers agreed with the QCA's views in the draft recommendation that declaration promotes investment in regional markets, producing flow-on effects for employment and regional development:

While, the South West Producers acknowledge that it could be argued that employment and regional development would still occur to some extent in the absence of declaration, they firmly believe that declaration provides additional employment and regional development. That is particularly clear at this point given the investment decisions in the coal industry and rail haulage industry that are to be made in the near future, and which are far less likely to be positive in the absence of declaration.⁷⁶⁴

The South West Producers quoted QRC data showing that in 2016–17 the resources sector in the West Moreton region paid approximately \$30 million in wages to 263 full-time employees, spent approximately \$7 million to the benefit of local businesses and community organisations, and contributed approximately \$70 million in gross regional product, which formed 2 per cent of the West Moreton region's total gross regional product.⁷⁶⁵

The South West Producers further argued that there are 'material public benefits to State financiers from declaration', saying 'that a rail line that either largely or wholly supports itself through commercial activities is preferable to one the State is required to heavily subsidise'. They contended that a reduction in coal volumes will increase the funding burden on the state (due to a proportional decrease in access charges received) and the effects will be compounded by the reduction in coal royalties.⁷⁶⁶

Linfox argued that a reliable and cost competitive rail service on the Central West system is 'critical' to the economic development of this region.⁷⁶⁷ It cited a Member of the Queensland Parliament in 2018 who stated in response to Aurizon's potential cessation of its Central West intermodal operations (prior to Linfox's acquisition) that 'closing this rail freight service will

⁷⁶³ Glencore, sub. 8, p. 19.

⁷⁶⁰ Queensland Rail, sub. 33, p. 75, para. 391.1.

⁷⁶¹ Glencore, sub. 8, pp. 18–19.

⁷⁶² Glencore, sub. 8, p. 17.

⁷⁶⁴ South West Producers, sub. 31, p. 22.

⁷⁶⁵ South West Producers, sub. 4, p. 49.

⁷⁶⁶ South West Producers, sub. 31, p. 22.

⁷⁶⁷ Linfox, sub. 50, p. 3.

cripple western Queensland', noting in particular that rail transport allows for the safe transport of goods classified as hazardous and dangerous into the region.⁷⁶⁸

QCA analysis

Employment and regional development

Glencore (Mount Isa Line), the South West Producers (West Moreton system) and Linfox (agricultural systems) commented on the economic benefits that are currently generated in each region. Economic benefits could also be expected to be generated based on the firms and industries that use the North Coast Line to transport inputs and outputs. The Kuranda Scenic Railway and other tourism services on the Tablelands system also contribute to the tourism economy in North Queensland—in particular, the passengers travelling on the Kuranda Scenic Railway accounted for 55 per cent of all customers travelling on Queensland Rail's regional passenger services in 2017–18.

However, the information stakeholders provided generally focuses on the *total* economic benefits delivered in the region. No estimates were provided of how much of this benefit would be foregone in the absence of declaration, or how much *additional* output and investment declaration may promote. These additional benefits in employment and regional development are difficult to quantify.

To the extent that declaration promotes competition and investment in markets and regions dependent on the Queensland Rail service, the QCA considers that declaration is likely to have a net beneficial effect in promoting employment and regional development in those markets and regions.

Finance and public funding

The South West Producers submitted that declaration has facilitated continued investment in the coal industry in the West Moreton region, which 'has relieved some of the financial pressure that would otherwise sit with the state government to fund maintenance of the West Moreton [system] solely for the purpose of passenger, livestock, grain and other freight services'.⁷⁶⁹ The South West Producers also considered that in the absence of declaration, the uncertainty of pricing and other terms of access may 'push coal users out of the market for the service', or reduce their demand for the service, such that 'a far greater proportion of the costs of operating and maintenance of the West Moreton corridor infrastructure would fall onto other sectors or directly on the government'.⁷⁷⁰

All Queensland Rail's systems with the exception of the Mount Isa Line receive TSC subsidies (see section 8.4.3), which the QCA understands are provided to meet the cost of the supply of the below-rail service. That is, where Queensland Rail cannot fully recover the cost of providing the below-rail service from the access charges⁷⁷¹, the TSC subsidy operates to 'fill the gap' between the amount of access charges received and the cost of providing the service. Therefore, if more access revenue is received on a system, it would be expected that the amount of the TSC subsidy would decrease proportionately.

⁷⁶⁸ Linfox, sub. 50, p. 3.

⁷⁶⁹ South West Producers, sub. 4, p. 50.

⁷⁷⁰ South West Producers, sub. 4, p. 53; South West Producers, sub. 16, pp. 24, 31–32.

⁷⁷¹ The Mount Isa Line is the only Queensland Rail system that is able to operate on a 'commercial basis' and does not receive below-rail TSC payments for its operation.

Queensland Rail's 2017–18 below-rail financial statements show the amount of revenue received via subsidies, and the subsidies' proportion of total revenue (Table 12). Queensland Rail does not disaggregate this data for each of its rail systems.

West Moreton Mount Isa region North Coast region Rest of network Total below-rail region \$0.743 million Nil \$152.339 million \$413.632 million \$566.714 million (0.61 per cent of all (75.4 per cent of (72.52 per cent of (63.3 per cent of revenue received) all revenue all revenue all revenue received) received) received)

Table 12 Queensland Rail revenue from TSC contracts, 2017–18

Source: Queensland Rail, Financial Statements for the Year Ended 30 June 2018, Below Rail Services Provided by Queensland Rail, p. 4,

https://www.queenslandrail.com.au/business/acccess/Compliance%20and%20reporting/2017-18%20Below%20Rail%20Financial%20Statements.pdf.

This data shows that with the exception of the West Moreton system and Mount Isa Line, Queensland Rail is highly dependent on TSC revenue to fund its operations. The other primary source of revenue is access charges—the West Moreton system received approximately 96 per cent of its revenue from access charges in 2017–18, while the Mount Isa Line received just over 98 per cent.

As a high proportion of Queensland Rail's costs are fixed, any reduction in revenue from access charges in any system will likely increase its dependence on TSC funding. To the extent that declaration promotes competition and investment in markets and regions dependent on the Queensland Rail service, the QCA considers that declaration may reduce the reliance of Queensland Rail on government subsidies. However, government policies relating to the TSC subsidies are at the discretion of the government, and may change at any time. As such, the QCA considers that declaration is likely to have a neutral effect in terms of finance and public funding issues.

13.7.4 Environmental and safety benefits

Stakeholder submissions

Queensland Rail submitted that 'efficiencies promoting increased competitiveness of rail results in environmental and safety benefits from increased rail modal share'.⁷⁷² It noted that stakeholders have argued that increased rail investment results in environmental and safety benefits as a result of shifting volumes from road onto rail.⁷⁷³ However, the key issue is whether this arises as a result of declaration. Queensland Rail considered that because it is incentivised to promote rail investment, and because of the lower regulatory costs that would arise in the absence of declaration, rail investment and increased modal share will be promoted without declaration.

Pacific National linked declaration to environmental and social benefits:

[D]eclaration of the QR infrastructure has supported the competitiveness of rail haulage (as against heavy vehicle road transport) on key freight corridors [including the North Coast Line]. This has in turn delivered a number of environmental and social benefits.⁷⁷⁴

⁷⁷² Queensland Rail, sub. 8, p. 7, para. 47.

⁷⁷³ Queensland Rail, sub. 33, p. 88, para. 443.

⁷⁷⁴ Pacific National, sub. 9, p. 14.

It gave examples of public benefits of rail:

- Road freight produces 14 times greater accident costs than rail freight per tonne kilometre.
- Road freight produces 16 times as much carbon pollution as rail freight per tonne kilometre.⁷⁷⁵

Pacific National calculated that the public benefit associated with its use of the North Coast Line, in terms of reduced carbon pollution and accident costs, were approximately \$17 million in reduced carbon pollution per annum and \$15 million in accident costs.⁷⁷⁶

The South West Producers considered that Queensland Rail's arguments regarding the existence of these benefits without declaration reflect Queensland Rail's views about the constraints and incentives it would face.⁷⁷⁷ The South West Producers did not consider that these constraints and incentives will be effective without declaration.

Watco disagreed with Queensland Rail's position that access as a result of declaration would not promote the public interest on the Western, Central Western and South Western systems, highlighting the safety and environmental comparisons between road and rail as cited by Pacific National.⁷⁷⁸

GrainCorp highlighted the benefits of transport by rail compared to by road, including reductions in accidents and road congestion, along with reduced fuel usage, which has significant environmental benefits.⁷⁷⁹ It estimated that road transport uses almost 250 per cent more fuel than rail over a similar distance. Rail transport also reduces road maintenance costs. It submitted that with the entry of Watco, GrainCorp intends to increase its use of rail to transport grain, which will further enhance these benefits.⁷⁸⁰ It attributed these benefits to a stable regulatory framework for below-rail access.

QCA analysis

To the extent that declaration promotes an increase in competition and investment in dependent above-rail markets, the QCA considers that above-rail operators are in turn likely to seek to attract end users to move their freight by rail instead of by road (or other means). This could therefore realise the additional environmental and safety benefits of rail transport, as highlighted by stakeholders, compared to a future without declaration. It is however, difficult to quantify such benefits.

13.8 Conclusions

13.8.1 The service as a whole

The QCA considers that declaration of the Queensland Rail service as a whole (as described in s. 250(1)(b) of the QCA Act) would:

• have a small beneficial effect on investment in facilities

⁷⁷⁵ Pacific National, sub. 9, p. 14.

⁷⁷⁶ Pacific National, sub. 9, p. 14.

⁷⁷⁷ South West Producers, sub. 40, p. 49.

⁷⁷⁸ Watco, sub. 48, p. 6.

⁷⁷⁹ GrainCorp, sub. 52, p. 14.

⁷⁸⁰ GrainCorp, sub. 52, p. 14.

- not have an effect on investment in dependent markets, to the extent that the QCA has been unable to identify a market dependent on the service as a whole, in which the QCA considers that declaration would promote investment
- cause Queensland Rail to incur administrative and compliance costs, although many of these costs are ultimately borne by users of the service.

Having weighed all of the costs and benefits, the QCA considers that there is a marginal net benefit.

Therefore, the QCA considers that access (or increased access) to the Queensland Rail service as a whole, on reasonable terms and conditions, as a result of declaration of the service, would promote the public interest.

13.8.2 North Coast Route service

The QCA considers that declaration of the North Coast Route service would:

- have a small beneficial effect on investment in facilities
- have a beneficial effect on investment in the dependent above-rail haulage market
- cause Queensland Rail to incur administrative and compliance costs, although many of these costs are ultimately borne by users of the service
- result in other small benefits.

Having weighed all of the costs and benefits, the QCA considers that there is a net public benefit.

Therefore, the QCA considers that access (or increased access) to the North Coast Route service on reasonable terms and conditions, as a result of declaration of the service, would promote the public interest.

13.8.3 Mount Isa Route service

The QCA considers that declaration of the Mount Isa Route service would:

- have a small beneficial effect on investment in facilities
- have a beneficial effect on investment in the North West Queensland minerals tenements market
- cause Queensland Rail to incur administrative and compliance costs, although many of these costs are ultimately borne by users of the service
- result in other small benefits.

Having weighed all of the costs and benefits, the QCA considers that there is a net public benefit.

Therefore, the QCA considers that access (or increased access) to the Mount Isa Route service, on reasonable terms and conditions, as a result of declaration of the service, would promote the public interest.

13.8.4 West Moreton Route service

The QCA considers that declaration of the West Moreton Route service would:

• have a small beneficial effect on investment in facilities
- have a beneficial effect on investment in the market for coal tenements in the West Moreton region
- cause Queensland Rail to incur administrative and compliance costs, although many of these costs are ultimately borne by users of the service
- result in other small benefits.

Having weighed all of the costs and benefits, the QCA considers that there is a net public benefit.

Therefore, the QCA considers that access (or increased access) to the West Moreton Route service, on reasonable terms and conditions, as a result of declaration of the service, would promote the public interest.

13.8.5 South Western, Western and Central Western Route services

The QCA considers that declaration of the agricultural systems services would:

- have a small beneficial effect on investment in facilities
- have a beneficial effect on investment in the dependent above-rail haulage markets
- cause Queensland Rail to incur administrative and compliance costs, although the level of these costs attributable to the agricultural systems would be relatively minor
- result in other small benefits.

Having weighed all of the costs and benefits, the QCA considers that there is a net public benefit.

Therefore, the QCA considers that access (or increased access) to each of the South Western Route service, the Western Route service and the Central Western Route service, on reasonable terms and conditions, as a result of declaration of the service, would promote the public interest.

13.8.6 Tablelands system service

The QCA received very limited information on the Tablelands system. The QCA considers that declaration of the Tablelands system services would:

- not have an effect on investment in facilities
- not have an effect on investment in any dependent markets, including the dependent market for above-rail passenger services
- cause Queensland Rail to incur administrative and compliance costs, although the level of these costs attributable to the Tablelands system would be relatively minor.

Having weighed all of the costs and benefits, the QCA considers that there is not a net public benefit.

Therefore, the QCA considers that access (or increased access) to the Tablelands system service, on reasonable terms and conditions, as a result of declaration of the service, would not promote the public interest.

14 DECLARATION PERIOD

The QCA recommends that the following parts of the service be declared for a period of 15 years:

- the North Coast Route service
- the Mount Isa Route service
- the West Moreton Route service
- the Central Western Route service
- the Western Route service
- the South Western Route service.

The QCA is satisfied about all of the access criteria for each part of the service for this period.

The matters to which the QCA has had regard in determining that 15 years is the appropriate period to recommend for declaration are set out in the QCA's analysis of the period for assessing total foreseeable demand for the purpose of criterion (b) (section 11.5).

APPENDIX A: THE HOLD-UP PROBLEM

This appendix elaborates further on the economic rationale for the 'hold-up' problem, which is discussed in the context of Queensland Rail. Specifically, it sets out some basic ideas, examples and ways to think about the issue, including applications to Queensland Rail and its customers.⁷⁸¹ It also includes a case study to provide further insight.

This discussion is not intended to be definitive or exhaustive, as there is a large literature in economics that addresses different aspects of the hold-up problem. Relevant references have been provided for further reading.

Overview

'Hold-up' is an economic problem that occurs where the value of an economic agent's relationshipspecific investment is potentially appropriable by that agent's trading partner(s). Relationship-specific investments are, by definition, particular to a given business relationship. For example, a supplier's purchase of specialised equipment or machinery to produce inputs specific to a buyer represents a relationship-specific investment.

A relevant feature of this type of investment is that, once made (sunk), its value in alternative uses is lower than its value in the current trading relationship. Further, the more specific the assets are to the current relationship, the more difficult it becomes for the investor to redeploy them to other uses. As a result, exit from the relationship is costly.

Accordingly, at the time of the initial investment decision, both parties have an incentive to make the relationship 'work'. However, once the investment is made (i.e. costs are sunk), the incentives of the parties change. This is because the gains from trade are only realised *after* the initial investment occurs.⁷⁸² As such, the parties have an incentive post-investment to behave strategically—should an opportunity arise—in order to appropriate a greater share of the gains from trade. The risk of this type of opportunistic behaviour is known as the *hold-up problem*.⁷⁸³

Given the potential for hold-up, the parties have an incentive, at the time of investment, to solve this dynamic problem, typically by committing to a contract prior to the investment that defines the terms and conditions of exchange.⁷⁸⁴ However, for reasons discussed later in this appendix, contracts are not always successful in resolving the hold-up problem, particularly when the initial investment is in an asset with a long expected life.⁷⁸⁵

This economic problem has several possible implications for efficiency. If a party perceives—at the time of making its investment decision—that it will not receive its expected return after the investment, that party will not have the incentive to choose an efficient level of investment in the first place. Specifically,

⁷⁸¹ The QCA uses 'customer' and 'user' interchangeably.

⁷⁸² Specifically, the basic timing of events is that i) parties make the required investments; ii) the good or service is produced; and iii) the gains from trade are realised and divided.

 ⁷⁸³ Goldberg, V, 'Regulation and administered contracts', *The Bell Journal of Economics*, vol. 7, no. 2, 1976, pp. 439–440. For an extensive discussion of the hold-up problem, see Williamson, O, *The Economic Institutions of Capitalism: Firms, Markets, Relational Contracting*, 1985.

⁷⁸⁴ Without a contract, investment in these circumstances will generally not be efficient. See Grout, P, 'Investment and wages in the absence of binding contracts: a Nash bargaining approach', *Econometrica*, vol. 52, no. 2, 1984, pp. 449–460.

⁷⁸⁵ In this case, the actual 'hold-up' could take the form of a forced renegotiation of the original contract or court enforcement of an imperfect contract.

the investor might delay, or reduce, its level of investment, even though the economic benefits of the investment—relative to its economic costs—indicate that it should proceed at a certain level and timing. For example, a prospective market entrant might decide not to enter an industry, even though the social benefits of entry exceed the social costs.

The investor might also expend substantial resources ex ante in trying to protect the potential investment from ex post appropriation—this activity itself can be inefficient, as the cost of protecting the investment involves the use of real resources. For example, an investor might spend excessive time and effort trying to write a 'complete' contract to protect its investment (discussed further below).⁷⁸⁶

As a result, relationship-specific investments can also have implications for the organisation of transactions in supply relationships. When investments are highly specific and uncertainty is high, contracting can be very costly and/or complex. In these circumstances, it might be more efficient for the transacting parties to internalise the transaction via vertical integration.⁷⁸⁷

Why does hold-up occur?

Investment 'hold-up' cannot be ascribed to any single reason. Rather, it is the presence and interplay of several economic factors that play a role in creating a hold-up opportunity. Klein suggests three economic pre-conditions are required for hold-up:

- asset specificity
- incomplete contracts
- incentives for wealth maximisation.⁷⁸⁸

Asset specificity

Asset specificity refers to the extent to which an investment is specific to a particular transaction, such that the opportunity cost of the investment is significantly lower. In general, economic exchange between two transacting parties requires them to make some investment in either assets or information that are specific to their business relationship. The more specific the asset or information is to the relationship, the more this feature 'locks in' the parties to the particular relationship—it becomes more costly to switch trading partners. In contrast, the less specific the assets, the stronger is the investing party's option to exit the relationship, given the prospect of hold-up.

One way to measure asset specificity is with reference to the aggregate level of *quasi-rents* created by investment. For the purpose of this discussion, the quasi-rents associated with a specific investment in a project can be defined as simply the difference between the value of an asset in its present use and its value in its next best alternative use (i.e. its opportunity cost). As an investment becomes increasingly specific to the relationship, its 'go-alone' value is likely to decrease, as it has little value in alternative uses. Therefore, as specificity increases, the quasi-rents increase, all else equal.

Types of asset specificity

There are four common types of asset specificity:⁷⁸⁹

⁷⁸⁶ Holmstrom, B & Roberts, J, 'The boundaries of the firm revisited', *Journal of Economic Perspectives*, vol. 12, no. 4, 1998, p. 74.

⁷⁸⁷ Vertical integration does not always solve the hold-up problem. One view is that contracting within a firm can be as difficult as contracting between firms, because vertical integration changes ownership and, therefore, the allocation of residual rights of control. See Grossman, S & Hart, O, 'The costs and benefits of ownership: a theory of vertical and lateral integration', *Journal of Political Economy*, vol. 94, 1986, pp. 691–719.

⁷⁸⁸ Klein, B, 'The hold-up problem', in P Newman (ed.), *The New Palgrave Dictionary of Economics and the Law*, vol. 2, 1998, pp. 241–244.

 Physical asset specificity—one or both parties to a transaction make investments in machinery or equipment that involve design features specific to the relationship and which have lower values in alternative uses.

For example, boilers in a coal-burning electricity plant can achieve greater efficiency if they are designed for a specific type of coal; however, they are less efficient if they burn coal with a different heat, sulphur, moisture, or chemical content.

• *Site specificity*—investments in productive assets are made in close physical proximity to each other in order to reduce inventory, transportation and processing costs.

For example, a utility constructs a 'mine-mouth' electricity plant adjacent to several mines and anticipates obtaining all coal supplies for the plant directly from these mines.⁷⁹⁰

 Human-asset specificity—refers to the accumulation of knowledge, training and/or expertise that is specific to one trading partner.

For example, the design and development of a new automobile is complex and time-intensive, involving very close collaboration between the automobile company and its parts suppliers; these suppliers acquire specific knowledge about the production of the components.⁷⁹¹

• *Dedicated assets*—investments are made in general capital to meet the demands of a particular buyer; the assets are not specific to the buyer, but if the buyer decides not to purchase, the input supplier would have substantial excess capacity.

For example, NutraSweet was the largest producer of the artificial sweetener, aspartame, with a worldwide market share of close to 95 per cent; as the largest buyers of its product were soft drink manufacturers—Coca-Cola and Pepsi—NutraSweet's investment in aspartame capacity was a dedicated asset.

Queensland Rail declaration review

The two most relevant types of asset specificity in the Queensland Rail declaration review are site specificity and physical asset specificity.

Site-specific investments typically involve costs for large, sunk assets that have low, or no, value in alternative uses.⁷⁹² Site-specific assets are typically highly immobile and therefore cannot be readily redeployed to alternative tasks. As a result of these features, asset specificity for site-specific assets is likely to be high.

A number of examples of site specificity are relevant for the current review. The sunk assets associated with existing investments include rail infrastructure and mine-related development.⁷⁹³ Most rail infrastructure, such as track and tunnels, is largely fixed (i.e. in situ); once put in place, the infrastructure cannot be redeployed. Likewise, above-rail assets can also be specific assets, such as rollingstock configured for the specific gauge or axle loads of the specific track it is designed to run on.

⁷⁸⁹ Williamson, O, The Economic Institutions of Capitalism: Firms, Markets, Relational Contracting, 1985.

⁷⁹⁰ Joskow, P 'Vertical integration and long-term contracts: the case of coal-burning electric generating plants', Journal of Law, Economics & Organization, vol. 1, no. 1, 1985, pp. 33–80.

⁷⁹¹ Montevarde, K & Teece, D, 'Supplier switching costs and vertical integration in the automobile industry', *Bell Journal of Economics*, vol. 13, no. 1, 1982, pp. 206–213.

⁷⁹² Large, site-specific infrastructure assets could have an opportunity cost if their salvage value is positive. Of course, if the cost of salvaging (or scrapping) the asset outweighs its potential salvage value, then a firm would not salvage it.

⁷⁹³ The potential, but not yet, incurred (sunk) expenditures associated with future investment in mining and aboverail operations are also relevant.

The underlying value of a mining development, once established, resides in its potential output. While mining equipment and machinery might be transferable, these assets are secondary to the primary asset, the mine itself.

While the investments in mine development and rail network development might not have occurred contemporaneously as part of the same ex ante decision, the two sets of investments over time have become effectively 'locked in' to each other, as rail transport in most cases is the most financially viable option for network customers to transport coal and minerals to their destinations.

Other assets related to the network have an element of physical asset specificity. For instance, Queensland Rail's network uses narrow gauge track, which differs from the gauge of track in other Australian states. This characteristic means that a potential entrant (i.e. a prospective above-rail operator) from outside Queensland would have to incur costs in procuring rollingstock exclusively for use on narrow gauge track. Investment costs of buying rollingstock for this purpose represent a type of physical asset specificity because they reduce the transferability, and hence value, of such assets in other uses.

Contracting

Incompleteness

The economics literature on contract theory demonstrates that, in the absence of a credible ex ante contract, a firm would not have a sufficient incentive to invest in a relationship-specific project due to the prospect of hold-up by the other party.⁷⁹⁴ Accordingly, contracts are fundamental instruments for addressing performance.

In relationships with large, specific investments (i.e. high asset specificity), contracts are arguably even more important. A primary purpose of a contract is to establish an agreement ex ante—before the relationship-specific investment is made—that specifies the terms of exchange, including the ex post distribution of the investment value between the parties. A relevant question is why a contract does not always protect parties from hold-up. After all, a contract specifying the terms of trade and subject to court enforcement would seem prima facie sufficient.⁷⁹⁵

As uncertainty is prevalent, writing a 'complete' contract is prohibitively costly.^{796,797} Complete contractual specification would require expansive 'search and discovery' of all relevant contingencies and incur costs for negotiating with the transacting party about the distribution of value for each contingency (as well as agreement on non-price terms and conditions). Given the very low probability of most contingencies, complete contractual specification would not be cost-effective.

Contract structure

Investments in essential infrastructure tend to be highly asset-specific, because fixed and sunk costs are large. Transaction cost economics suggests that, all else equal, increasing asset specificity is positively

⁷⁹⁴ In particular, it can be shown that no efficient bargaining process exists i) if both project value and costs are private information; ii) if gains from trade do not arise with certainty; and iii) as long as the parties are free not to trade (i.e. they can 'walk away' with no loss). See Myerson, R & Satterthwaite, M, 'Efficient mechanisms for bilateral trading', *Journal of Economic Theory*, vol. 29, 1983, pp. 265–281.

⁷⁹⁵ While contracts are significantly helpful in addressing performance, including the hold-up problem, they are nonetheless imperfect. The efficiency of contracts generally requires i) perfect information; ii) no transactions costs; and iii) the absence of externalities toward third parties. See Tirole, J, *The Theory of Industrial Organization*, 1988.

⁷⁹⁶ If a contract is 'complete', then it will never require revision, and enforcement is always possible. In particular, the contract specifies all possible contingencies. As a complete contract contains no gaps or missing contingencies, it would perfectly govern the exchange between parties as circumstances unfolded over time.

⁷⁹⁷ Transactional costs of writing a contract include i) writing the contract in sufficiently precise terms such that it can be understood as intended by a court of law; ii) monitoring; and iii) enforcement.

correlated with the use of long-term contracts.⁷⁹⁸ If an investment is long-term and specific (i.e. its value in alternative uses is low) then both parties want long-term assurances of their potential gains from the relationship prior to committing the investments. Accordingly, long-term contracts frequently characterise these types of situations.⁷⁹⁹

However, it is likely that there will be a number of uncertainties—factors that cannot be adequately anticipated or measured—at the time of investment, and their resolution will not occur until after the investment. This possibility suggests that a contract should be sufficiently flexible so that adjustments can occur after the investment in order to accommodate realisations of uncertainty. The expected costs of ex post adjustments in such circumstances could be less than the expected losses from a rigid contract.⁸⁰⁰

Therefore, there is a trade-off between mitigating opportunistic behaviour and flexibility. Inflexible pricing in long-term contracts can be hazardous in the face of uncertainty, and long-term prospects, by their nature, face the most uncertainty.⁸⁰¹ For instance, uncertainty about demand and supply conditions 10 years from today is greater than uncertainty about such conditions one year from today—uncertainty grows with time. As a result, long-term contracts require some flexibility despite concerns with opportunism. Accordingly, most long-term contracts contain some form of provision for price adjustment.⁸⁰²

Methods of price adjustment typically take one of two forms: redetermination provisions or renegotiation provisions. Redetermination provisions establish prices via formulae. They might reference an explicit, predefined schedule of prices ('definite escalator') or relate contract prices to market conditions as those conditions unfold ('indefinite escalator').⁸⁰³ Renegotiation provides substantial flexibility as it enables the parties to take into account a full range of information before agreeing on an adjusted price. At the same time, renegotiation provides greater scope for strategic behaviour (discussed below).

⁷⁹⁸ Williamson, O, 'Transaction costs economics: the governance of contractual relations', *Journal of Law and Economics*, vol. 22, no. 2, 1979, pp. 241–242. Modern transaction cost economics originates with Williamson. Williamson's work (especially his later work) focuses on the idea that if one can characterise transactions using key attributes (e.g. frequency of transaction, uncertainty and asset specificity), then any transaction can be 'mapped' into an efficient institutional arrangement. Williamson suggests that each characteristic has a positive relationship to the need for adopting internal governance. See especially Williamson, O, *The Economic Institutions of Capitalism: Firms, Markets, Relational Contracting*, 1985.

⁷⁹⁹ Empirical evidence confirms that asset specificity is an important determinant of vertical relationships, including the use of long-term contracts. Data on the duration of contracts between electric utilities and coal mines in the United States shows that, as asset specificity becomes more important, these contracts reflect longer-term commitments (see Joskow, P, 'Contract duration and relationship-specific investments: empirical evidence from coal markets', *American Economic Review*, vol. 77, no. 1, 1987, pp. 168–185). The same conclusion applies for longterm contracts between natural gas producers and pipelines in the United States (see Crocker, K & Masten, S, 'Efficient adaptation in long-term contracts: take-or-pay provisions for natural gas', *American Economic Review*, vol. 75, no. 5, 1985, pp. 1083–1093).

⁸⁰⁰ In addition, an overly rigid contract might not be breached easily or cheaply if there are major, unexpected changes in market conditions (see the General Motors–Fisher Body case study at the end of the appendix).

⁸⁰¹ For example, if there is substantial uncertainty over price and volumes, parties might prefer shorter contracts, all else equal.

⁸⁰² Crocker, K & Masten, S 'Pretia ex machina? Price and process in long-term contracts', *Journal of Law and Economics*, vol. 34, no. 1, 1991, pp. 69–99.

⁸⁰³ For an indefinite escalator, the process remains deterministic, but the formula ties the contract price to a price index or to the price of a relevant input or substitute. Adjustment provisions, such as most-favoured-nation and right-of-first-refusal, relate contract prices to prices from similar transactions or best alternative offers. In this way, these clauses make use of information more closely related to the transaction. See Crocker, K & Masten, S 'Pretia ex machina? Price and process in long-term contracts', *Journal of Law and Economics*, vol. 34, no. 1, 1991, pp. 74– 76.

Renegotiation and bargaining power

In the presence of asset specificity, one purpose of a long-term contract is that it avoids the need for frequent renegotiation. Aside from the costs of negotiating mutually acceptable terms, repeated bargaining can dissipate the value of the initial investment and have adverse impacts on efficiency.⁸⁰⁴ However, it can also benefit the parties if it means that the re-negotiating leaves no 'surplus on the table'.

Renegotiation provides for the most flexibility in the face of changes in market circumstances, as the negotiations are able to take full advantage of current information. The drawbacks are that renegotiations are typically more costly and that they are prone to strategic behaviour—there is greater temptation to argue about 'how to share the pie'.⁸⁰⁵

Relative bargaining powers are central to negotiations and their outcomes. If a buyer has a maximum willingness to pay (reservation price) of v and a seller has a cost of c such that $v \ge c$, then it is efficient for trade to occur.⁸⁰⁶ One way to think of bargaining power is the ability of each party to obtain a final price within this range that is as far from its reservation value as possible. For example, a seller would like to obtain a price p = v but will settle for p such that $v > p \ge c$. Relevantly, bargaining power can also shift the bounds of the bargaining range.⁸⁰⁷

Contracting and reputation

A final consideration about contracts is the interaction between contracting and reputation. Macaulay (1963) reports that relations between firms tend to be more informal than predicted by contract theory. The explanation is that, as contracts are costly to write and enforce, firms view reputation as a substitute for contracting specification. Ultimately, contractual performance is secured by the threat of termination of the relationship by one party.⁸⁰⁸

Economic theory suggests that an effective reputation mechanism requires several conditions to be satisfied:

- Information about the firm's past behaviour must be available to all potential trading partners.
- The firm must have a sufficiently long time horizon.
- The firms must believe in collective punishment and reward (product boycott for example), and there must be a common perception of cooperative vs non-cooperative behaviour.

In the context of monopolies, the effectiveness of a reputation mechanism will likely be weaker, as the mechanism relies on stakeholders punishing the company for 'bad' behaviour. However, in a monopoly situation, the costs of boycotting the monopoly might be too high given there are no real alternatives. As

⁸⁰⁴ Macleod, B & Malcomson, J, 'Contract bargaining with symmetric information', *Canadian Journal of Economics*, vol. 28, no. 2, 1995, p. 358.

⁸⁰⁵ Goldberg, V & Erickson, J, 'Quantity and price adjustment in long-term contracts: a case study of petroleum coke', *Journal of Law and Economics*, vol. 30, no. 2, 1987, p. 387.

⁸⁰⁶ This cost reflects production cost and any other relevant opportunity costs, such as the cost of capital.

⁸⁰⁷ There are a number of important factors that can potentially affect bargaining power: i) demand and supply conditions; ii) market concentrations; iii) informational advantages; iv) risk aversion; v) patience; and vi) negotiating tactics. For further details, see Choi, A & Triantis, G, 'The effect of bargaining power on contract design', *Virginia Law Review*, vol. 98, no. 8, 2012, pp. 1675–1678.

⁸⁰⁸ Macaulay, S, 'Non-contractual relations in business: a preliminary study', *American Sociological Review*, vol. 28, no. 1, 1963, pp. 55–67.

a consequence, the net benefit of establishing and maintaining a good reputation is likely to be less for a monopoly than for a firm in a non-monopolistic market.⁸⁰⁹

Queensland Rail declaration review

The QCA does not have transparency of commercial-in-confidence contracts between Queensland Rail and its customers. Based on stakeholder submissions, a number of these contracts tend to possess two key features. They are generally longer-term contracts—with a duration of at least five years, and in some cases 10 years. In addition, a number of these contracts contain take-or-pay provisions.

While a number of these contracts are long-term contracts, the lives of the assets are typically longer. Coal mine investments typically have expected economic lives of at least 10 to 30 years. Likewise, aboverail rollingstock typically has an asset life of at least 20 years. For instance, the New Acland Stage 3 coal development in the West Moreton system (and new rollingstock investments) would have longer asset lives than the existing assets. The implication is that a number of these assets will have longer asset lives than the term of a typical take-or-pay contract for rail access and therefore will require renewal (on one or more occasions) in the future.⁸¹⁰

As a third party, the QCA does not have knowledge of how these renewal negotiations might unfold. However, the QCA considers it likely that removal of the declaration, all else equal, would alter the relative bargaining powers in favour of Queensland Rail. Queensland Rail is the only provider of rail network access. In systems in which rail is the most economically viable transport option, and in systems in which customers have already made sunk investments in reliance on access to the below-rail service, customers have no other alternative. In bargaining parlance, the customer's 'no-agreement' option is very limited.^{811,812}

If this bargaining power is substantial (at least relative to a future with declaration), Queensland Rail could potentially obtain advantages over price and non-price terms in negotiations with customers. Queensland Rail may, or may not, choose to exercise this bargaining power (see the next section on opportunism and reputation effects). However, the problem is that it does not seem possible for Queensland Rail to credibly commit, at the current time, to not exercise greater bargaining power in a future without declaration.

While investments by Queensland Rail and its customers are already sunk, there are economic implications for dynamic efficiency. Existing customers or potential market entrants might either delay or forgo new investment. Further, stakeholders might expend considerable time and resources trying to protect their share of investment value. This type of protective activity is potentially wasteful. Finally, actual reapportionment of the present value of the initial investment might raise issues of fairness (although this is not an efficiency issue per se).

⁸⁰⁹ Graafland, JJ & Smid, H, *Reputation, corporate social responsibility and market regulation*, Tilburg University, Netherlands, 2004, available online as Munich Personal RePEc Archive (MPRA) paper no. 2072, posted 18 February 2010, https://mpra.ub.uni-muenchen.de/20772/.

⁸¹⁰ South West Producers, sub. 40, p. 11.

⁸¹¹ Choi, A & Triantis, G, 'The effect of bargaining power on contract design', *Virginia Law Review*, vol. 98, no. 8, 2012, pp. 1675–1676.

⁸¹² As Pacific National notes, 'since PN must acquire access to the QR infrastructure in order to make use of [PN's] own infrastructure [e.g. rollingstock], we have little choice but to accept the terms dictated by the monopolist. PN cannot credibly make a threat to withdraw from negotiations with QR ... consider the credibility of a threat by PN to stop using QR's below-rail infrastructure and redeploy its rolling stock to offer freight services by road – such a threat clearly would not be credible, given the sunk, asset-specific and long-lived nature of PN's investment in rolling stock and other above-rail infrastructure' (Pacific National, sub. 37, p. 10).

Incentives for 'wealth maximisation'

Opportunistic behaviour

The third pre-condition for hold-up relates to the incentive for 'wealth maximisation'. Given a material degree of asset specificity and the incomplete nature of contracts, there will be rents ex post investment that each party in the relationship would like to acquire. An important consideration is determining the conditions under which a party to a transaction is likely to undertake a hold-up to appropriate these rents.

American economist, Oliver Williamson, explains hold-up behaviour by reference to 'opportunism', and defines opportunism as 'self-interest seeking with guile'.⁸¹³ Accordingly, Williamson's definition admits a broad range of opportunistic behaviour such as lying, cheating, and deception. Setting aside Williamson's definition, the concept of opportunism can manifest itself in a number of forms, including adverse selection, moral hazard, shirking, incomplete disclosure of information and other forms of strategic behaviour.⁸¹⁴

While specific assets and contract incompleteness are both prevalent in transactions, hold-up does not always occur. For example, consider investment in designing specialised automobile parts. It is normal practice for Japanese suppliers to make specific investments and retain ownership of the specialised machines. This model would seem to present the manufacturer with an opportunity for hold-up. Once the supplier has made an investment in the machines, the manufacturer could demand a lower price. Alternatively, as the Japanese manufacturers tend to have a small number of suppliers for any particular part, a supplier might be in a position to attempt an opportunistic renegotiation using the threat of withholding supply.^{815,816}

Reputational capital (private sanctions)

Given these real-world observations, Klein considers that the 'opportunism' rationale by itself is not a sufficiently complete or satisfactory explanation of hold-up.⁸¹⁷ Klein proposes that transactors are more deliberative and calculating when making a decision about whether to hold up a transacting party. In particular, the benefits from undertaking the hold-up must outweigh the short and long-term costs imposed by the transactor being held up. Klein refers to these costs (collectively) as 'reputational capital' (or a 'private sanction') that the party being held up can impose on the offending party. This reputational capital comprises two primary costs:

• *future losses from termination of the relationship*—given the transaction-specific capital in the relationship, terminating the relationship means the potential loss of the discounted value of the (future) quasi-rents from the investment

⁸¹³ Williamson, O, The Economic Institutions of Capitalism: Firms, Markets, Relational Contracting, 1985, p. 47.

⁸¹⁴ Some of these problems arise from the fundamental asymmetry of information between transacting parties. Adverse selection ('hidden information') refers to the incentive for an economic agent to hide its true cost profile. Moral hazard ('hidden action') refers to the lack of the incentive for an economic agent to operate or invest efficiently.

⁸¹⁵ Holmstrom, B & Roberts, J, 'The boundaries of the firm revisited', *Journal of Economic Perspectives*, vol. 12, no. 4, 1998, pp. 80–81.

⁸¹⁶ In contrast, the traditional US model typically involves the auto manufacturer owning any physical assets specific to its production needs. This model also extends to include any assets that a supplier to the manufacturer uses in that supplier's factory. For example, the stamping machines used in making a particular automobile part will belong to the automaker, even though the supplier uses the machines in its plant. This business pattern is consistent with hold-up theory, in that the designs are highly specialised and difficult to fully protect in a contract. Accordingly, external contractors are not prepared to enter into such relationship-specific investments (Holmstrom, B & Roberts, J, 'The boundaries of the firm revisited', *Journal of Economic Perspectives*, vol. 12, no. 4, 1998, pp. 80–81).

⁸¹⁷ Klein, B, 'Why hold-ups occur: the self-enforcing range of contractual relationships', *Economic Inquiry*, vol. 34, no. 3, 1996, p. 444–463.

 future losses from foregone business due to reputational damage—the transactor perpetrating holdup will confront higher costs of doing business in the future as potential transacting partners will demand more favourable and explicit contract terms.⁸¹⁸

Contractual self-enforcement

Essentially, each party to a transaction at a point in time compares the potential hold-up gain from breaching the contract with the reputational capital of the 'held-up' party. Hold-up only occurs if the prospective gain outweighs the losses from the other party imposing its reputational capital in response to the hold-up. However, if the potential gain is less than the reputational capital, then hold-up does not occur.

Importantly, the reputational capital that each party can potentially impose on the other party acts as an implicit bound on the other party's behaviour with respect to the contract. Specifically, as long as circumstances do not move the value of hold-up outside these bounds, a hold-up will not take place (i.e. the value of the reputational capital is greater).

Accordingly, these bounds function as the 'self-enforcing range of the contractual relationship'. This range represents the extent to which market conditions can change without precipitating a hold-up by either party. Put another way, only when changes in market circumstances move the transacting parties outside of this range does the threat of hold-up become credible.⁸¹⁹

This probabilistic explanation is consistent with real-world observations of both 'hold-up' and 'no hold-up' under similar conditions, namely high asset specificity and incomplete contracts. It is also consistent with the observation that contracts are imperfectly specified, with parties expecting to perform their obligations in a way that is consistent with the *intent* of the contract.⁸²⁰

Queensland Rail declaration review

Applying this framework, a key issue is whether a future without declaration is likely to move Queensland Rail and transacting parties outside of the self-enforcement range of their contracts—relative to a future with declaration. In other words, the issue is whether it is more likely, or less likely, that hold-up would occur in a future without declaration.

In a future without declaration, Queensland Rail would likely have greater bargaining power at a renegotiation stage, given that in most cases customers have already made sunk investments in reliance on continued access to the below-rail service, and in many cases access to the network would be the most economically viable means for transporting products to their final destination.⁸²¹

With sufficient bargaining power, Queensland Rail could impose a hold-up on transactors, if it chose to do so. Specifically, it could potentially extract all of the surplus value from a transacting relationship, leaving

⁸¹⁸ Klein, B, Crawford, R & Alchian, A, 'Vertical integration, appropriable rents, and competitive contracting process', *Journal of Law and Economics*, vol. 21, no. 2, 1978, pp. 306–307.

⁸¹⁹ While reputation contributes to 'self-enforcement' of business relationships, transactors are not indifferent to contract terms. This is because a firm's reputational capital is limited. Consequently, firms use contract terms to get 'close' to desired performance and then use their reputational capital to ensure their relationship is 'self-enforcing' over the widest range of future market conditions. As a result, court enforcement and self-enforcement are complementary mechanisms to ensure transactor performance.

⁸²⁰ Klein, B, 'Why hold-ups occur: the self-enforcing range of contractual relationships', *Economic Inquiry*, vol. 34, no. 3, 1996, p. 449.

⁸²¹ While it is possible that other determinants of bargaining power might give users some countervailing power, these effects seem unlikely to be as strong.

the other party marginally committed to remaining in the market. Accordingly, the expected value of hold-up would likely be high relative to its expected value in a future with declaration.⁸²²

The second consideration relates to transactors' reputational capital. Expected losses from the termination of a relationship (i.e. a user 'quits' the relationship due to hold-up) and the expected losses of potential future business (e.g. a user chooses not to enter the market) determine the value of this reputational capital.

For the Queensland Rail systems, in most cases, it is not a credible threat for users to terminate their relationship with Queensland Rail in the event of hold-up. A threat to terminate would generally not be credible because, with few exceptions, existing users have already made highly specific, sunk investments. For prospective entrants, hold-up could have some reputational implications for their decisions; however, this depends on the availability of public information about Queensland Rail's past behaviour. While there might be some adverse effects for Queensland Rail from undertaking hold-up, it would seem that the net effect would be that reputational capital would be relatively 'small', particularly if current parties have no other economical option but to contract with Queensland Rail.

This qualitative analysis suggests that the expected value from hold-up could be large and that any reputational capital (i.e. private sanction) could be small. This framework suggests that a change from a future with declaration to a future without declaration could increase the probability of moving the parties outside of their self-enforcement range (i.e. hold-up could occur).

Case study: General Motors–Fisher Body⁸²³

The classic example of hold-up in the economic literature is the General Motors–Fisher Body case.⁸²⁴ The original production process for automobiles involved individually constructed, open and largely wooden car bodies. General Motors sourced these from independent suppliers. However, by 1919, the production process began to shift toward using closed-metal bodies.

In that year, General Motors and Fisher Body signed a contract for Fisher Body to supply General Motors with the novel, closed-metal automobile bodies. In order to produce the bodies, Fisher Body had to make substantial investments in highly specialised stamping machines that were specific to General Motors.⁸²⁵

The transactors attempted to design their contract to address the potential for hold-up in their relationship. To protect Fisher Body's investment in specialised assets, the contract included an exclusive dealing clause that required General Motors to purchase all of its closed-metal automobile bodies from Fisher Body for a period of 10 years—General Motors could not source bodies from any other suppliers during that time.

The contract also protected General Motors from hold-up by Fisher Body. It set the price equal to labour plus transportation costs, plus a 17.6 per cent mark-up to cover capital costs.⁸²⁶ The contract also contained a most-favoured-nation clause that provided General Motors would not be charged a price higher than Fisher Body charged other customers supplied with similar auto bodies.⁸²⁷ The outcome of

⁸²² The extent to which Queensland Rail could appropriate this value would depend on the presence of declaration and the efficacy of regulation (and possibly other factors).

⁸²³ Klein, B, Crawford, R & Alchian, A, 'Vertical integration, appropriable rents, and competitive contracting process', *Journal of Law and Economics*, vol. 21, no. 2, 1978, pp. 297–326.

⁸²⁴ Queensland Rail's consultant, HoustonKemp, refers to the General Motors–Fisher Body case in its arguments (HoustonKemp 2019, p. 12).

⁸²⁵ These stamping machines or presses for individual automobile parts are called 'dies'.

⁸²⁶ The pricing terms likely related to variable, rather than total, cost, as it would have been difficult for Fisher Body to identify the capital and overhead costs attributable only to General Motors (and not to other customers).

⁸²⁷ Church, J & Ware, R, *Industrial Organization: A Strategic Approach*, 2000, p. 78.

the agreement was a long-term cost-plus contract intended to address the potential for hold-up by either party following Fisher Body's initial investment.

All three pre-conditions for hold-up were present in this case. First, the specialised stamping machines designed by Fisher Body were highly specific to General Motors. The machines could not be easily used to make bodies for any other car manufacturer. As such, these machines represented a type of physical asset specificity. The fact that the investment was highly specific to its relationship with General Motors made Fisher Body vulnerable to hold-up. Specifically, General Motors could threaten to reduce its demand for car bodies or terminate the relationship altogether unless Fisher Body agreed to reduce the price.

Second, the agreed contract involved an exclusive dealing arrangement, whereby General Motors agreed to buy all of its closed-metal automobile bodies from Fisher Body for 10 years. At the time of the original contract, the dominant production process involved using wooden open bodies; Fisher Body's closed-metal bodies were a novelty. Unforeseen by either party at the time of contracting, the demand for closed-metal bodies (and automobiles in general) rose substantially afterwards. By 1924, more than 65 per cent of General Motors' automobiles had closed-metal bodies.

General Motors became increasingly dissatisfied with the pricing terms of the contract. Fisher Body had adopted a highly inefficient, labour-intensive production process. Rather than make relevant capital investments, Fisher Body employed additional workers and placed a 17.6 per cent margin on its workers' wages. While opportunistic and inefficient, this practice was nonetheless consistent with the original contract terms, which permitted Fisher Body to charge General Motors its variable costs of production plus 17.6 per cent.⁸²⁸ As a result, the imperfect terms of the contract enabled Fisher Body to hold up General Motors.⁸²⁹

Third, unanticipated circumstances pushed General Motors and Fisher Body outside of their contractual self-enforcement range. The unexpected increase in demand for car bodies was so substantial that the expected profit to Fisher Body from exercising hold-up materially outweighed the reputational capital that General Motors could impose. Accordingly, Fisher Body took advantage of an imperfect, cost-plus contract to the detriment of General Motors. By 1924, General Motors found the relationship intolerable and commenced negotiations to purchase equity in Fisher Body. The result was a final merger agreement in 1926.⁸³⁰

⁸²⁸ Fisher Body also used the contract to locate its production facility far from the General Motors assembly plant. It then charged General Motors its variable transportation costs plus 17.6 per cent for transport of the car bodies to the General Motors assembly plant. However, a positive argument for this decision is that acceding to General Motors' request to locate its plant adjacent to the General Motors plant would have exposed Fisher Body to some degree of site-specificity. See Klein, B, 'Why hold-ups occur: the self-enforcing range of contractual relationships', *Economic Inquiry*, vol. 34, no. 3, 1996, p. 446.

⁸²⁹ Church, J & Ware, R, Industrial Organization: A Strategic Approach, 2000, p. 78.

⁸³⁰ Klein, B, Crawford, R & Alchian, A, 'Vertical integration, appropriable rents, and competitive contracting process', *Journal of Law and Economics*, vol. 21, no. 2, 1978, pp. 309–310.

APPENDIX B: SERVICE AND FACILITY DEFINITIONS

For the reasons set out in the analysis of criterion (a), the QCA has identified seven parts of the existing declared service, each of which is itself a 'service' within the meaning of s. 72 of the QCA Act.

Each service is the use of a facility.

In each case the facility is rail transport infrastructure for providing transportation by rail if the infrastructure is used for operating a railway for which Queensland Rail Limited, or a successor, assign or subsidiary of Queensland Rail Limited, is the railway manager (where all terms have the meanings given to them in the QCA Act as at the date of this final recommendation). For convenience, the relevant rail transport infrastructure is identified by reference to the widely described and understood components or rail systems of Queensland Rail's network.

Accordingly, each service and facility is described as follows:

- the Tablelands system service, which is the use of the Tablelands system (where the facility is referred to in this final recommendation as the Tablelands system)
- the North Coast Route service, which is the use of the North Coast Line and the Metropolitan system (where the facility is referred to in this final recommendation as the North Coast Route)
- the Mount Isa Route service, which is the use of the Mount Isa Line and those parts of the North Coast Line that interconnect the Mount Isa Line and the Port of Townsville (where the facility is referred to in this final recommendation as the Mount Isa Route)
- the Central Western Route service, which is the use of the Central Western system, the Metropolitan system, and those parts of the North Coast Line that interconnect the Central Western system and Metropolitan system and the Port of Mackay (where the facility is referred to in this final recommendation as the Central Western Route)
- the Western Route service, which is the use of the Western system, the Metropolitan system and those parts of the West Moreton system that interconnect the Western system and the Metropolitan system (where the facility is referred to in this final recommendation as the Western Route)
- the South Western Route service, which is the use of South Western system, the Metropolitan system and those parts of the West Moreton system that interconnect the South Western system and the Metropolitan system (where the facility is referred to in this final recommendation as the South Western Route)
- the West Moreton Route service, which is the use of the West Moreton system and the Metropolitan system (where the facility is referred to in this final recommendation as the West Moreton Route).

The geographic description of each facility is set out in each criterion (a) chapter for that respective service.

Where use of the Metropolitan system is referred to as part of the relevant facility, it is a reference to the whole of the Metropolitan system. The QCA considers this is appropriate as this approach provides flexibility to Queensland Rail and its users in managing the use of the Metropolitan system. The Metropolitan system is unique among Queensland Rail's rail systems in that it is used to carry large numbers of above-rail commuter passenger services. The QCA notes that above-rail passenger services

are subject to particular statutory requirements, the operation of which may require flexibility in the scheduling of above-rail freight services on the Metropolitan system.⁸³¹

The QCA does not consider that it is necessary or appropriate to attempt to identify the specific parts of the Metropolitan system that are utilised by Queensland Rail in conjunction with the other identified parts of the network. Such an approach would risk producing a service definition that is unduly narrow, by excluding rail infrastructure that is or could be used by Queensland Rail to operate the services described in this Appendix alongside its passenger services.

⁸³¹ Queensland Rail, sub. 33, p. 38, para. 186; Queensland Rail, *Draft Queensland Rail Access Undertaking* 1, explanatory submission, March 2012, p. 18, https://www.qca.org.au/wp-content/uploads/2019/06/8354_r-qrail-qrail2012dau-expdocs2012dau-0412.pdf.